## The progression over time of profiles of children with Developmental Coordination Disorder

Victoria Ann McQuillan

Submitted in accordance with the requirements for the degree of

# **Doctor of Philosophy**

The University of Leeds School of Education

March 2019

The candidate confirms that the work submitted is her own and that appropriate credit has been given where reference has been made to the work of others.

This copy has been supplied on the understanding that it is copyright material and that no quotation from the thesis may be published without proper acknowledgement.

## Assertion of moral rights:

The right of Victoria Ann McQuillan to be identified as Author of this work has been asserted by her in accordance with the Copyright, Designs and Patents Act 1988.

© 2019 The University of Leeds and Victoria Ann McQuillan

## Acknowledgements

Jonathan Richard McQuillan (15.2.1972 – 13.7.2017)

Gerald Spinks (Gerry) McQuillan (18.1.1932 – 8.1.2018)

I am so proud and privileged to have been inspired by two such generous and enthusiastic men, who loved life and led by example. Words cannot express how much they are loved and missed.

First, I would like to thank all the children, families and school staff involved for their enthusiasm and generosity with their time, making this project possible.

I owe special gratitude to my supervisors Professor David Sugden, Dr Mary Chambers and Professor Ruth Swanwick.

Professor Sugden gave me constant encouragement. His sheer breadth of knowledge was awe-inspiring and his patience in allowing me to discover things for myself, with guidance, was great training. His enthusiasm and support were boundless and I am forever grateful.

Dr Mary Chambers provided me with sound advice and encouragement. Her knowledge of DCD is extensive and has been so helpful in giving perspective to the project.

Professor Ruth Swanwick provided essential advice and encouragement. Particularly advice on the structure and logic of the thesis and asked me incisive questions that were challenging and invaluable to the academic rigour of the project.

I am indebted to all three for challenging me, expanding my critical thinking and helping me to complete this project.

I would like to acknowledge all the love, support and encouragement that I have received from my family, which has both sustained and inspired me. I especially want to thank my children Emma, Laura and Edward, whose unstinting practical help, encouragement and belief in me has been wonderful. Laura has special thanks for the role reversal in the great care she took of me. Her special consideration for my nutrition enabled me to keep writing! I also want to thank my parents Pat and Gerry McQuillan for their unceasing and bountiful support in all my endeavours. My gratitude and thanks to Adrian, Cathy, Rachel and Lucy and also to Lucy, Pete, Alex and Sophia and also Hannah for all their help and support. I would like to thank my brother Jonathan for his love, generosity and the way in which he enriched all our lives.

I also want to acknowledge and thank my dear friends, Glenys Waring, Josie Collier and Tara Daniels, whose continuous moral support, friendship and practical help is invaluable. I also owe a debt of gratitude to my colleagues at the University of Liverpool, especially Dr Bethan Collins and Prof. Kate Bennett.

I am grateful to the School of Education at the University of Leeds for the scholarship that made this study possible. I also want to thank the Royal College of Occupational Therapy and the University of Liverpool School of Health Sciences for grants that helped to fund the project.

I would like to add a special dedication to Professor David Sugden. All who knew him and had the privilege to work with him so keenly feel the loss.

I first knew him as great scholar, then as mentor, and came to count him as dear friend. I owe him a great deal as he has enriched my life and taught me so much, with such patience and humour. His anecdotes peppered and enlivened every meeting and his enormous energy and enthusiasm were infectious. He always encouraged me to strive for better and was always so inspiring and supportive. His generosity with his time and expertise mark him out among professionals and for those reasons and many more he will be greatly missed.

## Abstract

**Aim:** To investigate the characteristics of children with DCD with different severity of motor ability and compare motor progression over time.

**Introduction:** Children with DCD have difficulty in the development of motor coordination and learning new motor skills. Empirical and professional evidence suggests that they differ from typically developing children, but also suggest that they are heterogeneous in nature. DCD commonly overlaps with other developmental disorders. Studies have used different cut-off points for motor impairment and yet little is known whether children's motor progression differs for these different cut-offs or the impact on participation in physical activity (PA).

**Method:** An ecological mixed methods study design was used. Children had detailed profiles compiled of their individual characteristics and context. DSM-5 criteria were applied to identify children with and without DCD. Repeated motor measurements over 2 years measured motor stability or change. A case study approach identified a subset of children to interview about their experiences of participation in PA. Data analysis considered group comparison and individual motor progression through a dynamical systems lens.

**Results:** Children were categorized according to motor ability:  $\leq 5^{th}$  percentile, 6-16th percentile and  $\geq 25^{th}$  percentile on MABC2. Children in the lowest motor ability group had distinct characteristics. They had significant differences between their motor performance and the other groups, both at baseline and over time. It was characterized by stable, persistent and poor motor performance, while the other two groups were more variable over time.

**Conclusion:** Children  $\leq 5^{\text{th}}$  percentile of MABC2 appeared to have special characteristics in motor and non-motor domains. These were less variable and different to both typically developing children and to children with milder motor impairment. The results point the way to differential intervention according to the nature and severity of the characteristics in DCD in both the motor and non-motor domains.

# **Table of Contents**

# Table of Contents

ACKNOWLEDGEMENTS	III
ABSTRACT	V
TABLE OF CONTENTS	VI
LIST OF TABLES	XIII
LIST OF FIGURES	XV
ABBREVIATIONS	XVIII
CHAPTER 1 INTRODUCTION	2
<b>1.1 RESEARCH CONTEXT AND MY INTEREST IN DCD</b>	2
<b>1.2 PURPOSE AIMS AND OBJECTIVES</b>	3
<b>1.3 STRUCTURE OF THE THESIS</b>	5
CHAPTER 2 DEVELOPMENTAL COORDINATION DISORDER	9
2.1 DEFINITION OF DEVELOPMENTAL COORDINATION DISORDER (DCD)	9
2.2 THE NATURE OF DEVELOPMENTAL COORDINATION DISORDER (DCD)	11
2.2.1 Symptoms	12
2.2.2 HETEROGENEITY OF SYMPTOMS IN DCD	13
2.2.3 NON-MOTOR SYMPTOMS AND OVERLAPPING CONDITIONS	14
2.2.4 PROBLEMS ASSOCIATED WITH DCD AND SECONDARY PROBLEMS	15
2.3 IMPACT OF DCD ON FUNCTION AND PARTICIPATION IN DAILY LIFE	17
2.3.1 USE OF THE ICF TO DESCRIBE PROBLEMS IN DCD	18
2.3.2 Physical activity limitations and withdrawal from physical activity	19
2.4 DIAGNOSIS OF DCD	20
2.4.1 ICD -10 (WHO, 1993)	21
2.4.2 DSM IV (APA 1995, 2000)	22
2.4.3 CLARIFICATION AND THE NEED FOR CONSENSUS	23
2.4.4 DSM 5 (APA, 2013)	29
2.4.5 CLINICAL PRACTICE GUIDELINES: EACD (2012)	32
2.5 CHAPTER SUMMARY	33

CHAPTER 3 DEVELOPMENT OF MOTOR CONTROL AND MOTOR LEARNING	35
<b>3.1 DEFINING MOTOR DEVELOPMENT</b>	35
<b>3.2 THE NATURE OF CHILD DEVELOPMENT</b>	36
<b>3.3 DEFINITIONS: MOTOR ABILITY, MOTOR SKILL</b>	37
3.4 MOTOR DEVELOPMENT THEORIES	38
3.4.1 HIERARCHICAL NEURO-MATURATIONAL THEORY	38
3.4.2 INFORMATION PROCESSING THEORY	41
3.4.3 NEURONAL GROUP SELECTION THEORY	43
3.4.4 Ecological psychology	44
3.4.5 DYNAMICAL SYSTEMS APPROACH	45
3.4.6 SUMMARY OF THEORIES OF MOTOR DEVELOPMENT	49
3.5 MOTOR SKILL ACQUISITION/MOTOR LEARNING	50
3.5.1 Closed loop theory and the role of feedback in motor learning	50
3.5.2 Open loop or schema theory in motor learning	51
3.5.3 Ecological relevance and task constraints in a dynamic systems approach to mo	TOR
LEARNING	52
3.5.4 THE ROLE OF PRACTICE IN MOTOR LEARNING FOR DCD	53
<b>3.6</b> Assessment of motor change over time and variability in motor control	55
3.6.1 VARIABILITY AND THE STAGES OF SKILL ACQUISITION	56
3.6.2 THE ROLE OF VARIABILITY IN TYPICAL AND ATYPICAL MOTOR DEVELOPMENT	58
3.7 Embodied cognition	60
3.7.1 EXECUTIVE FUNCTION	61
<b>3.8</b> Adolph's five principles of motor development	62
<b>3.9</b> The influence of self perceptions of motor competence in physical activity	63
<b>3.10</b> PARENTAL SUPPORT AND ATTITUDE TO PARTICIPATION IN PA	65
3.11 Chapter Summary	67
CHADTED A CUDDENT ISSUES IN DEVELODMENTAL COODDINATION DISODDED	60
CHAI TER + CORRENT ISSUES IN DEVELOT MENTAL COORDINATION DISORDER	
4.1 PREVALENCE OF DCD	69
4.2 GENDER RATIO	70
4.3 MEASUREMENT ISSUES AND FULFILING DIAGNOSTIC CRITERIA FOR DCD	71
4.3.1 MEASUREMENT TOOLS FOR CRITERION A:	71
4.3.2 CHECKLISTS TO ASSESS IMPACT OF MOTOR IMPAIRMENT ON FUNCTION FOR CRITERION B	73
4.3.3 Onset of symptoms in early developmental period Criterion C	74
4.3.4 MOTOR IMPAIRMENT NOT BETTER EXPLAINED BY KNOWN NEUROLOGICAL DISORDER, VISUAI	L OR
INTELLECTUAL IMPAIRMENT FOR CRITERION D	75
4.4 LONGITUDINAL EVIDENCE: COURSE AND OUTCOME OF DCD	77
4.4.1 DIFFERENT MOTOR OUTCOMES FOR CHILDREN WITH DCD	79

4.4.2 HETEROGENEITY IN DCD, VARIABLE OUTCOMES AND POSSIBLE LINES OF INQUIRY	82
4.5 Evidence from experiences of adolescents and adults with DCD and the	
IMPORTANCE OF PARTICIPATION	88
4.6 PARTICIPATION AS AN OUTCOME FOR CHILDREN WITH DCD	89
4.6.1 REDUCED PARTICIPATION IN PHYSICAL ACTIVITY IN CHILDREN WITH DCD	90
4.6.2 INVESTIGATION OF PERCEIVED SELF COMPETENCE IN PHYSICAL ACTIVITY IN CHILDREN WITH	I DCD
	90
4.6.3 EVIDENCE FROM INVESTIGATION OF DETERMINANTS OF ACTIVITY IN CHILDREN WITH DCD	92
4.7 Evidence from longitudinal intervention studies in DCD	93
4.8 THEORETICAL PERSPECTIVES IN DCD RESEARCH	96
4.8.1 THEORETICAL FRAMEWORK	98
4.8.2 Morton's causal model	99
4.9 SUMMARY OF CURRENT ISSUES IN DCD	101
4.10 CONSIDERATIONS FOR DATA ANALYSIS IN DCD RESEARCH	103
4.10.1 GROUP COMPARISON	103
4.10.2 Cluster analysis	104
4.10.3 DEVELOPMENTAL LANDSCAPE	104
4.10.4 RANDOM COEFFICIENT MODELS	105
4.10.5 Analysis of longitudinal data and the problem of variability	105
4.10.6 COMBINING METHODS TO AUGMENT DATA	106
CHAPTER 5 METHODOLOGY	108
5.1 QUESTION: HOW DO THE PROFILES OF CHILDREN WITH DCD CHANGE OVER TIME?	108
5.2 DESIGN	110
5.3 ETHICAL APPROVAL	113
5.4 PROCEDURE	114
5.4.1 PROCEDURAL STEPS	116
5.5 ETHICAL ISSUES ASSOCIATED WITH THE STUDY	117
5.5.1 Consent	117
5.5.2 Wellbeing	117
5.5.3 Confidentiality	118
5.5.4 Privacy	118
5.6 RECRUITMENT PROCEDURE	118
5.6.1 Stakeholders	118
5.6.2 GATEKEEPERS	119
5.6.3 Children's assent	119
5.6.4 CASE SELECTION FOR INTERVIEW	120

121

**5.7 PARTICIPANTS** 

5.7.1 INCLUSION CRITERIA	123
5.7.2 Exclusion criteria	126
5.8 SUMMARY OF THE STUDY DATA COLLECTION PROCEDURE	127
5.9 DETAILED DATA REQUIREMENTS FOR ANALYSIS	128
5.9.1 PART I: IDENTIFICATION, CLASSIFICATION AND PROFILING OF SCHOOLCHILDREN	128
5.9.2 PART II: STABILITY AND CHANGE IN MOTOR PERFORMANCE ABILITY OVER TIME	133
5.9.2.1 PART II: STABILITY AND CHANGE IN CHILD SELF PERCEPTION OF ENJOYMENT, ADEQUACY A	ND
PREDILECTION TO TAKE PART IN PHYSICAL ACTIVITY OVER TIME	134
5.9.3 PART III EXPLORATION OF THE IMPACT OF CHILD CHARACTERISTICS, PERCEPTIONS AND MOT	ГOR
STABILITY OR CHANGE ON EXPERIENCES OF PARTICIPATION IN PHYSICAL ACTIVITY	134
5.10 INSTRUMENTS	135
5.10.1 Measuring instruments for Part I: profiling children with and without DCD	135
5.10.2 Instrument for Part II: measuring motor stability and change over time	144
5.10.3 Instrument for measuring stability and change in perception over time	146
5.11 DATA ANALYSIS	147
5.11.1 ISSUE SURROUNDING CONVENTIONAL STATISTICAL ANALYSIS	147
5.11.2 Inferential statistical analysis	148
5.12 CASE STUDY	156
5.12.1.USE OF THEORY AND PREVIOUS RESEARCH TO GUIDE DATA COLLECTION	158
5.12.2 FRAMEWORK FOR DETERMINING CASES:	159
5.12.3 RATIONALE FOR THE INTERVIEW SCHEDULE:	161
5.13 REFLECTIONS ON THE UNIQUE FEATURES OF THE STUDY	163
CHAPTER 6 RESULTS	166
6.1 FINDINGS FROM RESEARCH QUESTION: HOW DO THE PROFILES OF CHILDREN WITH DCD	
CHANGE OVER TIME?	166
6.2 PART I: SAMPLE DESCRIPTION	168
6.2.1 SAMPLE CHARACTERISTICS	168
6.2.2 IDENTIFICATION OF THE SAMPLE	171
6.2.3 THE CONTEXT AND CHARACTERISTICS OF THE SAMPLE AT BASELINE	172
6.3 COMPARISON OF THE CHARACTERISTICS OF MOTOR IMPAIRED AND NON-IMPAIRED CHILDR	REN
	174
6.4 FINDINGS FROM COMPARISON OF IDENTIFICATION OF MOVEMENT DIFFICULTY	180
6.4.1 Closer analysis of the identification and assessment of DCD at baseline	182
6.4.2 DISCREPANCY BETWEEN TOOLS FOR IDENTIFICATION OF DCD	184
6.4.3 IDENTIFICATION OF DCD BY USE OF CHILD REPORT (CSAPPA)	185
6.5 THE SEVERITY OF MOTOR IMPAIRMENT IN DCD	186

6.5.1 GROUP CLASSIFICATION BY MOTOR PERFORMANCE ABILITY AT BASELINE: RED, AMBER AND GREEN 187 6.6 DIFFERENCES IN CHARACTERISTICS OF RED, AMBER AND GREEN MOTOR GROUPS AT BASELINE 187 187 6.6.1 AGE IN MONTHS: 189 6.6.2. PRESENCE OF ASSOCIATED CHARACTERISTICS 194 6.6.3 COGNITIVE ABILITY **6.7 MOTOR CHARACTERISTICS OF EACH GROUP** 204 6.7.1 GROUP COMPARISON OF MABC2 TOTAL SCORE AT BASELINE 207 209 6.7.2 GROUP COMPARISON OF MANUAL DEXTERITY AT BASELINE 6.7.3 GROUP COMPARISON OF AIMING AND CATCHING AT BASELINE 210 6.7.4 GROUP COMPARISON OF BALANCE AT BASELINE 212 6.7.5 SUMMARY OF GROUP MOTOR ABILITY PERFORMANCE AT BASELINE 213 6.8 GROUP COMPARISON OF THE PARENT RATED DCDQ (07) 214 215 6.8.1 THE RED GROUP (SDCD): ALL SCORED  $\leq 5$  TH PERCENTILE ON MABC2 6.8.2 THE AMBER GROUP (MDCD): ALL SCORED 6-16<sup>TH</sup> PERCENTILE ON MABC2 215 216 6.8.3 THE GREEN GROUP (TDC): ALL SCORED ABOVE 25<sup>TH</sup> PERCENTILE ON MABC2 6.8.4 STATISTICAL ANALYSIS OF DCDQ (07) 216 6.9 GROUP COMPARISON OF THE CHILD RATED CSAPPA 218 6.9.1 THE RED GROUP (SDCD)  $\leq 5^{TH}$  PERCENTILE ON MABC2: 219 6.9.2 THE AMBER GROUP THE 6-16TH PERCENTILE ON MABC2: 219 6.9.3 THE GREEN GROUP (TDC) > THE 25TH PERCENTILE ON MABC2: 220 6.9.4 GROUP COMPARISON OF GROUP CSAPPA SCORES AT BASELINE 220 **6.10 SUMMARY OF CHARACTERISTICS AT BASELINE WITH PARAMETRIC TEST RESULTS** 222 6.11 NON-PARAMETRIC TEST RESULTS AT BASELINE 223 224 6.12 CORRELATIONS AT BASELINE 6.13 SUMMARY OF PART I RESULTS 226 6.14 RESULTS PART II: CHANGE OVER TIME 228 **6.15 PROSPECTIVE STUDY OF MOTOR ABILITY AND ATTRITION RATE** 229 6.16 INVESTIGATION OF STABILITY OR CHANGE IN MOTOR PERFORMANCE ABILITY OVER TIME 230 230 6.16.1 SMALLEST DETECTABLE CHANGE (SDC) IN MABC2 TOTAL SCORE 231 6.16.2 MOTOR CHANGE IN THE RED GROUP (SDCD) 6.16.3 MOTOR CHANGE IN THE AMBER GROUP (MDCD) 231 6.16.4 MOTOR CHANGE IN THE GREEN GROUP (TDC) 232 6.17 ANALYSIS OF THE DISTRIBUTION OF ASSOCIATED CHARACTERISTICS ACROSS GROUPS 233 6.17.1 PERCENTAGE OF CHILDREN WITH ASSOCIATED CHARACTERISTICS WITH MOTOR STABILITY OR 234 CHANGE 6.18 ANALYSIS OF GROUP MEMBERSHIP OVER TIME 235

6.18.1 DIAGRAMMATIC REPRESENTATION OF MOTOR PERFORMANCE ABILITY CHANGE OVER THREE	Ε
TIME POINTS BY EACH GROUP	236
6.19 REPEATED MEASURES ANOVA FOR GROUP DIFFERENCE IN MOTOR PROGRESSION	241
6.20 REPEATED MEASURES ANOVA FOR GROUP DIFFERENCE IN CSAPPA PROGRESSION	246
6.21 REPEATED MEASURES ANOVA OF MANUAL DEXTERITY OVER TIME	249
6.22 REPEATED MEASURES ANOVA OF BALANCE OVER TIME	253
6.23 REPEATED MEASURES ANOVA OF AIMING AND CATCHING OVER TIME	255
6.24 REPEATED MEASURES ANCOVA MABC2 TOTAL SCORE COVARIATE FOR IQ COMPOSITE	259
6.25 SUMMARY OF RESULTS PART II	261
6.25.1 CORRELATIONS TO EXAMINE POSSIBLE EXPLANATORY RELATIONSHIPS	264
6.26 RESULTS PART III: IMPACT OF CHARACTERISTICS, STABILITY OR CHANGE	266
6.27 GROUP ANALYSIS OF RATES OF PARTICIPATION IN EXTRA CURRICULAR PA	266
6.28 CROSS CASE ANALYSIS	269
6.28.1 Environment (exosystem and mesosystem levels)	270
6.28.2 FAMILY (MICROSYSTEM AND MESOSYSTEM LEVELS)	271
6.28.3 Child	273
6.28.4 Selected cases	276
6.29 THEMES FROM CHILD'S PERSPECTIVE INTERVIEWS	282
6.30 OVERALL SUMMARY	287
CHAPTER 7 SUMMARY, DISCUSSION AND CONCLUSION	288
7.1 Study aims	288
7.2 SUMMARY OF STUDY FINDINGS	289
7.3 PROFILE OF CHARACTERISTICS IN DCD	291
7.3.1 What are the characteristics of children with different motor severities of DC	D?
	291
7.4 HOW STABLE ARE THE MOTOR CHARACTERISTICS OR DO THEY CHANGE OVER TIME?	301
7.4.1 The children in the group with lowest 5th percentile of motor ability	301
7.4.2 The children in the group $6-16^{\text{th}}$ percentile of motor ability	302
7.4.3 The children in the group with $25^{ ext{th}}$ percentile and above of motor ability	304
7.5 HOW STABLE ARE THE SELF-PERCEPTIONS OVER TIME?	305
7.5.1 Ecological analysis with Bronfenbrenner's model:	307
7.6 THE CHILD'S PERSPECTIVE OF EXPERIENCES OF PARTICIPATION IN PHYSICAL ACTIVITY	310
7.7 ISSUES OF CONSENSUS WITH THE IDENTIFICATION OF MOTOR DIFFICULTY	311
7.8 DISCUSSION OF STUDY VARIABLES AND LIMITATIONS OF THE STUDY	312
7.9 THEORETICAL AND CLINICAL IMPLICATIONS FROM MOTOR CHANGE OVER TIME	319
7.10 CONCLUSION	328
	334

APPENDIX 1 ETHICAL APPROVAL	382
Appendix 2 Study flyer	385
APPENDIX 3 LETTER TO HEAD TEACHERS AND INFORMATION FOR TEACHERS	386
APPENDIX 4 STUDY INFORMATION SHEETS AND CONSENT FORMS	390
APPENDIX 5 RESPONSES TO ETHICS COMMITTEE	399
APPENDIX 6 PARENT QUESTIONNAIRE	404
Appendix 7 Interview schedule	406
APPENDIX 8 PUBLICATIONS AND CONFERENCE PRESENTATIONS ARISING FROM THE STUDY	407

# List of Tables

Table 2.1 ICD-10 (WHO, 1993)	21
Table 2.2 DSM IV (APA, 1995, p56)	22
Table 2.3 DSM5 criteria (APA, 2013, p74)	30
Table 5.1 How DSM5 criteria were investigated in this study	125
Table 5.2 Selection of children for interview based on classification	126
Table 5.3 Classification of participant movement impairment	130
Table 5.4 Total DCDQ '07 score predictor of DCD status used in this study	137
Table 5.5 Reliability coefficient scores for subdomains of SNAP IV	140
Table 5.6 Cut-off scores for subdomains of SNAP IV used in this study	141
Table 5.7 Internal consistencies for CSAPPA	147
Table 5.8 Comparison of DCD studies and effect sizes	150
Table 5.9 Effect size and corresponding sample size	152
Table 5.10 Effect size, power and sample size for two-tailed hypothesis	152
Table 5.11 Sample size recruited to the prospective study	153
Table 5.12 Investigations and main analysis for this study	155
Table 5.13 Rationale for selecting cases for interview and number of children interview	ed in
each category	159
Table 5.14 Case selection criteria for case study methodology	161
Table 6.1 Characteristics of total sample	169
Table 6.2 Comparison of sample neighbourhood	173
Table 6.3 Characteristics of the study sample	174
Table 6.4 Comparison of movement impaired and non-movement impaired groups at	
baseline	175
Table 6.5 Comparison of school performance between the groups	176
Table 6.6 Summary of Parent Questionnaire based on MABC2 category at baseline	178
Table 6.7 Different methods to identify motor impairment	181
Table 6.8 Three group comparison of type of associated characteristics	190
Table 6.9 The number and type of associated characteristics per group at baseline	191
Table 6.10 The ecological context per group	192
Table 6.11 Pooled results for all motor impaired children compared to TDC	194
Table 6.12 Three group comparison of KBIT2 IQ scores	197
Table 6.13 Summary of mean & (SD) group characteristics at baseline	204
Table 6.14 Motor characteristics of three groups at baseline	206
Table 6.15 Summary of comparison of group motor characteristics at baseline	214
Table 6.16 CSAPPA scores by Red, Amber and Green group	220

Table 6.17 Summary of group comparison of DCDQ '07 and CSAPPA at baseline	222
Table 6.18 Correlations strongly significant (p= .01 two tailed)	225
Table 6.19 Correlations weakly significant (p=.05 two tailed)	225
Table 6.20 Motor change T3-T1 (MABC2 total score) over time by group	231
Table 6.21 Analysis of number of children with AC per group over time	234
Table 6.22 Number of children per group changing (SDC +/- 10 points total MABC2 score	) or
remaining stable at T3	235
Table 6.23 strength of correlation	265
Table 6.24 Motor subgroup participation rates in extra curricular PA	267
Table 6.25 Comparison of contextual characteristics of participating and non-participating	ng
children	271
Table 6.26 Summary of stability or change for TDC/Green group	274
Table 6.27 Summary of stability or change for moderate/Amber group	275
Table 6.28 Summary of stability or change for severe/Red group	275
Table 6.29 Mixed methods analysis grid for children who participated in PA (After Cresw	/ell
& Plano Clark, 2011)	283
Table 6.30 Mixed methods analysis grid for children who did not participate in PA (after	
Creswell & Plano Clark, 2011)	284

# List of Figures

Figure 3-1 Newell's model of interacting constraints adapted to illustrate the resulting	
effects on variability of physical performance (David's et al., 2003)	58
Figure 4-1 Proposed causal modelling of DCD (Sugden & Wade, 2013 p244, adapted from	m
Morton, 2004)	101
Figure 5-1 Flow diagram of the study procedure in three parts	115
Figure 5-2 Recruitment and classification procedure	122
Figure 6-1 Distribution of ages of the sample at baseline	169
Figure 6-2 Distribution of associated characteristics	170
Figure 6-3 Sample identification and classification	171
Figure 6-4 Numbers of children identified by parents, teachers & MABC2	182
Figure 6-5 Group means for age	188
Figure 6-6 Box plot of median ages by severity of motor impairment	188
Figure 6-7 Q-Q plot of composite IQ for whole sample	195
Figure 6-8 Frequency distribution of composite IQ for whole sample	195
Figure 6-9 Group means and SD for IQ	199
Figure 6-10 Box plot of group KBIT2 composite IQ scores	199
Figure 6-11 Group mean and SD for verbal IQ	201
Figure 6-12 Box plot of group KBIT2 verbal IQ	201
Figure 6-13 Group mean and SD non-verbal IQ	202
Figure 6-14 Box plot of group KBIT2 non-verbal IQ scores	203
Figure 6-15 Q-Q plot of MABC2 total score for sample	205
Figure 6-16 Frequency of MABC2 total score for sample	205
Figure 6-17 Box plots of three group median of MABC2 total score at baseline	209
Figure 6-18 Box plots of Manual dexterity score at baseline	210
Figure 6-19 Box plots of Aiming & catching at baseline	211
Figure 6-20 Group mean and SD of Balance at baseline	212
Figure 6-21 Box plots of group median balance at baseline	213
Figure 6-22 Group mean DCDQ '07 scores	215
Figure 6-23 Box plots of group median DCDQ '07 scores	217
Figure 6-24 Group mean CSAPPA scores at baseline	218
Figure 6-25 Box plots of group CSAPPA at baseline	221
Figure 6-26 Kruskall-Wallis non-parametric test summary	224
Figure 6-27 Representation of group membership of the children from the Red group o	ver
T1, T2 & T3	236

Figure 6-28 Representation of group membership for teh children in the Amber group ov	/er
time T1, T2 & T3	237
Figure 6-29 Representation fo group membership for the children from the Green group	
over time, T1, T2 &T3	238
Figure 6-30 Summary of group's pattern of motor performance change over time	238
Figure 6-31 Red group motor change over time	239
Figure 6-32 Amber group motor change over time	240
Figure 6-33 Green group motor change over time	240
Figure 6-34 group mean and SD of MABC2 tottal score over time	244
Figure 6-35 MABC2 total score over 3 data points (T1, T2, T3) by group	245
Figure 6-36 CSAPPA group means and SD change over time	247
Figure 6-37 Estimated marginal means of CSAPPA	249
Figure 6-38 Group means and SD Manual dexterity over time	252
Figure 6-39 Plot of progression of manual dexterity over time	252
Figure 6-40 Group means and SD of Balance over time	254
Figure 6-41 Plot of group progression of Balance over time	255
Figure 6-42 Aiming and catching means & SD over time	256
Figure 6-43 Plot of Aiming and catching group means over time	258
Figure 6-44 ANCOVA MABC2 covariate for IQ	260
Figure 6-45 Overview of group motor progression and participation in PA	267
Figure 6-46 Bronfenbrenner's bio ecological model used as a framework to explore	
differences in children's perception of participation in PA	270
Figure 6-47 Classic profile of MABC2 & CSAPPA over time for TDC	273
Figure 6-48 Classic profile of MABC2 & CSAPPA over time for DCD+AC	274
Figure 7-1 Sufficient experience for motor learning for typically developing children	323
Figure 7-2 Morton's model incorporating study findings (adapted from Sugden & Wade,	
2013)	325

# Abbreviations

- ADHD: Attention deficit and hyperactivity disorder
- APA: American Psychiatric Association
- ASD: Autistic spectrum disorder
- BOTMP: Bruininks- Oseretski Test of Motor Proficiency
- CCC2: Children's Communication Checklist 2
- CP: Cerebral palsy
- CNS: Central nervous system
- CSAPPA: Children's self perception of adequacy in, and predilection for physical activity
- DAMP: Disorders of attention, motor control and perception
- DCD: Developmental coordination disorder
- DCDQ: Developmental coordination disorder questionnaire
- DSM: Diagnostic and statistical manual
- EACD: European academy for childhood disability
- GLM: General Linear model
- ICD: International Classification of Diseases
- ICF: International Classification of Function, Disability and Health
- IQ: Intelligence Quotient
- IT: Information Technology
- KBIT2: Kaufman Brief Intelligence Test 2
- LI: Language impairment
- MABC2: Movement Assessment Battery for Children 2
- MD: Muscular dystrophy
- MID: Minimum important difference

- MLD: Moderate learning difficulty
- MLM: Multi level model
- NTT Neuromotor task training
- PA: Physical activity
- PAC: Perceived athletic competence
- PE: Physical education
- PDD: Pervasive developmental disorder
- RD: Reading disorder
- SDC: Smallest detectable change
- SDD: Smallest detectable difference
- SDDMF: Specific developmental disorder of motor function
- SES: Socio-economic status
- SLI: Specific language impairment
- SNAP: Swanson, Nolan & Pelham Questionnaire
- SPSS: Statistical Package for Social Sciences
- TDC: Typically developing children
- WHO: World Health Organisation

## **CHAPTER 1 INTRODUCTION**

#### 1.1 Research context and my interest in DCD

My interest in Developmental coordination disorder (DCD) developed during my career in the National Health Service as an occupational therapist. My work in child development teams entailed a comprehensive follow up for all babies born prematurely. It was apparent that these children were at greater risk of all kinds of developmental problems and required access to a range of health and education professionals. I encountered children with cerebral palsy (CP) who were such a diverse group in terms of their array of perceptual and motor problems, type (spastic, dystonia, ataxia), level and distribution of disability (for example diplegic, hemiplegic or quadriplegic) and fluctuating and abnormal muscle tone, all of which had a huge impact on their functional abilities. In addition many of these children had associated difficulties in learning, vision and or hearing, but conversely many others had no sensory problems and normal intelligence.

I encountered another group of children, who despite average intelligence and no reported CNS damage, encountered many of the motor and perceptual problems that the children with cerebral palsy met but to a lesser degree, yet with functional impairment significant enough to impact on everyday life. These children were labelled 'dyspraxic' (later to be called developmental coordination disorder). They too appeared to be a diverse group. Some had poor balance and poor gross motor skills whilst others mainly had difficulty with fine motor skills. Some encountered great difficulty planning an action, whilst others could plan but had great difficulty executing the action or task sufficiently well. Handwriting emerged as a huge problem for a significant proportion of the children and was the most frequent reason for referral. Yet when the children were examined more closely many were found to have additional specific learning difficulties, particularly in reading and spelling which only compounded the writing difficulty. Many also had attention and social difficulties. Further interest led me to undertake a school-based project investigating a group of primary school children with DCD and poor handwriting. The findings appeared to support the hypothesis that the children with DCD had a different profile of results compared to the control poor hand writers without DCD and that the nature and severity of their handwriting difficulties was much more severe. However, small sample size should lead to the results being interpreted with caution.

After changing jobs and establishing an independent practice, I discovered an even more diverse range of profiles, with many more of the children meeting the criteria for two or even three developmental conditions (including DCD). This presented more questions. Why was it so difficult for parents to get the full extent of their child's needs acknowledged by statutory services, particularly if their child met criteria for more than one condition? Why did some children improve and gain specific skills but were not able to generalize them? Why did some children make rapid gains in motor development and others small incremental gains? Some children appeared to have severe motor impairment scores yet succeed in certain sports, why? Could there be different subtypes of DCD? Would they be associated with the type and number of other developmental conditions? Would they change over time with maturity? How much did family or school influence participation outcomes for children with DCD? These questions are important when considering policy for intervention for children with DCD. Study for a PhD seemed to be the logical strategy to answer some of these questions.

### 1.2 Purpose aims and objectives

The aim of this thesis was to examine the characteristics of children as potential underlying reasons for heterogeneity and the varied motor course found in children with DCD.

The study adopted an ecological approach using Bronfenbrenner's bio ecological framework to guide data collection in a prospective longitudinal study over two years. DSM5 criteria were used to identify 7-14 year old schoolchildren with and without DCD and they were categorized according to the severity of their motor impairment using the MABC2 at the start of the study.

Repeated measures of motor performance were collected alongside contextual data for each child. The results were analysed using both the conventional general linear methods for group comparison and non-linear case analysis.

A dynamical systems theoretical approach to motor development was used to interpret the findings, with particular note of the inter- and intra-individual variability in motor progression and proposed as a means to identify children at risk of a more persistent problems and worse prognosis. A differential approach to intervention may be appropriate for this group of children.

#### Key words:

DCD, longitudinal, motor variability, ecological

In order to address these objectives the follow research question was adopted

#### Question: How do the profiles of children with DCD change over time?

- 1 What are the characteristics of children with different severities of DCD compared to children without DCD?
  - These included their:
  - Motor characteristics
  - Associated characteristics
  - IQ
  - Ability to take part in everyday activities
  - Socio-economic and family context
  - Self-perception of their adequacy, enjoyment and predilection to participate in physical activity

- 2.1 How stable are the motor characteristics or do they change over time?
- 2.2 How consistent are their self-perceptions of their adequacy, enjoyment and predilection to participate in physical activity or do they change over time?
- 3. How do the characteristics and stability or change in motor ability and self-perceptions impact experiences of participation in physical activity from the child's perspective?

## **1.3 Structure of the thesis**

Chapter 2 introduces current knowledge of the characteristics of DCD. This includes the complex nature of DCD, the myriad of motor and non-motor symptoms associated with it and the heterogeneity often encountered in clinical practice and research. There is a synopsis of the secondary consequences and some of the serious outcomes that children with DCD may encounter without appropriate intervention. The role of overlapping conditions as a possible cause for heterogeneity is also discussed. The difficulty of distinguishing between the impact of DCD, it's secondary consequences, or the presence of overlapping conditions in terms of motor and other outcomes is highlighted. The importance of consensus agreements of criteria for the identification and diagnosis of DCD is also discussed in relation to the past and current DSM diagnostic criteria. As DCD is primarily a disorder of the development and acquisition of skilled motor control, the next chapter explores some of the theory behind these concepts

Chapter 3 introduces a definition of motor development and motor learning. Reviews of some of the theories that have informed our understanding of motor control and motor skill acquisition are explored. There is a discussion of how some of these theories have influenced research in DCD. The case is made for an ecological approach to help capture some of the constraints and affordances that may influence the development of motor skills in DCD. This enables analysis from a dynamical systems perspective. The importance of active participation in physical activity for motor learning is highlighted. The role of variability in motor development is also discussed and proffered as a potential diagnostic and prognostic tool in atypical development. Without understanding typical motor development and how it progresses, it is difficult to establish whether children with DCD follow typical or atypical patterns of motor development over time

Chapter 4 discusses the course and prognosis for DCD by exploring some of the longitudinal studies in DCD research that have informed current understanding about the nature and outcomes of DCD. Issues are explored involving potential confounding influences on the motor and other outcomes in DCD research. This is discussed along with current understanding of selfperception of motor ability and the role it may have in participation in physical activity (PA) for children with DCD. Study design, sample identification, measurement issues, choice of data analysis and theoretical interpretation are also discussed in relation to current understanding of DCD and any potential gaps in research are highlighted. As both the course and prognosis for DCD has been reported as varied, it is important to examine the progression over time to determine which variables are important. An ecological approach is suggested to investigate motor progression in DCD from a dynamical systems perspective to capture some of these important variables.

Chapter 5 outlines the methodology for the prospective longitudinal ecological study using mixed methods in three parts.

Part I describes the method of identification and classification, by severity of motor ability, of school children with and without DCD according to DSM5 criteria. A description follows of the battery of tests and questionnaires adopted to undertake fine-grained analysis of the characteristics of children with DCD.

6

Part II describes methods for measuring any stability and change in motor ability and self-perception over time for typically developing children and those with DCD. It also describes how the children from each classification from part I were tracked and compared.

Part III describes how data from parts I & II may impact the children's experiences of participation in extra curricular physical activity (PA). Detail follows of the case study method, using a Quant-Qual priority sequence approach to determine the cases for interview. This is followed by description of cross case analysis to augment the data and help establish possible influences on participation in PA for children with DCD.

Chapter 6 outlines the results in three parts.

Part I describes the fine-grained characteristics of the children, grouped by classification of motor ability on MABC2  $\leq 5^{th}$  percentile, 6-16th percentile and  $\geq 25^{th}$  percentile. Group comparisons of the baseline characteristics are presented.

Part II describes the stability or change in the motor abilities and selfperception of the children over time and changes in classification are reported. Trajectories of motor ability of the different group classification are compared with respect to the variability in motor performance between them.

Part III describes comparison of the participation in PA between the different groups of children, classified by motor ability. Self–perception, family attitude and resources are also compared to help determine the ecological circumstances of the children. Selected cases representative of typical and atypical cases from each group are reported and themes from their interviews are discussed in relation to their experiences of participation in PA.

Chapter 7 summarizes the study findings and discusses the practical and theoretical implications in relation to the existing body of knowledge on the nature of DCD with possible areas for future research. There is a discussion of the ecological design. This permits analysis to facilitate comparison at both the

group and individual level, and enable the results of motor stability or change to be interpreted through the lens of a dynamical systems approach to motor development. As a result, the importance of the role of variability and active participation in motor learning, for different groups of children, is highlighted and discussed. Suggestions are given about the possible need for differential intervention resulting from the variability. Findings associated with facilitating participation in PA, despite low motor ability are explored within the theories used in current DCD research and clinical practice.

# CHAPTER 2 DEVELOPMENTAL COORDINATION DISORDER

This chapter begins by explaining current understanding of the complex nature of developmental coordination disorder (DCD). This will include discussion of how children with DCD present to teachers and clinicians, their many and diverse symptoms and the consequences of the symptoms, and secondary problems associated with these, on the children's daily lives. There is also a discussion of the difficulties of the identification of DCD and the need for consensus for diagnosis for both clinical intervention and for research. Some of the key features of DCD are highlighted, followed by a discussion of how DCD is currently identified and diagnosed along with some associated ongoing issues.

### 2.1 Definition of Developmental Coordination Disorder (DCD)

Developmental coordination disorder is classified as a neurodevelopmental disorder and subcategorized as a motor disorder in the current (2013) iteration of the American Psychiatric Association's (APA) Diagnostic and Statistical Manual of Mental Disorders (DSM-5). To receive the diagnosis, a child with poor motor control has to fulfil four criteria of diagnostic features. The first criterion is that a child's acquisition and execution of motor skills is substantially below that expected, given both the chronological age and opportunity for skill learning of the child. The second concerns the significant and persistent interference these poor motor skills have on activities of daily living, productivity and leisure and play. The third requires the onset of the motor symptoms to manifest in the early developmental period and the fourth concerns the cause of the motor problems, which states that the motor skills deficits should not be better explained by intellectual disability, visual impairment or a neurological condition affecting movement (e.g. cerebral palsy, muscular dystrophy or a degenerative disorder) (DSM-5, APA, 2013).

However, DCD has not always been recognized as a neurodevelopmental disorder and historically the children with the condition were referred to as 'clumsy' or maladroit (Dare & Gordon, 1970; Hulme & Lord, 1986; Losse et al., 1991). This terminology was not particularly helpful for either the children's self esteem or for correctly identifying children with specific motor learning difficulties, as 'clumsy' tended to be an umbrella term for any type of motor difficulty with potentially different underlying causes. This was problematic for research as it led to difficulty in comparing groups of children. It also caused problems in clinical settings, as different intervention is required for conditions with different underlying causes. For example, a degenerative neurological condition causing poor motor control requires a different approach to intervention than poor motor control due to a developmental condition.

Another term commonly used to label children with these motor difficulties is dyspraxia. However, this is terminology borrowed from adult neurology literature, which describes a loss of function for planning motor behaviour, usually associated with damage to the parietal–occipital areas of the brain (Goodgold–Edwards & Cermak, 1990) and does not accurately reflect the developmental nature of the condition.

However, more helpful terminology was introduced when the World Health Organization (WHO) in 1992 and the APA in 1994 formally recognized the condition with agreed specific criteria for diagnosis. These different diagnostic criteria will be examined in turn and discussed later in the chapter as they reflect a progression in the understanding of the nature of DCD through evolving research. The term developmental coordination disorder subsequently became the internationally adopted term by researchers and practitioners (Sugden & Wade, 2013), which has enabled clearer identification of study participants and study comparison.

## 2.2 The nature of Developmental Coordination Disorder (DCD)

Despite Developmental Coordination Disorder (DCD) having been formally recognized since 1992, it is not well understood or recognised by the general population and could be classified as a 'hidden' disability. A recent international online survey of 1297 parents, teachers and physicians revealed that it is also poorly recognized by professionals (Wilson et al., 2013), as only 41% of paediatricians and 23% of General practitioners had knowledge of the condition. Harris et al. (2015) also suggested that although DCD is reasonably prevalent in the population healthcare professionals under-recognize it.

It is a developmental disorder and presents in early childhood, and although the children appear 'normal', parents often notice subtle differences in their children. For example, they may sit, crawl or walk later than other children and may have great difficulty acquiring basic dressing skills. However, without proper recognition or diagnosis it is difficult for parents to get help for their children.

Nevertheless, DCD is relatively common and is seen in about 5-6% of all children (APA, 2013), although not all children present the same clinical picture, and this may add to the difficulties with recognition and diagnosis. DCD is identified as a specific developmental disorder of motor skills that cannot be explained by a known medical disorder or intellectual retardation, but can interfere with daily life and so merits remediation (Henderson & Barnett, 1998). It is a condition characterized by difficulty learning new motor skills and difficulty in the development of motor coordination (Blank et al., 2012), which are considered core features.

Importantly, it is now accepted as a chronic disability, which significantly impacts on a child's ability to perform everyday motor-based tasks (Missiuna & Campbell, 2014) and so is beginning to be acknowledged as a serious disorder. There is evidence too that the motor coordination problems significantly interfere with school and everyday function (Wang, Tseng, Wilson & Hu, 2009;

Summers, Larkin & Dewey, 2008; Missiuna et al., 2008) and mounting evidence that these difficulties and their social and emotional consequences appear to persist into adulthood (Rasmussen & Gillberg, 2000; Missiuna et al., 2008, Kirby et al., 2008; Kirby, Edwards & Sugden, 2011). Although DCD is a disorder involving coordinated movement, it is often associated with a range of psychological problems (Missiuna & Campbell, 2014) and other non-motor difficulties (Dewey et al., 2002; Leonard & Hill, 2015; Wilson et al., 2017) that can also have serious negative consequences. Every class is likely to have a child with DCD and so is a condition of great importance to therapists and teachers alike.

#### 2.2.1 Symptoms

Children with DCD are slower, less accurate and more variable in their motor skills than their typically developing peers (Geuze & Kalverboer, 1994; Cantin et al. 2007) (see Zwicker et al., 2012) for a review. This can impact most aspects of daily life and draw unwelcome attention from peers. Strength, power, speed and endurance are also lower in DCD than typically developing children (Hands & Larkin, 2002), leading to lower health-related fitness, and can present problems with play, sports and keeping up with their peers. The cause of DCD is unknown, but difficulties with perceptual skills have been a popular line of inquiry for researchers, but not without contention. For example, a meta-analysis by Wilson and McKenzie (1998) found a clear association between visual–perceptual deficits and poor perceptual–motor skills in children with DCD, but Geuze (2005) suggested that it is unclear how these visual-perceptual deficits affect perceptual-motor skills.

Wilson et al. (2013) conducted a meta-analysis of research literature from 1997-2011 and concluded that children with DCD present with patterns of deficits in predictive control of movement (internal modelling), rhythmic coordination and timing, dynamic control of posture and gait, catching and manual interception (interceptive action) and deficits in executive function. Practical examples of these deficits in a school child would be difficulty positioning him or herself on a sports field to receive a ball, difficulty keeping time in music or dance, slumping in a chair, walking into doorways or objects in a classroom and difficulty organizing self and belongings. The suggestion is that children with DCD have a reduced ability to form internal models for action, instead having to rely on feedback (Wilson et al., 2013). A further review by Wilson et al. (2017) of DCD research on cognitive and neuroimaging findings found a broad cluster of deficits in motor learning processes, anticipatory control of movement and cognitive control. However, they concluded that current data do not allow prediction of whether a child with mild, moderate or severe DCD will present with a particular cluster of motor or cognitive issues (Wilson et al., 2017). The condition is not widely understood or recognized by teachers or medics and many of the children remain undiagnosed, left with a 'hidden disability' to cope with quite significant impairments in everyday function (Blank et al, 2012) and lower quality of life (Zwicker, Harris & Klassen, 2013).

#### 2.2.2 Heterogeneity of symptoms in DCD

However, not all children with DCD present with the same symptoms. They can present with differing characteristics, which can include deficits in timing of motor response, deficits in sensorimotor integration and visual perceptual difficulties (Visser, 2007). This is acknowledged by researchers and widely accepted that children with DCD form a heterogeneous group (Dewey & Wilson, 2001; Jongmans et al., 2003; Wilson & McKenzie, 1998; Martin, Piek et al., 2010) with varying profiles of difficulties in the motor domain (Wilson et al., 2003), for example, poor balance but relatively unaffected fine motor control or vice versa, or indeed poor performance across all domains. Moreover, it is not unusual to find children with DCD who perform either at the same level or above their peers on a specific sensorimotor skill, yet have severe problems in another sensorimotor area (Visser, 2007). This is often confusing for parents and teachers who are unaware of DCD. Clearly, descriptive analysis of the various characteristics lacks authority and the identification and diagnosis of DCD is complex.

#### 2.2.3 Non-motor symptoms and overlapping conditions

The presence of non-motor problems has also been well documented in DCD in areas such as attention and specific learning difficulties, perceptual difficulties, problems with self esteem, making friends at school and under achievement (Losse et al., 1991; Cantell et al., 2003; Jongmans et al., 2003). Indeed. attention problems are so commonly found accompanying motor coordination difficulties this led to the term DAMP (disorders of attention, motor control and perception) being introduced by Swedish researchers (Gillberg, 2003). Attention problems are also strongly associated with ADHD and the comorbidity of ADHD and DCD is reported to be as high as 50% (Missiuna et al., 2011; Pearsall-Jones et al., 2010). Many studies have been undertaken to establish the affect of attention on motor performance (Piek et al., 2004; Piek et al., 2007; Castelnau et al., 2007). A population cohort study provided further evidence for non-motor problems co-occurring with DCD and reported findings that children with DCD were at increased risk of wide ranging difficulties outside the motor domain (Lingam et al., 2010). However, there is some debate whether they represent separate but overlapping conditions (see Cairney, 2015, p18 for an overview).

Another consideration is that motor clumsiness has also been identified in children with other developmental conditions, for example dyslexia, attention deficit hyperactivity disorder (ADHD), specific language impairment (SLI), autistic spectrum disorder (ASD) (Piek et al., 2004; Hill, 1998; Hill, 2010; Kaplan et al., 1998; Green & Baird, 2005; Martin, Piek et al., 2010) and children with low birth weight or born prematurely (Foulder-Huges & Cooke, 2003). However, these findings have important implications and raise both theoretical and practical considerations, such as aetiological origins of the overlapping behaviours and call into question the specific nature of DCD. There have been previous arguments for a diffuse brain dysfunction termed 'atypical brain development' or 'minimal brain dysfunction' to explain these overlaps, rather than view DCD as a specific disorder, but 'pure' DCD does occur (Peters & Henderson, 2008). The concept of co-occurrence, continuum and comorbidity

is not without controversy (see Kaplan et al., 1998; Kaplan et al., 2001; Kaplan et al., 2006 for a discussion). However, the potential impact that the presence or absence of co-occurring conditions can have on the assessment and prognosis for children DCD should be accounted for in research. This will be discussed further in chapter four.

### 2.2.4 Problems associated with DCD and secondary problems

The effect of poor coordination and other non-motor problems can lead to children with DCD encountering significant difficulties in everyday life with profound impact. For example, children and adolescents with DCD are thought to be at increased risk for mental health problems, poorer psychosocial functioning and lower educational attainment (Losse et al. 1991; Rasmussen & Gillberg, 2000). Children with DCD have been reported to experience greater levels of anxiety than their typically developing peers (Pratt & Hill, 2011), lower self worth (Skinner & Piek, 2001) and poorer psychosocial adjustment (Dewey, Kaplan, Crawford and Wilson, 2002). However, it is not clear whether the motor difficulties, the non-motor deficits or the secondary difficulties relating to the impaired function produced by these difficulties are the cause. Missiuna & Campell (2014) proposed that sufficient evidence exists in the literature supporting negative mental health outcomes for children with DCD. However, they suggested the role that comorbidity of other disorders might play as primary or secondary stressors is largely unexplored.

#### 2.2.4.1 The Environmental Stress Hypothesis and DCD

Cairney et al. (2013) proposed an environmental stress hypothesis model with which to explore the relationship between some of the primary and secondary factors that may mediate and mitigate risk factors associated with psychological problems and DCD. The model proposes that the primary stressor of poor motor coordination leads to a series of secondary stressors over time, which in turn lead to poor mental health (Harrowell et al., 2017). The model has been tested with a number of different samples, age groups and research designs, some of which have been replicated and provide support for the framework as a

causal network for the association between motor skills and internalizing problems (see Mancini et al., 2016 for a review).

One such study used the prospective longitudinal data from the UK Avon longitudinal study (ALSPAC) to assess the relationship between DCD and mental health outcomes in adolescents aged 16-18 years (Harrowell et al., 2017). Advantages of this population based study, apart from the large sample size (n=6902), is that children identified with DCD were defined by DSM -IV-TR criteria, assessed with three sub-tests of the Movement Assessment Battery for Children (MABC) (Henderson & Sugden, 1992) and included if they had an IQ > 70 and motor score ≤15<sup>th</sup> percentile. Thus some confounding variables could be addressed within the study and furthermore, detailed socio-economic status (SES) information and family medical and employment history was also available for analysis. They found support for the Environmental Stress Hypothesis and demonstrated longitudinally in a population cohort that poor mental health was an important consequence of DCD. Additionally, Harrowell et al. (2017) found that there were sex differences, adolescent girls with DCD were more likely to report emotional difficulties and depressive symptoms, and boys more likely to report peer problems. Importantly, they also found a mediating effect of social communication skills and self-esteem and posited that clinical support targeted at these domains may improve mental health outcomes for children with DCD, although this is yet to be tested empirically. Clearly this could have important implications for future intervention design and requires further study. One issue not reported in the study was any differential effect of the severity of motor impairment on outcome as all children ≤15<sup>th</sup> percentile of motor ability were included in the analysis.

#### 2.2.4.2 The potential role of co-occurrence in DCD secondary problems

A recent pilot study by Dewey & Volkovinskaia (2018) aimed to address the question of the potential role that co-occurrence played in quality of life outcomes. They compared a group of adolescents with co-occurring DCD plus ADHD with a group of DCD alone, and with a group of typically developing children (TDC) to investigate the impact of the disorders on health-related
quality of life. They found no significant difference between the groups in total quality of life scores, but the group with ADHD and co-occurring DCD had significantly lower subscale scores for mood, emotions, school environment and financial resources and reported higher victimization. Although this was only a small study (n=44), it appears to suggest that children with DCD plus ADHD have more negative mental health outcomes than with DCD alone. Missiuna, Cairney et al., (2014) also found that children with both DCD and ADHD had significantly more symptoms of depression and anxiety than TDC in their population-based sample. It therefore appears to support the hypothesis that children with co-occurring DCD and ADHD may be at greater risk of negative mental health outcomes. A finding also supported by Rasmussen & Gillberg (2000), who found higher rates of mental health problems, alcohol misuse and ADHD in adults at 22 years with DCD diagnosed at 7 years old. The identification of disorders co-occurring with DCD is therefore imperative for future research and should also be considered in intervention design as there is the potential they may require different forms and intensity of intervention.

# 2.3 Impact of DCD on function and participation in daily life

DCD can affect efficient function in many areas in the daily life of a child (APA, 2013), which can cause frustration and lead to under achievement. Cermak, Gubbay & Larkin (2002) suggested that when considering the functional difficulties of children with DCD, it is not just whether the child can perform the task, but how much effort and time it takes to complete should also be considerations as these are critical elements for school success.

Some of the areas affected include impact on academic achievement, particularly handwriting, spelling and mathematics (Losse et al, 1991) and children with DCD are much less likely to achieve five or more GCSEs (Harrowell et al., 2018). Handwriting problems are frequently highlighted in children with DCD. They are reported to produce less legible writing (Smits-Engelsman et al., 2001) and fewer letters when copying (Prunty, Barnett,

Wilmut & Plumb, 2013). Leisure is also affected (Zwicker et al., 2012), particularly difficult for boys participating in team sports (Poulson et al., 2007) or being left out of physical activities (Zwicker et al., 2018). Children with DCD also experience impairment in the ability to undertake self-care activities, which can include doing up fastenings, cutlery and utensil use (Zwicker et al., 2018), grooming activities such as cutting nails, or brushing and styling hair, for example (Stephenson & Chesson, 2008; Missiuna et al. 2007; Wang, et al., 2009). These every day struggles can negatively impact on quality of life for the child with DCD (Zwicker et al., 2013 & 2018). Indeed, Cantell & Kooistra (2002) cautiously summed up potential long-term prognosis most likely ascribed to DCD as 'withdrawal, passivity and isolation'. Undoubtedly, the implications are that DCD should be assessed and appropriate intervention started early in order to avoid some of the serious negative consequences.

#### 2.3.1 Use of the ICF to describe problems in DCD

The International Classification of Functioning, Disability and Health (ICF) WHO (2001) is useful when considering where the problem lies and where to target intervention as it characterizes differences on three levels: 1) body structure or functions (impairments), 2) activities or whole body movements (activity limitations) and 3) involvement in life situations (participation). For example, a child with DCD might encounter difficulties at all three levels in the following ways: 1) body structure - impairment (poor balance and incoordination), 2) activity- activity limitation (unable to ride a bike) and 3) participation participation restrictions (unable to play out with friends on bikes and so misses future experiences). Missiuna, Rivard & Bartlett (2006) reviewed assessment and intervention practice for children with DCD using the ICF and found tools that measured extents of impairment or activity limitation were useful for identifying children with DCD and determining optimal type of intervention, whereas evaluative tools measuring activities or participation (not impairments) were more useful for determining change over time. Improvement in functional tasks tends to be environment-specific and participation or engagement in activities directly impacts the quality of life of the individual (Missiuna, Rivard &

Bartlett, 2006). Therefore, evaluative measures that capture this information are more useful to evaluate participation levels and efficacy of intervention, whereas impairment measures are useful to quantify any change in the extent of impairment. This is also important in research, as there is a dearth of information in the field of DCD about how the severity of motor impairment, or indeed how specific impairments, impact specifically on participation. The intervention work reported by Dunford (2011) demonstrated that improved function could occur without improvement in impairment scores. Thus by examining the interaction between a child's strengths and weaknesses and particular environmental factors it can help in understanding the complexity of some of these factors in DCD, such as presence or absence of co-occurring conditions (Green et al, 2008).

# 2.3.2 Physical activity limitations and withdrawal from physical activity

Of particular concern for children with DCD are the long-term effects of physical activity limitation and withdrawal from physical activity (PA). Reduced physical activity may lead to risk of secondary problems (Missiuna, Rivard & Bartlett, 2006) such as decreased strength and fitness (Schoemaker & Kalverboer, 1994) and obesity, although empirical results are still equivocal on obesity (Cairney, 2015). Rivilis et al. (2011) conducted a systematic review to synthesize data on fitness and PA in children with DCD and although body composition, cardiorespiratory fitness, strength, endurance, PA and anaerobic capacity had all been negatively associated with poor motor proficiency, methodological challenges led to mixed and inconclusive results. Clearly children with poor motor competence will be at greater risk of hypo activity and potentially at greater risk of later cardiovascular disease. However, Rivalis et al. (2011) concluded that participation in PA is influenced by a multitude of factors, such as self-perception, social and environmental pressures. Therefore a better understanding of the factors, including environmental contexts that influence children's participation in PA, is required for research and appropriate intervention design.

### 2.4 Diagnosis of DCD

DCD is diagnosed by behavioural characteristics, as there are no known biomarkers for DCD. Therefore clear diagnostic criteria are vital, as the danger of it becoming an umbrella term for the myriad of symptoms with which it is associated is evident. Baird (2013) considers this reliance solely on observation of surface features of behaviour contentious, since diagnostic systems have moved beyond medical classification and are used to set eligibility criteria for services and benefits.

Two classification systems exist for DCD for this reason and these have gone through several changes over the last three decades. They are the World Health Organisation (WHO) International Classification of Diseases and related Health Problems (ICD-10: WHO, 1992, 1993) and the American Psychiatric Association (APA) Diagnostic and Statistical Manual for Mental Disorders (DSM- III-R (APA, 1987); DSM-IV (APA, 1994); DSM-IV-TR (APA, 2000) and DSM-5 (APA, 2013). The additional credibility this gives the disorder has been noted by Chambers et al. (2005) and probably accounts for the exponential rise in research interest in the condition in the last three decades.

However, despite the two systems having common characteristics, there are significant differences in the terminology and criteria between the two. For example, even the titles differ, in ICD-10 (WHO, 1992, 1993) the disorder is known as "specific developmental disorder of motor function" while in DSM-III (1987), DSM-IV (1995), DSM-IV-TR (APA, 2000) and DSM 5(APA, 2013) it is referred to as "developmental coordination disorder". Although there are currently no medical tests for the diagnosis of DCD, the classification systems ICD-10 and DSM-IV offer specific inclusion and exclusion criteria that must be met to achieve a diagnosis of DCD. Each system has four criteria, but they contain subtle differences. ICD-10 will be discussed first and compared to DSM IV. Then the changes leading to DSM 5 will be discussed in relation to subsequent empirical evidence and expert consensus agreement. ICD-11 is currently under review and will not be implemented until 2022 and so will not be discussed here.

# 2.4.1 ICD -10 (WHO, 1993)

#### Table 2.1 ICD-10 (WHO, 1993)

Criterion A	The score on a standardized test of fine or gross motor coordination is at least 2 standard deviations below the expected level for a child's chronological age.
Criterion B	The disturbance described in criterion A significantly interferes with academic achievement or with activities of daily living.
Criterion C	There is no diagnosable neurological disorder
Criterion D	Most commonly used exclusion clause. IQ is below 70 on an individually administered standardized test.

Thus in ICD-10 (WHO, 1993) the first two criteria are inclusion criteria. Criterion A acknowledges the developmental aspect and relates motor function to chronological age and also states a specific level of impairment in either fine or gross motor coordination at least 2 standard deviations below the expected level on a standardized test. This assumes that all standardized tests are equal and measure exactly the same constructs, something that will be challenged and discussed later. Criterion B is the same in both classification systems as they both agree that there must be significant interference with academic achievement and everyday activities, although these are not specified.

The remaining criteria are exclusionary; criterion C excludes any neurological disorder and criterion D excludes intelligence quotient (IQ) below 70, as motor incoordination is known to be associated with both neurological disorders and lower IQ.

Let us turn our attention to DSM-IV, as this too has four criteria, two inclusionary and two exclusionary but they differ in the following ways.

#### 2.4.2 DSM IV (APA 1995, 2000)

#### DSM IV (APA, 1995) & DSM IV- TR (APA, 2000)

#### Table 2.2 DSM IV (APA, 1995, p56)

Criterion A	Performance in daily activities that require motor coordination is
	substantially below that expected given the person's
	chronological age and measured intelligence. This may
	manifest in marked delays in achieving motor milestones (e.g.
	walking, crawling, sitting), dropping things, "clumsiness", poor
	performance in sports, or poor handwriting.
Criterion B	The disturbance in criterion A significantly interferes with
	academic achievement or activities of daily living.
Criterion C	The disturbance is not due to a general medical condition (e.g
	cerebral palsy, hemiplegia, or muscular dystrophy) and does not
	meet existing criteria for Pervasive Developmental Disorder.
Criterion D	If Mental Retardation is present, the motor difficulties are in
	excess of those usually associated with it.

DCD is classified in DSM-IV under 'disorders usually first diagnosed in infancy, childhood or adolescence' as a 'motor skills disorder' (APA, 1995). Criterion A here also acknowledges the developmental aspect and that manifestations of poor coordination vary with age, and should be below that expected for chronological age, but also mentions measured intelligence. Significantly, no specific mention is made about standardized testing or of level of motor impairment required for diagnosis, or of level of intelligence.

Criterion B is the same as ICD-10 and states that there must be significant interference in academic achievement or daily activities, but again does not specify which ones. The exclusionary criteria C and D are significantly different from those in ICD-10. Criterion C is broadened to exclude a general medical condition, rather than just neurological disorders, and specifically excludes

Pervasive Developmental Disorder. Criterion D allows for a diagnosis of Mental Retardation, but unlike ICD-10, does not specify an IQ level and furthermore states that the motor difficulties must be in excess of those usually associated with it. Thus DSM-IV acknowledges the motor difficulties associated with lower IQ, but importantly does not exclude these children from a diagnosis of DCD.

The diagnostic criteria in DSM-IV can be interpreted as slightly more inclusive than those in ICD-10 and in addition allows for a dual diagnosis and will be discussed in more detail later. The DSM-IV definition also acknowledges that there are other disorders commonly associated with DCD that may include Phonological Disorder, Expressive Language Disorder, and Mixed Expressive Language Disorder (APA, 1995). This introduces the notion of co-occurrence of developmental disorders and DCD, something that has precipitated much debate amongst researchers, such as Hill (2001), Kaplan et al. (1998; 2001; 2006) and Green and Baird (2005) and will be discussed later. Furthermore, it mentions that the prevalence of DCD can be 5% of children aged 5-11 years and that the course can be variable (APA, 1995). However, subsequent empirical evidence from population studies by Wright & Sugden (1996) and Lingam et al. (2009) have raised some important questions about classification methods which will also be discussed later. Thus, under DSM-IV diagnostic classification, dual diagnoses may be given if ADHD is present, but not if Pervasive Developmental Disorder is present.

### 2.4.3 Clarification and the need for consensus

With two parallel classification systems being used to diagnose children, researchers began to question whether they were investigating the same type of disorder and could legitimately compare studies. One criticism was that the classification system had developed without any real empirical evidence (see Henderson & Barnett, 1998 for a discussion). This led to the development of consensus statements, agreed by the research community in the field of DCD.

#### 2.4.3.1 The London consensus statement (Fox & Polatajko, 1994)

The first of which was the London consensus statement (Fox & Polatajko, 1994) in which there was an agreement to adopt DSM-IV (APA, 1994) terminology Furthermore, some important assumptions based on and classification. empirical evidence were made explicit. For example, DCD was accepted as a chronic or permanent condition rather than something children outgrew, that there were poorly understood multifactorial aetiological factors, with evidence for contributory effects from gestational events and heredity influences with the probable existence of subtypes of DCD (Fox & Polatajko, 1994). The importance of a comprehensive diagnostic process was discussed in order to distinguish DCD from other diseases or progressive neurological disorders and sufficient consideration given to a child's unique experiences of their family, their environment and factors such as genetic endowment and their temperament. Secondary characteristics including self-esteem, motor planning and coping strategies were also highlighted as important considerations that affect motor performance (Fox & Polatajko, 1994). This would involve a comprehensive assessment process and individualized intervention focused on improving functional performance.

Despite the aim of this consensus to facilitate more specific research into DCD by stipulating inclusion and exclusion criteria, a criticism was that few researchers made explicit their adherence to each of the specified criteria (Geuze et al., 2001). Later, the Leeds consensus (Sugden et al., 2006) built on the previous work of the London consensus and debated issues that had arisen in research in the intervening years.

#### 2.4.3.2 The Leeds Consensus (Sugden et al, 2006)

Several themes had emerged from research into DCD, which the Leeds consensus addressed in relation to each criterion from DSM-IV. The themes include impact of motor impairment on functional performance, the diverse presentation of symptoms with a variable course, the importance of a child's

context and movement experience and the frequent co-occurrence of DCD with other developmental conditions.

DCD is described as an idiopathic condition with a varying, but significant impact throughout the lifespan recognized across culture, race, socio-economic status and gender. Although the onset is apparent in the early years, it does not recommend diagnosis of DCD before 5 years of age. This is also echoed by clinical guidelines Blank et al. (2012), which are currently being updated.

The persistent nature of DCD is also given emphasis, which draws on evidence from longitudinal research (Losse et al, 1991; Geuze & Borger, 1993; Rasmussen & Gillberg, 2000; Cantell et al., 2003). These acknowledge that the long-term prognosis can be variable, with a small proportion improving, whilst others continue to have motor difficulties into adulthood. However this poses important questions about identification and measurement issues in DCD, as we need to be sure that like children are being compared. Research needs to be explicit about how the criteria for diagnosis have been assessed and met, and indeed which measurements to take to establish change over time. This is important and will be discussed in detail in chapter four. The consensus statement also acknowledges potential serious negative outcomes, such as DCD being often accompanied by educational, medical and psychiatric problems (Sugden et al, 2006). This serves to highlight the need to identify which children are likely to be most at risk so that intervention may start earlier to avoid some of the secondary consequences. Thus the difficulties caused by DCD are acknowledged to start in early childhood and are likely to continue throughout the individual's life, although the nature of the difficulties may change over time.

Another significant factor in the Leeds consensus is the need to account for the importance of the individual's context i.e. their learning environment and learning experience. It states, "without adequate support and/or specific intervention within the family, school and work environments, an individual with DCD will be placed at a significant disadvantage" (Sugden et al., 2006, p5).

Thus emphasis is shifted from a within-child deficit to consider the wider social and environmental influence, which is more in line with knowledge about the acquisition of skills and learning (see Bronfenbrenner, 1979). Thus child– environment interactions are acknowledged as an important part of child development and for the acquisition of motor skills, which will also need due consideration before giving a diagnosis of DCD.

Other significant differences in the Leeds consensus relate to criteria C and D and co-occurring conditions. Evidence suggests DCD is a unique and separate neurodevelopmental disorder, which can and does co-occur with one or more neurodevelopmental disorders, commonly Attention deficit hyperactivity disorder (ADHD), Autistic spectrum disorder (ASD), and developmental Dyslexia. It therefore suggests that it is "inappropriate to exclude possibility of dual diagnosis of DCD and Pervasive developmental disorder (PDD), and both should be given if appropriate" (Sugden et al, 2006, p6). This is a clear amendment to DSM-IV-TR (2000), which excludes DCD if PDD is diagnosed. Furthermore, studies such as those by Peters & Henderson (2008) and Green et al. (2008) have since provided evidence for 'pure' DCD despite the existence of evidence for many children who meet the diagnostic criteria for more than one developmental disorder (Pearsall-Jones, Piek & Levy, 2010; Lingam et al, 2010; Hill, 1998; Hill, 2001; Green & Baird, 2005; Gillberg, 2000).

This adds an additional layer of complexity to the assessment and identification of problems for children with DCD and poses further questions about trajectories and functional outcomes for the children with additional diagnoses. Will children with DCD and one or more additional developmental conditions have a worse prognosis? Work using latent class analysis on questionnaire data about developmental disorders by Martin, Piek et al. (2010) indicated that the likelihood of comorbidity increased with symptom severity and is consistent with the findings from Kaplan et al. (2006). Thus it would appear to indicate that general screening for other developmental disorders is important if a child meets the criteria for one disorder.

# 2.4.3.3 Specific criteria from the Leeds consensus with potential impact on research

Exclusion criteria: The Leeds consensus is clear, however, about exclusion criteria. Specifically that known medical conditions (cerebral palsy (CP), hemiplegia, muscular dystrophy (MD)) should exclude diagnosis of DCD and that IQ level cut off is clearly stated, a child with IQ below 70 (either measured or presumed) should not be given diagnosis of DCD because of the higher prevalence of motor disorders. This is in contrast to DSM-IV-TR (2000), which does not state an IQ level. This should help to narrow the focus for clinical and research interest in DCD. However, potential issues are highlighted with each theme.

Assessment and cut-off point: Another clear directive from the Leeds consensus regards assessment, because of the influence the type of assessment has both on goals and on the process of intervention. Specifically it stipulates that in order to meet criterion A, an "individual, culturally appropriate assessment, norm referenced test of general motor competence for criterion A with recommended cut off level at 5<sup>th</sup> percentile for research and clinical settings" should be used (Sugden et al, 2006, p6). However much of the DCD research literature has included children up to the 15<sup>th</sup> percentile and has made no distinction between the children at the 5<sup>th</sup> percentile.

Criterion B suggests that the link between academic achievement and poor motor coordination is complex. It makes more explicit recommendations about assessment of daily living and suggests it should include culturally relevant selfcare, leisure and schoolwork (including tool use, P.E. and handwriting). Importantly, it suggests that the views of child, parents, teachers and relevant others should also be taken into account. This is a significant acknowledgement of the importance of child and family centred care, but it is expensive and lengthy to gain views of all stakeholders in research and so few studies have included this important information. Criterion C: it recommends a conventional neurological examination to rule out major neurological conditions. However, it is often rare to find input for a medical examination in the research literature.

IQ cut-off point: Criterion D suggests that ideally a measure of IQ should be obtained, but where not feasible, a teacher's opinion or national test data are acceptable to establish a general level of intellectual ability and below IQ 70 should not be given diagnosis of DCD. However, lower IQ level is more commonly encountered in a clinical setting and so this may be problematic in practice. Research tackling issues found in clinical populations may also be more useful.

Thus we can see that by following explicit recommendations potential ambiguity can be removed from research populations thereby ensuring easier comparison across future research. However, the issues highlighted above warrant some investigation. Specifically, there is a dearth of evidence whether the children with different severities of DCD progress differently and the debate continues over different cut-off points for motor assessment.

Finally, it addresses skill generalization and states that a number of individuals can contribute to enhance generalization and application in the context of everyday life. It should take account of family circumstances, be evidencebased and grounded in theories and nature of learning, and enhance environmental conditions that support skill acquisition. This supports current thinking in the World Health Organization International Classification of Functioning, Disability and Health ICF, (WHO, 2001) which acknowledges the social model of disability and recognizes the importance of person–environment interaction in any assessment of function. Indeed, as Green et al. (2008) and Dunford (2011) point out, it is not known whether it is the children who continue to have low motor ability who will go on to have the worst functional outcomes. Poor motor coordination per se may not be the most important factor in impaired function, thus it is vital for future research to examine the interplay of additional factors, such as the number and type of additional developmental conditions, the child's IQ, the child's context, movement experience and preferences.

This calls for innovative methods of research to capture and record this data for analysis. Case study research employing qualitative methods has started to appear in the DCD literature (Peters & Henderson, 2008; Summers, Larkin & Dewey, 2008; Mandich, Polatajko & Rodger, 2003; Missiuna, Moll, Law, King & King, 2006; Missiuna, Moll, King, King, Law, 2007; Missiuna, King, Stewart, Macdonald, 2008) and is starting to throw light on some of these areas.

# 2.4.4 DSM 5 (APA, 2013)

DSM-5 criteria are those currently used to diagnose DCD. These have incorporated updated research evidence and current understanding of the nature of DCD. It changed the criteria slightly from DSM IV (see table 2.3).

The important addition in criterion A in DSM 5 now emphasizes not just the delay in motor skill but also includes the acquisition of motor skills, even when given adequate opportunity for their learning and practice and that this extends to ongoing problems in the execution of the motor skills already learned (Missiuna, 2015). Clearly this has implications for intervention, as Schoemaker & Smits-Engelsman, (2015) note, in so much as repetition alone will not lead to significant improvement. However, Henderson & Geuze (2015) caution that changes in criteria introduced in DSM-5 require careful consideration, pointing out the introduction of 'opportunity for skill acquisition and use' in criterion A seems reasonable, until one considers how it may be evaluated. A certain amount of clinical judgment will still be required of the assessor.

#### Table 2.3 DSM5 criteria (APA, 2013, p74)

Criterion	Diagnostic features
A	Acquisition & execution of coordinated motor skills is substantially
	below that expected given individual's chronological age &
	opportunity for skill learning and use.
В	The motor skill deficits in A significantly & persistently interfere
	with activities of daily living appropriate to chronological age and
	impacts on academic/school productivity, prevocational &
	vocational activities, leisure & play
С	Onset of symptoms is in the early developmental period
D	The motor skills deficits are not better explained by intellectual
	disability or visual impairment and are not attributable to a
	neurological condition affecting movement (cerebral palsy,
	muscular dystrophy, degenerative disorder)

Criterion B now acknowledges the far-reaching impact of motor difficulties on leisure and play in addition to those already noted in self-care and school activities. DSM-5 now focuses on the child's engagement within the community in addition to the family and home environment (Missiuna, 2015). Furthermore, by noting a persistent interference with pre-vocational and vocational activities, as well as leisure activities, it acknowledges DCD as a serious chronic condition across the life span.

In addition, criterion C notes that the condition is present from birth and that the onset appears in the early developmental period indicates that DCD is considered a neurodevelopmental condition (Missiuna, 2015; Wilson, Smits-Engelsman et al., 2017).

Importantly criterion D addresses co-morbidity and dual diagnosis, opening the possibility of children with diagnosis of ASD also having a diagnosis of DCD. The notion of a diagnosis of DCD despite lower IQ is also possible, unless the

motor impairment is better explained by the intellectual delay, rather than as a co-occurrence of DCD. Furthermore, Williams et al. (2014) argue that now, as criteria D is more specific than in previous iterations, there is no reason that preterm birth should preclude a diagnosis of DCD unless CP is diagnosed.

#### 2.4.4.1 Commentary on DSM-5

Although the present iteration for DCD identification and diagnosis is clearer in DSM-5, some interesting challenges remain. Baird (2013) highlights some concerns regarding DSM5 for the diagnosis of DCD. The first concerns how developmental disorders are now viewed. Many disorders are now considered dimensional that were once considered categorical, which creates challenges for any classification system. The second concerns the reliability, validity and utility of any classification system. Baird (2013) points out that these are paramount for three reasons; they set criteria for diagnosis and research, they function as a clinical tool and they are the basis for remuneration in some countries. The third challenge concerns change over time, Baird (2013) notes that many of the children and young people encountered in a clinical setting frequently do not meet the criteria for certain disorders and often change from one diagnosis to another as they grow older. This can create problems for categorizing and comparing groups of children when tracking them in longitudinal research, as they are likely to change group over time. The fourth consideration concerns of risk factors. Current knowledge on neurodevelopmental disorders indicates a complex mix of genetic and nongenetic risk factors (Baird, 2013), although the cause of DCD remains elusive. DSM-5 tries to account for this as it allows account of the uniqueness of the individual by specifying the pattern of onset, course and associated descriptive features (e.g. intellectual ability). The fifth concerns consideration of the symptom severity in relation to the impairment of everyday function. Baird (2013) suggests that by adopting a quantitative approach to symptom severity, dimensions will be recognized, and use of the framework of ICF will consider individual strengths and needs as well as the enablers and barriers in the environment.

So we see that the identification of DCD is complex. Over time the criteria for definition, diagnosis and classification have become more specific as more empirical evidence becomes available and an account of individual and environmental factors now form a central part of the process.

However, symptom severity has been under represented in research. In particular the severity of the motor impairment, whether mild, moderate or severe is important as this could have an impact on the reported prevalence rates, the perception of the problem by services and on the reported outcomes. An example of this is the group of children who are classified as having moderate motor impairment and are "at risk" of DCD and there is some contention over their importance. For example, the international clinical practice guidelines include this group in their classification, whereas the Leeds consensus does not. Nonetheless, the scrutiny of research is important, yet research serves little purpose unless the findings can be applied to practice and, in the case of DCD, inform identification and intervention. A group of international experts examined recent research on DCD to produce the clinical practice guidelines for DCD.

## 2.4.5 Clinical practice guidelines: EACD (2012)

The European Academy for Childhood Disability (EACD) developed the guidelines for DCD (CPG-DCD) led by Blank et al. (2012). They are currently under review for update. They apply to children "with long-standing, non-progressive problems of specific motor skill performance, not attributable to any other medical or psychosocial condition" (Blank et al, 2012, p59). One of the most striking differences between these and the Leeds consensus relate to the percentile cut-off used for diagnosis. The European EACD recommendations (Blank et al, 2012) state that limited sensitivity in the presently available motor test battery makes the conservative 15<sup>th</sup> centile cut-off safer. Furthermore, they suggest that any area-specific deficits that would be relevant to ADL (e.g. balance or dexterity) would be missed if only using the 5<sup>th</sup> centile (Blank et al,

2012). Moreover, they note that as reasonably good agreement exists between most motor measures using the 15<sup>th</sup> centile they suggest that it is "plausible to use a cut-off level of 15<sup>th</sup> centile in addition to criteria II and III" (Blank et al, 2012, p73) in order to be able to give children adequate support. It is possible too that by allowing a cut-off of 15th percentile insurance companies will permit treatment for children in this at-risk range.

Whilst this is relevant clinically, an important distinction between the different requirements of clinical and research settings should be noted here in order to properly classify the children. Blank and colleagues (2012) use the term 'moderate DCD (SDDMF)' to describe children scoring between the 5 and 15<sup>th</sup> centile on a motor test battery, whereas others have used the term 'at risk of DCD' to describe this group (for example Johnson & Wade, 2009). Correct classification of children for research assumes a much greater significance when considering patterns of dysfunction and possible trajectories as the children with moderate DCD may constitute a different group from those with severe DCD. However, this will require first classifying and then analysing the results from each group over time by use of longitudinal study design.

# 2.5 Chapter summary

DCD is complex, with a myriad of motor and non-motor symptoms, many of which can have serious long-term implications if left unrecognized. DCD has been the focus of a great deal of research over the last 30 years that has contributed significantly to the body of knowledge. However, identification and diagnosis are not straightforward and several issues remain with the interpretation of diagnostic criteria, despite the consensus agreements and international guidelines. However, since DCD is primarily a disorder of motor development and the acquisition of motor skills and they are core features of the condition, it is prudent to consider underlying motor theory. An understanding of motor development is central to the understanding of DCD and so the next chapter will outline current interpretation of the theories of

motor development and of motor skill acquisition and relate these to the research that has informed our current understanding of DCD.

# CHAPTER 3 DEVELOPMENT OF MOTOR CONTROL AND MOTOR LEARNING

The previous chapter outlined DCD as a disorder of motor development and this chapter begins by exploring the nature of child development and the development of motor control. The first part of the chapter explains what is known about motor development from a theoretical perspective, while the second part of the chapter explains current understanding of motor learning and skill acquisition. Motor development and motor learning are intrinsically linked, as development is age related and experience dependent, so generally speaking, motor control improves with age and experience and as we learn more we develop more. However, this is not necessarily the case for motor control in children with DCD. Brief overviews of some of the theories of motor development are discussed, in particular the emerging change in emphasis of the importance of the role of the environment and active movement experience in motor development theory. The capture of the nature of change in motor development is discussed from a dynamical systems perspective and the concept of motor variability is explored and related to functional participation in physical activity for essential movement experience. Some additional constraints with respect to motor learning and participation are discussed, such as self-perception and parental support along with the role of practice and the role of the environment are also considered when examining participation in physical activity (PA). The concepts are related to children with DCD in order to explore why they appear to participate less in PA than typically developing children.

# 3.1 Defining motor development

There are a number of definitions of motor development that contribute to our understanding of the term. Adolph, Weise & Marin (2003, p134) describe it as "changes in children's ability to control body movements from an infants first

spontaneous waving and kicking movements to the adaptive control of reaching to locomotion and complex sports skills." Sugden and Wade (2013) succinctly define it as adaptive change towards competence. Both definitions emphasize change and adaptation and these concepts will be explored within the discussion of some of the theories that have shaped our understanding of motor development. More recently Adolph (2018, p2) has broadened the scope and suggests that motor development refers to "improvements and decrements in motor skill over the life span and the processes that underlie these changes" and has identified five underlying principles. The principles will be better understood in light of the theories of motor development and therefore will be visited after a discussion of the main theories.

#### 3.2 The nature of child development

The term development refers to the process by which an organism grows and changes through its life span, but the most dramatic changes in human development take place during pre-natal development, infancy and childhood (Smith, Cowie & Blades, 2003). Generally developmental processes have been However, an important conceptualization of development related to age. considers the ecology of development, that is studying 'development in context' and Urie Bronfenbrenner proposed a model with which to do this. He described human development as the product of interaction between the growing human and its environment and an evolving process of interaction through which the behaviour of the child is instigated, sustained and developed (Bronfenbrenner, 1979). From this perspective, the process of development is bi-directional with both the child and the setting in which the child finds itself influencing the progress. The environment considered relevant to developmental process is not limited to an immediate setting (i.e. face to face interaction), but encompasses interconnections between settings (e.g. roles, inter-personal interactions) and external influences (e.g. cultural beliefs and practices). Bronfenbrenner (1979) conceived the ecological environment as a nested arrangement of concentric structures, referred to as micro-, meso-, exo and macrosystems to explain these settings. Let us consider how motor control develops, but first clarify some terms.

#### 3.3 Definitions: motor ability, motor skill

Almost every aspect of daily life and human interaction involves movement. This is true throughout the life span, from babies to adults, and facilitates active participation in necessary activities and social interaction. Movement can be described as the displacement of body parts and analysed as body movements or intended functions (Keogh & Sugden, 1985, p7). However, the context in which it takes place is important and concerned with the interplay of the mover's intent and function. The neuromuscular system is responsible for contracting muscles and producing movement, and movement is the observable behaviour (Keogh & Sugden, 1985, p8). However, the control of movement becomes more complicated when objects and other people are involved, as the movements are situation specific (Sugden and Wade, 2013). As typical children develop their motor control becomes more accurate and their movements more skilful.

The term ability is defined as the inherited, relatively enduring, stable trait of the individual, thought of as 'the hardware' (Schmidt & Wrisberg, 2000), and is used when referring to a capability or aptitude that an individual brings to performance situations (Sugden & Wade, 2013). Abilities are possibly genetically determined and underpin numerous skills, but are not easily influenced by practice. Schmidt and Wrisberg (2000) propose that skills, on the other hand, are chiefly developed by practice and experience. It is important to differentiate between skill and ability, particularly in atypical development and understand to what degree ability is a limiting factor and whether intervention should target abilities or skills (Sugden & Wade, 2013). It may be that children with DCD have deficits in both.

However, Burton & Miller (1998) refer to the low correlations between different motor skills and suggest that there is little evidence that motor abilities exist, however this remains contentious in sports and music (as high correlations will only occur where there are common features e.g. tennis and squash). They suggest that the term abilities can be interpreted to refer to a person's potential movement competencies, as opposed to their actual movement performances, and that the results of an assessment procedure only indicate what a person did, not what they could do in other circumstances (Burton & Miller, 1998). Clearly this has important implications when planning intervention, and Sugden & Wade (2013) advise that deciding what to measure is of paramount importance. Equally, the same could be said about measurement decisions in research into the motor problems of DCD. However, without a sound theoretical base from which to make these decisions the measurements are of little value.

Therefore, in order to more fully understand DCD, it is helpful to consider some underlying theoretical assumptions that have influenced our understanding of motor development and in turn have influenced research paradigms through which DCD has been investigated.

# **3.4 Motor development theories**

#### 3.4.1 Hierarchical neuro-maturational theory

In the mid 1900s motor development was considered to be a pre-determined gradual unfolding of patterns in the nervous system. Observational work by Gesell and Amatruda, Piaget and McGraw favoured the viewpoint that motor development was driven by cerebral maturation not experience. Gesell's work was heavily influenced by that of Coghill, an early behavioural embryologist, who studied the salamander and found correlations between changes in their movement pattern and corresponding changes in their nervous systems. This led Gesell to conclude that development was biologically driven as an orderly genetic sequence (Gesell, 1933). This viewpoint left virtually no place for the role of interaction with the environment in motor development (Hadders-Algra, 2000). The notion that a genetically determined sequence followed general developmental rules, such as cephalo-caudal and proximal to distal development, predominated clinical reasoning and led to 'developmental

diagnosis', which consisted of series of tests for the assessment of developmental milestones (Hadders-Algra, 2000).

This neuro-maturational approach to motor development asserted that motor milestones were reached at specific points in time as a direct result of the specific order of development in the central nervous system (CNS). The idea was that predetermined patterns of movement evolved as cortical control took over from lower brain stem and spinal reflexes (Sugden & Soucie, 2017). Evidence from observation of the emergence of motor milestones, such as standing and walking, in the absence of movement practice was thought to support this hypothesis. Examples such as McGraw's study of the twins Johnny and Jimmy (McGraw, 1935), where Johnny was confined to his crib for most of the day and Jimmy was given motor practice, yet both attained their motor milestones such as crawling and walking at identical times.

Although motor development is now thought to be more complex than this, detailed studies of infant motor development using photographs, movies and direct observation of movement by developmentalists such as Gesell, Shirley and McGraw have provided a great legacy of methodology for the systematic study of motor development (Thelen, 2000). They paved the way for close observation, including empirical data from measurements, taken over time from the same children. For example, Shirley (1931) studied the parameters of gait development by oiling the feet of infant and toddlers, allowing them to walk on paper and taking multiple measurements. Gesell established a vast data set of developmental norms for motor milestones by repeated testing of large numbers of the same children with standardized tests and many form the basis of tests still in current use, such as Baley Scales of Infant Development the Denver Developmental Screening test (Thelen, 2000). Both Gesell and McGraw studied twins to explore the contributions of genes and the environment to motor development and both agreed that cortical readiness was necessary to improve function, giving support in their view, to the primacy of maturation in motor development (Thelen, 2000).

Although ideas about motor development have moved on, the empirical work on motor milestones laid the foundation for contemporary practice. For example, delayed motor milestones are often used a marker for atypical motor development, such as cerebral palsy, where a lesion in the motor cortex directly impacts motor control. Children with DCD are also reported to have delayed motor milestones and inclusion of this as one of the criteria for diagnosis is recommended (Blank et al., 2012, p63 recommendation 2). Perhaps indicating that DCD is viewed as involving problems or differences in the CNS. Wilson et al., (2017) support this view following their systematic review of research including neuroimaging data. This led them to conclude sufficient evidence exists to support the hypothesis that children with DCD have differences in brain structure and function to typically developing children. However, Sugden (2018) suggests that current diagnostic criteria for DCD, as defined by DSM-5, do not involve atypical neurological substrates. Moreover, most of the existing neuroimaging studies have been carried out with relatively small sample sizes and this raises questions about how representative the samples are of the DCD population. Thus the tension in research can only be resolved with more studies and representative samples.

Another consideration is the acknowledgement that there are wider influences on the development of motor control. Wade & Kazeck (2018) suggest the underlying causes of poor motor control in DCD is more complex than problems in the CNS and favour an ecological approach involving task and environmental constraints, rather than use assumptions of brain mechanisms 'processing' information.

Although research in the field of DCD is rapidly expanding, it is difficult to draw conclusions about causal influences of the motor difficulties without understanding the theoretical position from which they have been investigated. Many studies in DCD have drawn heavily from IP theory, where the brain and central nervous system (CNS) are viewed as the controller of the motor system.

#### 3.4.2 Information processing theory

Information processing theory was derived from system analysis work used in communication systems, proposed by Shannon and Weaver (1949), and applied to behavioural systems in social sciences. The Shannon and Weaver model (1949) encompassed an information source, transmitter, signal, channel, receiver and destination and any interference or distortion in the signal was referred to as 'noise'. This theory has been so influential in motor development because psychologists compared human processing to computer processing and used the mathematical basis of information theory to make quantitative predictions about behaviour (Keogh & Sugden, 1985). Central to information processing theory is the assumption that information from the environment (e.g. visual, kinaesthestic, auditory, tactile etc.) requires interpretation by an internal mechanism in order to assign meaning and generate an appropriate response (Sugden & Wade, 2013). The brain is often depicted as analogous to a computer, where the input of sensory information is processed and triggers an appropriate output motor response. Sensory information entering the processing system is thought to be held in a working memory store so that it can be used to generate a plan of action, often referred to as a programme, and governs variables such as timing, force and spatial characteristics that may influence the movement outcome (Sugden & Soucie, 2017). This theory has given rise to a system of measurement and has had a huge impact on the way that processing has been investigated.

Adam's (1971) closed loop theory was highly influential in understanding skilled motor performance from an information processing stance and proposed that there were two states of memory, the memory trace and the perceptual trace. The memory trace was thought to initiate movement and influence choice of direction based on knowledge of results (KR) and through practice. Whereas, the perceptual trace (analogous to recognition memory in verbal tasks), thought to be responsible for guiding the limb to the correct location, and formed from past experience with feedback from earlier responses, is compared to incoming sensory feedback to detect errors and make necessary adjustments (Schmidt,

1975). Schmidt (1975) proposed that Adam's ideas were popular as they were testable, he had operational definitions for his constructs and suggested experimental variables, however, a criticism was that his theory could not explain generalization to rapid responses.

Schmidt addressed this with another model based on information processing theory. He proposed his schema theory (1975) to explain how humans execute rapid skilled body actions without conscious control. He suggested that sensory information modified a set of restructured movement commands, known as the motor programme, which defined and shaped the action produced (Schmidt & Wrisberg, 2000) based on a control system known as the open loop system. Input is delivered, processed and an action chosen, instructions are then sent to the effector to carry then out and the action is completed. There are no modifications while the movement is in progress as there is no feedback in the open loop system. Schmidt & Wrisberg (2000) use the analogy of a traffic light system to describe this system, the lights regulate traffic flow at a junction by a repetitive sequence of red, amber and green lights. If an accident occurs the system will still operate the lights, even if the road is blocked, as if nothing has happened because there is no feedback. Thus this system is ineffective in the face of unpredicted changes, but effective as long as the circumstances surrounding the action are unchanged, such as in a stable, predictable environment (Schmidt & Wrisberg, 2000).

Both the closed and open loop theories are underpinned by the information processing theory, based on the assumptions that motor performance and skill learning requires a motor programme to organize and execute an appropriate motor response to an environmental task demand (Sugden & Wade, 2013). Geuze (2018) terms these 'black box' theories, in which perceptual information is somehow linked into motor action.

Information processing (IP) theory has been the predominant theory in cognitive psychology and the study of motor behaviour for the last half century and has

led to a vast body of research carried out using this paradigm, particularly for research in DCD (Sugden & Wade, 2013). Indeed, Wilson and colleagues undertook two reviews of studies using this paradigm (Wilson et al. 1998; Wilson et al., 2017), both of which provided evidence that children with DCD were inferior on nearly all measures of information processing than TDC, seeming to suggest support for this approach. A further recent review of studies from 2011-2016 by Adams et al., (2017) suggested reasonably strong evidence to support the internal modelling deficit hypothesis (IMD), which is another cognitive neuroscience approach based on an information processing paradigm. The IMD hypothesis suggests that the movement difficulties in DCD are caused by deficits in predictive control when planning and executing movements (Wilson et al., 2017b).

However, Wade & Kazek (2018) argue that experimental design using an IP paradigm only searches for IP explanation for the underlying motor deficits. Although the studies using the IP paradigm have contributed to our current understanding about the nature of motor problems in DCD, they have tended to use a lab based experimental design and, one could argue, lack real world validity. Moreover, Geuze (2018) cautions that current theoretical explanations of DCD lack consensus, although other theories are also contributing to our understanding of DCD.

### 3.4.3 Neuronal Group Selection Theory

Neuronal group selection theory (NGST), developed by Edelman (1989), states that development starts with multiple neuronal groups forming primary neuronal repertoires, the cells and gross connectivity of which are determined by genetic information (Hadders-Algra, 2000). Experiential sensory information induces modifications in the strength of the synaptic connections within and between the neuronal groups, resulting in a variable secondary repertoire. This changed connectivity within the secondary repertoire allows for a situation specific selection of neuronal groups, which form the basis for mature variable behaviour that can be adapted to environmental constraints (Hadders-Algra, 2000), p567). Hadders-Algra (2000) argues that this suggests a complex

interplay of genes and the environment and points to the importance of selfproduced activity to develop optimal circuitries.

It is often found clinically that children with DCD self select to opt out of physical games and activities, possibly further restricting their self produced movement experience thus setting up a negative cycle. Geuze (2018) appears to adopt this view and suggests there are several causes for coordination problems in DCD, which may include deviant development, slower rate of learning, as well as physical and environmental constraints, some of which may be overcome in time. Future research should then incorporate an investigation of the child's motor abilities and environmental opportunities for self produced activity as the two areas have a reciprocal interaction.

# 3.4.4 Ecological psychology

Another theoretical perspective, ecological psychology comes from the work of Gibson (1979), which rejects the view that information has to be processed or translated by CNS in order to formulate a motor response. Central to the ecological approach is 'affordance', which is the term used to describe the reciprocal relationship between an animal and its environment (Gibson, 1979). Affordance is the resource or support offered by the environment, but the animal in turn must have the capabilities to perceive it and also to use it (Gibson, Adolph & Eppler, 1999). This implies another reciprocal relationship, of perception and action, whereby, perception provides the information for action and action (Gibson, Adolph & Eppler, 1999).

Gibson proposed that perception is direct, so that perception and action (movement) are viewed as directly linked, as opposed to Information processing theory where indirect perception is assumed (Sugden & Wade, 2013). The coupling of perception and action both guides and constrains future action, because the individual engaging in action detects new information and is sensitive to the properties in the environment and their opportunities for action (Sugden & Soucie, 2017). Sugden & Soucie (2017) suggest that no action can effectively occur without consideration of the person doing the action and the context in which it takes place. For example an approaching ball may afford catching, but will depend on the skill of the catcher, the speed of delivery, the size of the ball and the prevailing wind conditions if thrown outside. From this perspective the developing child is an active participant in the process that affords change in the control of movement through the repeated cycle of exploration-perception-action (Sugden & Soucie, 2017).

Experimental evidence appears to support this view, for example, adults passing through apertures, reaching for objects with limbs or tools and judging stair heights to support the notion that adults perceive affordances in task constraints and bodily requirements and can adjust their actions accordingly (Gibson, Adolph & Eppler, 1999). Children with DCD are often seen to struggle generalizing motor skills or adapting to known tasks in a different environment, perhaps indicating difficulties with perception-action coupling. This line is taken by Wade & Kazek (2018), who argue that DCD can be viewed as a deficit in perception-action coupling with reduced sensitivity to the action opportunities present in the environment. This has been investigated using a dynamical systems approach with some promising results and will be discussed below.

### 3.4.5 Dynamical systems approach

Nicolai Bernstein (1967) rejected a top-down approach to motor control by suggesting that the body, which consists of hundreds of bones and joints and millions of muscle fibres, could work together in synergies, rather than the brain sending many individual messages to coordinate movement. He suggested that the brain recruited an appropriate pattern to accomplish a functional task, rather than individual muscles, and so movement was function specific not muscle specific, citing the classic example of a person's signature remaining the same

whether signing with pen on paper or using a broomstick on a blackboard (Thelen, 2000). He therefore proposed that control of movement was not reliant on selection of specific motor programmes, but largely self-organizing (Sugden & Wade, 2013). Bernstein noted that the body has multiple degrees of freedom that have to be controlled. For example, the many joints and muscles in the arm and trunk that must be controlled in order to reach for a glass of water without spilling it. He believed that groups of muscles are constrained to act together as a unit synergistically (Shumway-Cook & Woollacott, 2001), constrained but not controlled by the CNS (Sugden & Wade, 2013). He believed there is no need for a centre issuing commands to achieve action. From this perspective, movements can emerge as a result of interacting elements without the need for motor programmes or specific commands (Shumway-Cook & Woollacott, 2001). Bernstein claimed that it was the dynamics of movement that instructed the nervous system, rather than the other way round, because the motor system also exploits the mechanical properties of limbs and the body (Thelen, 2000). This led to consideration of the contribution of biomechanics of the moving limb to the emergence of new skills, which is particularly pertinent in children, because growth and change in centres of inertia repeatedly bring new biomechanical challenges, therefore skill acquisition for a child is a continually interactive process (Thelen, 2000).

According to the dynamic approach, developing organisms are complex systems of many individual elements embedded within and open to a complex environment (Smith & Thelen, 2003, p343). This complex system can generate organized behaviour in response to constraints and opportunities in the environment. Smith & Thelen (2003) suggest that this self-organization is characterized by stable and unstable states and so development can be viewed as a series of evolving and dissolving patterns of dynamic stability in response to the solution of a problem. They use the example of a baby crawling, a means of locomotion for a baby with sufficient strength and coordination to assume weight bearing on all fours, but not sufficient strength or balance to stand upright. This is a stable behaviour for several months, but becomes destabilized when the baby learns to walk by patterns of standing and walking.

The system self organizes in response to a problem, to cross the room, later to be replaced by a more efficient manner (Smith & Thelen, 2003, p344).

# 3.4.5.1 A constraints based approach

Newell (1986) used a dynamical systems approach and explored factors in environmental constraints that guide motor control. Constraints are defined as 'the boundaries or features, which interact to limit the form of biological systems searching for optimal states of organization' (Davids et al, 2003, p. 247). Newell (1986) proposed a constraints based model of motor skill acquisition that took account of 1) the characteristics of the individual, such as genes, height, weight, cognitions, motivations etc. and referred to as organismistic constraints 2) environmental constraints which can be physical such as gravity, heat, light etc. or social 3) task constraints and include specific performance goals (Davids, 2010). For the first time the individual, the task and the environment were considered to be important in the emergence of motor control.

Thus the emergence of motor control and coordination arises from the selforganized movement patterns, constrained and influenced by the task, environment and the individual performers abilities (Sugden & Wade, 2013, p69). Moreover, development is reinterpreted from the Piagetian view of the child as passive stimulus-responder to Bernstein's view of the child as an active movement problem-solver (Thelen, 2000). This self-organizing open system approach permits exploration of the child's motor actions in a variety of tasks and settings and also allows explanation of changing or inconsistent results according to the context. This is particularly important in the field of DCD, as motor inconsistency is often a trademark of the motor performance of children with DCD.

# 3.4.5.2 The role of body size and scale

Newell & Wade (2018) develop the idea of constraints further and highlight the role of body size and scale in movement development from an ecological perspective. They suggest that change in the size and form of the body during

childhood is a major factor in the change of movement organization and its outcome in physical activities. They suggest that body scale and the timescale of its change through growth are shown to relate to the emergence and disappearance of fundamental movement skills in infancy, the perception of what an environment affords functionally for action together with the emergent pattern of movement coordination (Newell & Wade, 2018, p205). They argue that a central issue for development is the mapping of the timescale of change in physical growth to the timescale in change of acquisition of perceptual motor skills. Further, they develop the hypothesis that an imbalance in these timescales can lead to performance deficits in perceptual motor skills and cite adolescent clumsiness during the growth spurt as an example.

Visser (1998) conducted a longitudinal study to examine motor performance in relation to growth in height and weight in adolescent typically developing boys and boys with DCD. He predicted that both groups would show a decline in performance during their rapid growth spurt. However, he found both positive and negative effects from the growth spurt and differences between the TDC and boys with DCD. The TDC, for example displayed negative growth effects that included balance, estimation of distance and muscle control consistency, but the positive effects included accuracy in target aiming. However, this was not the case for the boys with DCD, who did not appear impacted by the growth spurt, yet showed wide intragroup variability. This led him to conclude that only well coordinated boys were at risk during the growth spurt (Visser, 1998, p153). This was an interesting study, and could imply that boys with DCD experienced different sensorimotor processing. However, the sample size was relatively small (DCD n=15 and TDC n=16) and pre-dated the consensus statement on diagnostic criteria for DCD, so we cannot be sure that the DCD sample would meet DSM criteria for DCD or therefore extrapolate the results wider. Nevertheless, it exposed important differences between the groups and posed the question whether TDC are sensitive to the constraints of the effect of growth, causing a temporary disruption in their perception action coupling, whereas the boys with DCD did not have the same sensitivity?

Other DCD research has tested body scaled information affordances in perception-action experiments for example, Wilmut, Du & Barnett, (2017) found differences between DCD and typically developing participants (aged 7-29 years) when asked to make visual estimates and then actually passing through an aperture. The DCD participants had difficulty with the dynamic perception-action relationship. This appears to be supported by the experiences of individuals with DCD, who often report walking into doorframes and other objects. Johnson & Wade (2009) also found differences between the judgments of action capabilities ability of children with DCD and TDC when adapting to reach during experimentally controlled constraints. Both studies add support to proposal by Wade & Kazeck (2018) to consider DCD as a deficit in the perception-action relationship. So we see, from this perspective, the unit of analysis becomes the child, the task and the context (environment) in which it takes place (Sugden & Soucie, 2017). This lends further support for future studies incorporating an ecological approach.

#### 3.4.6 Summary of theories of motor development

We have seen that the understanding of motor development has evolved from theories at the beginning of the 1900s, viewing development as a gradual unfolding of abilities determined by maturation of the CNS at particular points in time, with a strong genetic influence. Followed by the prevalence of the so-called 'black box' theories that emphasize the importance the CNS for information processing and the development of schema to formulate a motor response. The emergence of a dynamic systems approach combined with ideas from ecological psychology then served to de-emphasize the role of the developing child as a passive recipient of information and suggest an active role of the child learning from the environment and their own actions. However, tensions remain between researchers who adopt different paradigms on motor development, about causal influence of motor difficulty in the field of DCD research. The cause of motor problems in DCD is still not fully understood. It may be possible that in future an ecological perspective could encompass an IP approach within the design.

Although in practice it can be hard to distinguish between motor development and motor learning, for ease of presentation they are dealt with separately in this chapter. DCD is characterized by a deficit in the acquisition of motor skills, often referred to as motor learning, so it is pertinent now to explore how have these theories have influenced understanding of motor learning.

# 3.5 Motor skill acquisition/motor learning

Schmidt and Wrisberg (2000) draw attention to important differences between motor performance and motor learning. They suggest that motor performance is always observable and influenced by many factors (such as fatigue, motivation and attention), whereas motor learning is an internal process that reflects the level of a person's performance capability. Thus performance may be thought of as one shot that can be influenced by temporary variables, whereas motor learning is more generic. Furthermore, they state than in order to learn a motor task, a person must engage in practice or performance attempts (Schmidt and Wrisberg, 2000, p12). Schmidt and Lee define learning as a relatively permanent increase in capacity to change behaviour brought about by practice and or experience. Sugden & Wade (2013) emphasize this change in behaviour and state that motor learning involves acquiring a set of processes that are instrumental in changing motor behaviour and echo that it is a relatively permanent state. Children with DCD encounter difficulty with motor learning and their motor performance is also likely to be impaired and so it is important to understand some of the ideas behind current understanding of motor skill acquisition.

### 3.5.1 Closed loop theory and the role of feedback in motor learning

We return to Adam's Closed loop theory (1971), whereby learning was considered to be facilitated by refinement through a series of perceptual-motor feedback loops, using assumptions based on information processing theory (Rosenbaum, 2010). For example, the inexperienced learner approaches a task with a crude movement and perceptual feedback indicates that the movement was not effective, so that subsequent movements are performed to

reduce the error, using a perceptual trace feedback loop. The perceptual trace is the internal reference with which to compare movement and detect error through repeated practice (Zwicker & Harris, 2009). Rosenbaum (2010) suggests, that according to closed loop theory, feedback should help people perform tasks more effectively and points to the improved performance of novice skill learners given explicit verbal feedback (called knowledge of results, KR) over the performance of those who are not told how well they have performed. However, the theory falls down when applied to complex motor skills, skill generalization and skills that can be performed when feedback is removed, but Rosenbaum (2010, p103) asserts that it 'does a good job of explaining specificity of practice'. An example of this is described by Schmidt & Wrisberg (2000, p221), who explain that practitioners can enhance development of learner's error detection by verbally drawing attention to their movement-produced feedback by asking them to estimate or describe aspects of their movement, thus helping them become sensitive to it leading to improved capability. This approach has been used with children with DCD, but has been criticized, as there is little skill generalization to other tasks.

#### 3.5.2 Open loop or schema theory in motor learning

Schmidt (1975) proposed an open loop theory where generalized motor programmes, or schema, created from past movement patterns influence future motor performance (Zwicker & Harris, 2009). These are thought to consist of pre-programmed muscle commands, reducing the need for feedback control. They take the form of stored responses for specific movements, including information on conditions such as speed and force, in addition to information on the sensory consequences of the intended movement (Hill, 2005, p48). Rosenbaum (2010) suggested that open loop theory provides a better explanation of skill generalization and the accounting for variability and novelty of performance. Hill (2005) suggested that this evidence points to use of both open and closed loop control in skilled movement. The idea is that the general motor programme is adapted for each situation in parallel to the execution of the movement itself, implying that on-line changes can be made to existing

programmes (Hill, 2005). Wolpert et al., (1998) proposed a model involving the role of the cerebellum in inverse and forward models of the motor system, so that the body is capable of learning to produce appropriate motor commands under a variety of contexts and can switch rapidly between controllers to adapt as the context changes. These assumptions led to a line of enquiry in DCD research that proposed that the internal models (controllers) for specific motor skills might be disrupted in DCD (Mon Williams, 2005) and led some researchers to implicate the role of dysfunction in the cerebellum, which Missiuna et al. (2006) suggested had some face validity. However, the experimental design has been quite task specific with little attention paid to environmental constraints so far.

# 3.5.3 Ecological relevance and task constraints in a dynamic systems approach to motor learning

The constraints approach in dynamical systems was introduced earlier, but now specific comments on constraints will be addressed with respect to learning. A further analysis of some of the terms in this field of study may help with the conceptual understanding from a theoretical stance. Newell, (1991), for example, distinguished between motor learning, motor control and motor development, which he considers as related subdomains to motor skill acquisition. However, he acknowledges that the study of all three areas share considerable common theoretical ground and refers the reader to Wade & Whiting (1986) and suggests that this is also supported by theoretical developments in the understanding of perception and action (Newell, 1991).

Furthermore, Newell highlights that contemporary study of motor skill acquisition implicitly focuses on self-generated motor performance enhancement within a variety of contextual and task constraints and deemphasizes the role of instructional concepts. As a consequence of this, he suggests that, models and theories of motor skill acquisition have tended to be task and context specific and so any development of broad theoretical perspectives on the acquisition of motor skills is dependent on a general
understanding of task constraints (Newell, 1991). He critiqued the information processing approach to the understanding of motor skills acquisition, stating that largely emphasis has been on how information is processed and little emphasis has been given to what information is processed. He also states that the interest in learning from the cognitive perspective, during the 1980s, was largely orientated to so–called cognitive tasks. Thus neither of which fully addressed the understanding of motor skill acquisition, but he suggests that a synthesis of the two could be facilitated by an understanding of the influence of task constraints on motor skill learning.

Newell argues that the traditional approaches to motor skill acquisition have failed to capture many of the dynamic qualities of the stages of motor skill acquisition exhibited by novice and expert performers. "Skill is as reflection of a dynamic exploratory activity, not the stereotypical reproduction of a static representation of action" (Newell, 1991, p 233), furthering his argument for the understanding of ecologically relevant aspects of task constraints and the dynamic interaction of information and movement. This may be an important focus for our understanding of the motor skill acquisition problems faced by children with DCD.

However, Schmidt & Wrisberg (2000) argue that the most important contributor to motor learning is the act of proper physical practice and the quality of that practice is crucial, so the role of practice in motor learning in DCD is discussed below.

### 3.5.4 The role of practice in motor learning for DCD

We know that children with DCD do not learn by just providing them with opportunity to practice, as they have a low level of transfer from one acquired skill to the next and the acquired skills are highly contextual (Schoemaker & Smits-Engelsman, 2005). Sugden & Wade (2013) advise that practice is more formal than experience and includes variables such as how to present the tasks

(partial or whole task), the distribution of practice (e.g. blocked or random, constant or varied), the type and amount of feedback and the transfer and retention of skills (Sugden & Dunford, 2007). Therefore, the structure of practice will have added importance for children with DCD because of their difficulty in motor learning. However, the most effective method of practicing or providing instruction and feedback depends on the stage of learning of the child.

Fitts & Posner (1967) describe three stages of learning: the first is the cognitive stage, which is required to detect which movement pattern will be required to achieve the goal; the second, is the intermediate level, where the basic movement pattern is mastered but refinement is still needed and errors occur; the third stage is one of automaticity, where movement patterns can take place without conscious attention (Schoemaker & Smits-Engelsman, 2005).

Blocked practice, where an individual performs a single task repeatedly (sometimes called constant practice) is often used in the early stages of learning. Random practice, where an individual attempts several different tasks in an intermingled order, has been found to be more effective for higher rate of transfer and retention of the skill and used after the first stage of learning (Schmidt & Wrisberg, 2000). This varied practice is also thought to facilitate adaptability by developing stronger schemas (a set of rules relating to various outcomes of the individual's actions) to allow them to generalize the experience to other motor performance (Schmidt & Wrisberg, 2000).

However, this generalization, or as Hadders-Algra (2000) describes it 'adaptive variability' (where children can adapt their behaviour to specific demands of a situation), is particularly problematic for children with DCD. Neuromotor Task Training (NTT) (Schoemaker & Smits-Engelsman, 2005) is one intervention approach that has applied principles of motor learning. It is a task orientated approach to training motor skills in children with DCD, whereby skills required in daily life are taught directly and encourage active practice in ecologically relevant tasks, underpinned by dynamical systems theory (see Schoemaker &

Smits-Engelsman, 2005 for a description). The EACD guidelines (Blank et al., 2012) recommend that it may be a useful intervention for DCD, as there is some empirical evidence to support its efficacy to help the children develop the motor competency to complete functional goals. A systematic review by Preston et al. (2017) also found strong evidence to support this approach for DCD.

# 3.6 Assessment of motor change over time and variability in motor control

Motor development has been defined as 'adaptive change towards competence' Sugden and Wade (2013, p2), which emphasizes the transactional relationship between the individual and the environment, but change is not always represented as a smooth trajectory. Sugden & Wade (2013) argue that the diverse nature of observable behaviours associated with motor change includes variability, continuity, discontinuity, accelerations and spurts, rather than the smooth linear graph that is often portrayed. From this stance, the variability of motoric change can be explained. They contend that conventional methodological design often miss critical points of developmental change by using methodological designs such as cross-sectional or generalized longitudinal designs and so tend to represent change as linear. Perhaps by following the motor change in children with DCD more closely, the notorious intra individual and intra group differences may reveal more information about the nature of DCD.

Adolph and colleagues argue that all children's movements are variable and that intra individual variability is a signature component of motor development and of goal directed behaviour (Adolph, Cole & Vereijken, 2014). However, depending on the nature of the motor skill, some aspects of motor skill intra individual variation decreases over development, other aspects of motor skill variability continues and increases over development and for yet others, the structure of variability changes over development. This is illustrated by case examples of the research on acquisition of infant walking and concludes that intra individual variability can refer to very different aspects of motor behaviour occurring at very different temporal and spatial sequences (see Adolph, Cole & Vereijken, 2014 for a review of the research). It is therefore important to make distinctions between the nature of the motor skills under investigation; otherwise false conclusions can be drawn.

Adolph and colleagues observe that this poses a problem for research in motor development, as studies tend to focus attention on one area to the exclusion of another. For example, in the investigation of the development of motor control, the focus is on the decrease in intra individual variability and on standardized tests and laboratory tasks where consistency is paramount. Whereas, in contrast, the investigation of the developmental of improvement in adaptability to novel conditions research tends to focus on increase in intra individual variability or continuance, in which behavioural flexibility is more important. Additionally, studies investigating developmental changes in the organization of movement tend to focus on the structure of variability in order to uncover optimal levels of predictability and complexity (Adolph et al., 2014). The picture is therefore complex and is dependent upon which elements of motor development are under scrutiny.

### 3.6.1 Variability and the stages of skill acquisition

In the early stages of skill acquisition intra individual variability is thought to reflect poor motor control (Deutsch & Newell, 2005; Vereijken, 2010 cited in Adolph et al., 2014) and so development involves acquiring sufficient control to reduce variability. This is thought to be due to an increase in control arising from better and faster use of perceptual feedback and greater use of feed forward mechanisms (Deutsch & Newell, 2005 in Adolph et al. 2014) and less noise in the sensory motor system (Schmidt & Lee, 2011 in Adolph et al., 2014). Thus researchers consider decrease in intra individual variability to be a sign of skilled performance and control and decrease in variability accompanies more consistent, accurate and economical performance.

Successful performance of most everyday activities in the real world requires some variability (Adolph et al., 2014). This is because local conditions are in a state of flux and motor behaviour has to adapt to these variations in the body, the environment and task. Simply repeating the same movements in the same way would not result in successful execution of the task. Variable movements and a variety of strategies allow the tailoring of motor actions to current situations and allow behaviour to be flexibly adapted to current constraints because, outside the laboratory, local conditions are never constant (Adolph et With sufficient experience in a variety of everyday situations al., 2014). children can adapt movements simultaneously to variable changes in their body and environment. New strategies arise in the course of coping with the task and the more ways to tackle a problem increases the probability of selecting a satisfactory solution. Adolph et al. describe these as 'raw materials to learn' and suggest that regardless of intent, continuance or increase in intra individual variability serve as an exploratory function (Adolph et al., 2014; Hadders Algra, 2000). Furthermore, Adolph et al. (2014) suggest that opportunities for learning variable movements in action create a variety of strategies that, in turn, generate information about the environment, the self and, importantly, the relations between them. They also suggest that this increase in intra individual variation is imperative from the perspective of learning and development because a selection process requires many variants to act upon.

From this perspective intra individual variability is viewed as an important aspect of development.

Davids et al. (2003) propose that dynamical systems framework has influenced the way in which we can view inter- and intra-individual variability in motor performance, as a function of learning and development across the lifespan. They explain that skilled performers can freeze and unfreeze the degree of freedom (DOF) in response to constraint demand within the chain of movement. Whereas, less skilled performers tend to rigidly fix DOF and show variability that is not functional due to their inferior adaptability to task constraints. Furthermore, they urge us to consider functional variability of motor behaviour as a key criterion of successful performance and that motor patterns emerge under different task constraints to achieve stable task outcomes, rather than pre-determined invariant (Davids et al., 2003, p249). See Fig. 3.1 for a diagrammatic representation. Perhaps the difficulty in adapting to different tasks and situations often displayed by children with DCD could be due to their difficulty unfreezing the degrees of freedom.



Figure 3-1 Newell's model of interacting constraints adapted to illustrate the resulting effects on variability of physical performance (David's et al., 2003)

### **3.6.2** The role of variability in typical and atypical motor development

Recent work suggests that suboptimal levels of intra individual variability, that is too much or too little, can distinguish children with typical development from those with disabilities and delays (Adolph et al, 2014). For example, children with Down syndrome frequently show increased amounts of intra individual variability in basic motor skills relative to typically developing children (Looper et al., 2006 in Adolph et al. 2014). Conversely, children with cerebral palsy have

less variable and more stereotyped movement and a limited capacity to vary motor behaviour, i.e. limited variability (Hadders-Algra, 2010).

Successful adaptive movements require different body parts to work together with coordination of joints and limbs and different aspects of the movement to be controlled simultaneously. Thus a challenge for skill acquisition is learning to coordinate the body as a single unit, rather than a collection of separate parts (Adolph et al. 2014) and so any child who encounters difficulty coordinating their limbs with movement of their body will have difficulty learning skilled movement. Observation of change in variability of motor performance should reveal information on the progress of motor skill acquisition and highlight any difficulties.

However, Smith and Wade (2015) urge us to consider how variability is analysed, as it can be viewed as 'good' or 'bad' depending upon the technique used and the theoretical stance taken. For example, variability interpreted as the ability to make ongoing adjustments in response to changing environmental and task constraints, as observed in skilled performers, is seen as flexibility and therefore 'good'; whereas variability analysed with standard linear techniques interprets variability as 'noise' or output errors, does not reveal the ongoing dynamics and is interpreted as 'bad' (Smith & wade, 2015, p50).

Adolph et al. (2014) suggest that a major shortcoming in research is our lack of knowledge about how different types of intra individual variability promote and hinder healthy development. Most research findings are consistent with the idea that some amount of some types of intra individual variability is conducive to healthy learning and development of basic motor skills. However, they state that there is a paucity of experimental evidence to support this notion, but that identifying differences in the amount/structure of variability between children with typical development and those with disabilities is a "promising first step".

Therefore, future research in the field of DCD should identify the amount of variability in motor development between TDC and children with DCD, to see if this could be a useful tool to distinguish those with particular difficulties.

Of course, variability in the development of motor control can be influenced by many factors, whether within the child, the task or the environmental context. The bidirectional relationship of the developing child and the environment has already been discussed in relation to child development, but there is also thought to be a bidirectional relationship between the development of motor control and the development of perception and cognition (Adolph, 2018). The consideration of how cognitive development influences motor development and vice versa and how this is explained is addressed by an approach called embodied cognition.

### 3.7 Embodied cognition

In direct contrast to the information processing theory and other classical cognitive theories, embodied cognition emphasizes the importance of the body for cognitive processing, rather than focusing solely on the brain. It is posited, in embodied cognition, that through the body interacting with the environment the sensorimotor experiences help with the acquisition and representation of conceptual knowledge (Wellsby & Paxmen, 2014). Thus motor development has come to be viewed as a process that goes hand in hand with cognitive development, rather than as a function of independent biological processes (Sugden & Wade, 2013). Embodied cognition, includes dynamical systems theory and social behaviour theory, and purports that cognition emerges from the interaction of sensori-motor actions of the baby with his or her environment (Sugden & Soucie, 2017). Leonard (2016) further suggests that motor development provides opportunities for the development of a range of perceptual, social and cognitive skills and is also influenced by these abilities. From this perspective, children with any motor or perceptual motor difficulties are going to be disadvantaged, not only in their cognitive development, but also in social and other domains. Thus children with DCD, because of their poor motor control, are likely to face a double developmental disadvantage.

An area of cognitive function that has been the focus of attention of research in DCD has been that of executive function. Executive functions involve planning, organization, decisions and response inhibition and they have a direct impact on behaviour and performance.

### 3.7.1 Executive function

Executive function (EF) difficulties have been widely reported to be problematic for children with DCD (Wilson et al., 2013), yet some of the research has been questioned. For example, Leonard & Hill (2015) have critiqued methodological issues with some of the research and caution that future research investigating EF should delve deeper into the underlying causes of performance differences by carefully considering methodological factors such as the domains and samples tested, the informational constraints of the task, as well as the neural underpinnings of the behaviour observed.

Nevertheless, a child's confidence, attention and memory (executive functions) all affect their motor performance (Sugden, 2018), and can influence their participation either positively or negatively. Clinically we find that children with DCD often lack the confidence to engage in physical games with their peers and so will miss out on the social and perceptual skills associated with that play. As all appear to agree that there is a close association between motor skills and perceptual, cognitive and social abilities over the course of development, this lack of physical engagement could have serious implications for children with DCD, if it is not addressed. Indeed, if we now return to Adolph (2018), her five principles of motor development provide us with another way of viewing motor development and provide scope for potential opportunities to intervene at many levels to influence it's course.

### 3.8 Adolph's five principles of motor development

Adolph (2018) suggests that the study of motor development has a broad scope, because it refers to improvements and decrements in motor skill over the life span. She proposes five core principles. The first principle concerns the nature of motor development, not characterized as a series of stages, but by it's inherent equifinality (multiple pathways to the same endpoint) and multifinality (multiple outcomes from the same starting point). She gives the examples of these with the manifestation of mature skills in young adults often taking different forms and the different ways that elderly adults compensate for decrements in skill. She also points to individual differences and intra-individual variability as evidence of this.

The second principle concerns body-environment reciprocity, whereby motor actions depend upon the physical constraints of both body and environment. Here the development of motor actions reflects changes in the fit between variations in the environment and changes in body characteristics, for example strength, endurance, leg length etc. relative to step height and number of steps, as changes in either alter possibilities for stair climbing action.

The next principle is perception-action coupling; Adolph argues that motor development involves changes in psychological functions in addition to changes in biomechanics, because the continually varying physical constraints requires perception (and sometimes cognition) in order to adaptively guide the motor actions. She argues that movements are controlled optimally prospectively (e.g. whether a narrow gap is passable) and require perceptual information about possibilities for action and that exploratory activity and feedback from the active movements generate the requisite perceptual information.

The fourth principle is experience; Adolph argues that motor development is only age related, not age determined, because experience affects the ages that skills first appear, the form they take and which skills are acquired. For example, the onset of walking from 9-18 months, sitting with legs out or in deep squat, and whether an individual crawls, rides a bike, swims or drives a car is all experience dependent. She further contends that active self-generated experience is more powerful than those received passively. Cultural expectations and child rearing customs play a large role in determining the variety of experiences and skills.

Finally, she proposes the principle of developmental cascades, whereby changes in one motor skill bring about changes in other motor skills and lead to changes in other domains. For example, independent locomotion brings about changes in spatial cognition and memory and walking ability stimulates changes in adult-child social interactions leading to changes in emotional independence.

So we see that successful motor development has the potential for wide ranging positive impact on the rest of a child's development. However, this also serves to emphasize the disadvantage that children with DCD can face through poor motor control, but equally challenges us to consider the different levels at which some of the constraints could be ameliorated. One of which could be the provision of active positive movement experience. This leads us to consider some of the constraints in active movement experience for children with DCD, in particular, participation in physical activity. The first of these is the perception a child has about his or her movement competence.

# 3.9 The influence of self perceptions of motor competence in physical activity

It is widely accepted that participation in physical activity influences children's motor skill development and a widely held belief that the more proficient their skills become the more likely they are to engage in physical activity. Stodden et al. (2008) conceived of a model to test whether perceived movement competence was a mediator to influence levels of physical activity in children.

Slykerman et al. (2016) conducted a cross sectional study of 109 young children (mean age 6.5) that included a test of actual motor ability, a test of their perceived movement skill and accelerometers to measure intensity of PA and found that actual, rather than perceived motor skill competence, was more important for engagement in moderate to vigorous PA (MPVA). These results contrasted previous results with older children, whereby perceptions of physical competence were found to predict MPVA (Babic et al., 2014). Although Slykerman et al. (2016) suggest that for children less than 8 years old perceptions of their competence tend to be inflated as movement competence is still developing. However, Slykerman and colleagues concluded that the relationship between skill type and PA behaviour was not clear, although it was influenced by age, sex and instrumentation differences.

It is important to note that the model proposed by Stodden et al., (2008) acknowledged the developmental changes in children's self-perceived competence. They suggested that young children lack the cognitive ability to distinguish accurately between effort and actual motor skill competence and over inflate their ability, because they equate expenditure of effort with skill. Whereas, by middle childhood, when cognition has developed further, they are able to compare themselves with their peers so that their actual motor competence relates more closely to their perceived motor competence. That is, those children with lower motor competence will have lower perceived motor competence and therefore, Stodden et al. (2008) predict a lower engagement in They argue that this leads to vulnerability of children with low motor PA. competence to withdraw from PA and run health risks of being sedentary, for fear of publicly displaying their low motor competence. However, Babic et al. (2014) suggest that the relationship between self-perception of motor competence and engagement in PA is less strong in late adolescence. One suggestion is that team sports become less important and teens engage more in resistance training, walking, aerobics and other non-traditional activities, possibly less likely to be captured on traditional scales and so effecting the results (Babic et al., 2014).

Children with DCD participate less in physical activity (Rivalis et al., 2011) and also have been reported to have lower self-perception of athletic competence and social acceptance (Skinner & Piek, 2001). A positive self-perception of motor skill competence is going to be important for engagement of children in PA, but the exact nature and relationship requires more study. Other factors may be equally important.

Barnett et al. (2016) conducted a systematic review to identify potential correlates of gross motor competence in TDC aged 3-18 years using an ecological approach. Barnett et al. (2016) suggested that ecological models could be used to frame potential influencing factors, as they emphasize environmental contexts of the behaviour as well as the psychological and social influences. This way multiple spheres of influence can be understood in-depth and can help guide the development of intervention. They found that, of the 59 studies published between 1995 and 2014, the most examined correlates were biological and demographic. Findings included that a healthy weight status, being male and coming from a higher socioeconomic background were all consistent correlates, but only for certain aspects of motor competence, depending upon how it was operationalized. Only increasing age was consistently a correlate of all aspects of children's motor competence. PA was a positive correlate of motor coordination, but there was indeterminate evidence for PA being a correlate of object control or locomotor skill competence. Moreover, few studies investigated cognitive, emotional and psychological factors, cultural and social factors or physical or environmental factors (Barnett et al., 2016). This suggests that more research is required, using an ecological framework, as there are gaps in current understanding of perceived motor competence and potential correlates of motor skill competence.

### 3.10 Parental support and attitude to participation in PA

An area of study thought to influence children's participation in PA is parental attitude to PA and parental behaviour towards PA. However, before exploring this, it will be helpful to define and explain participation. Participation is often described as a key rehabilitation outcome and has been shown to be an

indicator of overall health and well being across the lifespan (Bedell et al., 2013). The ICF defines participation as 'involvement in a life situation' and participation restriction as 'problems and individual may experience in involvement in life situations' (WHO, 2001). Imms et al. (2016) conducted a systematic review of the language, definition and constructs used in research on participation of children with disabilities, as there was no universally accepted definition of participation. They found emergence of five themes from their analysis of studies on participation: preferences, attendance and involvement (related to 'being there' and subjective 'in the moment experiences'), activity competence and sense of self. They suggest that the themes are linked, because preference for a particular activity is likely to lead to attendance to engage in it and then involvement becomes possible. Attendance and involvement may lead to competence in the activity and improve self-confidence, which in turn may influence preferences (Imms et al., 2016).

This may well be the case for children with DCD, as they are thought to have reduced participation in physical activity. Cairney et al. (2005) found that boys with DCD had significantly lower participation rates in recreational PA than typically developing boys, a finding also supported by Batey et al. (2014). Moreover, Poulsen et al. (2007) found that children with DCD were unsuccessful in engaging in team sports and reported loneliness. Clearly, children with DCD are at risk of missing out on the social and physical benefits of engaging in PA and, from an embodied cognition perspective, missing potentially improving their experience and competence in motor and a number of other domains.

Welk (1999) studied the determinants of physical activity in children and concluded that they were multifactorial. He suggested that the family plays an important role in shaping children's activity habits through reinforcing factors, such as taking children to parks and exercising with them as well as parental encouragement. He also proposed a model for a conceptual framework for

understanding factors that may predispose, enable and reinforce a child to be physically active. It included multiple levels of influence (intrapersonal, sociocultural and environmental) and supports the idea of reciprocal interaction between the child, the environment and the behaviour (Welk, 1999). King et al. (2003) agree that participation is a complex concept and is influenced by personal factors related to the child and family and to environmental factors and even more complex for children with motor difficulties. Anaby et al. (2014) found that some of the most common facilitators of participation for children with disabilities involved social support of the family and friends and geographic locations and the most common barriers included attitudes, physical environment, transportation, policies and lack of support from service providers. Therefore, any research examining participation in PA should include an ecological approach to understand some of the family and environmental factors influencing the child.

### 3.11 Chapter Summary

Current understanding of motor development has shifted from considering the child in isolation to include a more transactional approach to the development of the child within his or her environment. Dynamical systems theory facilitates consideration of the nature of the role of the environment (physical, social and cultural) in providing affordances or barriers for children's self generated motor experiences, for the optimal development of motor and other skills, from our current understanding of embodied cognition. Traditional views about the nature of change in motor ability have also been discussed in relation to variability in motor performance and it's potential to distinguish typical from atypical motor development. It has also been noted that motor skill acquisition in children not only relies on practice, but is also influenced by experience, perceptual, cognitive and social factors and others such as perceived competence and parental and cultural influence. However, without exploring the progression of motor abilities from an ecological perspective, some of these important influences could be overlooked. Bronfenbrenner's bio ecological model is a useful framework with which to examine some of these factors. DCD is a disorder of motor development, yet we still do not fully understand how and

why the motor development of children with DCD progresses differently over time. The next chapter will explore some of the issues in DCD research that seeks to answer some of these issues.

# CHAPTER 4 CURRENT ISSUES IN DEVELOPMENTAL COORDINATION DISORDER

Understanding the course and prognosis of a disorder helps with the planning of services for appropriate intervention. However, this is not straightforward in DCD, as tensions exist associated with the identification and assessment of children with DCD and the way these influence how DCD is studied over time. Longitudinal study is a way to chart developmental progress and a number of such studies have been undertaken in the last three decades for DCD. This chapter will begin with a discussion of the prevalence of DCD and current knowledge of the course and prognosis of the disorder. Some longitudinal studies in DCD will be discussed and a number of methodological issues will be considered with the implications that they have on the study findings.

### 4.1 Prevalence of DCD

The prevalence of DCD is often reported as 5-6% of the population (DSM-5, APA, 2013). However, since there are no biomarkers for DCD and the diagnosis relies on interpretation of behavioural criteria, a few issues have arisen. For example, different studies have reported widely different prevalence rates ranging from 1.8 to 19.0% (Lingam et al., 2009; Tsiotra et al., 2006; Wright & Sugden, 1996), exposing discrepancy in adherence to the diagnostic criteria and measurement issues concerning the use of specific assessment tools (Zwicker et al., 2012; Barnett, 2008). Wright & Sugden (1996) questioned the prevalence rate in DCD, as it was rarely based on epidemiological studies and was subject to definitional difficulties. They chose a random population in Singapore and found the rate of DCD at 16% of children, 12% "at risk" with moderate impairment and 4% seriously impaired. Later, an epidemiological study by Lingam et al., (2009) used DSM-IV criteria and found a prevalence rate, in a UK birth cohort, of 1.8% severe DCD and 4% moderate impairment at age 7 years. However, not all DCD research can be based on epidemiological studies due to the expense involved. Clearly, prevalence rate will be dependent on both strict adherence to the diagnostic criteria adopted for DCD and the cutoff levels of the motor assessment used and so any future research needs to make these explicit. To add a further problem, not all motor assessments measure the same things, as different tests will identify different children and no motor test covers the full range of motor skills (Albaret & de Castelnau, 2007).

A further complication is that there is still no agreement among experts about the threshold for motor cut-off for DCD diagnosis, as EACD recommend 15<sup>th</sup> percentile whilst the Leeds consensus recommends 5<sup>th</sup> percentile. Furthermore, different published studies used different motor cut-off thresholds, for example 5<sup>th</sup>, 10<sup>th</sup> or 15<sup>th</sup> percentile and then did not necessarily analyse the results separately, making comparison across studies difficult. However, Larkin & Cermak (2002) described cut scores (cut-off thresholds) as an arbitrarily designated point of motor impairment and suggested that few tests can reliably distinguish between examinees with adjacent scores, further warning that motor test scores devoid of context are a naïve way to identify motor problems. Therefore, in order to appropriately identify children with DCD, an appropriate motor assessment (with explicit cut points) in addition to children's contextual information is necessary.

### 4.2 Gender ratio

Another consideration is the disproportionate ratio of males to females (3:1) (APA, 2000) often quoted in DCD research. Nevertheless, this has been recently questioned by employing population- based sampling and finding more equal sex ratios (Missiuna et al, 2011). A wider question is whether there is actually a lower prevalence of DCD in girls or, whether it is an under representation of girls in the published research or just reflects a higher rate of referral of boys to health and educational support services for movement problems. Either way, this warrants further investigation to establish if the 'gatekeepers', such as parents and teachers are inadvertently under referring girls for DCD assessment, as cultural expectations of girls motor performance is lower.

# 4.3 Measurement issues and fulfiling diagnostic criteria for DCD

The importance of internationally agreed criteria for the identification and diagnosis of DCD has already been discussed, however, the type of screening and assessment tools employed will also have a significant bearing on the identification and classification of children with suspected motor difficulties for the DSM criteria.

### 4.3.1 Measurement tools for criterion A:

Many tools identify motor problems in children suspected of having DCD and help to fulfil criterion A in the DSM diagnostic criteria. However, one criticism is that the motor tests employed for the diagnosis of DCD have tended to emphasize motor proficiency rather than the development of coordination (Larkin and Rose, 2005). Examples of these include the Movement Assessment Battery for Children (MABC) (Henderson & Sugden, 1992), the McCarron Assessment of Neuromuscular Development (MAND) (McCarron, 1982) and the Bruininks Oseretski Test of Motor Proficiency (BOTMP) (Bruininks, 1978). However, the high correlation between the total test scores of the MABC the MAND and the BOTMP and the validity studies of all three tests provides evidence for a general construct of motor ability or motor coordination (Larkin and Rose, 2005) and therefore validates their use in the identification of DCD. The BOTMP2 and the MABC2 are revised versions of their precursor tests and include extended age ranges to cover adolescence, which is an important addition for longitudinal research. It is often said that there is no 'gold standard' motor test, however the EACD guidelines endorsed the use of the MABC2 in the assessment of DCD (Blank et al., 2012) because it has good validity and is widely used.

Kaplan et al (1998) raised some important issues about identification and classification of the developmental disorders. They used three different assessments to classify DCD, namely two motor assessments the Bruininks-Oseretski Test of Motor Proficiency (BOTMP) (Bruininks, 1978) and the

Movement Assessment Battery for Children (MABC) (Henderson & Sugden, 1992) and a parent questionnaire the Developmental Coordination Disorder Questionnaire (DCDQ) (Wilson, Dewey & Kaplan, 1998). This study not only highlighted interesting points about choice of cut-off scores, but also recognized that different tests identified different children with DCD. The 15<sup>th</sup> percentile was chosen as a cut-off for MABC and <42 standard score for BOTMP and less than 1 SD below the group mean for the DCDQ. However, this led to a DCD prevalence rate of 21.4%, which included both the index group and the typically developing controls, and is much higher than previously reported by other studies, which stated rates from 2.7% to 15.6% (Wright, 1997). Furthermore, even 13% of the control group was classified as DCD, which seems disproportionately high and perhaps suggests too lenient inclusion criteria.

It also highlights the concepts of sensitivity (how accurate) and specificity (ability to determine DCD children from non DCD) in testing, which can lead to over diagnosis or 'false positives', identified by a test, and may well have been the case here. Moreover, Kaplan et al. (1998) recruited subjects between 1992 and 1997, which indicates that the start date of the study preceded any consensus on diagnostic criteria for DCD. It is unclear if this sample represented what we now understand as DCD or a more heterogeneous group. There are myriad reasons for motor impairment, including those resulting from a medical condition or neurological damage. Medical reasons can explain underlying aetiology and therefore the children can differ considerably from children with motor impairment without a known reason. This highlights again the importance in research about clear agreement on diagnostic criteria and is the reason that the research community has now adopted DSM-5.

Albaret & De Castelnau (2007) recommend that motor cut points should not be the only basis for decision for diagnosis, as the nature and effects on everyday life may be more pertinent. One way of obtaining this information is by way of a criterion-referenced checklist.

### 4.3.2 Checklists to assess impact of motor impairment on function for criterion B

Information gleaned from someone that knows the child well and can observe how motor impairs function on a daily basis will fulfil criterion B. A checklist provides a relatively quick and inexpensive way of collecting this information.

Wright & Sugden (1996) described a two-step procedure that provided a cost effective way of screening children with movement difficulties, so that only those at risk would need to undergo full norm referenced motor assessment. The MABC checklist was completed by parents and teachers to indicate where there were functional problems and any child, whose score indicted problems severe enough to suspect DCD, was then assessed with the MABC. Screening like this permits the easy assessment of large numbers of children and has been adopted by large-scale population studies, such as the Canadian study by Missiuna et al. (2011).

Several checklists are available for use with children suspected of DCD. These include the Developmental Coordination Questionnaire (DCDQ) (Wilson, Kaplan, Crawford, Campbell & Dewey, 2000), the DCDQ (07) (Wilson & Campbell, 2007), the MABC checklist (Henderson & Sugden, 1992), the Early Years Movement Skills Checklist (Chambers & Sugden, 2002), the Children Activity Scales for Parents and the Children Activity Scales for Teachers (Rosenblum, 2006) and the MABC2 checklist (Henderson, Sugden & Barnett, 2007). Screening tests must have sound psychometric properties, with appropriate levels of sensitivity (how many children are correctly identified with impairment) and specificity (how many children are correctly identified as free from impairment). Unfortunately, these tests had variable results relating to their sensitivity, with correct identification of children with DCD ranging from 50-100% (See Barnett, 2008 for a discussion). A usual minimum expected level of sensitivity is 80% (APA, 2000). Despite this, checklists offer valuable observations of the child's performance in their everyday context and identify areas for intervention to help improve the child's daily function.

Delayed motor milestones (such as sitting, crawling and walking) are often early symptoms of DCD (APA, 1995, p56). Parents frequently report that their pre-school children with DCD had great difficulty acquiring self-care skills, such as using cutlery, dressing and grooming and that any play requiring balance and coordination was problematic. This information can be obtained by parent interview or questionnaire and is an important part of the diagnostic process.

However, many of these symptoms are also present in children that are born prematurely or with very low birth weight, which has led to a debate about whether these children should receive a diagnosis of DCD or not. Both subtle learning problems and motor clumsiness have been noted in children born prematurely or with low birth weight (Foulder-Hughes & Cooke, 2003). Furthermore, an association between the motor competences at six years old and the duration of flares in the periventricular white matter of preterm babies has been found (Jongmans et al, 1998), inferring subtle damage to the brain. Moreover, an increased prevalence of DCD at aged 8 has been reported in extremely low birth weight and extremely premature babies (Roberts et al, 2011). This has led to debate about the appropriate application of the DCD diagnostic criteria and inclusion/exclusion criteria in the DSM IV (see Barnett, 2011 for a discussion). Given the known increased risk of cerebral palsy (CP) in this group of children there is also now some question about the notion of a cerebral palsy-DCD continuum (Pearsall-Jones, Piek & Levy, 2010a), suggesting that DCD in this group may be a mild form of CP. A detailed birth history should be included in any assessment and noted in any samples for research, in order to determine if these children constitute a different group in relation to symptoms and intervention.

Williams et al. (2014) reviewed existing research on risk factors and neurophysiological evidence to support understanding of underlying neural mechanisms for CP and DCD. They concluded that, although there are common peri- and neonatal factors for CP and DCD, the neurophysiological data are conflicting. There is, however, evidence of overlap in micro- and macro-structural abnormalities of the two disorders, but only when DCD samples include children exposed to pre- and peri-natal adversities. They established that no firm conclusions can be drawn on the CP-DCD continuum at present, but the most likely candidates for a CP–DCD continuum would be a subgroup of DCD born preterm, given their significant overlap in risk factors. However, more research comparing samples of children with DCD with and without pre- and perinatal adversity is required.

### 4.3.4 Motor impairment not better explained by known neurological disorder, visual or intellectual impairment for criterion D

The presence of intellectual impairment in DCD causes some tension with this criterion. For example, the Leeds consensus recommended that if a child had an IQ, presumed or measured, below 70 they should not be given the diagnosis of DCD (Sugden et al. 2006, p7). The argument presented was that within the typical range of IQ the relationship between motor competence and IQ is low, but as soon the IQ drops below 70 the relationship begins to rise (Sugden & Wade, 2013). Therefore, children with moderate learning difficulties (IQ 50-70) are more likely to have motor difficulties than TDC.

However, DSM IV TR (APA, 2000) does not stipulate an IQ threshold, merely stating that if mental retardation is present the motor difficulties should be 'in excess of those associated with it'. DSM5 (APA, 2013) is even less specific, stating that the motor skills deficits should 'not be better explained by intellectual retardation'. Furthermore, the EACD guidelines concluded that a specific IQ level does not seem helpful to distinguish between children with DCD and those with coordination problems due to mental retardation (Blank et al., 2012, p63). The situation thus appears to be open to the professional interpretation of a clinician or researcher. The only caveat is that an idea of intellectual ability should be obtained for any child with a potential diagnosis of DCD, in order to gauge whether the motor deficits are in excess of those expected for the IQ level. Ideally, a measure of IQ should be made; where not

feasible a teacher's opinion or national tests results may be acceptable (Sugden & Wade, 2013, p232).

The second issue concerns ruling out other diagnoses such as visual impairment or neurological conditions. Zwicker & Mickelson (2017) undertook a prospective cohort study to determine which other conditions could present like DCD using DSM5 criteria. They adopted a two–step procedure using the DCDQ checklist (Wilson et al., 2007) completed by a caregiver, a standardized motor test completed by an OT, a neurological examination and developmental history completed by a paediatrician. Over 100 children suspected of DCD were assessed and 71% met the criteria for DCD and 29% met the diagnostic criteria for another disorder. Of the 29% who met the criteria for other conditions, 50% were diagnosed with a neurodevelopmental disorder (e.g. ADHD, foetal alcohol syndrome, ASD or learning disability), 25% were diagnosed with a neurological condition (e.g. CP, hypotonia, seizure disorder, Chiari I malformation) and the remaining 25% were diagnosed with a genetic or medical condition (e.g. Ehlers-Danlos syndrome, Neurofibromatosis Type I, microdeletion syndromes).

This study highlights the importance of professional awareness of differential diagnoses for DCD with known neurological disorders, other neurodevelopmental disorders and medical conditions and for them to recognize the red flag symptoms so that the child can be referred on to a specialist for the appropriate assessment and intervention. Zwicker & Mickelson (2017) suggest that a paediatrician should assess all children suspected of having DCD and, whilst this is essential in a clinical setting, it is not always possible in a research setting.

A recent review by Smits-Engelsman et al. (2015) investigated how well authors, who had published between 2010 and 2015, had complied with DSM criteria in the description of the children they selected with DCD. They summarized 176 papers and found that most papers provided detailed

information on motor performance using standardized scores with reported cutoff values, but few provided sufficient detail about how IQ or other diagnoses had been ruled out and only 12% of papers had reported that a paediatrician had assessed children. This can be problematic in a number of ways as it makes it difficult to compare studies. We cannot be sure that all those labelled DCD actually meet the DSM criteria and it also does not rule out confounding variables, such as level of IQ and presence of other conditions. This is particularly important in longitudinal research investigating the course and outcome of DCD, as it is difficult to attribute the outcomes to DCD without first ruling out these confounding variables.

#### Summary

Diagnosis for DCD is not straightforward, despite internationally agreed diagnostic criteria. Although research in the field of DCD has greatly increased awareness of the disorder, future research should make explicit how the diagnostic criteria have been met in order to permit more detailed comparison across studies. Identification and diagnosis are an important first step, but given the complexity and heterogeneity of the disorder, understanding the course and prognosis of DCD are equally important in order to facilitate the development of appropriate interventions. Longitudinal studies are an important way to establish the nature of a disorder and how it may progress over time. A number of longitudinal studies have contributed to our current understanding of DCD, but they are not without methodological issues, which need to be clarified.

### 4.4 Longitudinal evidence: course and outcome of DCD

Longitudinal studies allow the opportunity to assess the same children over a number of occasions to monitor the changes that occur and relate the changes in one skill to changes in another (Hulme and Snowling, 2009). This facilitates the description of patterns of change and allows researchers to establish the direction (positive or negative) and magnitude of causal relationships (Menard, 2002). Moreover, it would be impossible to disentangle developmental (age),

historical (period) and cohort effects on change without longitudinal data (Menard, 2002) and only longitudinal data reflects intra-individual change.

However, analysis of the course and outcome of DCD is not straightforward due to many methodological constraints in the published longitudinal studies. For example, different studies used different diagnostic criteria to identify DCD, and pre- 1994 there were no consensus agreement about which criteria to use. Earlier studies from the 1980s, then followed up in the 1990s have informed our understanding of DCD, and although some had DSM criteria retrospectively applied, there is still an element of uncertainty about how all the criteria were met. Even after the 1994 consensus agreement, not all studies identified how they had met DSM criteria (Geuze et al., 2001). Smits-Engelsman et al. (2015) also noted that the situation had not changed much over a decade later, even following the 2006 Leeds consensus, as few published studies stated how DCD diagnostic criteria were met. The situation is compounded by use of different measurement and screening tools, which make direct comparison between studies more problematic, as different tools measure different items and can potentially identify different children (Albaret & de Castelnau, 2007).

Longitudinal studies concerning DCD also have different lengths of follow up time, ranging from 1.5 (Pless et al., 2002) to over 15 years (Rasmussen & Gillberg, 2000). This also makes comparison difficult, because of the different developmental stages of the children studied. For example, Knuckey & Gubbay (1983) found that children with mild to moderate motor difficulty aged 8-12 years had improved by age 16-20 years, and proffered developmental lag as a potential cause for DCD. However, there is some debate about the role of adolescence in motor progression in DCD (Cantell & Kooistra, 2002; Visser et al., 1998) and whether children actually improved during this period or whether it was a ceiling effect of the different motor measures that were only standardized up to age 12 years. Moreover, difference in the theoretical approach taken by the different studies inevitably affects the method and type of data collected and how it is interpreted.

Cantell & Kooistra (2002) summarized the longitudinal studies pre 2000 and suggested that they fell broadly under two different categories. The first was the optimistic approach, which considered DCD largely a disorder of childhood and on the lower end of a continuum of motor severity. The suggestion was that DCD was possibly due to a developmental lag, as study findings showed that some children had improved by late adolescence (Hall, 1988; Knuckey & Gubbay, 1983). The second was more pessimistic and was based on study findings that showed persistent impairment in the children with more severe symptoms (Ahonen, 1990; Geuze & Borger, 1993; Hellgren et al., 1993, Losse et al., 1991). Furthermore, Cantell et al. (1994) and Cantell (1998) suggested that severity of symptoms related to the persistency of DCD, since they found that children with mild to moderate difficulties improved over time to a typical level, whilst those with severe problems had persistent DCD.

### 4.4.1 Different motor outcomes for children with DCD

This differential motor outcome for children with DCD deserves closer investigation, since confounding variables such as IQ, the presence of other conditions and even the different experiences due to socio-economic context can influence outcome. However, the possible significance of children with milder symptoms, often referred to as a moderate DCD/at risk DCD group, could be important and further examination of findings from longitudinal studies can help uncover more detail.

In order to understand the nature and impact of a developmental disorder it is necessary to relate it to typical development (Hulme & Snowling, 2009), otherwise any changes noted may be attributed to the disorder rather than the change as the child develops. A number of longitudinal studies comparing children with DCD to typically developing children (TDC) have been reported (Geuze & Borger, 1993; Losse et al., 1991; Cantell, Smyth & Ahonen, 2003; Rasmussen & Gillberg, 2000; Lingam, Golding, Jongmans, Hunt, Ellis & Emond, 2009; Cairney, Hay, Veldhuizen, Missiuna & Faught, 2009). There is clear empirical evidence that DCD persists into adolescence and adulthood and is not a benign disorder.

However, findings of particular note from the earlier studies indicate that children with DCD, significantly differ from typically developing children (TDC) five or even 10 years later, but nevertheless show a wide variation from each other. For example, Geuze & Borger (1993) tested children aged 6-12 years matched for age, sex and school and found that 50% of the children identified with DCD still displayed motor impairment 5 years later (aged 11-17 years) and scored below 15<sup>th</sup> percentile on a test for motor ability, whereas the other 50% scored above the 15<sup>th</sup> percentile, with some even achieving a score in the typical range. This wide variation, despite a small sample size (DCD n=12; TDC n=14), warrants explanation as it could indicate different trajectories for children with DCD. However, several questions remain about the sensitivity of the test used, as it was only standardized for children up to age 12 years, and may not have been able to detect differences in the older age groups. The DCD group also reported less social contact and less participation in sport, perhaps indicating ongoing difficulties, despite achieving motor competence in the test. Unfortunately, the children were not matched for IQ or socio-economic status (SES), both of which could have an impact on social and sport participation.

Yet, other studies report similar findings, Cantell, Smyth and Ahonen (2003) found significant group difference between DCD and TDC children, but that the DCD children appeared to follow different trajectories from each other in their study, which followed children from 5 to 17 years of age. The difference in this study was that a distinction was made between the severities of perceptual motor difficulty (significant or minor) of the children with DCD. Those with severe perceptual motor problems appeared to continue to have difficulties, whereas those with only minor perceptual motor problems (the intermediate group) appeared to perform at a level close to the control group by age 17. Cantell et al (2003) suggested two possible explanations for the apparent catch up; either the tasks used to distinguish the groups lost adequate power with increasing age, or the intermediate adolescent group really caught up to their typically developing peers. They reviewed other follow-up studies and concluded that all the DCD children studied reported developmental change,

but not in a homogeneous pattern. Some studies reported improvement or "repair" by the onset of puberty (Sooranami-Lunsing et al., 1994; Visser et al., 1998), with suggestions of the benefits of physical growth, adaptation of the central nervous system and increasing movement experience with age, as possible explanations for the improvement in motor performance (Cantell et al., 2003). In addition, they questioned the confounding role of IQ in outcome and noted that the children in their study with lower IQ had a poorer school outcome.

However, there may be a more complex interaction at play here, as the results from another longitudinal study by Losse et al. (1991) found that even the DCD children with high IQ aged six had poorer academic outcomes than controls 10 years later. Another finding, indicating that multiple factors are involved in the functional outcomes for children with DCD, was that variations in the selfconcept and motor ability of the DCD group; their self-concept bore no pattern to their motor ability or PE attainment (Losse et al., 1991). Another important finding was that the children reported intense feelings of personal failure and many had additional problems with handwriting, concentration and behaviour and reported less enjoyment in sport, leisure and school. It may be that associated problems in non-motor areas have a larger impact on the developmental trajectory and functional outcome for children with DCD, or possibly different subtypes of DCD, which could account for such heterogeneity.

#### Summary

There is strong empirical support that children with DCD differ from TDC in their motor performance over time, but equally there is considerable intra group variability between children with DCD. Even within groups of DCD children with severe motor impairment heterogeneity has been reported (Kirby, 2005). It is therefore not clear which factors have the greatest influence on motor progression for children with DCD, but co-occurrence of other disorders, existence of subtypes of DCD, varying levels of IQ, self perception of their motor abilities, family context and even the presence of secondary problems have all been the focus of previous investigation, but methodological issues

have clouded the picture. Questions such as these will remain unanswerable unless there is strict adherence to the diagnostic criteria for DCD. Moreover, valid and reliable tools need to be used for measurement with clear cut-off levels indicating severity of motor difficulty, so that it can be taken into account and identify separate groups of children. Furthermore, confounding variables such as IQ and SES should also be reported.

# 4.4.2 Heterogeneity in DCD, variable outcomes and possible lines of inquiry

A potential problem in DCD research is that investigations have relied on comparison designs that measure levels of performance on a number of variables between the DCD and TDC. Fisk & Rourke (1979) point out that this strategy assumes implicitly that the children under investigation (in this case the DCD group) constitute a homogeneous population. Clearly, this is not what empirical evidence shows. They therefore suggest investigating subtypes to explore the heterogeneity. Several studies have explored the possibility of subtypes in DCD using an empirical technique known as cluster analysis.

### 4.4.2.1 The search for subtypes to explain heterogeneity

Wright and Sugden (1996) found large intragroup differences in the children with DCD and used cluster analysis to search for subtypes in DCD. In this approach a set of measurements is obtained and the participants are statistically grouped together on the similarities of their profile of scores (Visser, 2007). Several studies have been conducted using this approach (Hoare, 1994; Wright & Sugden, 1996; Miyahara, 1994; Dewey & Kaplan, 1994; Macnab et al., 2001; Green et al., 2008). However, it has been difficult to compare the results because the participants have been drawn from different countries (Australia, Singapore, USA, Canada and UK), different populations (3 from clinics, 3 from schools) using different diagnostic criteria, different tests of motor ability, different statistical tests, different theoretical positions leading to the choice of different variables to form the clusters (see Macnab et al., 2001 and Visser, 2007 for reviews). Despite this, there is some overlap between the

clusters identified in the studies and all have found a subtype with generalized sensorimotor deficit, but as Visser (2007) explained this is the only group that will show up regardless of the variables used in a study. Macnab et al. (2001) set out to replicate the study by Hoare (1994) and, despite the differences previously mentioned, were able to replicate five clusters fairly well. Green et al. (2008) adopted a similar procedure and also found five clusters, two of which showed a similar pattern to those of Hoare (1994), suggesting that this may be a valuable technique for future studies, providing clear information is stipulated about the population sample and the variables chosen.

However, Visser (2007) argued that subtype studies have contributed little to our understanding of DCD, as it does not explain aetiology and suggested that it would be more fruitful to distinguish subtypes in terms of underlying deficits and examine the developmental trajectories. Furthermore, he suggested that the presence or absence of comorbidities has a direct relevance to the way subtypes of DCD are defined and understanding will only improve if we know why comorbidity is linked to a particular subtype (Visser, 2007, p98).

This is supported by a recent Australian twin study that examined the relationship between movement problems and four developmental disorders by Martin, Piek et al., (2010). They found that the likelihood of comorbidity increased with symptom severity and that the results varied with the presence or absence of reading disorder. Despite the need for caution when extrapolating results from the twin population to the wider population (because of the higher incidence of developmental disorders in twins), future research should at least note the presence of additional developmental disorders in children with DCD.

### 4.4. 2.2 The influence of Comorbidity/co-occurrence of other disorders with DCD

Comorbidity is described as the simultaneous occurrence or two or more unrelated conditions and epidemiology shows that it is extremely common in child psychiatry (Caron & Rutter, 1991). Comorbidity is considered when the observed rate of comorbidity exceeds that expected by chance alone; this can be obtained by multiplying the base rates of each of the separate conditions involved (Caron & Rutter, 1991). Common developmental disorders associated with DCD are ADHD, SLI, RD and ASD. According to DSM IV (APA, 2000) the reported prevalence rates of these disorders are DCD 6%, ADHD 3-5%, RD 4% and ASD greater than 1% (Baron Cohen et al., 2009) respectively.

The effect of comorbidity of one developmental disorder with another can have significant implications for learning and progression and, therefore, on the choice of the mode of intervention. Attribution of a problem to a disorder without closer examination and better understanding about the nature and cause of the problem could potentially be attributed to the wrong cause and lead to educational or clinical decisions that may not be appropriate for the individual. Therefore, as comorbidity has been noted in DCD (Kaplan et al., 2006; Martin, Piek et al., 2010), as explained in chapter two, careful consideration of the impact of each disorder should be taken when examining progress and outcome in DCD.

However, Kaplan, Crawford, Cantell, Kooistra and Dewey (2006) questioned use of the term comorbidity associated with DCD and suggested use of the term co-occurrence instead, because the aetiology of DCD is unknown. They argued that the term co-morbidity implies that the nature of underlying cause for each diagnosis are separate from each other, whereas continuum suggests linear severity and probable related causality and co-occurrence implies a purely temporal concept that may or not reflect underlying aetiology (Kaplan et al, 2006). Therefore, the term co-occurrence will be used interchangeably with comorbidity here as it relates to the literature, but no underlying aetiology is inferred.

#### 4.4.2.3 The co-occurrence of ADHD with other disorders

Two of the most common developmental disorders are ADHD and Dyslexia (RD) which occur in approximately 5% of the population, but their rate of cooccurrence has been reported between 18-45% and both molecular genetic and behavioural studies support a partly shared genetic aetiology between ADHD and RD (see a review by Germano, Gagliano and Curatolo, 2010). Moreover, children comorbid for both conditions appear to have more severe cognitive deficits, leading Germano et al. (2010) to suggest that RD could be a marker for a group of children with ADHD with more severe cognitive deficits and worse academic and behavioural outcomes.

Another common comorbidity is ADHD and language impairment (LI). Cohen et al. (2000) examined this in a 2x2 (ADHD, LI) study adhering to strict DSM III criteria and IQ  $\geq$  85. Their findings demonstrated that the children with ADHD plus LI were associated with worse academic achievement measures and lower performance on working memory tests, leading them to state that this group were at greater risk and call for routine assessment of language competence, as it could have implications for intervention.

Moreover, studies in the area of specific language impairment (SLI) appear to indicate considerable overlap between SLI and RD. For example, McArthur et al. (2000) found that over 50% of 212 children studied met the criteria for both SLI and RD. Furthermore, studies on the developmental trajectory and functional outcome for children with SLI make interesting, if pessimistic reading. Knox (2002) investigated academic achievement and followed 100 school children with SLI and, even after accounting for level of support and special access arrangements, found results similar to Conti-Ramsden (1992), that these children performed much worse than the national average in all subjects, especially English. Therefore, children with SLI are at considerable risk of global academic problems (Tallal et al., 1997).

This highlights the danger of attributing poor outcomes and poor progress in school to DCD alone, without first accounting for the co-occurrence of other disorders, such as ADHD, SLI or ASD in children, as the interaction of different

developmental conditions may have a significant impact on the academic and behavioural outcomes for children.

### 4.4.2.4 DCD and co-occurrence with other disorders

Co-morbidity is widely reported in DCD. Moreover, a population based cohort by Lingam et al., (2010) found that children had an increased risk of difficulties in attention, short-term memory, academic and social skills if they also had DCD.

ASD and DCD are commonly reported comorbid. For example, Green, Baird et al. (2002) found that the incidence of motor impairments reported in the literature on ASD ranged from 73-90%. By using DSM IV criteria for ASD and DCD they tested a group of boys with Asperger's syndrome and a group with DCD, and reported that their motor performance was quite similar. Despite considerable variability being present in this group, the most severe cases of motor impairment were found in the Asperger's group and these boys clearly met the diagnostic criteria for both conditions.

Hill (2001) reviewed the literature on neurodevelopmental disorders of language and movement and concluded that there is substantial comorbidity between SLI and DCD. Gaines & Missiuna (2006) also found high comorbidity of SLI and DCD in kindergarten children and called for motor screening, as children with SLI appeared to be at greater risk of DCD than TDC.

DCD is commonly reported to be comorbid with ADHD. Indeed Kadesjo and Gillberg (2001) found, in a population based study in Sweden, that 87% of children who met full diagnosis for ADHD (DSM –III criteria) also had one or more comorbid diagnoses and 67% at least two diagnoses. The most common types of comorbidities reported with ADHD include oppositional defiant disorder; conduct disorder, affective and anxiety disorders, reading disorders and DCD (Kadesjo & Gillberg, 2001). In addition, Pearsall-Jones, Piek & Levy (2010)

found ADHD commonly associated with mood disorder. Therefore, it is possible that a child with DCD comorbid with ADHD could easily experience additional comorbidity. Although Martin, Piek et al. (2010), using latent class analysis, found that motor problems and ADHD were not found in the same class unless other disorders were also present. However, they did find a distinct class of children with motor only problems, but primarily associated with gross motor problems, they also found that fine motor problems appeared closely linked with reading problems rather than ADHD. Comorbidity is clearly prevalent in many developmental disorders including DCD, but more research is required to ascertain the exact nature of the association.

#### Summary

Co-morbidity is relatively common in developmental disorders and studies indicate that it can have deleterious effects, indicating that children with co-morbidities are likely to have more severe problems. Therefore, when examining the course and prognosis of DCD it is important to understand the course and prognosis of the disorders most likely to co-occur with DCD.

### 4.4.2.5 Evidence from the trajectories of other developmental disorders

DSM-IV specified three subtypes for ADHD, the combined, the inattentive, and the hyperactive/impulsive. Larsson et al. (2011) used data from a population based longitudinal twin study to examine the trajectories of these subtypes based on parent ratings. They found instability in the subtypes over time and children shifted from combined to inattentive and hyperactive to combined, which suggests that ADHD subtypes cannot be viewed as stable or discrete categories. Furthermore, a review of literature on attention problems and academic achievement concluded that inattentive ADHD is associated with poorer academic outcomes (Polderman et al., 2010). Thus the subtype category of ADHD in the formative school years may be of key importance.

Subtypes of DCD have not yet been agreed. However, because of the heterogeneity of both motor and non-motor symptoms and the vast differences

in severity, detailed records are required to help our understanding. It is vital that future research records these characteristics and those associated with cooccurring disorders, sometimes referred to as associated characteristics (AC), in order to establish an account of progression over time.

Previous research suggests that DCD does not have a uniform pattern of problems. However, empirical evidence supports the persistence of DCD through childhood into adulthood and is has been confirmed as a lifespan condition (Barnett, 2008). This provides another source of information about the course and outcome of DCD, as attention has turned to the experiences of people with DCD. Opportunity for first hand experiences of DCD symptoms, including those of co-occurring conditions, and the type of activity and participation restrictions that adults and children with DCD encounter can be used to inform better understanding of the nature and course of DCD and inform intervention.

# 4.5 Evidence from experiences of adolescents and adults with DCD and the importance of participation

Data emerging from experiences of adolescents and adults with DCD demonstrates that they encounter persistent difficulty in both motor and nonmotor areas. For example, Kirby, Sugden, Beveridge & Edwards (2008) questioned students with DCD and other developmental conditions in Further and Higher Education. They found that students with DCD reported on going problems with writing and self-care and were more likely to live at home than other students. Students with DCD also reported non-motor difficulty such as executive function difficulties greater than those experienced by students with dyslexia. More recently, Kirby and colleagues developed the Adult Dyspraxia/DCD Checklist (ADC, Kirby, Edwards, Sugden & Rosenblum, 2010), a structured questionnaire for adults with DCD. Young adults (n=19) aged 17-25 years meeting DSM IV criteria for DCD were questioned with the ADC, and nearly all had continuing coordination difficulties affecting areas such as writing
fast, playing team games, driving and parking. They also reported executive function problems impacting areas such as money management and organizing belongings. Importantly, they also reported greater isolation, avoiding team games, avoiding going to clubs or dancing and 50% reported participating in leisure time alone (Kirby, Edwards & Sugden, 2011).

These studies not only highlight the persistent nature of DCD, but also the serious impact the motor and non-motor symptoms associated with it have restricting participation in the lives of young adults with DCD. Therefore, careful consideration needs to be given to outcomes that are relevant to children and adults with DCD for both intervention and research. The ICF provides a useful framework with which to approach this.

# 4.6 Participation as an outcome for children with DCD

The International Classification of Function, Disability and Health (ICF: WHO, 2001), emphasizes functional performance in real life contexts. Disability is not only considered in terms of impairment to body structures and functions, there is also a need to identify limitations in activity and restrictions in participation in different environments. The importance of this was discussed in chapter two. However, a systematic review by Magalheas, Cardoso & Missiuna (2011) examined activities and participation in children with DCD from 1995-2005 and found that the majority of articles focused on body functions of children with DCD. Many studies reported activity restriction, such as difficulty with handwriting (52%) and use of hands in the classroom (45%) and for self-care (47%). However, the most frequent activity limitations and participation restrictions reported were play related activities (74%), such as riding a bike, roller blading, using playground equipment and participating in free play. 54% reported problems with running, jumping, skiing, swimming and limited participation in organized sports, whilst 43% reported poor performance on ball games. Thus participation in physical activity appears to be the area of most restriction for children with DCD. Limited opportunity for self-generated movement practice puts children with DCD at a disadvantage in terms of their motor learning and motor development, from a dynamical systems perspective.

Examination of some of the factors discussed in the literature that influence participation in PA could be helpful.

# 4.6.1 Reduced participation in physical activity in children with DCD

It has frequently been reported that children with DCD are not as physically active as their typically developing peers (Cantell, Smyth & Ahonen, 1994; Hands & Larkin, 2002; Shoemaker & Kalverboer, 1994; Poulsen et al., 2008) and that the children have reported that they lack confidence in physical activity (Losse et al., 1991; Shoemaker & Kalverboer, 1994; Cantell, Smyth & Ahonen, 1994). Self-efficacy (belief in one's ability or competence) is thought to impact Cairney et al. (2006) suggested that these factors participation in PA. contribute to make children with DCD less likely to participate in physical activity and put them at future risk of hypo activity and the health risks associated with it. Cairney et al. (2005) found that children with DCD had lower self-efficacy in PA than TDC; however, this may not be a universal finding. Stodden et al. (2008) found developmental differences in self-efficacy, as discussed in chapter three. However, it is important to understand some of the underlying assumptions in perceived self-competence and how it relates to research in DCD.

# 4.6.2 Investigation of perceived self competence in physical activity in children with DCD

Bandura (1982) defined self-efficacy as the sense that is developed when children are successful completing a task and attribute their success to their own abilities. This sense helps them to persist to succeed in tasks, whilst children with low sense of self-efficacy tend not to persist to solve the problem. Harter (1987) developed a theory of competence motivation with similar assumptions. She suggested that the level of mastery of a skill, referred to as competence, can range from poor, through adequate to superior and so children's level of perceived competence affects their continued interest in a task and any further mastery attempts. A major goal of achievement behaviour is the feeling of competence.

Skinner & Piek (2001) used Harter's theory to test perceived competence in children with DCD. They predicted that children with DCD were likely to experience low perceptions of competence in physical domains, because they were likely to have experienced repeated failure at movement skills and, as a result, limit their physical activity, thereby limiting their opportunities for practice. They tested a school population of 8-10 year old and 12-14 year olds. Children with MABC score <15<sup>th</sup> percentile were included in DCD group and these were matched for age and gender with TDC >50<sup>th</sup> percentile MABC and all had IQ >80. They found, as predicted, the DCD group had lower perceived athletic competence, but also found they had lower perceived social support and higher state and trait anxiety than the control group. These findings suggest that their problems were more complex than perceived athletic competence alone. Perceived athletic performance (PAC) is the way in which children perceive their sports ability and athletic performance (Harter, 1982).

More studies also found lower perceived athletic competence in children with DCD (Losse et al., 1991; Silman et al., 2011; Cairney et al., 2005; Cocks et al., 2009; Poulsen et al., 2008; Shoemaker & Kalverboer, 1994; Cantell, Smyth & Ahonen, 1994). However, there are studies, such as (Pless et al., 2002), that do not report lower perceived athletic competence in children with DCD. Furthermore, some studies report no difference between the PAC of children with DCD and TDC (Noordstar et al., 2014; Fliers et al., 2010). It is therefore possible that the poor motor ability in children with DCD may not be the over riding influence for their PAC. It is important to investigate which others factors may have a positive influence on the PAC for children with DCD, as it is thought that children with higher PAC are more likely to engage in physical activity, and an understanding of these factors could inform intervention to improve their participation.

# 4.6.3 Evidence from investigation of determinants of activity in children with DCD

Cairney et al. (2005) suggested that the potential pathways linking DCD to reduced PA were unclear, but nevertheless important. They investigated the determinants of activity levels by using the a self rated questionnaire, the Children's Self perceptions of Adequacy in, and Predilection for Physical Activity (CSAPPA) Scale (Hay, 1992) and a physical activity scale with 44 children aged 9-14 years who scored below 10<sup>th</sup> percentile on BOTMP. Unfortunately, no other DCD criteria were tested and so referred to the children as probable DCD (pDCD). The findings suggested that 28% of the variance in children's PA was predicted by their generalized self-efficacy and DCD. Furthermore, they found children with pDCD had much lower generalized self-efficacy than TDC and that higher self-efficacy was associated with greater participation in free play and organized activities. It therefore appears that perceived self-efficacy could have an important role to play in participation in PA for children with DCD, but the exact nature of which is unclear.

Kwan, Cairney, Hay & Faught (2013), compared 13-14 year old boys with pDCD and control TDC and found pDCD had lower PA, but that their attitudes to PA only accounted for 25% of the variance. Interestingly, both studies found a similar level of variance suggesting the self-efficacy can be significant, but that the influence of other factors may be equally important in participation in PA for children with motor difficulties.

#### Summary

It is evident from the literature that participation in physical activity is problematic for children with DCD. What is less clear is which determinants are associated with more positive outcomes for participation in PA for children with DCD. However, it is clear that poor motor control is a barrier to participation and reasonable to assume that targeted intervention is likely to improve motor control. However, it is difficult to predict which children with DCD are more likely to improve with intervention, as studies have reported mixed results. Intervention studies with DCD and longitudinal design have noted variable responses to treatment. The advantage longitudinal intervention studies have over cross sectional studies is that they provide opportunity for follow up after the intervention to determine which children have maintained any gains made during the intervention. It is however unclear whether intervention improves participation in PA for children with DCD.

# 4.7 Evidence from longitudinal intervention studies in DCD

It is difficult to predict which children with DCD will progress and are more likely to have a positive outcome. The motor outcomes appear to fall into one of three groups; those which improve to typically developing level, those children that improve but never reach typically developing level and those who continue to have severe motor impairment (Cantell et al, 2003; Geuze & Borger, 1993; Knuckey & Gubbay, 1983; Sugden & Chambers, 2006; Green et al, 2008). However, more interesting, perhaps in terms of future direction for research is the reported differential response to intervention of children with DCD. Three intervention studies are of particular note:

Sugden & Chambers (2006) followed 7-9 year old children (n=26), who met DSM IV criteria for DCD over a 4-year period until they were 10-13 years old. They used parent and teacher interventions, with an ecological approach, and noted that after intervention the group mean showed that the children had improved motor performance. However, intra-group analysis revealed that the children fell into three subgroups after intervention: some children improved and maintained their motor improvement to typically developing level (>15<sup>th</sup> percentile); some showed an uneven profile and improved but slipped back to the impaired motor range once intervention ceased and some children showed no motor improvement. Unfortunately, there was no control group of TDC and the presence of co-occurring disorders were not reported, but this study provided empirical evidence for the differential response to intervention by children with DCD.

Green et al (2008) studied subtypes of DCD, who met DSM IV criteria, and randomly allocated children (n=46) to intervention groups using a cognitive orientation to daily occupational performance approach (CO-OP, Missiuna et al., 2001) for a 20-week intervention. Children with disorders co-occurring with DCD were identified and their SES was also noted. They found that some children got worse or showed little or no improvement, either with or without intervention; some responded well to intervention, but of those children some had difficulty sustaining their improvement. The group with problems in all areas remained with the most difficulties even after intervention. Significantly, more children with a comorbid diagnosis (two thirds) did not make good progress. Furthermore, comparing clusters of subtypes provided limited predictability of who would make progress without intervention and only better verbal ability seemed to contribute to a better prognosis. Interestingly though, progress in motor skill appeared unrelated to initial severity or subtype, as they found that some children with a milder presentation did not necessarily improve without additional help, which appears to be counterintuitive and warrants further investigation. This is an important study, as it not only identified children who met criteria for DCD; it took into account co-occurring disorders and symptom severity in the data analysis and noted SES. However, it did not have a control group of TDC and did not account for developmental and environmental factors over the same period.

A differential response to treatment has also been reported in other studies. For example, Van der Waelvelde et al (2010) investigated motor stability in a group (n=49) of young clinic-referred children at risk of ASD, ADHD and DCD according to DSM IV TR criteria. All had IQ >70, motor ability  $\leq 15^{th}$  percentile and were aged 4-6 years at initial assessment on MABC. The group contained nine children born prematurely. They argued that most research on stability and change in motor ability in DCD had concentrated on children over six years and wanted to investigate younger children. However, not all the DSM criteria were fulfilled for DCD and so the children could only be grouped in those with ASD, ADHD and no diagnosis. Analysis of variance (ANOVA) was used to determine differences between the groups at baseline and post hoc analysis

found the group with ASD had the lowest motor performance. Change in motor ability was determined as the difference in MABC score between initial and subsequent measurement 2-3 years later. Some of the children had received therapy by the time they were re-tested and 22 children were no longer ≤15<sup>th</sup> centile MABC. Interestingly, they found that children who received no therapy improved significantly more than those who did receive therapy. They also found that the ASD group made the least motor progress and concluded that poor motor performance in young children was not always stable. Although they identified some children who met the criteria for both ADHD and ASD diagnosis, they did not analyse the results for comorbidity separately. Unfortunately, there was no TDC comparison group over the 2-3 year period and so developmental aspects could not be taken into account. As it was not a random sample and the power and effect size of the study were not quoted, it is difficult to extrapolate from the results to the wider population. However, the study did find differences in motor performance between the groups initially and at follow up and highlighted that children with ASD were at greatest risk of poor motor progression.

This led Reinhart and McGuinley to question that if problems are stable over time (e.g. performance on MABC) it may indicate fundamental disruption to brain circuitry and would thus be more pertinent to compensate or work around the problem or work with modalities less likely to be disrupted by these circuits (Reinhart & McGuinley, 2010). This obviously will have a large bearing on the direction of future research on the role and efficacy of intervention for motor interventions, but more studies are required accounting for the disorders comorbid with DCD.

#### Summary

Some children, despite intervention, do not show improvement and have persisting motor problems, which poses several questions. Does it depend on the initial motor severity of the disorder DCD, or do the comorbid conditions create additional problems? Is it dependent on the age at testing and the type of tool used to measure the impairments, or does the reported improvement in adolescence for some children actually occur as they find ways round their difficulties? Are the primary problems of more importance than the secondary problems that arise from living with motor coordination problems? Does a supportive environment affect motor outcome by influencing perceived self competence or is it the individual characteristics of the child, such as IQ, or motivation that is more important? Is there a difference in their motor learning between subgroups of DCD?

These important questions require a comprehensive methodology grounded in coherent motor theory in order to attempt to unravel some of the complexity.

An ecological approach to examine the various levels of influence would be one way to address some of these questions, but first it would be helpful to review some of the current theoretical perspectives in DCD.

## 4.8 Theoretical perspectives in DCD research

Work on DCD has been conducted from two different perspectives; these are the cognitive neuroscience and ecological approaches. The cognitive neuroscience approach is concerned with understanding biological processes that underpin cognition and action in order to understand any causal relationship between brain function, cognition and behaviour (Wilson et al., 2013). Whereas the ecological approach is concerned with the dynamic interaction between the individual, the task and the environment, which result in particular levels of motor skill or movement patterns.

A tension exists between different theoretical approaches to the understanding of the underlying causes for DCD. Wade & Kazek (2018) argue that the DCD has a history of causation viewed through the lens of an information processing (IP) model, whereby the brain is viewed a computer or machine that processes information, as described in chapter three. Motor deficits seen in DCD, from this viewpoint, are assumed to emanate from deficits in this processing system, but that this overlooks other causal explanations (Wade & Kazek, 2018). That is, the deficits are assumed to lie within the child or adult with DCD and the corresponding research focuses on the characteristics of the child and his or her motor and perceptual abilities. However, Geuze (2018) observes that strength of IP models is that they use a systems analysis approach to break down the system into functional parts that may be tested on functional limitations or deficits and these limitations are assumed to contribute to the deficits seen in DCD. Wade & Kazek (2018) argue that this constitutes a reductionist approach and despite burgeoning literature adopting this approach, they suggest that little empirical evidence supports a 'valid and reliable causation' for DCD (Wade & Kazek, 2018, p495).

In contrast, an alternative explanation for the underlying cause of the motor deficits seen in DCD is viewed from an ecological approach, which is supported by Wade & Kazek (2018). Here concepts are taken from the perception-action and dynamic systems literature, whereby progress in motor performance is viewed as being constraints led, with the task, the resources of the child or adult and the environment all dynamically interacting (Sugden, 2018). Thus the deficits seen in DCD may be viewed as context dependent. Indeed, Sugden asserts that the child is never the unit of analysis, always the child in context (Sugden, 2018), because the transactions between the three components of task, child and environment continually change throughout the learning process in a non-linear self-organizing fashion. The literature reviewed in this and the progression of motor performance in DCD and for participation in PA that occur at different levels of interaction between the child and environment.

Therefore, in order to approach an understanding of the nature of DCD, it's motor trajectory and corresponding participation levels of the children it is necessary to examine the context in which these occur. An ecological approach to study design would permit this. However, a consideration of how the findings may contribute to an understanding of the causal influences in DCD and fit with the wider knowledge base is also required. A theoretical framework can help situate this knowledge.

#### 4.8.1 Theoretical framework

Diagnosis for DCD is not straightforward and a number of questions have been raised in this thesis about its identification, assessment, inclusion and exclusion, which have important ramifications for research, as previously discussed.

It is logical to assume that in order to research a condition one needs a group of individuals who have received a diagnosis. It sounds simple enough, but let us consider some implications of a child receiving a diagnosis of DCD. First, we must assume that diagnosis is valid, however, Morton takes issue with the current methods of diagnosis of developmental conditions. He suggests that a considerable amount rests upon a diagnosis such as the management, intervention and prognosis of a condition, and points out that, as a result either misdiagnosing or not diagnosing at all can carry equal dangers for a child (Morton, 2004). Furthermore, he suggests that current diagnostic systems such as the WHO ICD 10 (and also by implication the DSM5) are "unscientific" because they rely on a list of symptoms to categorize an individual rather than being based on any theoretical position about causation.

This leads to us to question the basis on which we have to diagnose children with DCD. In both ICD 10 and DSM5 we are provided with a list of criteria (mainly behavioural) but no explanation of possible causation. There is agreement that children with DCD are required to score below a level expected for their chronological age and measured intelligence on a standardized test of motor ability, however, different standardized tests have been shown to identify different groups of children (Spironello, Hay, Missiuna, Faught & Cairney, 2010; Cairney et al, 2009) which only serves to cause us to question further the basis for diagnosis. Furthermore, Morton suggests that behavioural manifestations are notoriously difficult to rely on because they may differ in different situations and tell us nothing about underlying cognition (Morton, 2004). We know that children diagnosed with DCD are expected to fall in the range of normal IQ (>70) but nothing is stipulated about their cognitive processes, yet researchers have reported a higher prevalence of specific learning difficulties, difficulties

with executive function (such as planning), difficulty with cross modal perception, to name a few, but to what extent should these be part of the diagnosis or might they represent different subgroups? Yet, as Morton (2004) reminds us, cognition has been shown to be important in the recognition and understanding of two other developmental disorders, namely deficits in theory of mind in autism and phonological processing problems in dyslexia. We also know that children with good intact intellectual abilities can circumvent their difficulties to some extent, for example children with dyslexia still have underlying problems yet learn to read (Snowling, 2008). Could this also be that case with children with good intact intellectual abilities and improvement in motor performance in DCD? This leads us then to question the importance of the role of learning and of what may constitute an appropriate supportive learning environment. Might there be critical stages in development or a particular type of environment that is facilitative?

There is also the important question of the high level of prevalence of cooccurring conditions with DCD. Take for example the high prevalence of cooccurring ADHD; we can hardly ignore the impact that up to 50% of children with DCD may also meet the diagnostic criteria for ADHD may have, as attention difficulties will inevitably interfere with learning of all kinds. Will the characteristics of a child with ADHD plus DCD and their functional outcomes be different to those of a child with DCD alone? Or the case of SLI, or indeed autistic spectrum disorder, where a high proportion of children with these conditions also exhibit motor impairment severe enough to meet the criteria for DCD, how will their characteristics and outcomes differ, if at all?

#### 4.8.2 Morton's causal model

Morton (2004) suggests that we should trust clinical intuition where disorders are not seen merely as collections of randomly occurring symptoms and use a multi level theory to explain variability and severity of the disorder at a biological or cognitive level or behavioural level rather than relying on observations at the behavioural level. Morton offers his 'causal modelling approach' to understanding developmental disorders (see Morton, 2004). Hill (2006)

99

advocated using this approach to help understand both the within group heterogeneity in DCD and the characteristics that are prevalent and homogeneous across other disorders. As both the homogeneous and heterogeneous characteristics of children with DCD are key areas of concern, Morton's multi level theoretical approach to developmental disorders is an appropriate approach to underpin a study examining these issues. Sugden & Wade (2013, p244) modified Morton's causal model to encompass a dynamical systems approach to examine DCD. They proposed adapting the levels of interaction to biology, constraints and behaviour, and included two-way influences between experience and biological variables being mediated by constraints, see Fig. 4.1. This approach can help us consider some of the complex influences on the motor and other characteristics that present in DCD.

Pearsall-Jones et al. (2010b) suggested that causal pathways of developmental disorders are indeed complex and are a combination of genetic, epigenetic and environmental factors that evolve over time. It is therefore important that future research takes these various factors into account. Case study design is one method that is suited to capture multiple types of data.

Although there is a lot of data on DCD and some sparse longitudinal data, there is very little indeed looking at the combination of motor and other characteristics that may influence stability or otherwise of children with DCD.

Biological

CNS, family, genetics, neuromuscular, birth events



**Constraints** Organismic, environmental, task (Newell, 1986) Degrees of freedom, coordination structures, memory, attention, perceptual



**Behavioural** Self care skills, recreational skills, fine & gross motor skills, classroom skills

Figure 4-1 Proposed causal modelling of DCD (Sugden & Wade, 2013 p244, adapted from Morton, 2004)

# 4.9 Summary of current issues in DCD

Some of the complexity surrounding the identification, presentation, course and outcome of DCD has been considered in this thesis. The first complex issue concerns co-occurring conditions. Empirical evidence indicates that co-occurring conditions and their associated characteristics are common and can have a potential negative impact on the motor and non-motor outcomes for children with DCD. However, there is evidence to support that DCD does exist as a separate disorder in 'pure' form. It is therefore important to try to ascertain which outcomes relate to DCD and which are more likely to be attributed to the characteristics of co-occurring disorders.

The second concerns IQ, as IQ may also have a role to play in the outcomes for children with DCD. For example, at a biological level whether low IQ may indicate more severe difficulties, or limit response to experience because of impaired learning. Closer investigation of IQ is required within the context of motor progression in DCD and should include observations of motor learning.

The third issue concerns the environment and the child's context. The role of the environment and environmental constraints have been discussed in relation to child development in general and motor development in particular, when considering a dynamical systems theoretical approach to motor progression in children with DCD. The emphasis from this approach is of observation of the child, the tasks and the child's context. The context includes determinants for PA, such as, opportunities for practice, family resources and attitude to PA.

The fourth concerns issues of measurement. The various tools for measurement of motor ability and their different cut off thresholds have come under scrutiny in DCD research, but without an understanding of the child's context, they are insufficient to fully understand the nature of DCD. However, the measuring tools and cut-off thresholds do need to be identified and noted in DCD research, as the children with different severities of motor impairment may have different trajectories and outcomes.

The fifth concerns variable outcomes within groups of children with DCD. Variable motor progress and large intra-group differences have been widely reported in empirical DCD research, which has led to investigation of subtypes of DCD, but with only limited success. It may be that motor variability is of importance and could potentially be used as a prognostic indicator for atypical development.

The sixth concerns response to intervention. Differential response to intervention has also been reported from empirical studies in DCD, the cause of which remains unclear. Literature indicates the potential importance of the role of genetic or biological difference, environmental constraints, the role of self-efficacy and opportunity for practice as factors for motor learning and motor

progress. These factors therefore need further exploration by careful systematic investigation in a natural environment, without intervention, in order to understand some of the differences between children with the different presentations of DCD and TDC.

#### Summary

Therefore, in order to help understand some of these influences in DCD, a systematic investigation of some of the components discussed using Bronfenbrenner's bio ecological model was undertaken. This provided a framework for data collection at different levels of interaction, so that subsequent analysis of motor progression from a dynamical systems approach to motor development could be attempted. The aim was to permit better understanding of the difference levels of influence on motor progression in children with DCD and of their experiences of participation in physical activity. Details of the study design and procedure are covered in the next chapter, but first the various methods of statistical data analysis require careful consideration. The method of analysis and interpretation of data are equally important, as the use of different methods of analysis can lead to different conclusions.

#### 4.10 Considerations for data analysis in DCD research

#### 4.10.1 Group comparison

A vast body of evidence has reported difference between children with DCD and TDC on a number of different perceptual and motor tasks, as discussed in chapter two. However, comparative study design comparing children with DCD and TDC has been criticized, as it fails to capture the variability in the performance of the children. For example, within group variability is widely reported in studies of children with DCD (Geuze et al., 2001; Sugden & Chambers, 2006; Green et al., 2008; King et al., 2011). However, group comparison between DCD and TDC, although they may be statistically different, do not tell us much about the heterogeneous individuals, or underlying causes (Cantin et al., 2014; King et al., 2011). Some authors have reported individual differences within groups of children with DCD that include some children who perform worse than their peers, others at the same level as their peers and even some who perform better than their peers at some tasks (Sugden & Chambers, 2006; Green et al., 2008; Van Waelvelde et al., 2009). Cantin et al. (2014) observed that, as a consequence, when group means are calculated large standard deviations are often reported, which threatens the assumption of homogeneity of variance, which are required by the majority of statistical tests used in comparative study design. Furthermore, it is then difficult to know if the reported group differences are truly representative of the group or due to a few outliers (Cantin et al., 2014, p219).

#### 4.10.2 Cluster analysis

Cluster analysis has been used repeatedly to investigate the heterogeneity within DCD and some limitations have been discussed previously in relation to prior research in DCD. King, Harring, Oliveira & Clark (2011, p1389) have further criticized the use of cluster analysis to describe the behavioural differences (sub-types) between groups based on relationships among multiple dependent measures for each individual, as it is sensitive to the measures chosen and therefore analysis/interpretation of subtypes is subjective. An analysis proffered in this paper is that, historically, the measures chosen in DCD studies have been influenced from an Information processing theory approach to motor development and the majority of studies have not taken account of the commonly co-occurring disorders.

#### 4.10.3 Developmental landscape

Another approach is the 'developmental landscape' approach, which uses a large cross-sectional age range of TD children that form an age-based regression. DCD children outside the upper prediction limits of the TD regression are considered to have impaired performance on that task. This approach emphasizes individualized performance and it compares DCD children to a much larger group of TDC with a more expansive age range. However, it does not examine population level effects and differences at the

individual level may be difficult to detect if the between-subject variability of TDC is large (King, Harring, Oliveira & Clark, 2011, p1389).

## 4.10.4 Random coefficient models

An alternative approach to is to use random coefficient models, as they can better characterize intra- and inter- individual variability in children with and without DCD than general linear models (GLM) (King et al., 2011). One random coefficient model reported in DCD studies is multi-level modelling (MLM) (Cantin et al., 2014; Veldhuizen & Cairney, 2015). The advantages of this design are that data points are considered individually and in addition; multilevel modelling has less stringent assumptions than other statistical methods and so can be used, for example, even if homogeneity of variance is violated. It also has the potential to handle problems in research such as missing data and unbalanced longitudinal data (in that the number and times of observations may differ between individuals), and so may be a more convenient method of field analysis (Snijders & Bosker, 2002). Furthermore, Veldhuizen and Cairney (2015) advocated that multi level modelling provides the opportunity to study cases of marginal or fluctuating function. Indeed, researchers in the field of DCD are beginning to emphasize the importance of individualized analysis as a useful alternative (Bo, Bastian, Kagerer et al. 2008). However, in order to have sufficient statistical power in MLM large sample sizes are required (Field, 2018) and is complicated by the fact that power varies with a function of effect size and intra-class correlations, it differs for fixed effects versus random effects and it changes depending on the number of groups and the number of observations per group. Despite the advantages of MLM, it is difficult to recruit sufficient children who meet DSM5 criteria for a large sample size.

# 4.10.5 Analysis of longitudinal data and the problem of variability

Analysis of longitudinal study design poses yet more challenges. The nature of DCD, as a heterogeneous condition that occurs in early childhood, can be problematic when researching such a condition. It is well documented that child development is a complex interplay of genetic endowment, parenting, environmental experiences and the opportunities for learning over time (Smith,

Cowie & Blades, 2003; Hulme & Snowling, 2009). Hence a child's progress with DCD is difficult to predict and complex to study because of multiple stochastic variables. Longitudinal data is therefore very important in order to investigate factors involved in the influence of a child's progression, but data analysis must be able to capture these stochastic and multiple levels of influence.

There is a dearth of information about change over time for children with DCD and many complex methodological issues as discussed previously, which will impact on analysis. For example, Veldhuizen and Cairney (2015) highlighted that use of repeated testing in longitudinal research might result in cases that change to non-cases and vice-versa, from measurement to measurement. In addition, children with DCD have also often been reported to be variable in their trial-to-trial performance (Sugden & Chambers, 2006; King et al., 2011; Cantin et al., 2014), but when this is averaged over the trials the characteristic variation is lost. Cantin et al. (2014) highlighted that this intra-individual variation might be an important feature of DCD; as such changes could be different for children with DCD than for their peers. In this thesis it is argued that the intra-individual variability between children with different severities of DCD could be important, as changes could be different for children with DCD with more severe motor impairment than their less motor impaired peers or their typically developing peers.

#### 4.10.6 Combining methods to augment data

One way to try to overcome some of the difficulties associated with each method of analysis is to combine methods to compensate for the shortcomings. For example, Sugden & Chambers (2006) chose both comparative study design using GLM statistics and individual case analysis to investigate group, subgroup and individual motor performance of children (n=26) with DCD pre- and post intervention over time. Whilst the comparative statistical analysis showed that there had been an overall improvement following intervention, the sub-group analysis based on motor percentile points on MABC, revealed important variable response to intervention, and the individual case analysis revealed idiosyncrasies with possible reasons for differential responses to intervention.

This study highlights how important data could easily be missed if the analysis relies solely on comparative design.

#### Summary

Each method for statistical analysis carries a potential drawback; whether it involves missing useful intra-individual data by the use of group mean data or limiting statistical power by using a sample size too small for a multi-level model statistical test. However, combining different approaches and augmenting data analysis facilitates a more holistic understanding and is less likely to miss important findings.

Finally, where there may be many determinants contributing to outcomes (such as motor development and the participation of children in PA), for example, parent attitude, school or social environment, they should be included in the analysis and defined as explanatory variables. However, they require a framework to facilitate a more holistic picture to be captured, analysed and interpreted. A case study approach that can combine different methods of data collection and analysis at different levels of interaction is one possible solution to this issue. Furthermore, Bronfenbrenner's bio ecological model lends itself to different levels of analysis, as previously discussed. Barnett (2008) summed up the notion of an integrated approach, when discussing comprehensive assessment of DCD, which could equally apply to research design and data analysis: "The assessor must be equipped with a sound understanding of motor development and DCD, knowledge of the basic psychometric principles and the ability to critically evaluate different assessments tools. A major challenge ...will be to maintain an integrated approach to assessment across different levels of analysis. Methods that draw explicit links between biological, cognitive, behavioural and environmental influences will assist in this endeavour." Barnett (2008, p124).

# **CHAPTER 5 METHODOLOGY**

Current literature has identified that the course of DCD is varied and that children with DCD may have very different motor outcomes over time, but it is not fully understood why. DCD is heterogeneous in nature and commonly cooccurs with other developmental disorders, but it is unclear whether children with DCD plus associated characteristics have poorer motor outcomes and participate less in physical activity (PA). However, children with DCD are reported to have poor self-perception of their motor ability and participate less in PA than typically developing children (TDC), but this may be context dependent. However, active self-generated movement is thought to be important for motor learning and so participation in physical activity is important for motor and general development. The aim of this study was to note each child's context and describe the different profiles of children with and without DCD and identify their stability or change in motor performance ability, alongside the consistency in the self-perception of their enjoyment, ability and predilection to participate in PA over time. This information was used to inform better understanding of the characteristics of the different severities of DCD and subsequent motor progression over time. The impact on the children's experience of participation in physical activity was also investigated, using the child's perspective.

# 5.1 Question: How do the profiles of children with DCD change over time?

- 5.1.1 What are the characteristics of children with different severities of DCD compared to children without DCD?
  - These included their:
  - Motor characteristics
  - Associated characteristics
  - IQ
  - Ability to take part in everyday activities

- Socio-economic and family context
- Self-perception of their adequacy, enjoyment and predilection to participate in physical activity
- 5.1.2.1 How stable are the motor characteristics or do they change over time?
- 5.1.2.2How consistent are their self-perceptions of their adequacy, enjoyment and predilection to participate in physical activity or do they change over time?
- 5.1.3 How do the characteristics and stability or change in motor ability and self-perceptions impact experiences of participation in physical activity from the child's perspective?

To answer these questions a study was designed in three parts:

Part I identified and examined in fine detail the characteristics of children both with and without DCD. This included their motor performance ability, cognitive ability, demographic details and family characteristics, as well as their self-perceptions. The children were categorized into three groups according to the severity of their motor difficulties. These are the baseline data to enable comparison between the groups of children with different severities of DCD and typical children to answer the research question 5.1.1.

Part II examined how we look at the stability or change in the children's actual motor ability performance over time using three data points. It included a novel approach by concurrently examining the stability or change in the children's perception of their enjoyment, adequacy and predilection to take part in physical activity over time at the same three data points. This addressed both research questions 5.1.2.1 and 5.1.2.2.

Part III, using data from the previous two parts, examined whether the characteristics or whether their self-perception impacted on the experiences of the children with DCD to participate in physical activity. This information was

used to integrate the observations, measurements and different perspectives to inform better understanding of stability and change in the motor performance ability of DCD over time and the impact on participation in physical activity. The trajectories of selected case studies were analysed and supplemented with interview data to understand impact on participation in physical activity, from the child's perspective, and addressed question 5.1.3.

## 5.2 Design

The investigation adopted a three-part prospective longitudinal study combining quantitative and qualitative analyses to examine the presentation and progression of children with DCD and the impact it had on their experiences of participation in physical activity (PA). The design was ecological in nature, reflecting the importance of the complex nature of the bidirectional interaction of the environment with the characteristics of the child with DCD. The context (or ecology) in which a child develops is an important part of the developmental process and Bronfenbrenner (1977) argued that developing children influence the people and institutions around them as much as they receive influence from Indeed Sugden & Wade (2013) suggested that Bronfenbrenner has them. shown how the ecological settings are the major drivers of overall child development. The environmental settings can be thought of as a series of nested concentric circles, with the child influencing and being influenced by each of the various levels. The main focus of this study included settings in the first three levels. The settings include the first layer, which is the child's immediate family environment (which Bronfenbrenner refers to as the microsystem). The next layer comprises major settings at a particular point in life, e.g. school (referred to as the mesosystem) and the importance of the child's interactions here. The subsequent layer (the exosystem) refers to environmental factors that may be remote from the child, but nevertheless impact on the child's development e.g. parent employment or school special educational need (SEN) provision. Finally, the outer layer (the macrosystem) refers to external factors such as government policies and initiatives, the community cultural climate, the media's portrayal of gender roles etc. that influence the child's participation and development.

Bronfenbrenner enhanced his original ecological theory from the 1970's to include the role of the child's individual biology, psychology and behaviour fused dynamically with the ecological systems described; he termed this his Bioecological theory (Lerner, 2005). Morton (2004) took a similar stance, but also suggested that in developmental disorders the biological factors can play a limiting role and that the problems can be at the neural, biochemical and genetic levels, but that higher IQ may ameliorate underlying deficits. The study therefore incorporated measurement of IQ in all the participating children to Lerner (2005) further elaborated that the adaptive take account of this. relations between an active individual and his or her active and multilevel ecology constitute the driving force of human development. Therefore, a finegrained analysis of the characteristics of all the children was undertaken, coupled with analysis of data surrounding their ecological interrelations. These included children with different severities of DCD, as well as those of typically developing children, to help our understanding of the nature and progression of DCD and how it may impact participation in physical activity.

The study design was implemented using Bronfenbrenner's theoretical framework, with data collected about the individual characteristics of each child supplemented by collection of their ecological data framed in the micro-, mesoand exosystem levels for subsequent analysis. Consequently, there were a number of different stakeholders in this study with a series of different roles to play in child development. These included and were not limited to the children's parents, siblings, friends, teachers, sports coaches and local community. Sugden (2014) argued that the accumulation of each role played by each stakeholder can have a profound impact on the overall outcomes for the child, if used to positive effect, and termed this the 'aggregation of marginal gains'. The converse could also be true, with small barriers at each ecological level accumulating to have a negative impact on overall development. It was therefore important to consider the roles and interactions of the children with and without DCD with the various ecological levels to determine potential impact on their development of motor ability and participation in PA.

Child self-perception and perceived self-competence are areas thought to be important for child participation in PA, as previously discussed in chapters two and four. Indeed, Poulson, Ziviani & Cuskelly (2007) found that self-appraisal of satisfaction and ability was an important mediating influence in the participation in physical activity and team sports for boys. Measures of child self-perception of participation in PA were thus repeatedly administered for all children during the study. Therefore, mixtures of quantitative and qualitative data from the child, the family and school contexts were included in levels of analysis within the study in order to address the research questions.

Morse's (1991) priority sequence model informed how the mixed study design was approached. Initial quantitative data from standardized assessment determined the categorization of the motor profiles of children, who were then followed up over time. A case study approach was adopted to examine stable and changing motor profiles of children from each of the initial categories and explore the nature of the children's characteristics and their ecological interactions.

A few children from each category were purposively sampled to interview about their experiences of participation in PA from their perspective. Morse (1991) termed this a 'Qant – Qual sequential mixed methods' design, which was adopted to inform purposive sampling for the interviews in this study. Furthermore, this study design included typically developing peers who were also assessed and followed up over time. Hulme & Snowling (2009) suggested that in order to understand the impact of a developmental disorder it is necessary to relate it to typical development. Therefore, the stability or change in motor development of typically developing peers was compared to those with varying profiles of DCD to more fully understand the impact of DCD.

#### Summary

This ecological study used a prospective longitudinal design, which carefully profiled both the children with DCD and children with typical motor development to answer research question 5.1.1. This was a novel approach, as previous

studies have not necessarily screened the children without DCD, as they are assumed to be typically developing. The study then documented their progression in motor performance ability, concurrently with their self-perception of motor performance, over time to answer question 5.1.2. Again, this was a novel approach as perceptions about participating in PA and observed motor performance have not been tracked simultaneously in DCD research. This also involved collecting (ecological) contextual data about their families, schools and community, along with parental attitudes to PA. Three repeated measures of the children's actual motor performance ability were recorded, whilst repeatedly surveying their perceptions about PA over time to explore their stability or change in relation to their context.

Finally, some children with particular patterns of profile and progression were chosen for more detailed case study. These children were assessed for a fourth time and individually interviewed about their experiences of participation in PA. These case studies provided richer contextual detail and insight into participation with DCD from the perspective of the child and this addressed question 5.1.3. To my knowledge this is the first study to profile children both with and without DCD for associated characteristics that commonly co-occur with DCD and to investigate both their self-perception of motor performance whilst measuring their actual motor performance ability over time, in order to better understand impact on participation.

# 5.3 Ethical approval

Ethical approval was obtained from Leeds University Research Ethics Committee. In addition the author obtained a National Health Service (NHS) research passport and an enhanced Disclosure and Barring Certificate in order to satisfy the requirements for working with children and vulnerable people.

# 5.4 Procedure

The procedure for the three-part study is outlined in the flow diagram figure 5.1 below and the steps are discussed in section 5.4.1 These will then be explained in more detail.

Analysis of parts 1, II and III contribute to an overall picture of the nature of DCD.

# Part I

Part II

Part III

- Ethical approval
- Preliminary discussion with stakeholders, letters and information sent out, in-service training in school
- Recruitment of participants, collection of questionnaire data & individual baseline testing of IQ, motor performance and self-perception.
- Identify children with characteristics of overlapping conditions and different severities of DCD & TDC
- $\bullet$  Detailed analysis of characteristics of children with and without DCD for baseline data T1 (to answer RQ 5.1.1)
- •September 2014-January 2015
- Repeated testing of motor perforamnce using MABC2 and selfperception using CSAPPA
- Identifying stability and change in motor performance and selfperception over time of children with and without DCD. Data points T2, T3 & T4 (to answer RQ 5.1.2)
- Identify cases of stable and changing motor profiles and stable and changing self-perception.

• January 2015 - February 2016

- Exploration of how the characteristics of children with and without DCD and their progression over time, impact their participation in physical activity (RQ 5.1.3)
- Compare cases from each motor group for stability or change and participation in extra curricular physical activity
- Compare selection of cases of children participating in extra curricular PA and non-participating children from each group
- Select cases for child interview of their experiences of participation • Analysis of results
- March 2016-March 2017

Figure 5-1 Flow diagram of the study procedure in three parts

#### 5.4.1 Procedural steps

## 5.4.1.1 Preparatory work

Prior to data collection it was necessary to gain ethical approval and generate local interest with stakeholders. Ethics applications were made (see 5.3). Discussions with SENCOs, teacher advisors, Child Development Centre staff and the medical director ensued. They advised on schools potentially receptive to the project so that emails could be sent to head teachers inviting them to participate. Subsequent meetings were arranged to meet staff and deliver inservice training on DCD with participating schools. Handouts on DCD were given to teachers (see appendix 3) and information and consent forms were given to school SENCOs to distribute to parents (see appendix 4) along with researcher contact details should the parents have any queries.

#### 5.4.1.2 Recruitment, baseline testing and classification

Experienced teachers identified typically developing children and their peers with movement difficulties but without a known medical diagnosis. Consent forms and questionnaires (see appendix 4 & 6) were sent to parents and returned via school SENCO. An experienced OT undertook individual baseline assessments (T1) and short summary reports of their child's results were sent to parents. The children were subsequently identified as those with and without movement problems and confirmed by Movement Assessment Battery for Children 2<sup>nd</sup> edition (MABC2). A comparison of the methods of identification by teacher, parent & MABC2 was performed. Classification by severity of motor impairment and the presence or absence of associated characteristics led to six classifications.

# 5.4.1.3 Repeat testing and analysis to inform sampling for cases and interviews

Repeat individual assessment of MABC2 & child self-report questionnaire (CSAPPA) (T2) was undertaken and analysis of results allowed tracking the children within group classifications. This was repeated again (T3). The profiles permitted purposive sampling of children for case studies and individual

interviews. A fourth repeat MABC2 and child self-report questionnaire (T4) were undertaken with the specific cases. Analysis of interview data combined with data from case studies and motor stability or change helped formulate a picture of DCD.

# 5.5 Ethical issues associated with the study

# 5.5.1 Consent

The schoolchildren were recruited via gatekeepers such as the head teacher and class teacher and there was a possible risk of coercion to participate. It should not be assumed that the children give their consent to participate by proxy; therefore, parent and child both need to give their consent (Alderson & Morrow, 2011). The children were therefore invited to give their assent in addition to written consent from their parents. Separate information sheets were available for the children, parents and teachers explaining the purpose and form of the research in terms that were understandable to each group. In addition, the author was available to answer any questions either in a group meeting or by email. All participants were also made aware that they could withdraw from the project at any time.

# 5.5.2 Wellbeing

The sessions of assessment were kept relatively short (around 60 minutes) and delivered in the format of 'games' to make them enjoyable. The children could take short breaks or stop at any time if they wished, in order to safeguard their wellbeing.

The nature of the assessments had the potential to highlight previously undiagnosed difficulties. There was a potential risk that families could feel that feedback about their child's difficulties and support for the way forward might not be sufficient. Whilst this was not an intervention study, general advice about priorities for action at home and school was available and short summary reports for each child were sent home to ameliorate these potential difficulties.

## 5.5.3 Confidentiality

Children have the same rights to confidentiality as adults and consideration of storage of data from interviews on digital recordings and written notes and transcripts needed to comply with the Data Protection Act (Great Britain, 1998) (Alderson & Morrow, 2011). The participants were informed about the goals of the research and about dissemination of the outcome in the form of publications.

# 5.5.4 Privacy

Respect for privacy was important when collecting measurements, observations and particularly when interviewing, so a comfortable, quiet and private environment was negotiated with each school. It was made clear to the children that the interview could be stopped at any time. As the interviews involved questions about participation at school and with leisure, they could inevitably uncover aspects of the child's social and family life. Therefore, care was taken not to intrude on their right to privacy and only follow questions directly related to the research aims in accordance with the Data Protection Act (Great Britain, 1998). Time was spent building rapport with the interviewee, establishing mutual trust and explaining that names would change in published reports in order to protect privacy. It was also important that a summary of the research outcome was available in lay terms for all the participants to access the findings.

# 5.6 Recruitment procedure

# 5.6.1 Stakeholders

Extensive discussions took place with Local Authority Special Needs Coordinators (SENCO), specialist teacher advisors and staff from a Child Development Centre about the proposed project and about which stakeholders would be receptive to approach. During this period the NHS Trust underwent reorganization and a rapid turnover of senior staff, which made it impossible to engage the NHS Child Development Centre in the project.

#### 5.6.2 Gatekeepers

Letters attached to emails were sent to head teachers outlining the project and inviting them to participate in the study. These were followed up with further emails and the offer of a face-to-face meeting with the researcher. Meetings were established with the responding head teachers and a short in-service presentation on DCD was arranged. This was followed up with brief handouts outlining the symptoms a child with DCD at primary school and a child with DCD at secondary school might encounter (appendix 3). Teachers were asked to identify children they suspected of having DCD (i.e. they did not have a known medical condition) and children who were their peers without movement problems or special educational needs. The parents were then given written information about the study (appendix 4) and invited to participate by the SENCO. All the parents that gave written consent for their children to participate were given a parent questionnaire (PQ) (appendix 6), and three screening questionnaires: the Developmental Coordination Disorder Questionnaire-revised (DCDQ '07) (Wilson, Kaplan, Crawford & Roberts, 2007), the Swanson, Nolan & Pelham-IV Questionnaire (SNAP IV) (Swanson et al., 2001) and the Child Communication Checklist 2<sup>nd</sup> edition (CCC2) (Bishop, 2003) and asked to return them via the SENCO. There were no financial incentives offered, but all the parents were presented with a brief summary report of their child's results.

#### 5.6.3 Children's assent

Children were asked for assent to participate each time before being individually assessed by the author, an experienced occupational therapist, at school. Each child was assessed using the Kaufman Brief Intelligence Test 2<sup>nd</sup> edition (KBIT2) (Kaufman & Kaufman, 2004), the Movement Assessment Battery for Children 2<sup>nd</sup> edition (MABC2) (Henderson, Sugden & Barnett, 2007) and the Child Self Perceptions of Adequacy in and Predilection for Physical Activity Scale (CSAPPA) (Hay, 1992). All data were stored according to the Data Protection Act (Great Britain, 1998) and the results were anonymized. The data were collected by means of 3 or 4 repeated assessments over two

academic years by the same therapist. All tests were administered according to the standardized procedures outlined in the test manuals. The measures used are described in section 5.10

#### 5.6.4 Case selection for interview

Children were classified at baseline into Red, Amber and Green groups using the traffic light system proposed in the Movement Assessment Battery for Children 2<sup>nd</sup> edition (MABC2) manual (Henderson, Sugden & Barnett, 2007), based on the percentile rank of the MABC2 total impairment the score. The process of classification of children at baseline is outlined in figure 5.2. The children's profiles and progression were then analysed in order to purposively sample specific cases to interview. The interviews were semi-structured and lasted 20-30 minutes. The interview schedule (appendix 7) included questions about participation in everyday activities and was informed by previous research (Dunford et al., 2005; Iverson et al., 2005; Mandich et al., 2003; Missiuna et al, 2006; Missiuna et al., 2007; Missiuna et al., 2008; Kirby et al., 2008; Lingam et al., 2013). The theoretical approach was phenomenological, aiming for the lived experience of DCD from the child's perspective. Semistructured interviews with grand tour questions such as "tell what it is like..." were used to explore issues such as progress in school: friendships, academic strengths and weaknesses, differences, frustrations, triumphs and participation in sports and physical activity. In addition, the type, importance and enjoyment of leisure pursuits were explored along with issues with self-care and the problems and strategies adopted to overcome them. The interviews were enhanced using participatory arts-based materials depicting a series of activities including self-care, leisure and schoolwork to assist when conversation lapsed.

The children were informed at the beginning of the interview that they did not have to answer any questions that they did not want to and could stop the interview at any time. The interviews all took place at school; individually in a private room away from other children. The interviewer was the author, an experienced OT used to interacting with children with and without DCD. All interviews were recorded verbatim with a digital voice recorder and transcribed. Additional field notes were also logged to note emotional tone or any special circumstances. Narrative data were thematically analysed. The Themes were coded independently by two researchers and then compared and revised until there was agreement. The themes were also compared to previous research to examine any commonalities.

# 5.7 Participants

A convenience sample of children was recruited from school populations, rather than clinic populations, in order to eliminate potential referral bias and to identify any undiagnosed children with DCD. Caron & Rutter (1991) found that clinic populations of children often have an increased prevalence of co-occurring conditions and more severe problems. This was further supported by Sinani et al. (2011), who found differences between school and clinic samples of children with DCD. Figure 5-2 Recruitment and classification procedure



By using a school population it permitted the opportunity to identify children with DCD who may be coping or who are under the radar of services and explore the factors that may contribute to their successful participation.

As this was a convenience sample, additional contextual information about the families and any co-occurring characteristics of other developmental disorders (that are not exclusionary under DSM5 for the diagnosis of DCD) was required to facilitate meaningful interpretation of the results. After ethical approval was granted Head teachers and Special Needs Coordinators were initially contacted and invited to participate in the study. Then, after extensive discussion about DCD with the Special Needs Coordinators and Head teachers, parents were invited to participate in the study. Data were collected over two academic years from 34 children in mainstream schools (whose parents had given their consent). The children were recruited from both primary and secondary schools in order to include children pre and post- puberty and transition to secondary education.

# 5.7.1 Inclusion criteria

# 5.7.1.1 Inclusion criteria for Parts I, II

Teachers in mainstream schools identified children they considered to have motor difficulties and those they considered to be typically developing. Children from junior level and above were included, as they have developed more independence than those in infant level. Both girls and boys, between the ages of 7-14 years old at the start of the study, were invited to participate because these age groups are able to express themselves. To be classified as DCD the children normally meet the DSM5 (APA, 2013) criteria, which usually includes a diagnosis from a paediatrician after gathering information from multiple sources for each criterion.

Please refer to Table 2.3 in chapter Two for the DSM5 diagnostic criteria.

A criticism of previous DCD research has been lack of transparency about adherence to DCD diagnostic criteria (Geuze et al., 2001; Smits-Engelsman at al., 2015; Farhat et al., 2016), yet strict adherence may not always possible. For example, an assessment from a paediatrician is not always readily available in school population studies, but this should not excuse lack of transparency in the inclusion criteria. Therefore, in order to adhere to the DSM5 criteria as close as possible in this study, without an assessment from a paediatrician, the results from an occupational therapist, teachers, parents and the children themselves were consulted in the following ways detailed in table 5.1.

Please refer to table 5.1

The fifth percentile cut-off score was used in both the DCDQ (07) and CSAPPA to identify children with impairments and ensure optimum sensitivity. Sensitivity measures the proportion of positives that are correctly identified (i.e. those with DCD identified correctly), whereas specificity measures the proportion of negatives that are correctly identified (those without DCD correctly identified as not having DCD). APA (2000) suggest that in population based screening a sensitivity of 80% and a specificity of 90% is required in instruments used to screen for DCD. In addition DSM 5 (APA, 2013) recommends an individual standardized motor assessment for each child but, following the EACD guidelines (Blank et al., 2012), both  $\leq 5^{th}$  percentile and 6-16<sup>th</sup> percentile on MABC2 motor assessment were recommended. Therefore both were included to facilitate the identification of children with DCD.
Criterion	Diagnostic features
А	The author, an experienced Occupational Therapist (OT)
	individually assessed each child with the MABC2 (Henderson,
	Sugden & Barnett, 2007), followed by a discussion with the
	special needs coordinator to ascertain if the motor skills were
	substantially below that expected given child's chronological
	age & opportunity for skill learning and use.
В	The teacher identified children and the parents were invited to
	complete the DCDQ 07 (Wilson, Kaplan, Crawford & Roberts,
	2007) to rate their child's movement ability in relation to
	everyday tasks to determine if the motor skill deficits in A
	significantly and persistently interfered with activities of daily
	living.
С	The parents and special needs coordinator confirmed that the
	onset was in the early developmental period and Parent
	questionnaires were completed to comment on their family
	history and circumstances.
D	The parents and special needs coordinator confirmed that
	there was no diagnosed visual impairment or neurological
	condition from school health notes (in lieu of a paediatrician
	assessment) and the Occupational Therapist assessed each
	child using the KBIT2 (Kaufman & Kaufman, 2004) to
	ascertain IQ scores.
Supplementary	The parents completed CCC2 (Bishop, 2003) and SNAP IV
information	(Swanson et al., 2001) to screen for any associated
(although not	characteristics in communication and attention.
required under	• The children were asked to rate their enjoyment and
DSM5)	inclination to engage in physical activities by using the
	CSAPPA questionnaire (Hay, 1992).

### Table 5.1 How DSM5 criteria were investigated in this study

### 5.7.1.2 Inclusion criteria for part III

Inclusion Part III (individual interviews):

The children selected for interviews were selected on the classification of their profiles at baseline and on their subsequent stable or changing motor profile overtime. At baseline the children were identified by the severity of their motor impairment and classified by the traffic light system proposed in the MABC2 manual (Henderson, Sugden & Barnett, 2007). Thus a child from each group (Red/severe, Amber/moderate & Green/typically developing, both with and without associated characteristics at baseline) was chosen to participate in an interview and describe their experiences of participation in physical activity (PA).

Table 5.2 Selection of children for interview based on classification

RED	RED	AMBER 6-	AMBER 6-	GREEN	GREEN
≤5 <sup>th</sup> percentile on MABC2	≤5 <sup>th</sup> percentile on MABC2 with AC	16 <sup>th</sup> percentile MABC2	16 <sup>th</sup> percentile MABC2 with AC	≥ 25 <sup>th</sup> percentile on MABC2	<ul> <li>≥ 25<sup>th</sup></li> <li>percentile</li> <li>on MABC2</li> <li>with AC</li> </ul>
N=1	N=1	N=1	N=1	N=1	N=1

Some children did not remain in the same motor category over time and so an additional 4 children with unusual profiles, who changed categories over time, were also interviewed. This resulted in a total of 10 children who were interviewed. A more detailed explanation of the case selection is covered later in section 5.12.2 that discusses a framework for determining the cases.

### 5.7.2 Exclusion criteria

The exclusion criteria comprised any child with a general medical condition or neurological diagnosis, such as cerebral palsy, muscular dystrophy etc. or any child with a visual impairment (DSM5, APA, 2013). This helped ensure not to attribute symptoms to DCD when they could be attributable to another condition. The exclusion criteria also excluded any child who did not attend a mainstream school, as the educational environment and level of support was considered too different to those in mainstream school. However, children attending an ASD unit attached to a mainstream school were included in the study.

It was decided not to include children below the age of 7 because they were not considered to be able to accurately express themselves in written self-perception questionnaires without a pictorial element (Missiuna, Pollack & Law, 2004). Children older than 14 years of age at the start of the study were also excluded; it was thought that the workload in preparation for external exams could possibly be disrupted by participation in repeated data collection over two academic years.

### 5.8 Summary of the study data collection procedure

Part I of the study involved teachers identifying children that they thought had movement difficulties and their peers who did not and who did not have any additional special educational needs. The teachers were briefed about DCD and given a handout detailing examples of how DCD may present at school. Parents, who consented to participate, completed a series of questionnaires about their child and family, but did not receive any prior briefing about DCD. Finally, an experienced qualified OT individually assessed each child at school. Parents and children received a short summary of their results. Part I of this study involved detailed analyses of this data collected at baseline.

Part II of the study entailed repeating the same motor assessment of the children in school by the same OT over two academic years and the analysis of any change in motor ability. Part II also involved repeated administration of a self-perception questionnaire with the children by the same OT, synchronous

with the motor assessments. In addition, all children were asked about their participation in extra curricular physical activity.

Finally in Part III analyses of child characteristics, motor stability or change and the stability or change of self-perception led to the identification of specific cases of children who were interviewed about their perceptions of participation in daily life and physical activity. This gave insights into some of the barriers and facilitators for participation in physical activity for children with DCD.

However, a crucial part of this study was consideration of each child's context. The context (or ecology) in which a child develops is an important part of the developmental process and Bronfenbrenner argues that developing children influence the people and institutions around them as much as they receive influence from them (Bronfenbrenner, 1977). Thus the complex nature of the characteristics of the developing child with DCD (biology, psychology and behaviour) and the dynamic interaction with the differentiated levels of the ecological system proposed by Bronfenbrenner's bio ecological model (Lerner, 2005) provide a useful model for levels of analysis in this study. There now follows an explanation of the various data requirements in order to undertake this analysis.

### 5.9 Detailed data requirements for analysis

# 5.9.1 Part I: identification, classification and profiling of schoolchildren

Part I provided analyses of the identification of schoolchildren by parents and teachers and the subsequent classification of children meeting the criteria in DSM5 (APA, 2013) using EACD guidelines (Blank et al., 2012) for DCD. Important detailed profiling of the children then followed the classification. This involved screening questionnaires, discussion with teaching staff and individual child assessments in order to compile a detailed profile of each child.

Information regarding commonly co-occurring characteristics of other developmental conditions, any medical conditions and special educational needs was also noted and analysed. The DCD population is known to be heterogeneous (Wright, 1997; Macnab et al., 2001; Hoare, 1994; Green & Baird, 2005; Visser, 2007; Green et al., 2008), yet little is known about how children with differing profiles progress. Therefore careful identification and fine grained analysis of each child's profile was undertaken in order to be able to differentiate the potential issues that could impact motor stability or change over time. The first of which was the severity of motor impairment.

### 5.9.1.1 Severity of motor impairment:

Previous studies have used different thresholds for the identification and inclusion as DCD making any comparisons difficult (see Smits-Engelsman et al., 2015). For example, it would be difficult to compare studies if they did not indicate whether they used children only in the lowest 5<sup>th</sup> percentile or those in lowest 6-16<sup>th</sup> percentile of motor ability on a standardized test, or whether both groups were amalgamated as one. As Geuze, Jongmans et al, (2001) suggested, the different cut- off scores used by different researchers poses problems in making inferences from the studies. Furthermore, Henderson & Barnett (1998) suggested this has potential far reaching consequences for clinical practice and education. This is particularly the case when examining developmental trajectories, as different levels of motor severity may experience different trajectories over time, especially important for any studies making Therefore, children in this study were claims about intervention effects. classified at baseline by motor severity to see if they exhibited differences in stability or change in motor ability over time.

However, different authors have suggested different methods of classification for severity of DCD. For example, Henderson, Sugden & Barnett (2007) proposed a traffic light system for scores on the MABC2 whereby children at or below the 5th percentile are identified in the red zone to have significant movement difficulty, those between the 5<sup>th</sup> -16<sup>th</sup> percentile are in the amber zone and are considered at risk of movement difficulty and require monitoring

and those above the 16<sup>th</sup> percentile are in the green zone and have no movement difficulty detected. Smits-Engelsman et al. (2015), on the other hand, proposed the classification of DCD as either severe or moderate DCD or typically developing for those groups respectively. Children in this study are classified by severity of their movement difficulties as follows:

Table 5.3 Classification	n of participant mov	vement impairment
--------------------------	----------------------	-------------------

	≤ 5 th percentile	6-16 <sup>th</sup> percentile	≥ 16 <sup>th</sup> percentile
Smits-	Severe DCD	Moderate DCD	Potentially TDC
Engelsman et	(sDCD),	(mDCD)	(TDC)
al. (2015)			
Henderson et	Red	Amber	Green
al. (2007)			
	Significant	At risk, requires	No movement
	movement	monitoring	difficulty detected
	difficulty		

It is of note that the Green or typical motor development group, should be referred to as potentially TDC, but will be classified as typically developing in this study. This is because they score on the typical range for motor ability but may exhibit other associated characteristics. In order to fully differentiate them from the other groups all children in this group scored  $\geq 25^{th}$  percentile on MABC2.

### 5.9.1.2 Co-occurring Characteristics:

It has been documented for some time that DCD co-occurs with other conditions, which led Kaplan et al. (1998) to suggest that it may not be a discrete disorder. However, DCD has been found to occur in pure form (Peters & Henderson, 2008). Nevertheless, Visser (2003) highlighted the importance

for future research to identify characteristics co-occurring with DCD so they could be investigated in terms of their developmental trajectories. This is salient since other research has identified that co-occurrence may lead to less favourable outcomes for children (Rasmussen & Gillberg, 2000; Germano et al., 2010; Cohen et al., 2000; Missiuna et al., 2011). Furthermore, Snowling (2008) argued that comorbidities must be explained and suggested that single deficit accounts for developmental disorders, such as dyslexia do not explain the heterogeneity observed. On a practical note, Kirby (2005) recognized the imperative to identify co-occurring conditions to facilitate effective management for individuals in both clinical and education settings. Indeed, Goulardins, Rigoli et al., (2015) described the comorbid occurrence of ADHD and DCD as a significant public health concern because of the increased risks of overweight, obesity and school achievement problems associated with both disorders. Therefore, screening for associated characteristics is an important aspect of the study. The associated characteristics chosen were difficulty with attention, social and language impairment. The instruments for screening these are described in section 5.10.

#### 5.9.1.3 Measurement of IQ:

The co-occurrence of learning difficulty with developmental disorders has also been well documented (Dewey, Kaplan et al., 2002; Kadesjo & Gillberg, 1998; Germano et al., 2010 to name a few). Furthermore, Sugden & Wann (1987) found that children with moderate learning difficulties (IQ 50-70) were more likely to display movement difficulties, with findings that 30% of 8 year olds and 50% of 12 year olds in their study scored in the lowest 5<sup>th</sup> percentile of motor ability. DSM5 (APA, 2013), unlike ICD 10 (WHO, 1992), does not stipulate a minimum IQ level in the criteria for DCD, leading to the possibility of a diagnosis of DCD for children with IQ lower than 70 (providing they meet the other criteria). Moreover, the EACD guidelines do not stipulate a minimum IQ for diagnosis of DCD (Blank et al., 2012, p63). However, Sugden & Wade (2013) suggest that, as soon as IQ level falls below 70 the relationship between IQ and motor impairment rises as IQ falls, and for this reason should not be given a

primary diagnosis of DCD. However, many of these children commonly present to clinics with functional difficulties that impact on their daily lives and require some form of intervention. Therefore, it is important to ascertain if the motor impairment is over and above that expected for the level of intellectual impairment. Consequently, a profile of each child's level of verbal and nonverbal IQ was documented alongside their motor and functional ability at baseline. Thus ensuring that IQ could be taken into account, before drawing conclusions from the analyses of motor stability or change over time.

### 5.9.1.4 Data on SEN and handedness:

Impaired school performance is often associated with DCD in the extant literature, particularly with handwriting and may impact self-perception and participation. Furthermore Cairney et al. (2008) found higher prevalence of left-handedness in DCD  $\leq 5^{\text{th}}$  percentile and suggested a possible role for cerebral lateralization in motor coordination problems and so handedness was noted in this sample.

### 5.9.1.5 Contextual data about the family:

Another potentially problematic issue arising in the field of DCD is that of the heterogeneity and its potential underlying explanations, whether these occur within the child or are external to the child. Child development is generally now agreed to involve a dynamic interaction between the child and his or her environment (Smith, Cowie & Blades, 2003; Morton, 2004). This idea also follows that of the framework set out in the International Classification of Functioning, Disability and Health (WHO, 2001). Therefore, by identifying, measuring and classifying any associated characteristics, the severity of motor impairment and verbal and non-verbal IQ level, some baseline within-child measures were obtained for discrimination and comparison.

Contextual information about the resources available to each family were also collected in order to obtain data for each child to help explore factors external to

the child that could help explain motor outcomes over time and impact child participation. The parents were asked to complete questionnaires, which included information regarding socio-economic background, parent's employment, and number of children in the family, car ownership and family history of developmental disorders (King, Law et al. 2003; Michelson, 2006). These data were considered with respect to classification of the children for baseline data and for later comparison between them over time.

Beliefs, interests and attitudes also have an important impact on participation in physical activity (Jarus et al., 2011; Silman, Cairney et al., 2011) and so questions regarding importance of physical activity to the parents and how frequently they participated themselves were also included.

## 5.9.2 Part II: Stability and change in motor performance ability over time

Part II was designed to investigate the stability or change in motor characteristics of children with DCD over time. Geuze et al. (2001) highlighted the need for strict adherence to agreed DCD criteria in research to allow comparison between studies. Part I therefore provided detailed analyses of the profiles of the children with and without DCD. As previously mentioned, there is a dearth of information about the progression of differing profiles of children with DCD over time, with previous motor tests only standardized up to 12 years of age, raising the question of a ceiling effect in the reporting of change in older children with DCD. The Bruininks-Oseretski Tests of motor Proficiency (BOT-2) (Bruininks & Bruininks, 2005) and MABC2 are both now standardized up to and over 16 years of age, which permits follow up and examination of motor ability at secondary school age. The MABC2 was chosen for this study as it was standardized on a UK population. Thus, with well-defined research sub groups at baseline, the measurement of motor change was able to be analysed alongside the contextual data and associated characteristics to permit examination of factors likely to have contributed to the stability or change in motor performance.

### 5.9.2.1 Part II: Stability and change in child self perception of enjoyment, adequacy and predilection to take part in physical activity over time

Part II was designed to investigate the stability or change in the children's perception of their enjoyment, adequacy and predilection to take part in physical activity over time simultaneously with an objective measurement of their motor ability over time. This allowed a longitudinal comparison of their self-perceptions with their actual measured performance over time. Self-perception is thought to impact participation in physical activity for some children with DCD (Batey, Missiuna et al., 2014) but it is not clear if there are differential effects.

### 5.9.3 Part III Exploration of the impact of child characteristics, perceptions and motor stability or change on experiences of participation in physical activity

Analyses from part I, II and III of the study allowed investigation of any relationships between child self-perception and actual motor ability over time and allowed exploration of any differential impact of particular subgroup characteristics on children's participation in physical activity. This was then explored in more depth in a selection of individual interviews with children who possessed specific profiles and who demonstrated stability or change in motor ability over time. Post hoc analyses examined change over time and any differences between the groups of children with different severity of motor impairment. The more detailed analysis of the context of the selected case studies of the children with specific profiles, together with insights from their point of view, helped illuminate the findings and indicated factors most likely to have contributed to barriers or facilitators of participation in physical activity for the children with DCD.

### 5.10 Instruments

# 5.10.1 Measuring instruments for Part I: profiling children with and without DCD

## 5.10.1.1 Movement Assessment Battery for Children 2<sup>nd</sup> edition (MABC2) (Henderson, Sugden & Barnett, 2007)

The MABC2 is the test most commonly used to measure motor performance in DCD, is the best examined (Blank et al., 2012, p71) and has good reliability and validity, which is why it was chosen for this study. It served two purposes in this study; the first was to identify the severity and extent of motor impairment for profiling the children with and without DCD and the second was to measure change in their motor performance ability over time.

Identifying severity of motor impairment:

This is a test to identify children with motor function impairment aged 3-16 years and takes 20-40 minutes to administer individually. The total test score provides a measure of motor performance, which can help identify any difficulties under criterion A of DCD DSM5 (APA, 2013). The second edition extends the upper age range from 12 to 16 years, permitting assessment of adolescent schoolchildren. It has three subscales for: manual dexterity, aiming and catching and balance. Although Brown and Lelor (2009) questioned the validity of the MABC2 it was subsequently confirmed by confirmatory factor analysis (Schulz et al., 2011).

Przysucha et al. (2016) and Holm et al. (2013) have both questioned the test retest reliability of age band two when using the total test score. However, Schulz and colleagues (2011) determined that the validity of the test becomes stronger with older children. This means that the profile of scores across the three subcomponents can be used with greater confidence in age bands two and three (Schulz et al., 2011). These two age bands were used exclusively in this study permitting an analysis of the motor profile of the children. Item scores can be converted to standard scores (mean =10; SD = 3). The total test score can be calculated by adding the sum of the 8 item standard scores and then a standard score can be obtained (mean =10, SD = 3) (Henderson, Sugden & Barnett, 2007).

The inter–rater reliability .79, intra- rater reliability .79 and test –retest reliability (.73 to .92) of MABC2 are good to excellent (Henderson, Sugden & Barnett, 2007). Validity is also good (Schulz et al., 2011; Ellinoudis et al., 2011).

### 5.10.1.2 Developmental Coordination Disorder Questionnaire- revised (DCDQ '07) (Wilson, Kaplan, Crawford & Roberts, 2007)

This screening instrument was chosen to fulfil criterion B in DSM5, as it asks parents about the impact of their child's motor difficulties on everyday life.

The revised DCDQ is a 15 item parent-completed measure designed to identify subtle motor problems in children, originally for children 5-15 years old. It requires parents to compare their child's motor coordination for functional skills in everyday contextual areas with typically developing peers of the same age. It takes about 5 minutes to complete and is therefore a useful tool to help identify issues relating to criterion B in DSM5 (APA, 2013) for DCD. Wilson, Crawford, Green et al. (2009) state that the internal consistency is high ( $\alpha$  = .94) and concurrent validity with MABC is acceptable (r= .55). A later study by Pannekoek et al. (2012) also found high internal consistency ( $\alpha$  = .95) and concurrent validity with the MABC2 as fair but significant (r=. 34, p=0.01). The test authors state that the overall quoted sensitivity = 85% and specificity = 71%, although separate cut- off scores have been developed for three age groups (see table 5.4). Total scores range from 15-75 with cut off scores to support 'suspect DCD' and 'probably not DCD'. A higher score on the DCDQ '07 indicates a higher performance level. Three factors were originally identified but confirmatory factor analysis fit was poor, as a number of items had substantial loadings on more than one factor (Cairney et al, 2008). This led Cairney and colleagues to advise that it is best used as a measure of general motor problems and not to discriminate between the different kinds of problems (Cairney et al, 2008), which are myriad in DCD given its heterogenic nature. Only the DCDQ total score has been found to be a predictor of DCD (Cairney et al, 2008) and so none of the sub scores and only the total score was used in this study.

A large population based study in Canada found both sex and age differences in their sample, with girls and older aged children having higher mean scores on the DCDQ (Rivard et al., 2014), which caused them to suggest separate norms were warranted. Interestingly Rivard et al., (2014) found the mean score sex difference to be more pronounced in the  $\leq 5^{th}$  percentile (boys to girls ratio of 2:1) compared with  $\geq 25^{th}$  percentile (boys to girls ration 1:1). Nevertheless, this Canadian study also confirmed the three-factor structure and the psychometric properties already reported (Rivard et al., 2014).

Age	Cut off	Sensitivity (DCD correctly id)	Specificity (id correctly non
			DCD)
<8 years	≤46	75%	71%
8-10 years	≤55	89%	67%
>10 years	≤57	89%	76%
Overall ages	≤53	85%	71%

Table 5.4 Total DCDO '07	score predictor of DCE	status used in this study
	beene predictor of D db	blacab abea in this blaay

(Wilson, et al., 2009)

However, as APA (2000) suggest 80% sensitivity is preferable, 90% specificity is preferred for diagnostic test (Wilson et al, 2009), the DCDQ'07 does not meet the standards for sensitivity for children under 8 years. Neither does it meet the specificity criteria for a diagnostic test for DCD. This was also confirmed in an Australian study of adolescents aged 12-15 years, with findings of high sensitivity (85%) and low specificity (77%) (Pannekoek et al., 2012). Furthermore, in a review of parent and or teacher questionnaires commonly used to assess participation in children with DCD, Kaiser, Albaret and Cantell

(2015) found the results for DCDQ'07 in the different studies 'divergent and inconclusive'. As a result they called for more psychometric and cultural validation to confirm DCD. Despite it's shortcomings, the DCDQ 07 is a useful screening tool, provided it forms part of a two-step identification process for DCD, and is followed by an individual motor assessment to confirm or refute the diagnosis such as the procedure described by Wright & Sugden (1996). In this study parents, and in some cases where questionnaires were not returned, teachers completed the questionnaire and the total score was used to screen the children for possible DCD and then the children were individually assessed using the MABC2.

# 5.10.1.3 The Children's Communication Checklist 2<sup>nd</sup> edition (CCC-2) (Bishop, 2003)

This test was chosen as a screening tool for the associated characteristics of either or language and pragmatic impairments in children.

This is a 70-item multiple-choice questionnaire that screens for communication problems in children aged 4-16 years. It is standardized for a UK population and usually takes 5-15 minutes to complete. Its purpose is to screen for language impairment, identify pragmatic impairments in children with communication problems and assist in identifying children who may merit further assessment for ASD (Bishop, 2003). By definition pragmatic language impairments are dependent upon context and therefore information from someone who can observe the child frequently over a long period of time is required (Bishop, 2003). An adult, who has regular contact with the child, usually a parent or teacher, therefore completes the CCC2.

The checklist is divided into 10 scales, each with descriptions of strengths and difficulties and a rating of the frequency with which the behaviours occur. The first four scales are associated with specific language impairment (SLI); the next four cover pragmatic aspects of communication and the last two scales assess social relations and interests (behaviours that are associated with ASD). It is

possible to derive a Social Interaction Deviance Composite (SIDC), which reflects a mismatch between the sums of different scales to help identify a communication profile characteristic of autism. The CCC2 is able to discriminate children with communication impairments from a typically developing population, suggesting it is a useful screening tool (Norbury et al., 2004)

It is highly possible that children with DCD could have co-occurring SLI or ASD and this checklist provides a quick screening tool to help identify characteristics associated with both these developmental disorders. In this study parents or teachers who knew the child well completed the checklist to screen for characteristics associated with SLI and ASD at baseline.

The CCC2 has robust reliability, with internal consistency scores ranging from .65 to .8 across all scales (Bishop, 2003) and the strongest subtest inter-rater reliability (r=0.79) with the Social Interaction Deviance Composite (Norbury et al., 2004). Thus the checklist was considered robust enough to distinguish children with communication characteristics associated with ASD and SLI for the purposes of this study.

### 5.10.1.4 Swanson Nolan and Pelham - IV Questionnaire (SNAP IV) (Swanson et al., 2001)

This screening test was chosen to identify any associated characteristics of attention and behavioural difficulties in children.

The long form SNAP IV assesses ADHD, oppositional defiant disorder (ODD) and overlapping symptoms of other psychiatric disorders of childhood listed in DSM –IV and is available at <u>http://www.ADHD.net</u>. A short 26 item SNAP-IV version, also referred to as the MTA version, assesses ADHD core symptoms of hyperactivity/impulsivity and inattention, along with symptoms of ODD. There are average rating indices for inattentive, hyperactive/impulsive, combined ADHD and ODD subscales, where only scores above the 95<sup>th</sup> percentile are labelled clinically relevant (Bussing et al., 2008).

The 26 item SNAP–IV include 18 ADHD symptoms (9 for inattention, 9 for hyperactive/impulsive) and 8 ODD symptoms specified in DSM –IV. Items are rated on a four-point scale 0 (not at all) to 3 (very much) to obtain subscale scores for inattention, hyperactivity/impulsivity and oppositional defiance domains. The SNAP-IV is a reliable checklist with coefficient alpha for overall parent ratings  $\alpha$  =.94 and coefficient alpha for each of the subdomains ranging from .79 to .90. (Bussing et al., 2008).

Table 5.5 Reliability coefficient scores for subdomains of SNAP IV

Sub-domain	Parent rating	Teacher rating
Inattentive	0.90	0.97
Hyperactive/impulsive	0.79	0.92
ODD	0.89	0.96

Bussing et al. (2008) also found that it has an acceptable internal consistency and 3-factor structure. Furthermore, the predictive validity of the SNAP-IV was found to accurately identify children with ADHD when parents score inattention above 1.8 and hyperactivity/impulsivity above 2.4 (Bussing et al., 2008). However the SNAP-IV ratings must be interpreted with caution, as they are not diagnostic in isolation and the test was normed on a US population (Bussing et al., 2008), but it was felt to be an appropriate screening tool for use in this study.

The scoring instructions are available at <u>http://www.ADHD.net/snap-iv-instructions.pdf</u> and the cut off scores used in this study are in the table 5.6.

Sub domain	Parent rating	Teacher rating
ADHD inattention	1.78	2.56
ADHD hyperactivity/imp	1.44	1.78
ODD	1.88	1.38

#### Table 5.6 Cut-off scores for subdomains of SNAP IV used in this study

### 5.10.1.5 Kaufman Brief Intelligence Test Second Edition (KBIT2) (Kaufman & Kaufman, 2004)

This test was chosen to identify children with learning difficulties and to differentiate those with greater difficulties in either the verbal on non-verbal domains. This test is an individually administered, brief measure of verbal and nonverbal intelligence for ages 4-90 years old, but is not a substitute for a more comprehensive measure (Kaufman & Kaufman, 2004). It was normed on a large population in the USA but was considered appropriate for use in this study, with a UK population, because the purpose was to identify children with potential learning difficulties rather than give a definitive assessment of intelligence or diagnosis.

It takes 15-30 minutes to administer. The verbal score contains two subtests, verbal knowledge and riddles. Verbal knowledge is a 60-item measure of receptive vocabulary and range of general information, which is delivered through an array of pictures, and the child had to point to their chosen response. The Riddles subtest is a 48-item measure of verbal comprehension, vocabulary knowledge and reasoning. Responses are either by pointing to a picture or by single word. The raw scores of the two subtests are added to give the verbal raw score. The non-verbal subtest, Matrices, is a 46-item non-verbal measure consisting of visual stimuli that require an understanding of the relationship between them. The child has to select the picture that fits best with the stimulus picture, some are abstract and some are everyday items. This section demands nonverbal reasoning and flexibility to problem solve.

It was chosen for this study because it is a robust assessment. The construct validity is reported to be good and correlates with the Wechsler intelligence scale for children (WISC) full scale IQ (Kaufman & Kaufman, 2004). Furthermore, the reliability coefficients quoted in the KBIT2 manual are high: Internal reliability for the verbal score for children and adolescents is .9 and for the nonverbal score is .86 (Kaufman & Kaufman, 2004). The reliability of the IQ composite score is .93 across the entire age range and the test re-test reliability is .91 for verbal and .83 for nonverbal, making it a robust test (Kaufman & Kaufman, 2004). However, the results have to be interpreted with caution on a UK population, because it was normed on a US population and there may be subtle cultural differences. Nonetheless, it was able to give a good estimate of each child's level of verbal and nonverbal cognitive function and identify potential learning difficulties for this study.

### 5.10.1.6 Parent Questionnaire

This is an 11-item questionnaire for parents, designed by the author using Burgess' guidelines (2001), to ascertain detail of family life. A guestionnaire that covers items thought to be associated with child participation; as well as family history of developmental disorder and markers of socio-economic status does not currently exist. Rather than burden the parents with numerous questionnaires and risk a poor response rate, one questionnaire that covered all the desired areas was deemed more likely to be returned. The construct underlying the development of the questionnaire was the influence of parental factors associated with child participation in PA, based on current literature. Welk, Wood & Morss (2003) suggested that two aspects of parental behaviour promote PA in children. The first concerns role modelling, where a parent has an interest in PA, as well as their own efforts to be active. The second is parental support, which includes encouragement by providing opportunities for the child to be active such as participating in PA with the child or transporting the child to parks or leisure areas. It was therefore important to capture this data because of the potential impact on children's participation in PA in the study and to help answer research question 5.1.3. Furthermore, familial and environmental factors have been found to influence children's PA. Levels of parent's education and income are particularly important factors (Zecevic, Tremblay, Losvin & Michel, 2010) and so these areas were also included in the questionnaire design.

The content of the first part of the questionnaire covers the family environmental factors. Items 1-8 require a tick box response and cover information about family history of developmental disorder, number of adults and children living at home, any parental health problems, parental employment status and indicators of resources available to the family, namely: maternal level of education, entitlement to free school meals and car ownership. The list was informed by findings from previous studies on participation by children with disabilities (King, Law et al. 2003; Michelson, 2006).

The second part of the questionnaire concerns parental attitude and support for PA. Items 9 & 11 involved parents ranking importance of physical activity (PA) to the family and the level that their child's coordination difficulty interferes with family life using a Likert scale: 1= not very to 4= very much. Item 10 asked the parents to rank how frequently they participated in PA on a 5 point scale, ranging from rarely to twice weekly. The questionnaire took less than 10 minutes to administer and provided a quick method to collect some data about each child's micro- and meso- system environments to supplement the ecological study design.

Whilst the questionnaire was not standardized and has no psychometric properties, it was piloted on two families in order to clarify the questions and subsequent changes were made before using it in the study. It was found to have ecological validity and was a useful tool to collect contextual data such as family history, parental attitude, numerical data on the family socio economic status and family environment, without having to use multiple pre-exiting questionnaires to collect all the data, thereby reducing the risk of non-response.

# 5.10.2 Instrument for Part II: measuring motor stability and change over time

## 5.10.2.1 Movement Assessment Battery for Children 2<sup>nd</sup> edition (MABC2) (Henderson, Sugden & Barnett, 2007)

Measuring change in motor performance ability:

The total test score from MABC2 was used to measure change in motor performance ability. However, as test scores only represents a snapshot of motor ability at time of testing there is some contention as to whether a change in total test score represents a detectable change in motor performance in everyday life or is possibly a measurement anomaly. Wuang et al. (2012) suggested that there is little consensus on how responsiveness in the MABC2 should be assessed and proposed to investigate this by means of internal and external responsiveness. They described internal responsiveness as the ability of a measure to change during a defined period and external responsiveness as the extent to which changes in the measure correspond to changes in a reference measure. The latter leads to the notion of the minimal important difference (MID) or smallest change in score that a client perceives as important (Wuang et al., 2012).

Motor change in this study was measured by the changes in the total score of the MABC2, but a decision was needed to identify the magnitude of meaningful change. Previous studies in DCD have reported the minimum important difference (MID) expressed as point changes in total test score. For example, Wuang et al. (2012) chose MID of a 2-point change in Total Test Score in a study for Taiwanese children. Wilson et al. (2016) chose another method to measure change in total test score post intervention, which they called the smallest detectable difference (SDD) and calculated it as SDD= 1.96 x  $\sqrt{(2 \times SEM)}$ .

One way of ameliorating measurement anomaly is to be aware of the standard error of measurement of a test. The standard error of measurement for the MABC2 total score at the 90% confidence interval (CI) is two and is three points at the 95% CI (Henderson, Barnett & Sugden, 2007, p136). Thus in order to assume that the measurements avoid any error, a change of greater than three points in MABC2 total test score was required. However, this may still not guarantee a detectable change in everyday motor behaviour.

Holm et al. (2013) described another method that encompassed detectable motor behaviour change whilst still referring to MABC2 test scores, and referred to it as the smallest detectable change (SDC). Holm et al., (2013) define the SDC as the magnitude of change necessary to exceed the measurement error and represent the smallest change that can be detected beyond the measurement error. Their study results indicated that a change of +/- 10 total score points, with MABC2 level two, was necessary to state that real change had taken place using the same assessor over two occasions with Norwegian children (Holm et al., 2013). The 10-point difference reduced the possibility of measurement error, enhanced the reliability of the results, is more rigorous and was therefore adopted in this study to determine motor change.

An alternative approach, in a Dutch study, used standard scores and operated a change in standard score of two points using Dutch norms (Ferguson et al. 2013), but this approach would not be appropriate for UK children as Dutch norms were used.

Therefore, change in motor performance over time in this study was reported as MABC2 total score, with an associated percentile rank in relation to age norms. High MABC2 total scores indicate a higher level of motor ability on this test and the children had data on their motor ability collected at T1 (baseline), T2 and T3. Change scores were calculated for each child as the difference between MABC2 total score at baseline (T1), (T2) and (T3). Therefore, to examine any change during the study T1 scores were subtracted from T3 scores; positive score indicated an improved performance and a negative score indicated

deterioration in performance. This approach was similar to that adopted by Wilson et al. (2016) who measured change using the MABC. Finally, in order to examine inter-individual change in motor outcome over time during this study, T1 scores were subtracted from T3 scores and comparisons were made for each child in addition to those for each group.

# 5.10.3 Instrument for measuring stability and change in perception over time

## 5.10.3.1 The Children's Self-perceptions of Adequacy in and Predilection for Physical Activity scale (CSAPPA) (Hay, 1992)

This self-report for children was chosen as it may provide an idea of how each child perceives their own motor performance and has been the subject of a number of 'encouraging studies' (Blank et al., 2012, p69).

The CSAPPA 20 item scale is designed to measure children's perceptions of their enjoyment of, adequacy in and likelihood of engaging in physical activity. It takes 5-10 minutes to complete and can be administered individually or in a group.

This questionnaire is designed for children 8-16 years old and consists of opposing pairs of statements for the child to choose which best describes them and indicate whether the sentence was "sort of true for me" or "really true for me"(Hay, 1992). The most inactive or inadequate response is scored at 1 and the most active is scored at 4. It has strong predictive and construct validity and importantly, for use in this study, high test re-test reliability (r=.84 to .90) (Hay et al., 2004). It was found to significantly correlate with aerobic fitness, physical activity and motor proficiency (Cairney et al., 2005; Cairney et al., 2006). Internal consistencies are high (Cairney, Missiuna, Veldhuizen & Wilson, 2008) see table 5.7. Separate cut-off scores are recommended for boys and girls if used as a DCD screening instrument; the male cut off is <47 and the female cut off score is <53 (Hay, Hawes & Faught, 2004). The sensitivity is 91% and specificity is 87% for detecting impairment in motor proficiency (Hay, 1992).

It was therefore considered a robust tool to measure perceptions in this study and the strong test–retest reliability made it a good choice to measure any change in perceptions over time.

### Table 5.7 Internal consistencies for CSAPPA

Total score	α = 0.92
Adequacy	α = 0.87
Predilection to participate	α = 0.87
Enjoyment	α = 0.89

(Cairney, Missiuna et al., 2008)

### 5.11 Data Analysis

### 5.11.1 Issue surrounding conventional statistical analysis

In order to answer research question 5.1.1 'What are the characteristics of children with different severities of DCD compared to children without DCD?' it was necessary to compare the children. One way of approaching this was through statistical analysis. However, there are a number of issues surrounding conventional statistical analysis in DCD research.

Conventionally researchers have used group means and inferential statistics to help draw conclusions about the nature of DCD. However, this may be problematic in three ways:

 Rigorous findings are not only dependent upon good study design but also on recruiting sufficient sample size, to allow adequate power for inferential statistical analysis. Yet, recruitment of sufficient numbers of children who meet the criteria for DCD is a challenge in any study. Recent published studies on DCD were consulted to review sample size and power, where they were quoted. Sample size typically ranged from 10-20 per group (see Table 5.10).

- 2. There is currently some contention about the meaningful use and presentation of inferential statistics. For example, what constitutes statistically significant results and which statistical theoretical approach to use (Trafimow & Marks, 2015; Perezgonzalez, 2015). However, with careful analysis of the data, using first descriptive statistics and then a clear choice of theoretical approach for hypothesis testing, inferential statistics can play a useful part (Perezgonzalez, 2015).
- 3. Group comparison between TDC and DCD children has been criticized, as described in chapter four. Even if statistically different, it does nothing to explain anything about heterogeneous individuals or underlying causes (King et al., 2011). Furthermore, the information gained about any individual differences is lost (Cantin et al., 2014). This suggests that alternative methods should augment or replace this approach.

An explanation of how each of these challenges was approached in this study will follow.

### 5.11.2 Inferential statistical analysis

### 5.11.2.1 Null hypothesis significance testing:

A further consideration in any study intending to use inferential statistics is the recent debate about the contentious use of p values to indicate statistical significance. Trafimow & Marks banned their use in 2015 in the editorial of Basic and Applied Social Psychology, arguing for stronger descriptive statistics and greater consideration of effect size to determine whether or not test results have any significance that might relate to real world differences (see Trafimow & Marks, 2015 for a discussion). However, Perezgonzalez (2015) suggested that the misinterpretation of statistical results has both statistical and theoretical sources and blamed a conceptual mix up between Fisher's and Neyman-Pearson's approaches to Null Hypothesis significant testing (NHST). Fisher's approach tests data on a null hypothesis and uses level of significance to

ascertain the probability of the data under the null hypothesis and seeks significant results.

By contrast, Neyman-Pearson's seeks to make a decision by testing data on a main hypothesis and decides in favour of an alternative hypothesis according to a cut-off calculated a priori, based on sample size (N), Type I error probability ( $\alpha$ ), effect size (MES) and power (1-  $\beta$ ), the latter two are provided by the main hypothesis (Perezgonzalez, p1, 2015). This study adopted the Neyman-Pearson's approach and so sample size and power were determined a priori.

### 5.11.2.2 Determining the statistical power required:

The ability of a test to find a significant effect is known as the statistical power (Field, 2018). This is the probability that a given test will find an effect assuming that one exists in the population. This is the opposite of the probability that a given test will not find an effect assuming that one exists in the population (the  $\beta$  level), i.e. the Type II error rate. Therefore, the power of a test can be expressed as 1- $\beta$ . Given that Cohen (1992) recommends that a 0.2 probability of failing to detect a genuine effect, the corresponding level of power would be 1-0.2, or 0.8. We therefore typically aim to achieve a power of 0.8, or have an 80% chance of detecting an effect if one genuinely exists.

There are a number of steps that are required in order to determine power and sample size and they are inter related, but before describing them it will be useful to examine the effect and sample sizes that other published research on DCD has used. Green (2007) adopted this approach and identified four studies from 1990 to 2001 that had identified effect size. Since then others have adopted this approach (see Table 5.8).

Study	Measure	Effect size	Sample size
Humphries et al. 1990	BOTMP	0.86	20
Shoemaker et al. 1994	ΤΟΜΙ	0.86	35
Miller et al. 2001	BOTMP	0.35	20
Polatajko et al. 2001	MABC	0.55	14
Ferguson et al. 2013	MABC2	0.5	17 per group
Wilson et al. 2016	MABC2	0.7	10 per group

#### Table 5.8 Comparison of DCD studies and effect sizes

Field (2018) advocates determining the power for a statistical test a priori and involves considering

- How big the effect is (bigger effects are easier to detect)
- The sample size If test statistics can be considered a signal-to-noise ratio, the bigger the sample (and better approximation of the population) the less 'noise' and easier to find the 'signal'
- (α) the probability of a type I error (mistakenly believing there is a significant effect when none exits), usually set at 0.05.
- (β) the probability of a type II error (mistakenly believing there is no effect when one actually exits), usually set at 0.2

The power of a test is expressed as 1-  $\beta$ , usually 0.8 in social sciences.

Given that there is an inverse relationship between  $\alpha$  and  $\beta$ , a trade off can be made to attempt to minimize the probability of making both types of errors. Hinkle et al. (1994) suggest that by minimizing  $\beta$  the power is maximized, but while  $\alpha$  is under the direct control of the researcher  $\beta$  is not. It is custom and practice in social sciences to set  $\alpha$  level of significance at 0.05. This is because in the social sciences the consequences of a type I error is deemed more serious than a type II error. The converse is true in medical research, where the  $\alpha$  level is set very low (0.001) due to potential harm to patients if a treatment is said to be more effective than a standard treatment when it is not.  $\beta$  to  $\alpha$  ratio of 4:1 is therefore suggested in Social Science research (Hinkle, Wiersma & Jurs, 1994). Thus power (1- $\beta$ ) is 1-4(0.05) = 0.8, which leads us to now consider effect size.

#### 5.11.2.3 Effect size:

Cohen defined the effect size as 'the degree to which a phenomenon exits' (Hinkle et al., 1994, p294) and so the larger the effect the easier it is to detect. However, the likely effect size in a population can be estimated by using data from past research and this can be used to inform the size of sample required for the study (Field, 2018). See Table 5.10 for DCD research that informed this study.

Effect size measures either the sizes of associations or the sizes of differences between the outcomes or groups under investigation (Walker, 2008). The most common measures of effect size are the correlation and regression coefficients r or R. As correlations cover the whole range of relationship strengths from (-1 through 0 to 1) it states how large the relationships are. Yet, importantly is independent of how many participants are tested. Cohen (1988) provided rule of thumb for interpreting effect sizes and these are generally accepted in research (Walker, 2008). A small effect would be Cohen's d of 0.2, a medium effect would be 0.5, a large effect size would be 0.8 and very large would be 1.5. However, effect sizes can help determine sample size a priori, for example Dunlop & Myers (1997) suggest approximate sample sizes for each effect size (see Table 5.9).

	Small	Medium	Large
Cohen's d	0.2	0.5	0.8
Sample size	402	66	28
Correlation	0.1	0.3	0.5
Sample size	800	88	32
2x2 table	0.1	0.3	0.5
Sample size	800	88	32

Table 5.9 Effect size and corresponding sample size

Thus by using  $\alpha$  of 0.05 and power of 0.8 a minimum sample of 32 is required for any correlation study and a sample of 28 is required to demonstrate large effect.

In this study a sample of 34 children was recruited. Allowing for 10% attrition over the course of the study left 29 children and therefore permitted statistical analysis.

However, for a statistical comparison of the groups (Red, Amber & Green), because smaller numbers of children were available per group, the effect size had to be increased to 1.5 in order to maintain a power of 0.8.

Hinckle et al. (p 637,1994) provide tables to determine sample size for known power using 2 tailed interval data (see Table 5.10).

Table 5.10 Effect size, power and sample size for two-tailed hypothesis

Effect size	d value	Power	2 tailed	Sample size
Medium	0.5	0.8	2 tailed	34
Large	0.8	0.8	2 tailed	15
Very large	1.5	0.8	2 tailed	8

By applying this to the children recruited to this study it was possible to ascertain whether it was viable to use inferential statistics in the analysis of the group data.

### 5.11.2.4 Determining sample size

It is possible to calculate the sample size required to detect an effect if we know the power, the effect size and the chosen values of  $\alpha$  and 1-  $\beta$  (Field, 2018).

Thus, by setting  $\alpha$  at 0.05 and power of 0.8 it was possible to use inferential statistics to examine medium to large effect sizes with the sample size recruited in this study (see Table 5.10). However, this was undertaken after first examining the data using descriptive statistics. Gorard (2003) cautions that the standard error is larger with sample size less than 20. Moreover, precedent exists for small sample size in DCD, a recently published study on DCD by Wilson et al. (2016) used power of 0.8 and large effect sizes r= 0.70 (d=1.5), which determined they could use a sample of minimum of 10 per group. Field (2018) suggests that a power of 0.8 is sufficient to detect any effects that exist.

Therefore, for this study, assuming large effect sizes r=0.70 (d=1.5) and  $\alpha$ = 0.05, a sample of minimum 8 per group would be sufficient for a power of 0.80 for a two tailed test, see Table 5.10 (Hinckle et al., 1994, p637).

Table 5.11 Sample size recruited to the prospective study

Data	Time	Sample	
point		size	
		Recruited	
T1		34	
T2		31	
Т3		29	

Change over time in this study posed more of a challenge because of small sample sizes due to attrition (see Table 5.11). However, if a power of 0.75 and large effect size of d=1.5 is accepted, a minimum sample of 9 per group would detect differences between 3 levels of measurement (Hinckle et al., 1994, p 638). Thus, it was possible to use inferential statistics to compare groups within the sample, but the power of the results was reduced from 0.8 to 0.75 to accommodate small sample sizes. However, the results would still have to be interpreted with caution.

Nonetheless, as previously discussed, group comparison can only give broad notions about group mean behaviour over time and does not provide us with detail about intra-individual motor change. Other researchers have highlighted this in the field of DCD and King and colleagues (2011) provide a useful critique of some of the previously adopted methods, please refer to the previous chapter for a discussion.

#### Summary

By exploring the sample size, power and choice of statistical analysis in peer reviewed papers published on DCD, it helped to distinguish whether using inferential statistical methods would be viable for this study and what effect size to expect. Adopting a Neyman-Pearson's approach to hypothesis testing, determined the power and sample size a priori, along with setting alpha at 0.05 and testing for a large effect made the statistical analysis more robust and transparent.

This influenced the choice of method of analysis for this study and it was concluded that the sample size recruited would be adequate to undertake statistical analysis, providing the effect size and power were quoted. The methods are detailed in table 5.12

Table 5.12 Investigations and main analysis for this study

<ol> <li>Sample characteristics which may influence analysis and outcome (e.g. Age, gender, IQ, SES, AC)</li> </ol>	Frequency distribution, Komomgorov – Smirnov, ANOVA of children with and without movement disorder	
2. Sample characteristics of RED, AMBER & GREEN groups at baseline	Descriptive comparison based on MABC2 total score and comparison with previous studies	
3. Motor change over time and group comparison	Individual case series total score of MABC2 ≥ 10 point change, repeated measures ANOVA	
4. CSAPPA change over time and group comparison	Total score of CSAPPA, Repeated measures ANOVA	
5. Contribution of additional factors that could influence motor development (e.g. car ownership, parent attitude, SES)	Post hoc analysis Case study contextual data Supported by case interviews	
<ol> <li>6. Impact of Associated Characteristics of children on their acquisition of motor skills</li> </ol>	Correlation analysis	
7. Impact of Associated Characteristics of children on their participation in extra curricular physical activity	Correlation analysis Supported by case interviews	

Another consideration is that ANOVA assumes equal group size and equal homogeneity of variance, which may not be the case with the data. Although these approaches have been frequently adopted in DCD research, they do not capture individual differences or individual change over time.

In order to address this a case study approach was also adopted to highlight individual cases and examine them over time.

### 5.12 Case study

The extant literature agrees that DCD is complex and frequently co-occurs with other developmental conditions such as ASD or ADHD, but how each family responds to these will differ. Each family is unique and will have differing attitudes and will place different values on physical activity. The families will also have different resources available to them. Furthermore, development of a child's motor proficiency depends upon numerous factors such as perceived self- competency (Skinner & Piek, 2001), motivation, interest and enjoyment (Cairney et al., 2007), the affordances endowed by a child's physique (Sugden & Wade, 2013, p.65), prior experience and the resources available (DSM5, APA, 2013). One measurement tool cannot therefore capture all of these factors. Moreover, as the change in motor proficiency over time in this study marked developmental change and we have already heard that child development is a dynamic interaction between a child and his or her environment, it was important to acknowledge this interaction by incorporating an ecological design through Bronfenbrenner's bioecological framework.

Case study methodology was therefore best placed to facilitate investigation between some of these factors and allow integration of the analysis of the layers of data of how and why change over time may have occurred. Yin (1984) described case study as an embedded design that is multiple levels of analysis in one study. The multiple levels in this study comprised questionnaires on parents' attitude and parent resources, child self-efficacy questionnaires, contextual data, child-standardized measures of IQ, direct observations and child interview, as well as objective measurements of change in motor performance ability over time. These represented the micro- and meso- system environments that each child encountered and Bronfenbrenner's bio ecological model provided a framework for analysis of these data.

Thus by combining multiple data collection methods in this study it was possible to examine the quantitative data for any relationships and use the qualitative data to understand the context for rationale or theory underlying those relationships. It is this ability to probe deeply into the peoples thinking, beliefs and feelings about their actions that allows us to understand what happens and why it happens (Arksey & Knight, 1999). Interview is an effective way of collecting this data and the selection of cases is crucial, as the population from which they are chosen defines the set of entities and the limits for generalization (Eisenhardt, 1989).

The population in this study was children attending mainstream school and, so importantly, the cases were selected from a school population rather than a clinical population in order to capture some undiagnosed children with DCD as it was hypothesized that they may have developed helpful strategies to remain 'under the radar' of services. The child interview data supplemented the detailed baseline profile of characteristics and measurement of change for specific cases. Silverman (1997) suggested that the strength of qualitative research lies in its flexibility and ability to reveal the unexpected, if the researcher is open to new data. Thus by interviewing the children in school and observing their interactions with staff, peers and the interviewer, a more detailed picture of each case was constructed. Furthermore, an additional advantage of adopting a case study method is that it allows this flexible and opportunistic data collection to examine evidence for the 'why' behind relationships and this helps build internal validity in the study (Eisenhardt, 1989). Case study method was therefore chosen to strengthen the internal validity.

### 5.12.1.Use of theory and previous research to guide data collection

As there are no clear pre-established set of outcomes about the nature and progression of different profiles of DCD, an exploratory approach to case study was used in this study. Stake suggested that a number of cases could be studied jointly to inquire into a condition, population or phenomenon; he called this a collective case study and classified it as an instrumental study extended to several cases (Stake, 1994). Each case in this study was specifically chosen in order to advance the understanding of the characteristics and progression of DCD over time. The cases were chosen using the framework described in Fig. 5.3 and their ecological context was also investigated to help conceptual and theoretical development about how different influences effected progression for children with DCD and without DCD and whether or not associated conditions impacted outcomes in their experiences of participation in extra curricular PA.

Yin (2009) advised the development of a theory prior to collection of any case study data to provide guidance about what data to collect and strategies for its analysis. The theory can be as simple as a hypothetical story predicting why events, thoughts and behaviours occur or consulting existing work, which may already provide a good theoretical framework. In this study existing work guided the pre-specified criteria of self-perception (Cairney et al. 2005; Hay, 1992; Skinner & Piek, 2001), friends (Poulsen et al. 2008), family support and resources (Welk, Wood & Morss, 2003; Zecevic et al., 2010), type of associated characteristics (Green et al., 2006; Missiuna et al., 2014;) and severity of motor impairment (Poulsen, Ziviani & Cuskelly, 2007) to explore in relation to the experience of children's participation in PA.

The benefits of using a theory are a stronger design and better ability to interpret the eventual data (Yin, 2009). Yin (2009) also acknowledged that for some topics there may be little existing available literature or knowledge providing a conceptual framework, as is the case for an exploratory study, and this was the case for the progression of children with the different severities of

DCD. In this case Yin (2009) advocated the use of a wider range of theories that may be relevant to the study, such as theories of cognitive or social development or learning and social interaction, self efficacy and play theories, as this is the level at which generalization of the case study results occur. This study drew on theories of motor development; dynamic ecological explanations for participation and child development and Bronfenbrenner's bioecological model provided the framework with which to integrate the data.

### 5.12.2 Framework for determining cases:

Please refer to Fig.5.2 for the procedure for identifying and selecting cases to ensure that the heterogeneity of the school population sample was represented.

The table 5.13 shows the rationale for selecting the cases of children who represented the full range of profile characteristics in order to interview and examine the experiences from each category. This sampling method was theoretical rather than statistical and allowed the possibility of comparing polar types (Eisenhardt, 1989). For example, comparing a child with typical motor development and associated characteristics with a child with DCD and associated characteristics in order to extend or replicate any emerging theory or hypothesis.

Groups	Plus Associated	Without Associated
	Characteristics	Characteristics
RED DCD≤5 <sup>th</sup> MABC2	N=1	N=1
AMBER 6-16 <sup>th</sup> MABC2	N=1	N=2
GREEN ≥25 <sup>th</sup> MABC2	N=2	N=3

 Table 5.13 Rationale for selecting cases for interview and number of children interviewed in each category

In order to answer the research question 5.1.3 'How do the characteristics and stability or change in motor ability and self perceptions impact on experiences of participation in physical activity from the child's perspective?' cases had to be selected from each of the groups that had particular characteristics and had either stable or changing profiles. A further caveat was whether the child participated in extra curricular activity or not.

As the design incorporated Morse's priority sequence model of Quant-Qual sequential mixed methods (1991), the children's characteristics and profiles were determined from analysis of the quantitative data from the MABC2, the DCDQ (07) and the CSAPPA and then specific children who demonstrated particular profiles were selected. Thus selections of children from all motor groups were chosen (see Table 5.14) and this is illustrated with an example of the profiles chosen for the Red group (severe DCD) that included:

- A child with severe DCD plus associated characteristics who did not participate in extra curricular PA (the norm for the Red group)
- A child with severe DCD plus associated characteristics who did participate in extra curricular PA (exception for the Red group)
- A child with severe DCD only who did participate in extra curricular PA (exception for the Red group).

This method also permitted cross case analysis for both similar or opposite cases, thereby extending the type of analysis possible.

The subjective account of how a child makes sense of their situation can illuminate reasons for their participation with people, situations, and tasks and determine how they progress. This was elicited by individual interview with the researcher and the procedure outlined in Table 5.14 informed the selection of children for the interviews.
A child in Red Group ≤ 5 <sup>th</sup> percentile MABC2		A child in A Group 6-16 <sup>th</sup> perce MABC2	mber ntile	A child in Green Grou ≥25 <sup>th</sup> percentile MAB	
With AC	No AC	With AC	No AC	With AC	No AC
Who	Who	Who	Who	Who	Who
Participate	Participate	Participate	Participate	Participate	Participate
s in PA	s in PA	s in PA	s in PA	s in PA	s in PA
Who does	Who does	Who does	Who does	Who does	Who does
not	not	not	not	not	not
participate	participate	participate	participate	participate	participate
in PA	in PA	in PA	in PA	in PA	in PA

Table 5.14 Case selection criteria for case study methodology

### 5.12.3 Rationale for the interview schedule:

Previous research in the field of DCD in children has highlighted the potential impact of DCD on numerous activities in daily life (Missiuna, Moll, Law, King & King, 2006; Missiuna, Moll, King, King & Law, 2007) and it was important to capture this information.

The ICF (WHO, 2001) provides a useful framework for examining participation for children, as it examines participation under categories of work, leisure and activities of daily living, and so provided logical areas to investigate in this study. Moreover, the ICF acknowledges the person and environment as dynamic and interactive dimensions in an ecological approach similar to that of Bronfenbrenner (Stewart & Rosenbaum, 2003) and so is compatible with the study design. Therefore, arts-based semi-structured interviews focusing on participation in self-care, work and leisure, and in particular, participation in physical activity were therefore chosen for this study. These areas were guided by categories from the ICF and from previous research in DCD (Missiuna, et al., 2006; Missiuna, et al., 2007). This resulted in a series of photographs depicting

activities for each category to act as a visual prompt if required to elicit responses. Furthermore, the social and cultural environments are also important as a child's participation in activities is also affected by attitudes, values and beliefs (Law et al., 1999) and therefore data on child and parent attitude supplemented the interviews. Observations on the wider cultural environment, such as media portrayal of gender roles and curricular time allocated to PA also augmented the data.

#### Please refer to the interview schedule in appendix 7

In addition to the areas described in the ICF the schedule, grand tour questions were asked aiming for insights from the children about difficulties and successful participation.

#### Summary

Qualitative research was used to enable the child to describe his or her context in detail to illuminate underlying processes, motivations and possible explanations for phenomena or set of behaviours, which may well be context dependent. Group means and statistical analysis could only reveal general trends and so more detailed analysis, with a case study approach augmented with interviews and examined particular cases within their context. As this study examined the progression of motor profiles of children with DCD it was crucial to include rich description of their everyday context and their subjective experiences in order to determine any potential causal explanations for patterns of change over two academic years and their outcomes. The questions most appropriate for case study research are concerned with how things happen and why (Anderson, 1998; Bryman, 2010). The research question was concerned with how the motor profiles of children with different severities of DCD change over time and the factors associated with why the particular changes occur and the impact they have on participation in physical activity, case study was therefore an appropriate framework.

## 5.13 Reflections on the unique features of the study

In summary the DCD literature has identified potential sources of heterogeneity, which may be problematic for research, and this design aimed to account for, and describe them. Data were documented at the baseline for each child and included:

- the severity of motor impairment (whether ≤ 5<sup>th</sup> percentile on MABC2 or 6-16<sup>th</sup> percentile)
- the presence or absence of co-occurring associated characteristics
- the level of IQ
- the age and sex
- the parent's opinion on the impact of motor impairment on daily life.

However, there is also complexity surrounding the identification of children with DCD and it is, to a certain extent, dependent upon the measures employed, their specificity and sensitivity and the cut-off scores chosen. There is also the potential for discrepancy between the measures and between the various people who associate with the child. Identification of psychometrically sound measures is essential to try and ameliorate this, but it is also apparent that consensus guidelines are very important. Both these approaches have been adopted in this study.

An important part of the study was to examine any change in motor performance of the children with DCD over time to help better understand their development. Development is a complex process involving dynamic interaction between the child and his or her environment. However, for any causal inferences to be made about atypical development a study must include typically developing peers for comparison. This study included peers who had been identified without movement difficulties, yet the screening process also identified that some of them had associated characteristics. This highlighted the importance of identifying the profile of all the study participants at baseline before tracking their progression over time. However, the measurement of change in motor performance is not itself a straightforward process, as there is no consensus about how meaningful change should be identified and different researchers have employed different criteria. This study adopted three approaches: the MABC2 traffic light system, motor group change and MABC2 total score  $\geq 10$  points (larger than the standard error measurement) called the smallest detectable change (SDC) (Holm et al, 2013).

Furthermore, the importance of environmental context in child development includes family context and the resources available to the child, as these have been found to impact progress. Therefore, the family context of each child was also documented and included:

- the parents employment status
- family car ownership
- number of school aged children in the family
- parents health status
- family history of developmental disorder

Additionally, literature has highlighted that family attitudes can have a role to play in participation rates in physical activity and that movement experience impacts upon motor performance. For this reason the parents were asked about:

- how important they ranked participation in physical activity
- how frequently they engaged in physical activity
- how severe the impact of their child's motor impairment impacted family life

Moreover, self-efficacy has been discussed in DCD literature and also been identified as a potential factor in child participation rates in physical activity. Children in this study were therefore questioned about their self –efficacy and enjoyment of participating in physical activity.

Statistical analysis has also come under scrutiny in DCD research and is particularly pertinent for such a heterogeneous population, as homogeneity of variance is easily violated rendering tests such as ANOVA less reliable. The utility of general linear models (GLM) and comparing group means are also questioned because small sample and unequal sample sizes are all problematic for these tests, yet much more likely in DCD research. In addition, the intragroup differences in populations with DCD may be more important than the inter-group differences with typically developing children. Individual case series approach to motor change permits examination of intra-group variability, which may be an important indicator in future research in DCD. A more flexible analysis of the data considering different aspects, such as the real world impact of impaired motor performance, participation in physical activity and daily life has been added to supplement statistical analysis in this study. Furthermore, the child's own perspective has been sought in interviews of selected cases of children with particular profiles in order to more fully understand the impact of different profiles on participation. Adopting an ecological approach facilitated a more comprehensive analysis of the nature of DCD, its change over time and the perceived impact on participation in physical activity.

## **CHAPTER 6 RESULTS**

# 6.1 Findings from Research Question: How do the profiles of children with DCD change over time?

- 6.1.1 What are the characteristics of children with different severities of DCD compared to children without DCD? These will include their:
  - Motor characteristics
  - Associated characteristics
  - IQ
  - Ability to take part in everyday activities
  - Socio-economic context and family context
  - Self- perception of their adequacy, enjoyment and predilection to participate in physical activity
- 6.1.2.1 How stable are the motor characteristics or do they change over time?
- 6.1.2.2 How consistent are their self-perceptions of their adequacy, enjoyment and predilection to participate in physical activity or do they change over time?
- 6.1.3 How do the characteristics and stability or change in motor ability and self-perception impact experiences of participation in physical activity from the child's perspective?

The following sections describe the participants who took part in the study and how they were identified. The results are then presented in three parts.

Part I describes the fine-grained detail of the characteristics of children with and without DCD at the baseline of the study. This included any known co-occurring conditions or characteristics associated with co-occurring conditions, an

estimation of their IQ (including verbal and non-verbal components), the extent to which their parents considered their motor difficulty impacted daily life, demographic characteristics, and the child's perception of their motor abilities. Since the identification of children with DCD is complex and can be problematic, an analysis of the consensus between the different sources of identification of the children with motor impairment is also examined. Analysis of the characteristics that brought the children with movement problems to the attention of teachers can help us understand part of the complexity of identifying DCD. So there is a description of the comparison of the characteristics of the children who were classified by those with motor impairment (i.e. ≤ 16<sup>th</sup> percentile) and without motor impairment (>25<sup>th</sup> percentile chosen to clearly differentiate the typical children) on MABC2 respectively and any differences are presented. The children were then further classified according to the severity of motor performance ability impairment into Red (severe ≤5<sup>th</sup> percentile), Amber (moderate 6-16<sup>th</sup> percentile) and Green (typical motor ability  $\geq 25^{\text{th}}$  percentile) groups for more fine grained analysis in order to answer research question 6.1.1. Any difference between the groups at baseline is described and presented.

Part II explores how we look at change in motor ability over time and change in self-perceptions of enjoyment, adequacy and predilection to participate in physical activity over time. The three groups are investigated for any differences in stability or change in their actual motor performance (tested by MABC2) and simultaneously investigated for stability or change in their perceptions of motor performance (tested by CSAPPA) over 3 data points. These relate to findings from research questions 6.1.2.1 and 6.1.2.2 respectively. The progress of children with different severities of DCD is described and compared to typically developing children.

Part III explores the impact of their characteristics and self-perceptions on the experiences of the children with DCD to participate in physical activity. Selected case studies are presented to illustrate the progression of some

children from each group and explore some of their contextual data in relation to Bronfenbrenner's Bio ecological model. This will help us understand the interactions of the child with DCD and their particular characteristics with their specific family and school environments. These are enhanced by analysis of some interview data from the children selected for case studies in order to answer research question 6.1.3 and provide insight into children's experiences of participating in PA from their perspective.

## 6.2 Part I: Sample Description

The sample was recruited from school populations (rather than clinic populations) to eliminate potential referral bias and to identify any undiagnosed children with DCD. The first section describes the entire sample characteristics, which is followed by a section detailing the comparison of children with and without motor impairment and their respective characteristics. Finally, detailed analyses of the characteristics of the Red (≤5th percentile), Amber (6-16th percentile) and Green (≥25th percentile) groups (classified by category of motor performance ability of total score on MABC2) are compared for any differences.

### 6.2.1 Sample characteristics

Data were collected over 2 academic years from 34 children in mainstream schools whose parents had given their consent. Children were recruited from primary and secondary schools, however, the majority of children (85%) attended one primary school with an ASD unit attached. This school was situated in an area of deprivation. The total sample consisted of 29 males and 5 females and ages ranging from 7-14 years (90- 169 months), mean age 120.8 months and standard deviation of 19.8. At the start of the study 14 children were aged 7-9 (41%), 16 children (47%) were aged 10 or 11 years and 4 were 12-14 years (12%).

It was harder to recruit children from secondary schools and more difficult to schedule testing in their timetable, which accounts for the majority of the sample being recruited from primary school.



Figure 6-1 Distribution of ages of the sample at baseline

The preponderance of 10 year olds in the sample may have arisen from the teacher awareness of imminent transition to secondary school and the importance of needs assessment.

	Total	ASD	SLI	ADHD	ODD
	N=34	N=12	N=1	N= 10	N=2
Male	29	12	1	10	2
Female	5	0	0	0	0
Primary school	29	11	1	10	2
Secondary school	5	1	0	0	0

#### Table 6.1 Characteristics of total sample

The children were screened for associated characteristics of additional developmental disorders, such as attention difficulties, language and

communication difficulties. One child (2.9%) screened positive for characteristics of specific language impairment (SLI) and 12 (35.3%) of the children either already had a diagnosis of ASD or screened positive for characteristics of ASD. In addition 10 children (29.4%) screened positive for ADHD, however 9 of these children additionally screened positive for ASD. That is 9 children (26.5%) had characteristics of both ASD and ADHD. One child had a hearing impairment and wore one hearing aid.

#### Figure 6-2 Distribution of associated characteristics



The sample consisted mainly of boys (85.3%) at a ratio of almost 6:1 boys: girls. This may have reflected a bias of teachers identifying movement difficulties more readily in boys, or may just be reflective of the fact that 35% of this sample had ASD, which is diagnosed 4 times more frequently in boys (DSM5, p57, APA, 2013).

However, most studies of DCD describe ratios of between 2:1 to 7:1 boys to girls (DSM5, APA 2013). Furthermore, 85% of the sample was drawn from a primary school with an ASD unit attached, which naturally led to a higher

concentration of children with ASD and their siblings attending the school. The usual prevalence of ASD in UK schools is 1-2% (Baron-Cohen et al., 2009).

### 6.2.2 Identification of the sample

The way the sample was identified and classified is illustrated in figure 6.3

Teachers used their professional judgment to identify children without motor difficulties or intellectual impairment and identify those with motor difficulties. This is often how children with DCD are first referred to services and it is important to ascertain how effective teachers are at identifying potential motor problems. Therefore, a comparison of results from the different tools used to identify the children with motor difficulties is also reported.

#### Figure 6-3 Sample identification and classification



The children were screened using parent and child questionnaires and finally assessed individually by a qualified OT with a standardized, culturally appropriate movement assessment to identify the children who scored in the lowest 16th percentile, as recommended by the EACD guidelines (Blank et al., 2012). Any child who scored at or below the 16<sup>th</sup> percentile is described as 'at risk' of motor difficulties and those who score at or below the 5<sup>th</sup> percentile are described as having 'significant' movement difficulties (Henderson, Sugden & Barnett, 2007).

The flow chart (Fig. 5.2) in the methodology details the procedure; the number of children identified at each stage and the numbers classified in each group at data point 1 (T1) or the baseline of the study.

## 6.2.3 The context and characteristics of the sample at baseline

The context and characteristics of the sample in this study at baseline are important because they are a little unusual. Thirty-five percent of the sample had ASD and the IQ range was large, from 55 (moderated learning difficulties) to 131 (above average). In addition 41% of the children had a family history of developmental disorder and at least a third of the children received free school meals, indicating a level of financial hardship for the families.

A summary is presented in table 6.3 about the characteristics of the children in the study sample. However, the ecological context of the neighbourhood, and the educational context of the school were also considered within Bronfenbrenner's framework.

### 6.2.3.1 Macrosystem level: the neighbourhood borough

The geographical area in the study had an average level of unemployment and an average utilisation of outdoor space for exercise, when compared to England as a whole. However, the number of children in low income families was higher than average (19.6 compared to 17.0 for England) and the income deprivation indices based on 2015 figures was higher than average (20.1 compare to 14.6 for England) (Marmot indicators, Public Health England, 2019).

### 6.2.3.2 Exosystem level: educational context of the borough

There was a higher incidence of learning difficulty and moderate learning difficulty when compare to the national average (PHE, 2019). There was also

twice the national average of free school meals in the area and a higher rate of children with special educational needs.

	Neighbourhood	England
Moderate learning difficulties	39.3 per 1000	28.9 per 1000
Autism	13.3 per 1000	13.7 per 1000
Leaning difficulty	44.1 per 1000	33.9 per 1000
Free school meals	26%	13.5%
Special Educational Needs at primary school	16.3%	13.8%

#### Table 6.2 Comparison of sample neighbourhood

### 6.2.3.3 The mesosystem level: the school

The primary school was an average size (234 pupils attended in 2014), mainly white British pupils, but well above average proportion of pupils were supported at school action plus or had a statement of special educational needs. The proportion of pupils eligible for support through pupil premium funding was almost three times the national average in 2014, with 67.5% of pupils eligible for free school meals. There were two specially resourced bases, one for each key stage, supporting up to 14 pupils with autism from across the borough.

The Ofsted report 2014 rated the school 'Good' and commented, "This is a happy and caring school. It provides a calm and nurturing environment, particularly for those pupils who attended the specially resourced bases. As a result, pupils achieve well given their differing abilities." (OFSTED, 2014).

#### 6.2.3.4 The microsystem level: the children and families

After considering all the information the children were initially categorized and divided into:

- Those with movement impairment substantially below that expected for their age and opportunity DCD (≤16<sup>th</sup> percentile of MABC2) and
- ii) Those with no movement impairment potentially typically developing (≥25<sup>th</sup> percentile of MABC2, pTDC).

This was to determine any differences between the motor impaired children and those without a motor impairment at baseline and may also allow some insight into the characteristics that alert teachers to identify children with DCD.

	Entire sample n=34	Mean (sd)
Age in months: range	Range 90-169	120.8 (19.8)
IQ: range	Range 55-131	95.2 (20.5)
Number with ASD	12	
Number with ADHD	9	
Number with ODD	2	
Number with other	2	
SES free school	11 (7 did not	
Family history of developmental	14 (41.2%) known but others did not	

#### Table 6.3 Characteristics of the study sample

# 6.3 Comparison of the characteristics of motor impaired and non-impaired children

The gender distribution was roughly equivalent between the motor impaired and non-motor impaired groups, although the mean age was slightly older for the non-impaired group. The children in the motor impaired group had a higher prevalence of associated characteristics and both their general IQ and verbal IQ scores were lower than the non-motor impaired group. However, the number of children with a significant difference between their verbal and non-verbal IQ scores was higher in the non-motor impaired group. A large discrepancy here is considered abnormal in the general population (Kaufman & Kaufman, 2004, p34) and this result may indicate that the sample of non-motor impaired children (TDC\*) may not be typically developing in domains other than motor.

	DCD (movement	TDC* (non-impaired
	impaired) n=17	movement) n=17
Gender	Male n=14	Male n=15
	Female n=3	Female n=2
Age	Mean = 117.6 (sd 18.8)	Mean = 124.1 (Sd 20.9)
	Range=90-161	Range=94-168
MABC2 TI score	Mean = 45.7 (sd 17.2)	Mean = 77.5 (sd 7.7)
	Range =16-67	Range 69-94
KBIT2 General IQ	Mean = 83.8 (sd 19.5)	Mean = 106.5 (sd 14.6)
	Range= 55-122	Range=71-131
KBIT2 Verbal IQ	Mean = 85.7 (sd 15.5)	Mean = 109.5 (sd 13.4)
	Range= 52-122	Range= 86-132
Difference between	N=7	N=9
VIQ & NVIQ	41.2%	64.3%
Sig. to 0.05 or 0.01		
Children with	Attention = 8	Attention=2
Associated	ODD =1	ODD =1
characteristics	ASD=11	ASD= 2 SLI=1

Table 6.4 Comparison of movement impaired and non-movement impaired groups at baseline

Thus some differences appear between the group means for motor impaired and non-motor impaired children at the start of the study for age, IQ and associated characteristics. These were investigated further with statistical analysis and are reported later.

A comparison of some characteristics related to performance at school was also performed between the motor impaired and the non-motor impaired groups. The results are presented in table 6.5.

	DCD (movement impaired) n=17	TDC* (non-impaired movement) n=17
Right handed	N= 11 (64.7%)	N= 11 (64.7%)
Poor Handwriting	N=15 (88.2%)	N=4 (23.5%)
Any SEN provision	N=13 (76.5%)	N=4 (22.2%)

Table 6.5 Comparison of school performance between the groups

However, there was equivalent distribution of right-handed children in each group in this sample, although the DCD group had a greater incidence of handwriting difficulty (82% compared to 23.5%). The DCD group also had a higher incidence of children with special educational needs (76.5% compared to 22.2%). Perhaps this was unsurprising given that the DCD group had a lower mean IQ than the non-impaired motor group (83.8 compared to 106.5). Perhaps, predictably, the DCD group had considerably more difficulties than the non-impaired group and it would be logical to hypothesize that they would have lower self-perception and participation in PA.

Behavioural and functional outcomes are not solely determined by biological characteristics but by the bidirectional interaction of child and the environment (Bergman, Eklund & Magnusson, 1994). Therefore, by using Bronfenbrenner's model, investigation of the microsystem i.e. the interrelations

between the developing child and the immediate settings such as the family allows another layer of analysis. So too does investigation of the interrelations of the child in the mesosystem, i.e. major settings at a particular point in life (e.g. school) and the exosystem (such as parent employment), which influence (delimit, or even determine) the child's developmental setting (Lerner, 2005). Therefore contextual data concerning each child and their family environment were analysed to help establish possible influences on the child's motor progress. A summary of the parent questionnaire is presented in table 6.5. This summary was based only on the responses that were returned. Overall response rate was 91.18% but not all were complete. Complete response rate was 76.5%.

#### 6.3.1 The Family context:

The families with the children who had motor impairment had a higher prevalence of developmental disorder in their family, perhaps indicating a familial or genetic predisposition to neurodevelopmental disorders. However, they had a lower prevalence of some of the indicators of deprivation, such as free school meals, unemployment and no family transport. Moreover, fewer of these families had two or more children less than 18 years in the household and fewer parental disability or health problems, indicating a slightly more favourable context than the non-impaired group. Despite this group having lower levels of maternal education, they generally had a slightly more favourable context, in terms of family resources, than the non motor-impaired group. An understanding of level of a child's resources and deprivation is important because previous studies have shown that these can impact levels of participation (see King, Law et al., 2003 for a review).

 Table 6.6 Summary of Parent Questionnaire based on MABC2 category at baseline

	Motor impaired (DCD) n=17	TDC* (non-impaired movement) n=17
1 <sup>st</sup> degree relative with developmental disorder	69.2%	50%
Single parent	31.3%	35.3%
2 or more children <18 years in the family	64.3%	79.5%
Parent health problems	23.1%	55.3%
Parents not employed	14.3%	41.2%
Mother education  >GCSE	66.7%	71.4%
Free school meals	33.3%	46.7%
No family car	30.8%	37.5%
Importance of PA to family >score of 2	75%	58.8%
Impact of child's motor difficulty on family > score of 2	75%	23.5%
Freq of parent PA weekly or more	66.7%	68.8%

Family attitude to physical activity is an important motivator for participation (NICE Public Health collaborating Centre-Physical activity, 2007). In this study the parents of children with DCD appeared to rank importance of physical activity higher than the families without motor impaired children, and participated themselves at a similar frequency to the parents of the non-impaired children. Therefore, the DCD group had fewer attitudinal barriers to participation in PA than the non-impaired group. However, unsurprisingly they ranked the impact of their child's motor impairment on family life as high. The severity of the motor impairment was considered to have the potential to impact participation in PA and was therefore also investigated and is reported later.

#### Summary

In terms of their family context, the children with typical motor ability appeared more disadvantaged in relation to their available resources than the children with motor impairment. This was because this group had more parents with health problems (53% compared to 23%), a greater proportion of unemployed parents (41% compared to 14%), higher proportion of families with more than 2 children under 18 years (nearly 80% compared to 64%) and more families entitled to free school meals (nearly 47% compared to 33%). Yet, both groups had similar rates of single parent households, maternal level of education and parent participation in physical activity. However, the parents of children in the motor impaired group rated importance of PA higher than the parents of children with out motor impairment (75% compared to 59%) and unsurprisingly, rated the impact of motor difficulties on family life much higher (75% compared to 23%). An additional potential barrier to participation for the families with the children with motor impairment was the higher rate of first-degree relatives with a developmental disorder (nearly 70% compared to 50%). Both these figures seem higher than expected, but majority of the sample was drawn from a school with an ASD unit attached and this may have impacted the results.

The children with typical motor ability had greater potential resource barriers to participation in extra curricular PA for and a lower ranking of importance of PA by their parents. However, the families of children with motor impairment had identified a negative impact of their child's difficulties on family life, despite ranking the importance of PA higher. They may have more barriers to participation in extra curricular PA for their children than captured by the parent questionnaire. More information over time was required to unpick some of the complexity, particularly regarding the severity of the motor impairment and impact on participation. Motor impairment is a core feature of DCD and the identification of the disorder in this study entails fulfilling criteria of DSM 5 (APA, 2013).

As previously highlighted, researchers have found that the identification specifically of DCD can be complex and involves gathering data from different sources (Blank et al., 2012). However, not all tools identify the same children (Barnett, 2008) and not all professionals can easily identify motor performance difficulty (Harris, Zwicker & Mickelson 2015). Therefore a comparison of data collected from the different sources in this study was undertaken, in order to better understand the issues, and the findings are discussed.

# 6.4 Findings from comparison of identification of movement difficulty

The EACD guidelines (Blank et al., 2012) advocate the importance of using multiple sources for the identification of children with DCD, but it is unclear whether all sources are of equal merit. Therefore, a comparison of the results from each source of identification was undertaken. As a result of following the above procedure

- Teachers initially identified 18 children that they thought had movement problems and 16 that they considered to be typically developing with no intellectual impairment.
- Parents using the DCDQ (07) screening tool identified 16 children they thought had difficulty with everyday tasks, who were suspected of having DCD

 Children using the CSAPPA self-rated tool identified only 8 children with low self-efficacy, enjoyment and predilection to take part in physical activity, who scored in the suspect DCD range.

Method of identification of movement problems	Movement impaired group	Non- impaired movement group	Total	Missing data
Teacher id	N=18	N=16	34	0
Parent rated DCDQ (07)	N=16	N=15	31	3
Child rated CSAPPA	N=8	N=26	34	0
Researcher (OT) MABC2 test total score ≤ 16th	N=17	N=17	34	0

Table 6.7 Different methods to identify motor impairment

In order to avoid false positive results the cut-off score for DCDQ (07) and CSAPPA was set at  $\leq 5^{th}$  percentile, as these are both screening tools. However, the MABC2 cut off score was  $\leq 16^{th}$  percentile (at risk range) in line with the EACD guidelines (Blank et al., 2012). Previous studies have identified that children who score in the at risk range experience difficulties in participation in daily living (Wang, Tseng et al., 2009) and for this reason they have been included.

It came as a surprise to the teachers that five children they had considered not to have any movement impairment tested in the impaired range and three children they thought had impairments tested in the typical range.

## 6.4.1 Closer analysis of the identification and assessment of DCD at baseline

The identification and assessment of DCD is not a straightforward process and involves collecting information from a number of different sources, which may not necessarily agree and potentially identifies different children, as this small sample illustrates.

As previously mentioned the EACD guidelines (Blank et al., 2012) recommend gathering data from different sources and additional testing with an appropriate standardized motor test. Table 6.7 shows how many children were subsequently verified by MABC2 (Henderson, Sugden & Barnett, 2007).



Figure 6-4 Numbers of children identified by parents, teachers & MABC2

Interestingly, parents and teachers identified 5 children suspected of movement difficulty who scored above 25<sup>th</sup> percentile on MABC2. Two of these children had a diagnosis of ASD, one had ADHD, one had ODD and one had unknown early development. However, over time it became apparent that 3 of these children did go on to demonstrate movement difficulty and associated problems. The other two also had associated characteristics, one with ASD and the other ADHD, but they consistently scored above the 16<sup>th</sup> percentile on MABC2 over

time. Conversely, one child was identified by his parent alone and confirmed by MABC2 with severe motor impairment, both at baseline and over time. Additionally, 3 children were identified by MABC2 alone, 2 of whom were girls and were not confident with motor skills but improved their MABC2 score over time and a boy who scored in the impaired range over time. He was without associated characteristics and continued to score consistently  $\leq 16^{th}$  percentile on MABC2 over time and yet, despite this, was not identified by parent or teacher as having movement difficulties.

Furthermore, two boys identified by parents, but not teachers, as having movement difficulty yet consistently scored in the typically developing range on MABC2 over time. One boy was very small for his age and the other had a diagnosis of ASD, both of which may account for the issues raised by the parents of these children.

Of the sample of 34 children, 17 were identified as having movement difficulties when tested with the MABC2. Of these only nine children, were identified by both teachers and parents as having movement difficulties, and were subsequently verified by MABC2 score (i.e. 52.9% of those correctly identified with movement difficulty) all of these children had a diagnosis of ASD, 8 of whom had additional symptoms of ADHD. This number may have been higher, as there was missing data from the parents of a further three children who all scored  $\leq 16^{th}$  percentile on MABC2.

#### Summary

Teachers and parents identified just over half of the children with movement difficulties from a school population, but interestingly those identified by both also had additional associated characteristics and scored in the severe range i.e. ≤5<sup>th</sup> percentile on MABC2. However, additionally both teachers and parents also identified children who did not score in the impaired range, but had additional associated characteristics. Crucially, however, parents correctly

identified two children with movement problems that their teachers had not identified, which were subsequently confirmed by the MABC2. This data illustrated the point raised about the requirement of 80% sensitivity and 90% specificity of screening instruments in population based samples (APA, 2000) and demonstrated that identification and assessment of DCD to DSM5 criteria is a complex process. The picture was further complicated by change in motor performance ability over time, as the subsequent data will show.

#### 6.4.2 Discrepancy between tools for identification of DCD

In this study the difference in the identification of movement difficulties by different assessments is highlighted by the fact that the DCDQ (07) identified different children to the CSAPPA and their movement difficulties were not necessarily observed using the MABC2. This presents some interesting issues about DCD, adherence to DSM 5 criteria and how they relate to participation and these will be discussed later.

For example, parents using the DCDQ (07) identified five children who scored above the 16<sup>th</sup> percentile on the MABC2 (i.e. in the typically developing range). However, their parents had sufficient concerns about them to rate them low on the DCDQ (07) questionnaire, indicating functional difficulties. Therefore, the profiles of results for these children were examined in detail as they progressed over time. Four out of the five children proceeded to display difficulties in subsequent movement tests (taking them to either the 'at risk' or 'severe' motor difficulties range) and the remaining one (despite remaining in the typically developing range) had difficulties with handwriting that required remediation. This indicated that the parents had correctly identified functional difficulties, which may have been more subtle, and only been revealed on standardized tests over time.

### 6.4.3 Identification of DCD by use of child report (CSAPPA)

The CSAPPA also provides useful information about how each child perceives their movement ability. In this study only eight of 34 children perceived that they had difficulties and only four of these said that they did not enjoy physical activities (PA). Interestingly, two out of the four scored above the 16<sup>th</sup> percentile on the MABC2. Nevertheless, it provided some useful data about how each child perceives their enjoyment and inclination to participate in physical activities. Despite quite severe impairment in movement ability, eight out of the 10 children who scored ≤5<sup>th</sup> percentile on MABC2 stated that they enjoyed physical activity and six stated they felt reasonably competent. This indicated that they did not perceive their motor impairment to impede their participation in PA. Only four children from the entire sample stated that they did not enjoy physical activity and two of these scored above the 5<sup>th</sup> percentile on the MABC2. The CSAPPA scores appeared not to relate to the severity of their motor performance and therefore the CSAPPA, as a screening tool for DCD, did not appear helpful for identifying DCD for this sample. A more detailed analysis will follow.

#### Summary

It is evident that identification of children with DCD can be problematic, with different tools identifying different children. Furthermore, parents, teachers and children identified different priorities as challenging and MABC2 assessed specific motor performance ability as a snapshot at a given time. Overlap of agreement for parents, teachers and MABC2 only occurred for 53% of the sample at baseline (leaving 47% without consensus initially). Monitoring of motor performance over time was therefore important in order to verify any difficulties not recognized by other stakeholders.

The children were further categorized into groups by their movement performance ability on MABC2 to facilitate a more fine-grained analysis by motor severity of DCD.

### 6.5 The severity of motor impairment in DCD

The DCD group covered a wide range of movement impairment and this group also had a large number of associated characteristics. Many studies include children who score below the 5<sup>th</sup> centile, as well as those scoring between the 5<sup>th</sup> and the 15<sup>th</sup>, yet refer to the whole group as DCD (Farhat et al., 2016,). Smits-Engelsman et al. (2015) reviewed studies from 2010-2014 to determine how criterion A was operationalized. They found that 48% of the studies used 5-15<sup>th</sup> percentile cut-off and employed terms such as 'at risk', 'moderate' and probable DCD' and 24% of the studies used the 5<sup>th</sup> percentile cut-off and referred to 'severe DCD' or 'DCD', whilst 6.2% of the studies used other cut-off scores < 15<sup>th</sup> percentile and 22% did not report a cut-off score at all. However, not enough is known about these categories and whether they behave differently over time. Therefore, more information about the severity of movement impairment is required to help determine the nature of interaction and it's influences for children in daily life over time. Of important note in this study, is that the group with no detectable movement impairment also contained children with associated characteristics. This could impact on the study results, so these children were closely followed in and the results will be discussed later.

As the core symptom of DCD is difficulty with coordinated movement, the children were classified into three groups according to the severity of impairment of their movement ability, as measured by the MABC2. This differentiation is important because, as Smits-Engelsman et al. (2015) found, over many pervious studies the results have been reported combining all levels of severity of movement impairment, yet we do not know if all types of severity behave the same over time. Therefore, in order to answer the research question 6.1.1 the children were classified specifically by their motor performance ability.

# 6.5.1 Group Classification by motor performance ability at baseline: Red, Amber and Green

The children were divided into groups using their total scores from the MABC2 at the initial testing point to ascertain the severity of the motor impairment to allow closer examination of the characteristics. The traffic light system advocated by Henderson et al. (2007) was used to classify by severity of their movement difficulties. Please refer to table 5.4 in the methodology.

In order to fully differentiate them from the other motor groups all children allocated to the Green group scored  $\geq 25^{\text{th}}$  percentile on MABC2. This was because some children scoring in the green or typical range for motor ability typical range for motor ability exhibited other atypical associated characteristics.

# 6.6 Differences in characteristics of Red, Amber and Green motor groups at baseline

The characteristics of the children in each motor group were documented and compared and are presented in Table 6.8. This permitted fine-grained differentiation of the children at baseline for later comparison over time.

To examine group differences across a variety of measures one-way analysis of variance (ANOVAs, or non-parametric equivalent) were used. Pairwise comparisons were used to break down significant group effects and significant main effects were analysed using post hoc tests. For all analyses, significance levels were set at .05 and Bonferroni corrections were used to control for multiple comparisons, (and Hochberg test for unequal group size).

#### 6.6.1 Age in months:

The mean ages of each group were examined for differences at baseline (see fig. 6.5), so that any developmental differences may be accounted for at the

start of the study. The box plot (fig. 6.6) showed overlap between the median ages of the groups, suggesting no significant difference between them. However, this was tested for statistical significance. The Kolmogorov-Smirnov test showed the distribution for age was normal for Amber and Green groups, but not the Red group. However, the Shapiro-Wilk test suggested that the age for all groups was normally distributed and after examining the Q-Q plots of each group it was considered close to normal, so permitting a univariate analysis of variance (ANOVA). One-way ANOVA revealed no significant difference between groups in age [F (2,31) = 2.646, p=  $.087 \eta 2$  partial = .146] power .487 and post hoc test supported this.

#### Figure 6-5 Group means for age

#### **Descriptive Statistics**

Dependent Variable: Agemonts

group	Mean	Std. Deviation	Ν
sDCD	109.7000	14.29880	10
mDCD	129.1429	19.34277	7
TDC	124.0588	20.97758	17
Total	120.8824	19.86739	34



Figure 6-6 Box plot of median ages by severity of motor impairment

Summary: there was no significant statistical difference in age between the groups at baseline. Therefore, any differences in change over time during the study may not be attributed to differences in the ages of the groups, as they were not significant.

### 6.6.2. Presence of Associated characteristics

DCD is known to co-occur with other developmental conditions, but the extent to which this occurs was thought to vary with the severity of motor impairment and so each group was analysed and compared specifically for characteristics of impaired communication and attention. The typically developing group was also investigated (see Table 6.8).

There were a significant proportion of children who had a family history of developmental disorders in this study sample (41%), but it due to incomplete data it was difficult to ascertain if they were more concentrated in one particular group. However, the Red group had several important features; it contained no girls and had the highest proportion of children with additional associated characteristics. Therefore, in this sample, a child with severe DCD was much more likely to have associated characteristics than a child in any of the other two groups.

## 6.6.2.1 Associated Characteristics of the Red group (≤5<sup>th</sup> MABC2) or severe DCD

The data was based on 90% response rate of completed questionnaires.

This group of children had a very high rate of associated characteristics in addition to a much more obvious movement disorder.

Further analysis of the composition of the groups revealed more details about the potential variability and the nature of the characteristics of children with DCD. The Red group appeared to have more children with each of the associated characteristics with attention, language and communication difficulties than either of the other two groups (see Table 6.9) Furthermore, the parents were able to identify this at a much higher rate than the other groups, 90% had their movement difficulties identified by their parents It may have been because their children already had an ASD diagnosis and 80% had two or more associated characteristics (e.g. DCD plus ASD plus ADHD characteristics). This group also had the highest rate (50%) of children identifying that they had difficulty participating in physical activity. However, one child in this group did not have any associated characteristics, was not identified with movement difficulty by his teacher, or by himself but was correctly identified by his parents. Further explanation was possible by more detailed investigation over time.

	RED group	AMBER	GREEN
	≤5 <sup>th</sup> MABC2	group 6-	group ≥25 <sup>th</sup>
		16 <sup>th</sup> MABC2	MABC2
SNAP IV ≤5 <sup>th</sup>	N=8	N=1	N=3
percentile	1 missing		
CCC2 ≤5 <sup>th</sup> percentile	N=9	N=1	N=3
	1 missing		
CSAPPA ≤5 <sup>th</sup>	N=5	N=2	N=2
percentile			
2 or more	N=8	N=0	N=1
Associated	1 missing		
characteristics			
Only 1 associated	N=0	N=2	N=4
characteristic	1 missing		

Table 6.8 Three group comparison of type of associated characteristics

Motor ability	<b>Red</b> ≤5 <sup>th</sup>	Amber 6-16 <sup>th</sup>	Green ≥25 <sup>th</sup>
MABC2	percentile	percentile	percentileMABC2
	MABC2	MABC2	n=17
	n=10	n=7	
Number in	10	7	17
group			
Females	0	3	2
Males	10	4	15
ASD/ADHD/DCD	8	0	0
ASD/DCD	1	2	0
ADHD/DCD	0	0	0
ODD/DCD	0	1	0
DCD/Hear Imp	0	1	0
DCD only	1	3	0
ASD/ADHD	0	0	1
ASD only	0	0	1
ADHD only	0	0	1
SLI only	0	0	1
ODD only			1
Typical	0	0	12
Development			
TOTAL	10	7	17

Table 6.9 The number and type of associated characteristics per group at baseline

There was also a noticeable difference in the ecological context between the groups. The Red group had a greater number of children with siblings with a developmental disorder and a higher proportion of families receiving free school meals (table 6.10) and so were at a greater disadvantage. Gillberg (2010) described this phenomenon in the ESSENCE study, whereby adversity was dose dependent. That is, the children with the more severe disorders were more likely to encounter additional disorders as well as social disadvantage.

Table 6.10 The ecological context per group

Motor ability	<b>Red</b> ≤5 <sup>th</sup>	Amber 6-16 <sup>th</sup>	Green ≥25 <sup>th</sup>
MABC2	percentile MABC2	percentile MABC2	percentileMABC2
	n=10	n=7	n=17
Free School	6 (60%)	2 (28.6%)	8 (47.1%)
Meals			
Sibling with dev	4 (40%)	2 (28.6%)	5 (29.4%)
dis			

## 6.6.2.2 Associated characteristics of the Amber group (6-16<sup>th</sup> MABC2) moderate DCD

Clearly, this group of children had fewer associated characteristics and, although they encountered movement difficulties, they were less severe and less obvious to their parents and to themselves. Only 28.6% had their movement difficulties identified by their parents. Perhaps they were able to circumvent the movement difficulties because most children did not have the additional difficulties in attention and communication. Only one child had difficulties with attention and only one child had difficulties with communication, but none had two or more associated characteristics. Two identified themselves as having difficulties in participating in physical activity. Again only more investigation over time could help explain the different issues.

# 6.6.2.3 Associated characteristics of the Green group (≥25<sup>th</sup> MABC2) typically developing\*

This group had a more mixed composition than the AMBER (moderate) DCD group. This was possibly explained by the presence of two children with a diagnosis of ASD, but who scored above the 25<sup>th</sup> percentile on the MABC2 and two children who had been identified by teachers as having movement difficulties (possibly because of another diagnosis) but who scored above the 25<sup>th</sup> percentile on MABC2. Only one child had two or more associated characteristics. However, there were also a greater number of difficulties identified by parents in this group then the previous one, perhaps with fewer additional difficulties identified by their parent. However, two children no longer remained in this group on subsequent testing, demonstrating that investigation over time was important to facilitate better understanding of the issues. Interestingly, 11.8% identified themselves as having difficulties in participating in physical activity

Thus it was apparent that the three groups classified by movement ability appeared to have some differences in composition and that the RED (severe DCD) group contained a very high proportion (80%) of children with two or more associated characteristics. Their teachers and parents much more readily identified their difficulties and 50% of the children themselves identified that they had a motor problem.

A broad overview of patterns was achieved by pooling the results for severe DCD and moderate DCD, as only small numbers were involved. See Table 6.11

#### Summary:

It was apparent that in this study the children with DCD (motor impairment  $\leq$  16<sup>th</sup> percentile) were (3x) more likely to associated characteristics with attention

control (measured by SNAP IV) and communication difficulties (measured by CCC2) and 8x more likely to have two or more associated characteristics. However, some of the children  $\geq 25^{\text{th}}$  percentile on MABC2 initially identified as TDC also experienced some associated characteristics and were monitored closely. These will be discussed as they progressed over time.

	DCD ≤ 16 <sup>th</sup> percentile	Children $\geq$ 25 <sup>th</sup>
	MABC2 n=17	percentile MABC2 n=17
≤5 <sup>th</sup> percentile MABC2	N=10	N=0
DCDQ ≤5 <sup>th</sup> percentile	N=10	N=6
SNAP IV ≤5 <sup>th</sup> percentile	N=10	N=3
CCC2 ≤5 <sup>th</sup> percentile	N=10	N=3
CSAPPA ≤5 <sup>th</sup> percentile	N=7	N=2
2 or more Associated	N=8	N=1
characteristics		
Only 1 associated	N=1	N=4
characteristic		

Table 6.11 Pooled results for all motor impaired children compared to TDC

### 6.6.3 Cognitive ability

Another important variable between different groups of children with movement difficulty in this study was their level of IQ. Differences in levels of cognitive ability between the groups could potentially account for differences in progression over time and so all participants (n=34) were tested with the KBIT2 for their general intelligence and to give scores for non-verbal and verbal ability. In this study three children scored 2sd below mean (three children with IQ below 70). However, their motor skills were below that expected for their age or

level of intellectual ability as agreed by the special needs coordinator, the parents and an experienced occupational therapist and, for this reason, were kept in the study.

The frequency distribution of IQ for the sample was however normally distributed (see fig. 6.8) and this was confirmed by the Kolmogorov-Smirnov test D (34) = 0.106, p=0.200



Figure 6-7 Q-Q plot of composite IQ for whole sample



Figure 6-8 Frequency distribution of composite IQ for whole sample

Red Group: The severe (sDCD) *Group* ≤5 th *percentile on MABC2* 

The mean composite IQ for this group was lower when they were compared to the other groups. Many (90%) also had associated communication and attention difficulties and previous studies have identified that groups with co-occurring conditions are likely to have a lower IQ (e.g. SLI and DCD Flapper & Shoemaker, 2013). Furthermore, children with IQ less than 70 are more likely to exhibit movement difficulties due to a developmental lag (Sugden & Wade, Indeed, Sugden & Wann (1987) found that 50% of 8 year olds with 2013). moderate learning difficulties had motor impairment compared to 5 % of TDC the same age. However, they were included in the study because their movement difficulties were over and above those expected compared to other children with similar IQ and impacted on their activities of daily living, by consensus of teacher and occupational therapist. The verbal IQ mean was the lowest in this group, perhaps not unsurprising since nine out of the 10 in this group had communication impairments as measured by the CCC2.
Group	Total	KBIT2	Non-vei	rbal	Verbal	
	score					
<i>Group</i> ≤5 th	Range	Mean	Range	Mean	Range	Mean
' Red aDCD	55-	78.6	52-112	81.4	52-99	80.7
Red SDCD	100					
n=10	100	Sd		sd (19.6)		sd
		(17.2)				(13.9)
Group5-16 <sup>th</sup>	Range	Mean	Range	Mean	Range	Mean
Amber	71-	91.3	68-127	91.7	79-	92.6
mDCD n=7	122	sd		sd (22.6)	122	sd
		(21.8)				(15.8)
Group≥25 <sup>th</sup>	Range	Mean	Range	Mean	Range	Mean
Green	71-	106.5	63-130	101.2	86-	109.5
pTDC	131	sd		sd (19.2)	132	sd
n=17		(14.6)				(13.4)
Entire	Range	Mean	Range	Mean	Range	Mean
Group	55-	95.2	52-130	93.5	52-	97.6
N=34	131	sd		sd (21.3)	132	sd
		(20.5)				(18.7)

Table 6.12 Three group comparison of KBIT2 IQ scores

The mean for this test 100 and standard scores between 85-115,

Amber Group: IQ results for mDCD (moderate) 5-16<sup>th</sup> percentile on MABC2

The Amber (mDCD) group had a higher mean IQ than the children in the Red group (sDCD), but the mean IQ score was still below average across all measures. However, the mean verbal IQ was marginally higher than the nonverbal mean IQ and the mean composite IQ.

Green Group: IQ results for pTDC (typical) ≥25<sup>h</sup> percentile on MABC2

This group had the highest mean IQ and the smallest standard deviations of the three groups, with all the children in the average or above average IQ range, as well as the highest mean motor ability.

These measures suggest that the 'within-child' or biological characteristics of the children in the three groups were different at baseline. The Red group not only had children with a greater number of associated characteristics, but also had more children with two or more associated characteristics. Their mean IQ was lower than the other two groups, as was their mean age. Thus they appeared more impaired than the other two groups because they had the most severe motor difficulties, some learning difficulties and 80% had additional attention and communication difficulties.

#### Summary

Thus the children who participated in this study exhibited distinct characteristics when they were categorized by their movement ability. Furthermore, the children who exhibited characteristics of severe DCD appeared to have other difficulties in addition to their movement problems, which may include but are not limited to their capacity for learning. In addition, we have some understanding of their family context. Therefore, these differences could be quantified at the start of the study and the trajectories for these children could then be tracked to investigate what difference, if any, they had on their stability or change in motor ability and their participation in physical activity. A fine-grained analysis of IQ followed for each group to ascertain any statistical differences.

### 6.6.3.1 Group comparison of KBIT2 composite IQ score

The composite IQ covered a wide range of scores and as the Red (sDCD) group had a lower mean IQ than the other groups (see Fig. 6.9). It was

therefore important to ascertain if there were significant differences in IQ between the groups.

#### **Descriptive Statistics**

Dependent Variable: IQcomp

group	Mean	Std. Deviation	Ν
sDCD	78.6000	17.19948	10
mDCD	91.2857	21.47645	7
TDC	106.5294	14.56072	17
Total	95.1765	20.49903	34

#### Figure 6-9 Group means and SD for IQ

The box plot (Fig 6.10) of median IQ composite scores showed overlap between the scores of the Red (severe DCD) and Amber (moderate DCD) and the Amber and Green (TDC) groups, indicating less likelihood of significant difference between them. The Red and Green groups, however, showed much less overlap indicating greater likelihood of a significant difference between their composite IQ scores.





The distribution of scores across each group was tested for normality and confirmed by the Kolmogorov-Smirnov test. This enabled group comparison by ANOVA.

One-way ANOVA revealed a significant difference in composite IQ between the groups [F (2,31) = 8.864, p= .001,  $\eta$ 2 partial = .364] power = .957

Pairwise comparisons revealed that the sDCD (Red) group had significant lower mean composite IQ than TDC group. Although the sDCD (Red) group also had a lower mean IQ than the mDCD (Amber) group it was not statistically significant (p= .137). Furthermore, the mDCD group also had a lower mean composite IQ score than the TDC (Green) group, but this only just reached statistical significance (p= .053).

Post hoc tests only revealed a statistical difference between the group means of sDCD (Red) and TDC (Green) groups (p= .001).

Therefore, in this sample the mean IQ was lowest for (Red) sDCD group and there was a significant difference between them and the TDC group. Perhaps this is not surprising since this group had the highest number of children with associated characteristics consistent with ASD and ADHD. Others have also found that these children have lower IQ scores (Flapper & Shoemaker, 2013; Kaplan, Crawford, Cantell, Kooistra & Dewey, 2006).

## 6.6.3.2 Group comparison of KBIT2 verbal IQ group score

The verbal IQ spanned a large range (52-132) across the sample and so the group means were investigated to determine if there was a statistical difference between the groups. See Fig. 6.11

#### Figure 6-11 Group mean and SD for verbal IQ

group	Mean	Std. Deviation	Ν
sDCD	80.7000	13.91282	10
mDCD	92.5714	15.79934	7
TDC	109.4706	13.44023	17
Total	97.5294	18.71515	34

Dependent Variable: IQV

The median for the sDCD (Red) and mDCD (Amber) groups appeared close, and, despite the large range for each group, there appeared to be some overlap between the error bars for the mDCD (Amber) and TDC (Green) groups (see Fig. 6.12)

Figure 6-12 Box plot of group KBIT2 verbal IQ



A one-way ANOVA of verbal IQ between the groups revealed a significant difference [F (2,31) = 13.726, p< .001,  $\eta$ 2 partial = .470] power = .996

Pairwise analysis revealed that the sDCD (Red) and mDCD (Amber) group means for verbal IQ were both statistically lower than TDC (Green) group

( $p\leq.001$  and p=.012 respectively), but not statistically different from each other (p=.097). Post hoc tests supported this ( $p\leq.001$ ).

This indicated that there was no significant difference between the verbal IQ scores for the children in severe and moderate DCD groups. This meant that the severe DCD and moderate DCD groups means were both significantly different from the typically developing group mean for verbal IQ (measured by the KBIT2), but not significantly different from each other.

### 6.6.3.3 Group comparison of KBIT2 non-verbal IQ group score

The non-verbal IQ group means were investigated to determine if there was a statistical difference between the groups (see Fig. 6.13)

#### Figure 6-13 Group mean and SD non-verbal IQ

Dependent Variable: IQNV

group	Mean	Std. Deviation	Ν
sDCD	81.4000	19.55164	10
mDCD	91.8571	22.57896	7
TDC	101.2353	19.24360	17
Total	93.4706	21.26633	34

The box plots showed overlap of error bars between all the groups and it looked unlikely that any difference would be statistically different for non-verbal IQ scores (see Fig. 6.14).

A one-way ANOVA of the mean non-verbal IQ did not reach significance between the groups [F (2, 31) = 3.119, p = .058,  $\eta$ 2= .168] power= .558 This was confirmed by the Post hoc test Hochberg (for unequal sample sizes) that did not reach statistical significance, although the pairwise comparison of difference between sDCD (Red) group and TDC (Green) group means for non-verbal IQ was close (p= .054). Therefore, for this sample, there was no statistical difference detected between the group means for non-verbal IQ measured by the KBIT2.

### Figure 6-14 Box plot of group KBIT2 non-verbal IQ scores



### Summary

The children in the Red group (sDCD) had lower mean composite IQ scores than the other two groups, but only the difference between them and the Green (TDC) group reached statistical difference.

The relationship between the differences in verbal IQ group mean scores was more complex. There was a statistical difference between Red and Green group means for verbal IQ, but also a statistical difference between the Amber (mDCD) and Green groups for verbal IQ. However, there was no statistical difference between Red (SDCD) and Amber (mDCD) group means. This indicated that the children with both severe and moderate DCD had difficulty with verbal IQ that was significantly different to the typically developing group.

A comparison of the non-verbal IQ group mean scores revealed more interesting detail. There was no detectable statistical difference between the mean non-verbal IQ scores for the three groups. This would appear to suggest that, in this sample, the children with DCD ( $\leq 16^{th}$  percentile on MABC2) had difficulty specifically with verbal IQ in addition to their motor difficulties and significantly differed from typically developing children. This may have been because 85% of the sample was drawn from a school with an ASD unit attached and siblings attended the main school.

	sDCD	mDCD	TDC	р
	(Red) n=10	(Amber) n=7	(Green) n=17	
Age months	109.7 (14.2)	129.1 (19.3)	124.1 (20.9)	.087
KBIT2 IQ	78.6 (17.2)	91.3 (21.5)	106.5 (14.6)	≤ .001*
KBIT2 VIQ	80.7 (13.9)	92.6 (15.8)	109.5 (13.4)	≤ .001*
KBIT2 NVIQ	81.4 (19.6)	91.9 (22.6)	101.2 (19.2)	.058

Fable 6.13 Summary of mean &	& (SD) group chara	acteristics at baseline
------------------------------	--------------------	-------------------------

Motor difficulties, however, are core features of DCD and so the fine-grained motor characteristics of each group was analysed in more detail.

## 6.7 Motor characteristics of each group

The motor characteristics of the groups are described in more detail, but first the frequency distribution of the whole sample was examined to determine the appropriate type of statistical analysis.

The Kolmogorov-Smirnov test D (34) =0.145, p=0.69 did not deviate significantly from normal for the frequency distribution of the MABC2 total score for the whole sample, suggesting that it was normally distributed.



Figure 6-15 Q-Q plot of MABC2 total score for sample

Figure 6-16 Frequency of MABC2 total score for sample

Closer investigation of the sub-groups was undertaken to determine whether inferential statistics could be used. Table 6.14 compares the descriptive statistics of each of the 3 groups.

Group	MD	MABC2	A&C	MABC2	Balance	e
	score		score		MABC2	2
<i>Group</i> ≤5 th	Range	Mean	Range	Mean	Range	Mean
Red sDCD	4-18	10.7	8-16	10	4-26	14.7
n=10		Sd (4.1)		sd (2.8)		sd
						(6.9)
Group5-16 <sup>th</sup>	Range	Mean	Range	Mean	Range	mean
Amber	12-24	17.9	10-20	15.1	24-35	28.9
mDCD n=7		sd (4.5)		sd (4.1)		sd
						(3.9)
Group≥25 <sup>th</sup>	Range	Mean	Range	Mean	Range	mean
Green	19-32	26.1	12-28	18.1	25-39	33.3
pTDC		sd (3.9)		sd (4.3)		sd
n=17						(3.6)
Entire	Range	Mean	Range	Mean	Range	Mean
Group	4-32	19.9	8-28	15.1	4-39	26.9
N=34		sd (7.9)		sd (5.2)		sd
						(9.5)

Table 6.14 Motor characteristics of three groups at baseline

 Red Group: The children who scored in the severe (sDCD) range had the lowest scores in all subtests. However, the relatively small standard deviation for aiming and catching (2.8) indicated that all scores were close to the very low mean for this subtest and showed that all the children performed poorly. This item depends, to a certain extent, on learned skills and experience and unsurprisingly the children in this group performed poorly. This group had mean scores in all of the subsections ≤5th percentile, which indicated motor problems in all areas.

- Amber Group: The children who scored in the moderate (mDCD) range performed better across all subtests. However, there was quite similar overlap to the scores for the children with severe (sDCD) in the range for manual dexterity and aiming and catching (both were ≤5th percentile), which indicated severe problems, but not for balance. This group had better mean balance.
- Green Group: The children who scored in the TDC range clearly had much better movement ability, shown by their higher mean scores. However, only their mean balance score attained the 50<sup>th</sup> percentile, indicating that mean manual dexterity and aiming and catching were below average.

Summary: The three groups appeared to have different motor characteristics at baseline. However, there was some overlap between the Red and Amber groups for manual dexterity, as well as between the Amber and Green groups, for this subtest. Additionally, there was large overlap between all three groups for aiming and catching, making distinction between them harder. However, there were clearer distinctions in the mean scores for balance between all three groups. Of course any differences needed to be verified by statistical analysis.

### Statistical analysis

Using a calculation with  $\alpha$  =0.05 and power of 0.8, a sample size of a minimum of 9 per group was required to compare groups for children in this sample (<u>http://statisticalsolutions.net/pss\_calc. phpsoftware package</u>). The Red and Green groups had sufficient numbers per group (10 and 17 respectively), but the Amber only had 7 at baseline. Although the tests were run the Amber group results were interpreted with caution because of the small sample in this group.

## 6.7.1 Group comparison of MABC2 total score at baseline

A test for normality of the MABC2 data was first undertaken for each group to determine if parametric tests were appropriate. The Kolmogorov-Smirnov test

showed the scores across all groups were normally distributed for the MABC2 total score and so an ANOVA was used to compare group difference.

An ANOVA assumes equal variance for each group, however this was not the case for the MABC2 total scores with this sample. Levene's test was significant, which indicated that equal variance had been violated. However, Field recommends using Welch or Brown-Forsythe statistic in this case to account for this (Field, 2013, p461). Significant group effects were broken down using pairwise comparisons and significant main effects were broken down using post hoc tests.

One-way ANOVA revealed significant differences between the groups

MABC2 total score [F (2,31) =71.351, p≤ .001, η2 partial = .822] power = 1

This difference is unsurprising, since the groups were classified using this score.

Pairwise comparison revealed sDCD<mDCD<TDC and post hoc analysis (Bonferroni and Hochberg for unequal sizes) showed significant differences between sDCD (Red) and mDCD (Amber) groups and sDCD (Red) and TDC (Green) groups ( $p \le .001$ ) and the difference between mDCD (Amber) and TDC (Green) groups was also significant (p = .002).



Figure 6-17 Box plots of three group median of MABC2 total score at baseline

## 6.7.2 Group comparison of Manual Dexterity at baseline

The manual dexterity scores were normally distributed at baseline and so permitted parametric tests.

A one-way ANOVA revealed significant difference between the groups for manual dexterity [F (2, 31) = 45.503, p≤ .001,  $\eta$ 2 partial= .746] power = 1

Pairwise comparison revealed sDCD<mDCD<TDC and post hoc (Bonferroni and Hochberg) analysis revealed significant differences between sDCD (Red) and TDC (Green) groups and mDCD (Amber) and TDC (Green) groups (p≤ .001) and sDCD (Red) and mDCD (Amber) groups (p= .004).

This can be clearly seen by very little overlap of the groups on the box plot (Fig. 6.18). Therefore, in this sample, the manual dexterity characteristic of the three groups is distinctly different at baseline.



Figure 6-18 Box plots of Manual dexterity score at baseline

Manual dexterity is required for school related tasks such as handwriting, tool use and self-care. A child experiencing difficulty in one of these areas is one of the most common reasons for teacher referral to community occupational therapy services. Interestingly, both the Red (sDCD) and Amber (mDCD) groups had difficulty with manual dexterity and were significantly different to the TDC (Green) group. However the Red (sDCD) group had the lowest mean and was still significantly different to the Amber (mDCD) group mean.

## 6.7.3 Group comparison of Aiming and catching at baseline

Aiming and catching are key skills for team play and most ball games, both of which are important for playground participation with peers, particularly for boys at primary school age. The extant literature has highlighted these as problem areas for children with DCD and so investigation of group difference here was important.

One-way ANOVA revealed significant differences between the groups [F (2,31) = 13.607, p $\leq .001$ ,  $\eta$ 2 partial = .467] power .996

Pairwise comparison with post hoc analysis (Hochberg test & Bonferroni) revealed significant difference between sDCD (Red) and mDCD (Amber) groups (p= .035) and sDCD (Red) and TDC (Green) groups (p≤ .001) but not between the mDCD (Amber) and TDC (Green) groups (p= .266 & p= .299).

Red (sDCD) group performed significantly worse than both the Amber (mDCD) group and Green (TDC) groups. However, the Amber and Green groups did not show a significant difference in their aiming and catching scores. This is demonstrated by the large areas of overlap on the box plots (Fig. 6.19) of the Amber and Green groups below. This could either mean that the Amber group is performing close to typically developing or that the Green group is performing at the lower end of normal. From the appearance of the error bars, the latter is more likely.



Figure 6-19 Box plots of Aiming & catching at baseline

These results showed that the Red (sDCD) group encountered the greatest difficulty with ball skills. This would be barrier to their participation in playground activities and team sports. Interestingly, there was no significant difference between the mean aiming and catching for the Amber (mDCD) and Green (TDC) groups, making their profiles more similar. However, both means were

quite low for these skills and may have reflected the context of deprivation and possible lack of opportunities.

## 6.7.4 Group comparison of Balance at baseline

Balance is a fundamental skill for controlled movement and impaired balance can negatively impact on many aspects of participation in physical activity. Group means are presented in fig. 6.20

Again the Amber (mDCD) group profile appeared more similar to the Green (TDC) group than to the Red (sDCD) group. This was confirmed by statistical analysis.

Dependent Variable: B1

group	Mean	Std. Deviation	Ν
sDCD	14.70000	6.929005	10
mDCD	28.85714	3.976119	7
TDC	33.29412	3.618783	17
Total	26.91176	9.449604	34

Figure 6-20 Group mean and SD of Balance at baseline

One-way ANOVA revealed significant differences between the groups [F (2,31) = 46.517,  $p \le .001$ ,  $\eta 2$  partial = .750] power = 1

Pairwise comparison with post hoc tests (Hochberg & Bonferroni) revealed significant differences between sDCD (Red) and mDCD (Amber) and sDCD (Red) and TDC (Green) groups ( $p \le .001$ ) but not between mDCD (Amber) and TDC (Green) groups (p = .144 & p = .154). The Red (sDCD) group performed significantly worse than both the Amber (mDCD) group and Green (TDC) groups. They also showed that the Amber group performed worse then the Green group, but this did not reach statistical significance. Therefore, the Red (sDCD) group balance was clearly distinguishable from the other two groups.



Figure 6-21 Box plots of group median balance at baseline

Balance is a fundamental movement skill essential for participating in PA. Interestingly the Amber and Green groups balance looked more similar to each other.

### 6.7.5 Summary of group motor ability performance at baseline

It is evident from these results that sDCD (Red) and TDC (Green) groups were statistically different on all motor subtests of MABC2. Moreover, the sDCD (Red) group was also statistically significantly different to the mDCD (Amber) group on all subtests. However, despite the mDCD (Amber) and TDC (Green) groups having statistically significant difference in their manual dexterity scores, they were more similar in their mean aiming and catching and balance scores. Only the Red group performed significantly worse than the other two groups in all motor scores at baseline, demonstrating problems across all motor domains. In other words, the Amber and Green groups, in this sample, were more similar to each other than to the Red group. This may have occurred because the Green group mean was low for aiming and catching (possibly as a result of most of the sample residing in an area of high deprivation). The sDCD (Red) group had a distinctive motor profile, which was clearly distinguishable from the

other two groups at baseline, with motor difficulties across all areas. The Amber (mDCD) group, although they clearly had distinct difficulties with manual dexterity, was much less distinguishable from the TDC (Green) group for balance and aiming and catching.

	sDCD	mDCD	TDC	р
	(Red)	(Amber)	(Green)	
MABC2 total score	34.4 (13)	61.9 (3.7)	77.5 (7.6)	≤ .001*
Manual dexterity	10.7 (4.1)	17.9 (4.5)	26.1 (3.9)	≤ .001*
Aiming & catching	10 (2.8)	15.1 (4.2)	18.1 (4.3)	≤ .001*
Balance	14.7 (6.9)	28.9 (4.0)	26.9 (9.4)	≤ .001*

Table 6.15 Summary of comparison of group motor characteristics at baseline

## 6.8 Group comparison of the parent rated DCDQ (07)

DCD is known to interfere with daily life and the DCDQ (07) invites parents to identify motor difficulties and rate the impact they are having on their child. The DCDQ (07) group mean for the Red (sDCD) group was lower than the other two groups, as expected, because they had more severe motor impairment. However, the Amber (mDCD) and Green (TDC) DCDQ (07) group means were very similar. This was not predicted, since the Amber group had poorer motor performance than the Green group and thus it was expected to have had a greater impact on their daily life. Statistical comparison was undertaken to determine significant differences.

Only the total score was used for analysis, since previous factor analysis of the subtests in the DCDQ (07) indicated poor reliability (Cairney et al., 2008). However, it allowed comparison of the parent ratings by group. The 5 % cut off scores for identifying potential DCD were:

Suspect DCD when total score is  $\leq$ 55 for under 10 and  $\leq$ 57 for over 10 years old.

#### **Descriptive Statistics**

Dependent Variable: DCDQ

group	Mean	Std. Deviation	Ν
sDCD	30.2222	12.23497	9
mDCD	62.4000	10.59717	5
TDC	58.2353	17.04600	17
Total	50.7742	19.76480	31

Figure 6-22 Group mean DCDQ '07 scores

## 6.8.1 The Red group (sDCD): all scored ≤5 th percentile on MABC2

The response rate of return of the DCDQ (07) was 90% for the Red group and 100% of those returned indicated identification of movement problems by the parents. The DCDQ (07) scores ranged from 15 - 56. The movement difficulties were subsequently confirmed by individual assessment with the MABC2. 90% of this group had two or more associated characteristics and the entire group was boys.

# 6.8.2 The Amber group (mDCD): all scored 6-16<sup>th</sup> percentile on MABC2

The response rate was 71.4% for this group, 5 out of 7 DCDQ (07) forms were returned. This group had the largest number of girls, 3 (42.9%). Of the returned forms one set of parents rated their child with movement difficulties and this child also had a diagnosis of ASD. The DCDQ (07) scores in the group ranged

from 46 – 72, despite 42.9% of the children only attaining 9<sup>th</sup> percentile on MABC2 at baseline. 42.9% (3 children) of this group had one known associated characteristic, but none had 2 or more.

# 6.8.3 The Green group (TDC): all scored above 25<sup>th</sup> percentile on MABC2

The response rate was 100% for DCDQ (07) for this group. 5 of the group had one associated characteristic (29.4%), 2 boys with ASD, one with ADHD, one with SLI and one with ODD. Three children (17.6%) were identified at risk of DCD by their parents. One of these had ASD, another ADHD and the  $3^{rd}$  was a looked after child with no diagnosis. Of the three, only the child (he had ADHD) remained in the Green group over time. The other two went on to display movement difficulties and changed category on MABC2. A third child in the Green group, not identified by DCDQ (07), went on to display movement difficulties and changed group over time. The range of DCDQ (07) scores were large (15 – 75) and this was possibly due to the presence of children with associated characteristics.

## 6.8.4 Statistical analysis of DCDQ (07)

The Kolmogorov-Smirnov test revealed that the DCDQ scores were normally distributed. However, the Shapiro-Wilk test suggested that the Green group with the typically developing children DCDQ scores were not normally distributed. However, Q plots suggested the frequency of scores were close to normal and therefore ANOVA was run to explore difference between groups.

One-way ANOVA revealed a significant difference between the groups [F (2, 28) = 12.060, p $\leq$  .001,  $\eta$ 2 partial= .463] power = .991

Although pairwise comparison showed a significant difference between Red and Amber and Red and Green groups (p=.001 and  $p\le .001$  respectively), the difference between Amber and Green groups did not reach statistical

significance (p= .589). Post hoc analysis with Bonferroni and Hochberg's test (for unequal size) supported this. Therefore, the Red group appeared quite distinctive from the other two groups on a measure of parent rated motor difficulties. The Amber and Green groups were indistinguishable from each other, with parents from both groups identifying some difficulties, but not as much as the parents from the Red group. This is illustrated in fig. 6.23



Figure 6-23 Box plots of group median DCDQ '07 scores

### Summary

The children with severe DCD (Red group) were distinct from the other two groups. Their parents could identify severe motor difficulty that interfered with daily life. The difference in mean scores for DCDQ (07) between the Red group and the other two groups was statistically different. The entire Red group was boys and 90% had two or more associated characteristics, which may have helped to highlight the motor problems to a greater extent. The Amber and Green groups did not appear to be distinct from one another in the way that the parents rated their children on DCDQ (07). This may be because the motor difficulties were subtler and there were fewer children with any associated characteristics. Of interest was that the children identified with motor problems by DCDQ (07) in the Amber and Green groups, all but one had associated characteristics, and he was fostered and had unknown early development. The

Amber group also contained three girls and perhaps parental expectations for motor competence is less for girls. Another possible explanation is the small number of responses (5) for the Amber group, making statistical comparison much less reliable. However, once again there was a difference between the Red group and the other groups at baseline and this was reflected in how the parents rated the children on the DCDQ (07).

## 6.9 Group comparison of the child rated CSAPPA

This measure of child self-perception towards physical activity (PA) is considered to identify those children with motor difficulties thought to be at risk from DCD by detecting the children with low scores in adequacy, predilection and enjoyment of PA (Hay, 1992). The extant literature indicates that children with DCD often have lower self-perceptions and this can influence actions and participation in PA. It was therefore predicted that Red (sDCD) and Amber (mDCD) groups would significantly differ from the Green (TDC) group in their self-perception scores on the CSAPPA. However, the group means in this sample were quite similar for all three groups (see Fig. 6.24).

#### **Descriptive Statistics**

group	Mean	Std. Deviation	Ν
sDCD	55.4444	12.35021	9
mDCD	56.4286	17.20327	7
TDC	62.7647	11.13289	17
Total	59.4242	12.95914	33

Dependent Variable: CSAPPA1

Figure 6-24 Group mean CSAPPA scores at baseline

The CSAPPA permitted the children to rate themselves in their enjoyment of, predilection to take part in and their perceived adequacy in PA. It was therefore important to investigate if there were any differences between the groups. The 5% cut-off scores for the CSAPPA were used. These were a score of <47 for

males and a score of <55 for females, to indicate possible motor difficulties. The maximum score was 75.

# 6.9.1 The Red group (sDCD) $\leq 5^{\text{th}}$ percentile on MABC2:

The total scores from the CSAPPA ranged from 32-74. Despite their poor motor performance on the MABC2 (range 0.1 -5<sup>th</sup> percentile), 80% of these boys indicated that they enjoyed physical activity, were happy to take part and felt that their performance was adequate. This may have reflected their age (7-10 years) and the motor demands of the activities they liked (all are primary school age), a lack of self-awareness or conversely, a real delight in physical activity. Only by noting stability or change over time would ascertain which reason was more likely. 20% of the boys described hating P.E. and rated their perception of enjoyment, predilection to take part and participation in PA poorly. Of particular interest was that their CSAPPA scores appeared unrelated to the MABC2 scores.

## 6.9.2 The Amber group the 6-16th percentile on MABC2:

The children who score in the 6-16<sup>th</sup> percentile on MABC2 presented a different picture. The total CSAPPA scores ranged from 43-75. This group contained three girls approaching or beginning adolescence. All the girls stated that they enjoyed physical activity, but one 11 year old expressed that she did not rate her adequacy highly and fell with in the suspect motor impairment range on the CSAPPA.

The other child who scored within the suspect DCD range was a child who had an ASD diagnosis and expressed that he hated P.E. The remaining five children all stated that they enjoyed PA and stated that they felt reasonably competent despite scoring in the at risk range for motor impairment. The ages in the group ranged from 9-13 years and again the CSAPPA scores and the MABC2 scores appeared unrelated for this group too.

## 6.9.3 The Green group (TDC) > the 25th percentile on MABC2:

All the children in the Green group expressed enjoyment and a positive attitude to predilection and adequacy in PA, with the exception of one child. This child scored 36, which specified that he did not enjoy physical activity and indicated suspect motor impairment, despite his score of 50<sup>th</sup> percentile on MABC2. He had a diagnosis of ASD and was afraid of being hit by a ball in the playground and over time he scored in the impaired range on MABC2.

	sDCD	mDCD (Amber)	TDC
	(Red)		(Green)
CSAPPA mean (SD)	55.4 (12.4)	56.4 (17.2)	62.8 (12.95)
Range	32-74	43-75	36-75

Table 6.16 CSAPPA scores by Red, Amber and Green group

These results were examined in more detail as the children progressed over time within the context of their family circumstances and are discussed later.

## 6.9.4 Group comparison of group CSAPPA scores at baseline

A test for normality of the data was carried out to determine if parametric tests were appropriate to compare the group means. The Kolmogorov-Smirnov test showed that the scores were distributed normally across all groups, and demonstrated by Q plots and so parametric tests could be used.

However, contrary to the prediction, one-way ANOVA revealed no significant difference between the groups on CSAPPA mean scores [F (2,30) = 1.190, p= .318,  $\eta$ 2 partial= .074] power = .240

Furthermore, pairwise comparison revealed no significant differences between any of the groups and post hoc analysis supported this. Thus there appeared to be no significant difference in self-perception of adequacy, predilection or enjoyment of PA between the groups, despite quite distinct differences in their actual motor performance ability. Motor performance ability may not have been the most important factor in determining self-perception for these children.

The box plots (Fig. 6.25) illustrated the median for the groups were similar. This was unexpected as much of the extant literature highlights low self-perception in motor and other areas for children with DCD. Further analysis over time revealed some interesting findings and will be discussed later.

### Summary

It appeared that, despite vastly differing motor performance ability, some children in each group indicated an enjoyment and predilection to take part in PA. Furthermore, that they rated their performance in PA as adequate, despite their low MABC2 scores. On an individual basis the MABC2 scores appeared unrelated to the CSAPPA rating and suggested the possibility of other factors at play. Only further investigation over time would help uncover potential causes.



Figure 6-25 Box plots of group CSAPPA at baseline

	sDCD (Red)	mDCD (Amber)	TDC (Green)	р
CSAPPA	55.4 (12.4)	56.4 (17.2)	62.8 (12.95)	.318
DCDQ (07)	30.2 (12.2)	62.4 (10.6)	58.2 (17.0)	≤ .001*

Table 6.17 Summary of group comparison of DCDQ '07 and CSAPPA at baseline

The Red (sDCD) group have very distinctive characteristics and significantly differed from the other two groups in a number of ways. Furthermore, the parents of the children in the Red group could more easily identify motor problems that their children encountered. However, the groups were indistinguishable from each other by the children's self-perception of their adequacy, predilection and enjoyment of physical activity. Whilst these results must be interpreted with caution due to the small sample size of the Amber (mDCD) group, preliminary findings suggest that the children in the severe DCD (Red) group not only possess significantly different characteristics from typically developing children (Green group), but also from children with moderate DCD (Amber group) for large effect size.

# 6.10 Summary of characteristics at baseline with parametric test results

1. Age: There was no significant difference between the groups in age

2. Self-perception of motor performance (CSAPPA): There was no significant difference between the groups CSAPPA scores.

3. Motor performance ability (MABC2): There was a significant difference between the all groups for MABC2 total score and manual dexterity score, but only a significant difference between Red and Green groups for aiming & catching and balance

4. IQ (KBIT2): There was only a significant difference between Red and Green groups for composite IQ score and between Red and Green and Amber and

Green groups verbal IQ scores. The non-verbal IQ scores did not reach statistical significant difference for any group.

5. Impact on daily life (DCDQ 07): There was a significant difference between Red and Amber and between Red and Green groups, but not Amber and Green groups.

# 6.11 Non-parametric test results at baseline

As previously described, some of the assumptions for parametric tests were not met in the data, for example, the sample size for the Amber group was small and the sample was not random, therefore the Kruskal-Wallis non-parametric test was also undertaken. However, the results supported those of the parametric tests (see Fig. 6.26).

Hypoth	esis	Test	Summary
--------	------	------	---------

	Null Hypothesis	Test	Sig.	Decision
1	The distribution of MABC2total1 is the same across categories of group.	Independent- Samples Kruskal- Wallis Test	.000	Reject the null hypothesis.
2	The distribution of IQcomp is the same across categories of group.	Independent- Samples Kruskal- Wallis Test	.003	Reject the null hypothesis.
з	The distribution of IQNV is the same across categories of group.	Independent- Samples Kruskal- Wallis Test	.071	Retain the null hypothesis.
4	The distribution of IQV is the same across categories of group.	Independent- Samples Kruskal- Wallis Test	.000	Reject the null hypothesis.
5	The distribution of MD1 is the sam across categories of group.	Independent- eSamples Kruskal- Wallis Test	.000	Reject the null hypothesis.
6	The distribution of AC1 is the same across categories of group.	Independent- eSamples Kruskal- Wallis Test	.000	Reject the null hypothesis.
7	The distribution of B1 is the same across categories of group.	Independent- Samples Kruskal- Wallis Test	.000	Reject the null hypothesis.
8	The distribution of CSAPPA1 is the same across categories of group.	Independent- eSamples Kruskal- Wallis Test	.273	Retain the null hypothesis.
9	The distribution of Agemonts is the same across categories of group.	Independent- Samples Kruskal- Wallis Test	.086	Retain the null hypothesis.

Asymptotic significances are displayed. The significance level is .05.

Figure 6-26 Kruskall-Wallis non-parametric test summary

# 6.12 Correlations at baseline

Pearsons correlations were performed to identify any relationships between the variables at baseline, since the data followed a normal distribution.

There was a significant correlation between MABC2 and IQ composite score r= 0.648, significance value 0.01. This was not surprising since the group mean

IQ was below average and it is well known that children with lower IQ are more likely to encounter poorer motor performance (Sugden & Wann, 1987; Sugden & Wade, 2013). A weaker correlation was found between CSAPPA and MABC2 r=0.358 significance value 0.05 and again this was not surprising as the content validity of both measures is similar (Hay et al., 2004). However, there was no significant relationship between CSAPPA and IQ.

The CSAPPA correlated weakly with the aiming and catching and balance components of the MABC2 but not the manual dexterity component, r=0.345 and r= 0.365 respectively at significance level 0.05. Both those components are required for physical play, whereas manual dexterity is less important for physical activity play.

The correlation between CSAPPA and IQ was non significant and the correlation between CSAPPA and age did not reach significance.

Correlations	Pearson's r	Significance p (two tailed)
MABC2 and IQ	r= .648	P< .001
MABC2 and DCDQ 07	r=. 686	P< .001
DCDQ 07 and Age	r=. 517	P= .003
CSAPPA and DCDQ 07	r=. 553	P= .001

#### Table 6.18 Correlations strongly significant (p= .01 two tailed)

#### Table 6.19 Correlations weakly significant (p=.05 two tailed)

Correlations	Pearson's r	Significance tailed)	р	(two
CSAPPA and MABC2	r= .358	P= .041		
CSAPPA and A&C	r=.346	P= .049		
CSAPPA and Balance	r= .365	P= .037		
CSAPPA and IQ	r= .058	P= .748 NS		
CSAPPA and Age	r= .341	P= .052 NS		

### 6.13 Summary of part I results

Part 1 of the results from this study addressed the research question 5.1.1 'What are the characteristics of children with different severities of DCD compare to children without DCD?'

In summary, by using Bronfenbrenner's Bio ecological model to assess the meso-system, the findings suggested that context of the motor impaired group did not differ greatly in their home and family circumstances from the non-impaired motor group. In fact, the motor impaired group appeared to have slightly more favourable environmental contexts in terms of parent employment, car ownership and positive parent attitude to PA.

However, with more detailed analysis (when classified by severity of motor ability) the three groups had differences in the number of associated characteristics present, their IQ composite scores, handwriting difficulties, and level of special educational needs. The children in the Red group (most severe motor problems) had the most additional difficulties, with 80% having two or more associated characteristics. They also had statistically significantly different motor profiles to the other two groups in their group mean manual dexterity, aiming and catching and balance scores on MABC2. Furthermore, their parents were easily able to identify their motor problems and the functional impact on their daily life. This appeared to indicate that the 'within-child' or biological characteristics of this group might have been distinct or different to those of the other groups.

However, there was no significant difference in age, non-verbal IQ or selfperceptions of participation in PA (CSAPPA scores) between the groups at baseline. Furthermore, there were no significant correlations between the children's CSAPPA scores and either their age or IQ. Which would seem to indicate that the scores reflected their perceived enjoyment, adequacy and predilection to take part in PA, rather than their actual motor ability or misunderstanding of the questionnaire or the notion that only younger children enjoy uninhibited PA. By noting these similarities and differences at baseline it was possible to follow the progression of children with and without DCD over time and compare the groups.

The children were then followed over two academic years to investigate stability or change in their motor characteristics and their self-perception of enjoyment, predilection and adequacy in PA.

# 6.14 RESULTS PART II: Change over time

This study aimed to describe the different profiles of children with and without DCD and identify any differential stability or change in motor ability over time between them. Group difference in motor change was investigated simultaneously with difference in the consistency in self-perception of motor ability, enjoyment and predilection to take part in physical activity over time. It was hypothesized that the children with the lowest motor ability would have the lowest scores in self-perception over time. The preceding chapter has described the fine-grained analysis of the profile of characteristics of children with and without DCD at the baseline in order to answer the research question "what are the characteristics of children with different severities of DCD compared to children with out DCD?" The baseline results showed that there were significant differences between the groups for children with different severities of motor ability. There now follows an analysis of how children with the different profiles performed over time in terms of their motor stability or change, together with consistency in their self-perception over time.

This will address research questions:

- How stable are the motor characteristics or do they change over time?
- How consistent are their self-perceptions of adequacy, enjoyment and predilection to take part in physical activity or do they change over time?

The investigations in part I of the study, which examined differences in profiles of the children, included presence or absence of associated characteristics and the degree of severity of motor difficulty at the baseline measurement. This led to the subsequent classification into groups: Red group where children scored  $\leq 5^{\text{th}}$  percentile on MABC2 (sDCD), Amber group where they scored 6-16<sup>th</sup> percentile on MABC2 (mDCD) and Green group where they scored  $\geq 25^{\text{th}}$  percentile on MABC2 (TDC) using the MABC2 traffic light classification, as previously described. The findings indicated that three groups had distinctive characteristics and some differences at baseline. The following section

explores the pattern of stability or change in motor performance ability over time of these groups, to establish whether the children with different severities of DCD progressed differently from each other, or from typically developing children. Change in motor ability was ascertained by repeated measures of the MABC2.

However, in order to avoid any contention over test re-test reliability of the MABC2, three different methods of analysis were undertaken. 1) Examining the smallest detectable change in motor performance, taken as a change  $\geq 10$  points in total score of MABC2 (Holm et al., 2013). 2) Examining any change in group membership and the distribution of children with associated characteristics in the groups and 3) Examining any statistically significant difference between the groups, by comparing the progression of the groups based on their membership at baseline.

The analyses first examined the smallest detectable change (SDC) in MABC2 total score over three data points. This is a measure thought to indicate an observable change in everyday motor performance, as opposed to only recording a small changed score on a standardized test. Holm et al., (2013) reported this change as 10 points or more on MABC2 total score when measured by the same tester. This was followed by any examining any change in group membership (Red, Amber and Green) over time, and an examination of the distribution of children with associated characteristics in each of the groups. Finally, any group difference in motor progression, using MABC2 scores, was analysed using repeated measures analyses of variance (ANOVA). Significant group effects were broken down using pairwise comparisons, significance levels were set at 0.05 and Bonferroni corrections used to control for multiple comparisons.

## 6.15 Prospective study of motor ability and attrition rate

The investigation of motor performance ability spanned three data points over two academic years for all the children and, additionally, specific cases were selected for a 4<sup>th</sup> data point for closer inspection. The same experienced occupational therapist took repeated measures of the MABC2, for all children remaining in the study.

Of the 34 children measured at baseline (T1), the attrition rate was 9% at data point two (T2) as two children at a secondary school were unable to be retested and one child left a primary school to be home schooled. It reached 15% by data point three (T3) with two more children who moved schools, leaving 29 children at the end of the study.

# 6.16 Investigation of stability or change in motor performance ability over time

## 6.16.1 Smallest detectable change (SDC) in MABC2 total score

The total test score from MABC2 was used to measure change in motor performance ability. However, the smallest detectable motor behaviour change was also noted and was referred to as the smallest detectable change (SDC).

This study adopted the method described by Holm et al. (2013), where change in measurement of + or - 10 points in the total test score of MABC2, when the re-tests were performed by the same tester. This method reduced the possibility of measurement error and enhanced the reliability of the results.

Change was measured between baseline MABC2 score (T1) and score at data point three (T3). Missing data was only included if two data points had been collected and in those two cases the difference between T1 and T2 was used. Table 6.20 displays the stability or smallest detectable change (SDC) in total MABC2 test score for each of the Red, Amber and Green groups.

	SDC ≥10 points on MABC2 total score (Holm et al., 2013)		
Groups	Improved	Same	Deteriorated
Red (sDCD) n=9	2	7	0
	(22.2%)	(77.8%)	(0%)
Amber (mDCD)	5	0	0
n=5	(100%)	(0%)	(0%)
Green (pTDC)	4	9	4
n=17	(23.5%)	(53%)	(23.5%)

 Table 6.20 Motor change T3-T1 (MABC2 total score) over time by group

Each group was analysed in turn and there now follows a description of the results.

## 6.16.2 Motor change in the Red group (sDCD)

One child was lost to follow up, but of the remaining nine children, the predominant picture for this group was stability of motor performance ability over time. Nearly 78% stayed the same (i.e. the MABC2 total score difference  $\leq$ 10 points). None of the children deteriorated and 22% improved their motor performance ability over time. However, their improvement was not sufficient for them to change motor category and so all the children remained in the Red group over time. In other words, the entire Red group remained in the lowest 5<sup>th</sup> percentile for motor performance ability over time.

## 6.16.3 Motor change in the Amber group (mDCD)

Two children were lost to follow up, but the overall picture for the remaining five was variability or change in motor performance ability over time. The entire group improved their motor performance ( $\geq$ 10 points MABC2 total score). Furthermore, the entire group had changed motor category by T3. All scored within the typically developing range and moved from Amber to the Green group (from 6-16<sup>th</sup> percentile to  $\geq$ 25<sup>th</sup> percentile). This group comprised 40%

girls and it is possible that they behaved differently to boys overtime. However, with such small numbers it was difficult to reach a safe conclusion. Detailed analysis of individual cases gave more insight into possible causes and will be discussed later.

### 6.16.4 Motor change in the Green group (TDC)

Two children from this group did not attend for data collection at T3 and so their scores at T2 were used. Just over half (53%) of this group stayed the same over time (MABC2 total score difference  $\leq$ 10 points) and equal numbers deteriorated (23.5%), or improved (23.5%). Four children deteriorated over time, of which three changed motor category to Amber by T3.

It was therefore evident that the three groups behaved differently in terms of their motor performance ability over time.

#### Summary

Data from children in the study with at least two data points (n=31) was analysed. 35.5% of the sample of children improved their motor performance ability between data point one and data point three, 51.6% stayed the same and 12.9% deteriorated. However, the group behaviour appeared to differ between groups. The most stable groups appeared to be the Red (sDCD), followed by the Green group (TDC). The children in the Red group all remained in the lowest 5<sup>th</sup> percentile of motor ability over time and nearly 78% had no change or change smaller than the SDC of 10 points. None of the Red group deteriorated. The Green group had a more even distribution of change, with just over half (53%) remained stable and equal numbers either deteriorated or improved (23.5%) over time. Consequently, there appeared to be differing degrees of variability in change in motor performance across the sample. The greatest fluctuation appeared in the 6-16<sup>th</sup> percentile MABC2 range Amber group (mDCD), which resulted in all of the members improving and changing group category over time to join the Green group ( $\geq 25^{\text{th}}$  percentile MABC2). By T3 all members of the Amber group had reached the level of typically developing motor performance ability. However, in addition two children from the Green
group deteriorated over time and moved to the Amber group by T3. Therefore, there appeared to be an overlap in motor characteristics between a few children in the lower end of motor ability ( $\leq 37^{th}$  percentile MABC2) in the Green group and all of the children in the Amber group.

The following section explores some of the variables across the groups to investigate how they may have impacted motor progression.

First the distribution of children with associated characteristics across the groups was explored.

# 6.17 Analysis of the distribution of associated characteristics across groups

Nearly half of the entire sample of children had associated characteristics (47%). This was a higher than expected prevalence and may have resulted from recruiting the majority of the sample of families from children who attended a school with an ASD unit attached. However, the Red group had a greater proportion of children (90%) with associated characteristics and all of those children had two or more associated characteristics. Both the Amber and Green groups also had members with associated characteristics, (40%) and (29.4%) respectively, but the majority only had one associated characteristic (AC). Only one child, who scored in the Green group at baseline, had two or more AC, yet he was exceptional and changed group twice over time. Table 6.21 shows the distribution of children with associated characteristics across the groups over time.

	Red	Red	Amber	Amber	Green	Green	Total
		+AC		+AC		+AC	
T1	1	9	5	2	12	5	34
T2	1	10	2	1	13	4	31
Т3	1	8	1	2	12	5	29

Table 6.21 Analysis of number of children with AC per group over time

However, as previously mentioned, the three groups appeared to have different patterns of progress depending upon the degree of motor impairment at baseline. Table 6.20 shows the number of children who displayed an improvement in motor performance ability greater than or equal to the smallest detectable change (SDC) of 10 points total score of MABC2. Table 6.22 shows the numbers of children in each category and those with associated characteristics are shown in brackets.

# 6.17.1 Percentage of children with associated characteristics with motor stability or change

Table 6.22 indicates that the children with AC were represented in all categories of progression.

The children with associated characteristics represented 45.5% of the children who improved motor performance ability over time. They represented 56.25% of the children who had no change in motor performance ability over time and represented 50% of the children who deteriorated their motor performance ability over time. However, the numbers involved are small and should be interpreted with caution, yet they would appear to be distributed across all categories of motor ability, whether they improved or remained stable. This would appear to suggest that associated characteristics per se were not the most influential factor in motor progression over time.

Group	Improved	Same	Deteriorated
category at	≥SDC		≤SDC
baseline			
Red N=9	N=2	N=7	N=0
<5 <sup>th</sup> MABC2	(With AC=2)	(With AC =6)	
Amber N=5	N=5	N=0	N=0
6-16 <sup>th</sup> MABC2	(With AC=2)		
Green N=15	N=4	N=9	N=2
>25 <sup>th</sup> MABC2	(With AC=1)	(With AC=3)	(With AC=1)
Total N=29 at	N=11	N=16	N=2
Т3	(With AC=5 or	(With AC=9 or	(With AC=1 or
	45.5%)	56.25%)	50%)

Table 6.22 Number of children per group changing (SDC +/- 10 points total MABC2 score) or remaining stable at T3

### 6.18 Analysis of group membership over time

Previous researchers, for example Cantell, Smyth & Ahonen (2003) and Sugden and Chambers (2007) had identified that some children with DCD improve motor performance to typical level over time. So an analysis to ascertain if the children in this sample remained in the same group over time was undertaken. This helped to determine if there was differential progression of motor ability between the groups of children with the different severity of motor impairment (i.e. severity of DCD) at baseline. The analysis of the Red, Amber and Green group membership over time, revealed some interesting information.

The children with severe motor difficulties, who were allocated to the Red group (n=10) at baseline measurement (T1), O% changed groups over time (T1 to T3). One child was lost to follow up, but of those remaining in the study all remained in the Red group over time. This was in direct contrast to the children

with moderate motor difficulties, who were allocated to the Amber group (n=7) at baseline measurement (T1), none remained in the same group over time. Two children were lost to follow up but 100% of the remaining children changed groups over time (T1 to T3).

The children with motor performance ability in the typical range, who were allocated to the Green group (n=17) at baseline measurement (T1), only three changed group over time to T3 (17.6%). The rest remained in the green group over time.

# 6.18.1 Diagrammatic representation of motor performance ability change over three time points by each group

The Red group (sDCD) showed little variability and even though some children improved their performance over time they still all (n=9) remained in the Red group. This group contained only boys and all but one had two or more associated characteristics. These findings suggested a pattern of stability in motor characteristics of the Red group over time and is represented in Fig. 6.27



Figure 6-27 Representation of group membership of the children from the Red group over T1, T2 & T3

The Amber group (mDCD) on the other hand demonstrated large variability and many children changed group. All who started in the Amber group had changed to Green group by T3. Three children with associated characteristics started in the Green group but fluctuated greatly and changed groups over time, finishing in the Amber group. One child even fluctuated more wildly and started in Green group but changed to Red and finished in the Amber group. This is represented in Fig. 6.28. This group also contained three girls, two who improved after the first data point and then remained in the Green group over time. Both girls had very poor ball skills, which may have reflected limited opportunity and low expectations placed on girls for these skills. The other girl however was a keen footballer but unfortunately was lost to follow up.



Moderate DCD (mDCD) at risk AMBER GROUP

Figure 6-28 Representation of group membership for teh children in the Amber group over time T1, T2 & T3

The Green group (TDC) showed variability in their motor performance ability over time, with some improving, some deteriorating and some remaining stable. However, all but one of the typically developing children (i.e. those without associated characteristics) did not change group over time, all remained in the Green group. The same experienced OT tested all the same children over time and the test retest reliability of the MABC2 is quoted as reasonable (Brown & Lalor, 2009) with Pearson Product Moment Coefficients ranging from 0.73 to 0.84 (Henderson et al., 2007), it can therefore be assumed that the test results

are a reliable representation of the children's performance. This indicated that some fluctuation in motor performance ability over time in typically developing children is to be expected. It also highlights the differences in patterns of change over time between the other groups. Fig. 6.29 represents the Green group over time.





Figure 6-29 Representation fo group membership for the children from the Green group over time, T1, T2 &T3  $\,$ 

The distinctive patterns in the different group's motor behaviour over time can be summarized as follows:

GROUP	Variability	Stability
RED Severe DCD (sDCD)	Small variability	Some stability
AMBER At risk Moderate DCD (mDCD)	Largest variability	No stability
GREEN typically developing (TDC)	Some variability but no large changes	Some stability

# Motor change over time

Figure 6-30 Summary of group's pattern of motor performance change over time

Thus, children with different severity of motor impairment at baseline (Red, Amber & Green groups) appeared to show differences in the motor progression over time. This was explored further by statistical analysis using to determine if the differences reached statistical significance between the groups.

However, there was a certain amount of intra-group variation within each group. Therefore, plotting the individual trajectories of the children from each group depicts this well.



Figure 6-31 Red group motor change over time

Although there was only minimal variation within the red group, one child stands out. The child depicted on the top line in Fig.6.31, was the only child without ASD and had the most consistent scores over time. However, all children in this group remained at or below 5<sup>th</sup> percentile MABC2 over time.



Figure 6-32 Amber group motor change over time

The child represented as the red line in Fig.6.32 had ASD and had the most inconsistent scores over time, but all attained at or above 25<sup>th</sup> percentile on MABC2 by data point 3.



Figure 6-33 Green group motor change over time

The child represented by the lowest blue line in Fig.6.33 had ASD/ADHD, showed the largest variability over time and finished in the impaired range by data point 3.

# 6.19 Repeated measures ANOVA for group difference in motor progression

Statistical significance was investigated with repeated measures analysis of variance ANOVA (SPSS version 22). All missing data were not included and analysis was undertaken on n = 29 and therefore only 15 children in the Green group.

However, representing group change over time was problematic since group membership changed over time. Change was therefore explored by assigning group membership at baseline (T1) and by statistically comparing the results of various dependent variables of those members over time.

For comparison of more than two groups ANOVA for parametric data or Kruskall Wallis for non-parametric data was used. Parametric tests are more powerful tests, providing the assumptions for parametric data are met (such as normal distribution and a sufficient sample size for power). According to previous calculations these assumptions were met and permitted the use of parametric tests. Please refer to the previous chapter.

Change over time was tested with repeated measures on the same subjects. For repeated measure General Linear Model (GLM) a one way repeated ANOVA to detect group difference along a combination of dimensions was used.

Repeated measures one-way ANOVA computes several means when those means have come from the same entities (e.g. same subjects provide data at multiple time points) (Field, 2013), as was the case in this study. If there are three or more repeated measures then there is an assumption of sphericity.

This has to be tested using Mauchly's test, if the value is less than .05 the assumption is violated and adjustments are made using different columns in SPSS output (Field, 2013, p565). The table labelled within- subject effects shows the main results of ANOVA. If assumption of sphericity is violated then use Greenhouse-Geisser and Huyn-Feldt otherwise read sphericity assumed row. Significant results are less than .05 (i.e. a significant difference). Similarly, for contrasts and post hoc tests a significance of less than .05 was used.

Bonferroni post hoc is robust in terms of Type I error rates and power (Field, 2013 p 547). In terms of test power (the Type II error rate) Maxwell (1980) found Tukey's wholly significant difference test (WSD) to be most powerful under conditions of non-sphericity in very small sample sizes (n=8) but this is reduced in larger sample sizes (n=15) (Field, 2013, p547). As the sample size of the Green group was 17, Bonferroni post hoc tests were used. Type I error occurs when we believe that there is a genuine effect in the population when in fact there isn't. Probability of error is .05 (or 5%) known as  $\alpha$  level.

Type II error occurs when we believe that there is no effect in the population, when in reality there is. Cohen (1992) suggests that the maximum acceptable probability of Type II error is .2 (or 20%) and is called the  $\beta$  level (Field, 2013 p68). Power is 1-  $\beta$  (1-0.2=0.8) an 80% chance of detecting an effect

Levene's test of equality of error variance equal across groups was violated at data point one. Hochberg's post hoc analysis was applied because of the unequal number of group members and Bonferroni adjustment was also used to account for repeated comparisons. Pot hoc multiple comparisons indicated significant differences between Red and Amber and Red and Green groups, but not between Amber and Green groups. The most severely impaired children (Red group) progressed differently from both the moderately impaired (Amber) and the typically developing (Green) groups. The results are now presented comparing the groups on each test component.

#### A comparison of MABC2 total scores over time

The repeated measures of MABC2 total scores (T1 to T3) were compared to determine if there were statistically significant differences between the three groups. A one-way between groups (Red sDCD, Amber mDCD, Green TDC), repeated measures (time 1, time 2, time 3) analysis of variance (ANOVA) was employed to examine the dependent variable of MABC2 total score. Box's test of equality of covariance was not significant, signalling equality across all groups. However, Mauchley's test indicated that the assumption for sphericity had been violated [ $\chi 2$  (2) = 6.67, p= .036], therefore Greenhouse-Geisser corrected tests are reported

#### Within subject effects:

The repeated measures ANOVA with Greenhouse-Geisser correction determined that mean MABC2 total scores differed significantly at data collection times (T1, T2 & T3), [F (1.620, 42.133 = 6.849, p=.005,  $\eta$ 2 =.209] observed power=.885

This indicated that change in motor ability performance had taken place over the two year time period.

Furthermore the group by time interaction was also significant [F (3.241, 42.133) = 3.772, p= .015,  $\eta$ 2 =.225] observed power =.797.

This indicated that there was a group effect.

#### Between subject effects:

There appeared be a difference between the groups in the way they progressed in motor ability over time. The repeated measures ANOVA determined that the groups differed significantly [F (2,26) = 54.905, p≤ .001,  $\eta$ 2=.809] observed power 1.0

Pairwise comparisons using Bonferroni corrections showed that Red (sDCD) group progressed significantly worse than both the Amber (mDCD) group, p

 $\leq$ .001 and the Green (TDC) group, p  $\leq$ .001. However, the Amber (mDCD) group did not differ significantly from the Green (TDC) group over time, p= .088.

	group	Mean	Std. Deviation	Ν
MABC2total1	sDCD	32.4444	12.12550	9
	mDCD	60.6000	3.36155	5
	TDC	77.6000	7.89937	15
	Total	60.6552	22.01019	29
MABC2total2	sDCD	34.6667	14.18626	9
	mDCD	67.2000	8.04363	5
	TDC	74.6667	11.14621	15
	Total	60.9655	21.41675	29
MABC2total3	sDCD	37.5556	11.98031	9
	mDCD	74.8000	4.54973	5
	TDC	76.2667	11.02897	15
	Total	64.0000	20.76054	29

**Descriptive Statistics** 



#### Summary

The results indicated that there were significant differences between the groups MABC2 total scores results over time, indicating that the groups had progressed differently. The Red group differed significantly to both the Amber and Green groups. The significance of these results show that the MABC2 total score change over time was significantly affected by severity of motor impairment at baseline. That is, children with the lowest motor ability, classified in the Red group (sDCD) at baseline, progressed significantly differently to children with moderate motor difficulty, classified in the Amber (mDCD) group at baseline and also significantly differently to typically developing children, classified in the

Green (TDC) group at baseline. The mean MABC2 scores (see Fig. 6.34) were consistently the lowest for the Red group over time (T3 mean= 37.4, SD=12.0). Therefore the Red group (≤5th percentile on MABC2 at baseline) was consistently shown to have the greatest degree of motor difficulty and this remained over time.

Although the children with moderate motor difficulty (classified Amber group) appeared to progress differently from the typically developing children (Green group), the difference did not reach statistical difference. This was not surprising since all of the children in the Amber group, which remained in the study, attained performance in the typical range as time progressed to T3. The mean total MABC2 score at T3 for the Amber group (T3 mean=74.8, SD=4.5) was similar to that of the Green group (T3 mean=76.3 SD=11.0).

The progression of the three groups is displayed in Fig. 6.35





# 6.20 Repeated measures ANOVA for group difference in CSAPPA progression

In order to determine any consistency or change in the children's selfperception of motor ability, enjoyment and predilection to take part in PA repeated measures of the CSAPPA were taken at the same time as testing with the MABC2. This was to address the research question "How consistent are their self-perceptions of adequacy, enjoyment and predilection to participate in PA or do they change over time?"

The groups were tested for any significant difference between them. A one-way between groups (Red sDCD, Amber mDCD, Green TDC), repeated measures (time 1, time 2, time 3) analysis of variance (ANOVA) was employed to examine the dependent variable of CSAPPA total score. The CSAPPA was a repeated measure of each child's self rated perception of their adequacy, enjoyment and predilection to take part in physical activity measured at the same time as their actual measures of motor performance (measures of MABC2) at T1, T2 and T3.

Box's test of equality of covariance was not significant, signaling equality across all groups. However, Mauchley's test indicated that the assumption for sphericity had been violated [ $\chi 2$  (2) = 9.298, p= .010], therefore Greenhouse-Geisser corrected tests are reported.

#### Within subject effects:

The repeated measures ANOVA with Greenhouse-Geisser correction determined that there was no significant difference between means for CSAPPA scores at TI, T2 and T3 [F (1.51, 37.845)=2.435, p= .114,  $\eta$ 2=.089] observed power .402

This indicated that the children's self-perception remained relatively consistent over time.

The group by time interaction was not significant [F (3.028, 37.845) = .251, p=.862,  $\eta$ 2=.020] observed power .093 This indicated there was no apparent group effect.

#### Between subject effects:

The repeated measures ANOVA also determined that the groups did not differ significantly [F (2, 25) = .468, p= .632,  $\eta$ 2= .036] observed power .118

So there was no significant difference between the groups over time in their self-perception of motor ability, enjoyment and predilection to take part in PA on repeated measures of the CSAPPA. Although interestingly, all three group mean scores were lower at T3 than at T1 indicating lower self-perception over time, although this was not statistically different.

Descriptive Statistics					
	group	Mean	Std. Deviation	Ν	
CSAPPA1	sDCD	56.1250	13.02127	8	
	mDCD	58.6000	11.63185	5	
	TDC	61.6000	11.36913	15	
	Total	59.5000	11.70470	28	
CSAPPA2	sDCD	55.3750	15.87395	8	
	mDCD	58.8000	15.83351	5	
	TDC	62.7333	10.54559	15	
	Total	59.9286	13.07225	28	
CSAPPA3	sDCD	53.7500	19.44039	8	
	mDCD	53.8000	17.81011	5	
	TDC	56.6000	14.75418	15	
	Total	55.2857	16.12189	28	

#### Figure 6-36 CSAPPA group means and SD change over time

#### Summary

There was no significant difference between the children with severe motor difficulty Red (sDCD) group, moderate motor difficulty Amber (mDCD) group and the typically developing children Green (TDC) group for the children's self-perception of their adequacy, enjoyment and predilection to take part in physical activity over time. This was somewhat unexpected, since there was a significant group difference in their observed motor performance measures over time. However, as there had been no significant difference in self perception between the groups at baseline, perhaps it was not so surprising that there was no difference in their self perception over time. This was an interesting finding on two counts: The first was that the mean group self-perception remained relatively similar over time. The second was that all the group mean scores deteriorated (i.e. were lower) by T3, please refer to Fig. 6.36. for means and standard deviations and Fig. 6.37 for plots of group means over time.

Another interesting finding was that the mean CSAPPA scores were relatively low at T3 for all groups (Red mean=53.8, SD=19.4, Amber mean=53.8, SD=17.8, Green mean=56.6, SD=14.8). The maximum CSAPPA score is 75 and the CSAPPA cut-off score (to indicate difficulty in physical activity) for girls is 53 and 47 for boys. So we can see that the mean scores for the three groups were all at the lower end.

Although there were significant differences in progression in motor performance between the groups over time, there was no significant difference between the groups for self-perception over time. The results indicate that the group means for self-perception of adequacy, enjoyment and predilection to participate in physical activity over time, in this sample of children, does not appear to relate to their actual motor performance ability measures (MABC2 total score) over time. Other factors may have been more important and so contextual data from case studies will be discussed later.



Figure 6-37 Estimated marginal means of CSAPPA

However, the components of motor performance ability, namely manual dexterity, balance and aiming and catching were also investigated to ascertain if stability or

change in these areas could offer explanation for the CSAPPA scores.

### 6.21 Repeated measures ANOVA of manual dexterity over time

Manual dexterity is required in all self-care tasks, handwriting and most leisure activities. Impairment in manual dexterity can have a negative impact on independence skills, schoolwork and participation in play. The groups were therefore investigated to determine if there was a statistical difference between the progressions of their manual dexterity performance ability over time. A one-way between groups (sDCD, mDCD, TDC), repeated measures (time 1, time 2,

time 3) analysis of variance (ANOVA) was employed to examine the dependent variable of MABC2 manual dexterity score.

Box's test of equality of covariance was significant, signalling inequality across all groups. However, Mauchley's test was not significant and indicated that the assumption for sphericity could be assumed.

#### Within subject effects:

The repeated measures ANOVA determined that there was a significant difference between means for manual dexterity scores at TI, T2 and T3 [F (2, 52)=5.523, p= .007,  $\eta$ 2=.175] observed power .832 This indicated that there had been a significant difference in manual dexterity over time.

The group by time interaction was also significant [F (4, 52) = 5.089, p= .002,  $\eta$ 2= .281] observed power .950 indicating a group difference over time.

#### Between subject effects:

There was a significant difference between the groups for manual dexterity over time. The repeated measures ANOVA determined that the groups differed significantly [F (2,26) = 56.608, p ≤ .001,  $\eta$ 2= .813] observed power 1.0

Pairwise comparisons showed that children with the lowest motor ability (classified as Red (sDCD) group) progressed significantly worse than both the children with moderate motor difficulty (classified as Amber (mDCD) group),  $p \le$  .001 and significantly worse than typically developing children (classified the Green (TDC) group),  $p \le$ .001 and this was confirmed by post hoc Bonferroni corrections p = <.001. However there was no significant difference between the children with moderate motor difficulty (Amber (mDCD) group) and typically developing children (the Green (TDC) group) over time, p = .103, which was confirmed by post hoc Bonferroni corrections p = .308.

#### Summary

There were significant differences between the groups in the way their manual dexterity progressed over time confirmed by post hoc tests. Again, children in the lowest motor ability group (Red (sDCD) group) who progressed significantly differently to the children with moderate motor difficulty (Amber group) and the typically developing children (Green (TDC) group). However, the difference between the Amber (moderate) and Green (typical) groups did not reach statistical significance.

Therefore the results showed that manual dexterity change over time was significantly effected by severity of motor impairment at baseline. That is, children in the Red group (sDCD) progressed significantly worse to children in the Amber (mDCD) group and also to children in the Green (TDC) group. The mean manual dexterity scores (see Fig. 6.39) were consistently the lowest for the Red group over time (T3 mean= 10.7, SD=3.0), so consistently had the greatest degree of difficulty in manual dexterity over time.

Although the children with moderate motor difficulty (Amber group) appeared to start differently from the typically developing children (Green group), there was no statistical difference between these groups for manual dexterity over time. The group mean for manual dexterity at T3 for the Amber group (mean=27.6, SD=3.8) was higher than the group mean of the Green group at T3 (mean=25.4, SD=6.3). See fig. 6.38

	group	Mean	Std. Deviation	Ν
MD1	sDCD	9.8889	3.40751	9
	mDCD	18.2000	4.60435	5
	TDC	26.4667	3.71996	15
	Total	19.8966	8.31240	29
MD2	sDCD	9.8889	4.16667	9
	mDCD	22.4000	4.21900	5
	TDC	25.6000	4.17133	15
	Total	20.1724	8.17692	29
MD3	sDCD	10.6667	3.00000	9
	mDCD	27.6000	3.78153	5
	TDC	25.4000	6.26555	15
	Total	21.2069	8.75614	29

Figure 6-38 Group means and SD Manual dexterity over time Descriptive Statistics

Figure 6-39 Plot of progression of manual dexterity over time



### 6.22 Repeated measures ANOVA of balance over time

Goal directed movement virtually always requires postural adjustment and a major goal of postural adjustment is to keep balance (Hadders-Algra, 2008). Therefore, any physical play, in which children normally engage, will require a level of balance. Difficulty with balance can be a major barrier to physical activity for children with motor difficulty. Therefore, differences between the groups in the way children progress with balance could have implications for engagement in PA.

A one-way between groups (Red sDCD, Amber mDCD, Green TDC), repeated measures (time 1, time 2, time 3) analysis of variance (ANOVA) was employed to examine the dependent variable of MABC2 balance score. Box's test of equality of covariance was not significant, signalling equality across all groups. In addition, Mauchley's test was not significant and indicated that the assumption for sphericity could be assumed.

### Within subject effects:

The repeated measures ANOVA determined that there was no significant difference between means for balance scores at TI, T2 and T3 [F (1, 26)=.804, p=.378,  $\eta2=.030$ ] observed power .139

The group by time interaction was also not significant [F (2, 26) =1.182, p= .323,  $\eta$ 2= .083] observed power .236

### Between subject effects:

There were however, differences between the groups. The repeated measures ANOVA determined that the groups differed significantly [F (2,26) = 41.041, p  $\leq$  .001, n2= .759] observed power 1.0

Pairwise comparisons showed that, once again, the children with the severe motor impairment (Red (sDCD) group) progressed significantly worse than both the children with moderate motor difficulty (Amber (mDCD) group),  $p \le .001$  and

the typically developing children (Green (TDC) group),  $p \le .001$  and this was confirmed by post hoc Bonferroni corrections  $p \le .001$ . However, there was no difference between the children with moderate motor difficulty (Amber (mDCD) group) and the typically developing children (Green (TDC) group) over time, p= .597 and was confirmed by post hoc Bonferroni corrections p=1.0.

	group	Mean	Std. Deviation	Ν
B1	sDCD	13.44444	6.023104	9
	mDCD	29.00000	5.244044	5
	TDC	33.00000	3.760699	15
	Total	26.24138	9.991622	29
B2	sDCD	13.6667	8.81760	9
	mDCD	30.2000	6.41872	5
	TDC	30.9333	5.04928	15
	Total	25.4483	10.27683	29
В3	sDCD	15.0000	8.97218	9
	mDCD	32.4000	2.19089	5
	TDC	31.6667	4.85014	15
	Total	26.6207	9.92261	29

#### Figure 6-40 Group means and SD of Balance over time Descriptive Statistics

### Summary

The results indicated that balance change over time was significantly affected by severity of motor impairment at baseline, as post hoc tests showed significant differences between groups for balance over time. There was a significant difference between the Red (severe) group and the Amber (moderate) group and also between the Red (severe) and the Green (typical) group, but not between the Amber (moderate) group and Green (typical) group.

Children in the Red group (sDCD) progressed significantly worse in balance than children in the Amber (mDCD) group and also to children in the Green (TDC) group. The mean balance scores (see Fig. 6.41) were consistently the lowest for the Red group over time (T3 mean = 15.0, SD=9.0). Although the children in the Amber group appeared to progress differently from the Green group, the difference did not reach statistical difference. The group mean for balance at T3 for the Amber group (mean=32.4, SD=2.2) was higher than the group mean of the Green group at T3 (mean=31.7, SD=4.9). See Fig. 6.40



Figure 6-41 Plot of group progression of Balance over time

# 6.23 Repeated measures ANOVA of aiming and catching over time

Skills in aiming and catching are required for many playground games and are fundamental skills for primary school age children. By secondary school age these skills are usually refined and enable children to participate in racquet sports and team sports. Impairment in these skills can have a profound impact on leisure pursuits and the ability to enjoy and participate team sports. An investigation of any statistical difference between the groups for the progression of aiming and catching was important. A one-way between groups (sDCD, mDCD, TDC), repeated measures (time 1, time 2, time 3) analysis of variance (ANOVA) was employed to examine the dependent variable of MABC2 aiming and catching score. Box's test of equality of covariance was not significant, signaling equality across all groups. Mauchley's test was also not significant and indicated that the assumption for sphericity could be assumed [ $\chi^2$  (2) = 2.249, p= .325].

	group	Mean	Std. Deviation	Ν
AC1	sDCD	10.2222	2.90593	9
	mDCD	13.8000	4.26615	5
	TDC	18.1333	4.53347	15
	Total	14.9310	5.30452	29
AC2	sDCD	11.1111	3.10018	9
	mDCD	12.6000	2.07364	5
	TDC	18.1333	5.93857	15
	Total	15.0000	5.66947	29
AC3	sDCD	11.8889	2.97676	9
	mDCD	14.8000	5.58570	5
	TDC	18.2000	4.88730	15
	Total	15.6552	5.20491	29

Figure 6-42 Aiming and catching means & SD over time Descriptive Statistics

#### Within subject effects:

The repeated measures ANOVA determined that there was no significant difference between means for aiming and catching scores at TI, T2 and T3 [F (2,52)= .947, p= .395,  $\eta$ 2= .035] observed power= .205.

The group by time interaction was also not significant [F (4,52) = .492, p= .742,  $\eta$ 2= .036] observed power .158

#### Between subject effects:

There was, however, a significant difference between the groups in the way they progressed in aiming and catching over time. The repeated measures ANOVA determined that the groups differed significantly [F (2,26) = 10.109, p = .001,  $\eta$ 2= .437] observed power .974.

Pairwise comparisons showed that once again it was the children with the most severe motor difficulty (Red (sDCD) group) that progressed significantly worse than the typically developing children (Green (TDC) group),  $p \le .001$ , confirmed by post hoc Bonferroni corrections  $p \le .001$ . However, there was no significant difference between the children with the severe motor difficulty (Red group) and the children with only moderate motor difficulty (Amber group) for aiming and catching, p= .223, supported by Bonferroni corrections, p= .669. However, pairwise comparisons revealed a difference between the children with moderate motor difficulty (Amber (mDCD) group) and typically developing children (the Green (TDC) group) over time, p= .034. However, this was not supported by post hoc Bonferroni corrections p= .101



Figure 6-43 Plot of Aiming and catching group means over time

#### Summary

The results indicated that the progression in aiming and catching over time was also significantly affected by severity of motor impairment at baseline. Post hoc tests showed significant differences between the groups for aiming and catching over time. There was a significant difference between the children with the most severe motor difficulty (Red (sDCD) group) and typically developing children (Green (TDC) group), but not between the Red group and the children with moderate motor difficulty (Amber group), or between the Amber and Green (TDC) groups.

Children with the most severe motor difficulty (Red (sDCD) group) progressed significantly worse than typically developing children (Green (TDC) group). The mean aiming and catching scores (see Fig. 6.42) were consistently the lowest for the Red group over time (T3 mean = 11.9, SD=3.0). Although the children with moderate motor difficulty (Amber group) appeared to progress differently

from typically developing children (Green group), the differences did not reach statistical difference on post hoc testing (p= .101). Incidentally, there had been no statistical difference between the Amber and Green groups at baseline for aiming and catching either. Interestingly, there was no statistical difference between the children with severe motor difficulty (Red) and the children with moderate motor difficulty (Amber) groups for aiming and catching over time (p= .669) either. This suggests that the children in the Amber group still experienced difficulty with aiming and catching. The group mean for aiming and catching at T3 for the Amber group was (mean=14.8, SD=5.6) and for the Green group at T3 was (mean=18.2, SD=4.9).

# 6.24 Repeated measures ANCOVA MABC2 total score Covariate for IQ composite

Since there had been a significant difference in IQ between the groups at baseline, a one-way between groups (sDCD, mDCD, TDC), repeated measures (time 1, time 2, time 3) analysis of covariance (ANCOVA) was employed to examine the dependent variable of MABC2 total score, using composite IQ score as covariant. Box's test of equality of covariance was not significant, signaling equality across all groups. Mauchley's test was significant and indicated that the assumption for sphericity had been violated [ $\chi$ 2 (2) = 6.284, p= .043]; therefore Greenhouse- Geisser corrected tests are reported.

#### Within subject effects:

The repeated measures ANCOVA with Greenhouse-Geisser correction determined that the group by time mean MABC2 total scores differed significantly [F (3.25,40.6=3.505, p= .021,  $\eta 2=.219$ ] observed power= .763.

#### Between subject effects:

The results showed that there was still a significant difference between the groups in motor performance ability over time. The repeated measures

ACNOVA determined that the groups differed significantly [F (2,25) = 19.887, p  $\leq$  .001,  $\eta$ 2= .614] observed power 1.0

Pairwise comparisons using Bonferroni corrections showed that Red (sDCD) group progressed significantly worse than both the Amber (mDCD) group,  $p \le .001$  and the Green (TDC) group,  $p \le .001$ . However, the Amber (mDCD) group did not differ significantly from the Green (TDC) group over time, p = .256.

The results from the ANCOVA were similar to those of the ANOVA, indicating that the difference in IQ between the groups had not changed the outcome; that there was a significant difference between the children with SDCD and typically developing children, but that there was not a significant difference between the children with moderate motor difficulty and typically developing children over time.





Covariates appearing in the model are evaluated at the following values: IQcomp = 97.3793

# 6.25 Summary of results part II

Summaries of the group results of each ANOVA, with Bonferroni corrections are quoted below. These compare the group means for total motor change and each component of the motor change over time.

### ANOVA MABC2 total score over time

Red	Amber	p≤.001
Red	GREEN	p≤.001
Ambe	erGreen	NS p= .088

## ANOVA Manual Dexterity over time

Red	Amber	p≤.001
Red	GREEN	p≤.001
Ambe	erGreen	NS p= .308

## ANOVA Balance over time

Red	Amber	p≤.001
Red	GREEN	p≤.001
Ambe	erGreen	NS p= 1.0

ANOVA Aiming & Catching over time

Red	Amber	NS p= .669
Red	GREEN	p≤.001
Ambe	erGreen	NS p= .101

ANCOVA for composite IQ MABC2 total score over time

Red	Amber	p≤.001
Red	GREEN	p≤.001
Ambe	erGreen	NS p= .256

In addition the ANOVA of self perception over time

CSAPPA overtime

All groups NS p= .632

The children in the Red group (sDCD), who started with MABC2 total score  $\leq 5^{th}$  percentile remained significantly different to the children who started in the Green group (TDC) >25<sup>th</sup> percentile in all motor domains over time. Furthermore, the Red group remained significantly different to the children who started in the Amber group (mDCD) 6-16<sup>th</sup> percentile on MABC2, for total MABC2 score, manual dexterity and balance, but not for aiming and catching over time.

However, the Amber group did not significantly different from the Green group on any of the motor domains over time.

There was a significant different between the groups in IQ and so an ANCOVA with covariance of composite IQ score was run and revealed that the difference in motor ability over time was significant between the Red and Green groups and the Red and Amber groups, but not between the Amber and Green groups over time.

It therefore appeared that the children who met criteria for severe DCD (≤5<sup>th</sup> percentile MABC2) were not only significantly different from typically developing children at baseline in their motor and associated characteristics, but also significantly differed from typically developing children in their change in motor

ability over the two year period of observation. They therefore appear to be a very distinct group in terms of their motor progression, IQ and associated characteristics. However, they did not differ significantly from typically developing children or from the children with moderate motor difficulties in their self-perception either at baseline or over time.

The children who started in the 6-16<sup>th</sup> percentile of MABC2 with moderate difficulty in motor ability (Amber group) differed significantly from the group with severe DCD (Red group) and the typically developing group (Green group) at baseline. The Amber group also differed from the severe group over time in all domains except aiming and catching.

However, despite differing from typically developing children at baseline, the Amber group did not differ from them significantly in any motor domain over time. This is because all the children in the Amber group improved their motor ability over time and achieved typically developing scores by T3. That is, they were indistinguishable from the typically developing group over time and apparently caught up.

In answer to the research question "How stable are the motor characteristics or do they change over time?" These data suggest that in this study the group of children with severe motor impairment differed both from the group of children with moderate motor difficulties and from the group of typically developing children at baseline. Furthermore, they progressed differently in their motor performance ability over time to both the typically developing children and the children with moderate motor difficulty. Their motor characteristics appeared to remain relatively stable over time, whereas the children with moderate motor difficulties changed over time.

In answer to the research question "How consistent are their self perceptions of their adequacy, enjoyment and predilection to take part in PA or do they change over time?" The data suggests that there was no significant difference in the group's self-perception either at baseline or over time, despite their difference in

motor progression over time. This points to consistency in the group mean children's self-perception over time for all groups.

### 6.25.1 Correlations to examine possible explanatory relationships

In order to investigate the unexpected Red group CSAPPA results further baseline correlations were undertaken to check for any relationships. It could have been possible that children with lower IQ did not understand the CSAPPA questions, in which case we could expect a significant correlation (i.e. p<0.05) between IQ and CSAPPA score at baseline. However, the correlation was non significant. It could also have been possible that younger children did not experience the impact of poor motor control on participation in PA to the extent that older children do. In which case we would expect a significant correlation between age and CSAPPA at baseline. However, the correlation did not reach significance.

Pearson Correlations were run at baseline to examine any significant relationships.

There were no significant correlations (i.e. p>0.05) between:

CSAPPA and age (r= .341) (p= .052)

CSAPPA and IQ (r= .058) (p= .748)

CSAPPA and MD (r= .267) (p= .134)

There was a weak significant correlation (p= .05) between baseline scores of:

CSAPPA and MABC2 (r= .358) (p= .041)

CSAPPA and A&C (r= .346) (p= .049)

CSAPPA and Balance (r= .365) (p= .037)

However, medium significant correlations (p= .01 two tailed) were found between:

DCDQ 07 and age (r= .517) (p= .003)

CSAPPA and DCDQ 07 (r= .553) (p= .001)

However stronger significant correlations (p= .01 two tailed) were found between:

MABC2 and IQ (r= .648) (p< .001)

MABC2 and DCDQ 07 (r= .686) (p< .001)

Non-parametric Spearman's rho and Kendall's tau were also run and the results were the same, except there was a weak significant correlation between age and CSAPPA with Kendall's tau 0. 247 (p= .47)

It was clear that self-perception was not solely dependent upon motor performance ability and in order to understand more a case study approach whereby selected cases were investigated in more detail.

### Correlations www.statstutor.ac.uk

Assumes normal distribution, interval or ratio data. If not use Spearman's Rank

.0019	Very weak
.2039	Weak
.4059	Medium
.5079	Strong
.80 – 1.0	Very strong

#### Table 6.23 strength of correlation

# 6.26 RESULTS PART III: Impact of characteristics, stability or change

The previous results described the different profiles of children with and without DCD and their stability and change in motor performance ability and selfperception over time. The results from analysis of these data addressed the first three research questions pertaining to the characteristics of children with and without DCD and the stability or change in their motor profiles over time. The children with different severities of motor difficulty progressed differently, yet little was known about whether this had an impact on their experience of participation in physical activity. Part III of the study addressed this with the research question:

How do the characteristics and stability or change in motor ability and selfperceptions impact experiences of participation in physical activity from the child's perspective?

# 6.27 Group analysis of rates of participation in extra curricular PA

The groups categorized by motor performance ability at baseline testing (T1) not only revealed different characteristics but also different patterns of stability and change in motor performance ability over time. Moreover, analysis revealed that the groups also had different rates of participation in extra curricular physical activity (PA). Please refer to Fig. 6.45. The children in the lowest 5<sup>th</sup> percentile of motor ability had the lowest participation rate of less than 6% of the sample. Those with moderate motor impairment did not fair much better, only just fewer than 9% of children in 6-16<sup>th</sup> percentile group participated in extra curricular PA. Whereas 26.5% of children in TDC motor range participated in PA. These results show greater contrast when examined at subgroup level in table 6.24.

Table 6.24 Motor	subgroup	participation	rates in (	extra c	urricular PA
	Subgroup	puillipullon	races m	chu u u	un neurun n m

≤5 <sup>th</sup> MABC2 (n=10)	6-16 <sup>th</sup> MABC2 (n=7)	≥25 <sup>th</sup> MABC2 (n=17)
2 (20%)	3 (42.86%)	9 (52.9%)

Figure 6-45 Overview of group motor progression and participation in PA



The children with the most severe motor impairment, ( $\leq$ 5<sup>th</sup> percentile MABC2) possessed distinctive characteristics and progressed very differently to children in the other two groups. They had greater areas of difficulty across domains such as daily activities, IQ and SEN and furthermore, had stable low motor performance ability over time. It is therefore not surprising that they participated much less in extra curricular PA than those with typical motor development and slightly less than those with a moderate motor impairment. However, this places the children at greatest risk by limiting opportunities for active participation and motor practice and could lead to secondary problems associated with inactivity.

The children with moderate motor impairment at baseline testing also had a much lower rate of participation in extra curricular PA than TDC, but surprisingly only slightly greater participation than the severe group, despite their variable motor progression and improvement to that of TDC over time. These children are therefore also at risk of the same secondary problems.

As predicted, the typically developing children, ( $\geq 25^{th}$  percentile MABC2) had the highest rate of participation in PA, as 53% of the group participated in PA. This group was therefore nearly twice more likely to participate in PA than the lowest motor ability group. These results indicated that the children with any motor impairment were at a distinct disadvantage for participation in extra curricular PA.

#### Summary

The extant literature indicated that participation in PA is dependent upon many factors, including the family context and resources. However, group contextual data from the children and families at baseline testing (T1) indicated that there was no difference in SES and family resources between those with motor impairment and those without. Furthermore, 80% of the sample attended the same school and lived in the same area with access to the same community resources. Moreover, there was no difference between the groups in their self-
perception of adequacy predilection and enjoyment of PA either at baseline or over time. The difference in participation rates appeared to be due to different motor abilities. Children with motor impairment, either moderate or severe in this study had very low participation rates (less than 9% and less than 6% of the whole sample respectively) and are therefore at risk of the secondary consequences of non-participation in PA. However, the TDC group only had 26.5% participation rate, which is relatively low and also a cause for concern.

Although group data highlighted differences between the groups, it did not offer explanation for the underlying cause of the differences. On the surface, motor ability appeared to be an important factor, but missed out any complexity in the individual context. Examination of individual data was therefore undertaken by case analysis to explore more fully the different participation rates of the children from each motor group. The case sampling method indicated in chapter five required modification as the stable and changing profiles method outlined was not possible, because all the children in the 6-16<sup>th</sup> percentile group had changing profiles and changed groups and the entire ≤5<sup>th</sup> percentile group had stable profiles to remain in the same group. However, measures of IQ, presence of associated characteristics, measures of child self-perception and an account of family and school context were available to augment the group data and provide detailed case analysis.

#### 6.28 Cross case analysis

Where group comparison explored the trends, the differences and similarities in the group averages, it told us nothing about specific individuals and their context. This study design was ecological, which permitted cases to be studied within the context of Bronfenbrenner's Bio ecological framework. A cross case analysis was undertaken to compare both the individual and the contextual characteristics of the children who participated in extra curricular activity in order to explore which factors might be facilitative. Children that were typical and atypical for their group were chosen from each category. It was predicted, from the extant literature, that the characteristics of the children who participated in PA would demonstrate a high level of motor performance ability, enjoy PA and have a positive self-perception about their ability. Additionally, that the families would rate PA as important, have adequate resources and that the parents would participate frequently in PA. However, the findings revealed some exceptional cases and by comparing these and typical cases some themes and points illustrated areas for future investigation and possible intervention.

Figure 6-46 Bronfenbrenner's bio ecological model used as a framework to explore differences in children's perception of participation in PA

Environment	<ul> <li>School &amp; SEN policy, P.E. policy, after school clubs</li> <li>proximity of local amenities</li> </ul>
Family	<ul> <li>SES &amp; car ownership</li> <li>Attitude to PA</li> <li>Siblings, resources</li> </ul>
Child	<ul> <li>DCD/AC/ motor ability</li> <li>IQ, CSAPPA</li> <li>interest, enjoyment</li> </ul>

#### 6.28.1 Environment (exosystem and mesosystem levels)

The majority of the sample (80%) attended the same primary school and provided data to consider the context of the children in more detail. All but one of the chosen cases (a girl) attended the same primary school. She was included as a case because she transitioned to secondary school during the study. The primary school was situated in an area of deprivation where 70% of the children received free school meals. The school had an inclusive ethos and ran a number of after school clubs and activities ranging from sedentary

activities such as chess and IT club to active physical activities such as athletics, cricket club and football. The school playground had parallel bars, tyres and stepping-stones as permanent fixtures and the children could also play with hoops and balls in playtime. P.E. lessons were inclusive and all the children were encouraged to participate in sports such as rounders, cricket and field sports. There was a large playing field next to the school and a community leisure centre with swimming pool close by, where the children had school swimming lessons. There was a good bus route between the leisure centre and surrounding areas, however, the roads were very busy and the leisure centre was on the opposite side of a dual carriageway to the school.

#### 6.28.2 Family (microsystem and mesosystem levels)

The family characteristics of the children who did not participate in extra curricular PA were compared with those of the children who participated to investigate any distinguishing features.

	Participating children (P) n=14	Non-participating (NP) children n=20
	(% of P children)	(% of NP children)
Low importance of PA ranked ≤2/4 by parents	N=1 (7.14%)	N=10 (50%)
Low parental participation in PA ≤2 x monthly	N=1 (7.14%)	N=8 (40%)
No car	N=1 (7.14%)	N=9 (45%)
No siblings	N=5 (35.71%)	N=4 (20%)
Unemployed parents	N=1 (7.14%)	N=8 (40%)

Table 6.25 Comparison of contextual characteristics of participating and non-participatingchildren

### 6.28.2.1 Summary of the characteristics and context of the children who did not participate in extra curricular PA

The children who did not participate varied widely across motor performance scores and were present in each subgroup. Although in general children with poorer motor performance participated less than typically developing children, motor performance ability did not appear to be the only factor in participation for these children. Economic resources and family attitude appeared to be important factors for children who did not participate in extra curricular PA. Car ownership was low, either no car or just one car per family. Additionally, the number of families with more than one child under 18 was high across the children who did not participate, suggesting that sufficient physical resources may be an important factor in participation in extra curricular PA.

The importance a family attached to PA and how often the parents participated themselves also appeared important, as the parents of the children who participated less generally ranked importance lower and participated less frequently themselves.

### 6.28.2.2 Summary of the characteristics and context of the children who participated in extra curricular PA

The families of the children who did participate in extra curricular PA all ranked PA high, the parents all participated weekly or twice weekly themselves. All but one family had one or more cars and the majority had two or fewer children under 18 in the family, so that they appeared to have adequate resources.

It appeared that family resources and characteristics were important for all children who participated in PA, but especially for those with motor difficulties (i.e. Red and Amber groups).

#### 6.28.3 Child

The characteristics of stability or change in motor performance (MABC2 scores) and self-perception (CSAPPA scores) were compared by subgroup of category of motor ability at baseline. Typical profiles are shown in Fig. 6.47 & 6.48.

#### 6.28.3.1 Children who participated

Despite their verbal IQ ranging from 79-122, some commonalities emerged across the children who did participate. All the children enjoyed or had predilection to take part in PA at 80% or above, they also rated their adequacy to take part in PA at 70% or above on CSAPPA subtests. All did not change their self-perception over time greater than four points on CSAPPA total score, but change in their motor performance ability over time was not a common factor. So there appeared to be more stability in self-perception across these cases, but not for motor ability.

#### Figure 6-47 Classic profile of MABC2 & CSAPPA over time for TDC



#### Classic Profile of MABC2 & CSAPPA over time for TDC

Vicky McQuillan

Figure 6-48 Classic profile of MABC2 & CSAPPA over time for DCD+AC



Vicky McQuillan

### 6.28.3.2 Stability and change of motor performance ability and self-perception

Table 6.26 Summary of stability or change for TDC/Green group

Stability or change >10 MABC2	Change in MABC2 total score T3-T1	Change in CSAPPA total score T3-T1
Stable n=9	Ranged from -16 to +17	Ranged from -23 to +15
Deteriorated n=4		
Improved n=4		

The change in CSAPPA scores did not necessarily follow the same pattern as the change in MABC2 scores over time for the TDC. However, the children with stable motor profiles tended to have more stable self-perception with individual change in CSAPPA scores ranging from -4 to 0 over time, although group range was large.

Table 6.27 Summary of stability or change for moderate/Amber group

Stability or change >10 MABC2	Change in MABC2 total score T3-T1	Change in CSAPPA total score T3-T1
Stable n=0	Ranged from +11 to +20	Ranged from -21 to +11
Deteriorated n=0		
Improved n=5		

The change in CSAPPA scores did not appear to have a pattern for the children with moderate motor impairment. Despite the motor scores improving for all five children, the self-perception for three of the children deteriorated over time.

Table 6.28 Summary of stability or change for severe/Red group

Stability or change >10 MABC2	Change in MABC2 total score T3-T1	Change in CSAPPA total score T3-T1
Stable n=7	Ranged from -7 to +18	Ranged from -26 to +11
Deteriorated n=0		
Improved n=2		

The change in CSAPPA scores appeared more erratic for the children in the severe group. The two children who improved their MABC2 score > 10 points actually both scored lower on their CSAPPA over time.

#### 6.28.3.3 Children who did not participate

The children who did not participate in extra curricular PA had IQ ranging from 55-131 and so IQ did not appear to be an important factor. However, there

were many more (14) children with associated characteristics who did not participate than who participated (two). There was less stability over time in their self-perception, with change in total CSAPPA score ranging from 1-26 points in the children who did not participate. A greater proportion of children with lower motor ability did not participate in extra curricular PA, but they were evident across all motor abilities.

#### 6.28.4 Selected cases

Selected cases from each group are described to examine and compare the patterns of progression and contextual details. Let us first consider the cases examples of the children who did not participate in extra curricular PA and characterise them by the Red, Amber and Green groups.

#### 6.28.4.1 Case 1 Non-participating child Red (≤5<sup>th</sup> percentile MABC2) group

The non-participating child case in the Red group is typical of the children in this group that did not participate in PA, all of whom had associated characteristics in addition to meeting the criteria for DCD. This 8-year-old boy had a diagnosis of autistic spectrum disorder and also had attention difficulties in addition to DCD. His verbal IQ was 95. He had very low motor performance ability in manual dexterity, aiming and catching and balance (all scores at or below the first percentile). He also had low enjoyment of PA (25%) and low selfperception of his predilection to take part in (58%) and of his adequacy in performance of PA (39%). This pattern was as predicted and it was unsurprising that he did not participate. Of important note is that none of the children in this group changed motor category over time, i.e. their motor performance ability remained stable over time. However, despite his motor performance remaining stable, he did not have stability in his self-perception over time. His enjoyment remained low, but his scores in predilection and adequacy to take part in PA halved over time. He clearly did not enjoy PA and it appeared that he noticed a gulf between his motor performance and that of his peers over time and became less inclined to participate in PA.

His contextual data sheds a little more light on his case. Despite his parents ranking importance of PA high (3 out of 4) and participating weekly, they did not have a family car and also had another child with DCD and ASD characteristics of primary school age. Thus, finding suitable extra curricular PA and being able to transport the children were two additional barriers for this family.

The Amber cases were more complex and appeared to fall into two categories; those that enjoyed PA and rated their self-perception over 80% yet still did not participate and those that clearly did not enjoy PA. However, some of this group was lost to follow up over time and so data is incomplete. Contextual data may be more important for these children and will be discussed.

# 6.28.4.2 Case 2 Non-participating child Amber (6-16<sup>th</sup> percentile MABC2) group

This 8-year-old boy had a verbal IQ of 79. His motor performance ability at baseline measurement put him in the 'at risk' category and he also had behaviour difficulties consistent with oppositional defiance disorder. His manual dexterity score was at 1<sup>st</sup> percentile, his aiming and catching was at the 50<sup>th</sup> percentile and his balance was at 25<sup>th</sup> percentile. This profile should not preclude him from participating in ball games. His self-perception of enjoyment of PA was 92% and his predilection to take part and his adequacy in were both 89%, so he rated himself high across the three categories. However, both his motor ability and his self-perception changed over time. Despite the fact that his motor performance improved to that of typically developing peers and his self-perception improved, he still did not participate in extra curricular PA.

His contextual family data helps illustrate some important factors. There were four children under 18 years old in the family and his father was caring for the children alone. The family owned one car, three of the other children had attention or behaviour difficulties and, due to few economic resources, the children were entitled to free school meals. In addition the family ranked the importance of PA low and his parents rarely participated in PA. Therefore, the contextual circumstances of this child add additional barriers to participation in extra curricular PA.

# 6.28.4.3 Case 3 Non-participating child Green (≥25<sup>th</sup> percentile MABC2) group

The child in the Green group without associated characteristics is typical of other cases in this category. This 8-year-old boy had a verbal IQ was 132 and he was bright and able at school. He rated enjoyment of PA at 75% (most rated lower than 80%) despite rating adequacy at 86% (most rated 80% or above). His overall motor performance was in the typical range. His aiming and catching was at 75<sup>th</sup> percentile, his balance at 25<sup>th</sup> and his manual dexterity was at 9<sup>th</sup> percentile. This sort of profile would not hinder participation in ball games or schoolyard activities. His motor performance improved over time but he did not change motor category over time (i.e. remained in the typically developing range). However, he did change his self-perception over time. Noticeably his adequacy and predilection scores reduced over time and lower enjoyment (75%) also appeared to be an important factor here.

In addition, his contextual data also helped to explain his non-participation. He was one of four children under 18 years old, cared for by his single mother. Importance of PA was ranked low (2 out of 4) and undertaken rarely. There was no family car and limited family finance.

#### 6.28.4.4 Case 4 Non-participating child an exceptional case

This 8-year-old boy had a verbal IQ of 112. He started in the Green group, but his motor performance deteriorated so much that he changed groups. He had associated characteristics, which included autistic features and attention difficulties. He was an atypical case. Despite his high verbal IQ and motor performance in the typical range across all subtests at the starting point, he rated his self-perception of predilection (33%), adequacy (29%) and enjoyment (42%) very low. His perception changed over time, as did his motor performance, both of which deteriorated over time. He was afraid of ball play and often refused to go into the playground. His very low self-perception was an important factor, but he also had some contextual barriers too. He was an only child living with his single mother. They had no family car and importance of PA was ranked low (2 out of 4).

#### 6.28.4.5 Case 5 Non-participating girl exceptional case

Another exception appeared with a girl who started in the Amber group and despite her motor performance improving over time, her enjoyment and selfperception started high but deteriorated over time. This 10 year-old girl had a verbal IQ of 122 and her manual dexterity and aiming and catching scores were both either at or below the 5th percentile and her balance was at the 75<sup>th</sup> percentile. She rated her enjoyment of PA at 100% and her predilection to take part and adequacy in PA at 94% and 82% respectively at baseline. Her parents rated PA very high (4 out of 4) and participated weekly themselves. There were two children under 18 years old in the family and they had one car. Her motor performance improved, so that she performed in the typically developing range over time, and moved to the Green group. So despite improving motor performance, adequate family resources and a positive parental attitude towards PA this girl did not participate in extra curricular PA. This pattern also occurred with two other girls of a similar age. Interview data revealed some interesting issues that were important to her and her age and gender may well be overriding factors.

Let us now consider the characteristics of the children who did participate in PA. It was predicted that the characteristics of children who did participate in extra curricular PA would be a good motor performance, especially in the aiming and catching and balance components of performance, as these skills are required for team games and most sports. Additionally, it was predicted that they would have high enjoyment and self-perception of predilection to take part in PA and adequacy in performance of PA. However the results showed some unexpected results.

Importantly, children from the Red and Amber lower motor performance ability groups did enjoy PA and participate and have a good self-perception. By examining their family context we are able to understand a little more.

### 6.28.4.6. Case 6 a participating child from Red (≤5<sup>th</sup> percentile MABC2) group (DCD plus AC)

This 10-year-old boy had additional associated characteristics of autism and attention difficulties as well as meeting the criteria for DCD. His verbal IQ was 89. His motor performance ability was low (1st percentile for manual dexterity, 2<sup>nd</sup> percentile for aiming and catching and 5<sup>th</sup> percentile for balance). However, he rated his enjoyment of PA at 92% and his predilection to take part in PA at 80% and his adequacy at 70%. Both his motor performance ability and his self-perception remained stable over time. Thus, despite his low motor performance ability, which did not change over time, he enjoyed and wanted to participate in PA and rated his performance quite well. Some of his contextual data and interview shed more light on his case. He was an only child and lived with his single working mother. She ranked importance of PA high (3 out of 4) and participated twice weekly herself. They also had a family car and therefore had independent transport.

### 6.28.4.7 Case 7 a participating child from Red (≤5<sup>th</sup> percentile MABC2) group (pure DCD)

This boy was 10 years old, had a verbal IQ of 85 and two parents in full time work. He was the youngest child and had brothers who were 18 years or older. His motor performance ability was low (manual dexterity was at the  $2^{nd}$  percentile, balance at the 16th percentile and aiming and catching at  $25^{th}$ 

percentile). However, he rated his enjoyment of PA and predilection to take part in PA as 100%, and his adequacy at 86%. Neither his motor performance ability nor his self-perception changed over time and he clearly loved PA. He was the tallest boy in the school. His parents rated PA as important and participated themselves weekly. They had two cars and, importantly, his older brothers had a keen interest in playing rugby and encouraged him to play. He was a member of a local rugby team, practiced and played in matches twice weekly and thoroughly enjoyed it.

## 6.28.4.8 Case 8 a participating child from Amber (6-16<sup>th</sup> percentile MABC2) group

This boy was 10 years old, had a verbal IQ of 79. His motor performance ability was low, but manual dexterity was lowest at 5<sup>th</sup> percentile, aiming and catching was at 25<sup>th</sup> percentile and his balance was at 37<sup>th</sup> percentile. He rated enjoyment of and predilection to take part in PA as 100% and his adequacy at 92%. He loved PA and was part of a community football team that played twice a week. His motor performance fluctuated over time from the Amber to Green group and back to the Amber, but his self-perception did not change over time. His family ranked the importance of PA high (4 out of 4) and participated twice weekly. His parents were divorced but shared his care. There were no other children under 18 years in the family and they had a car. Again the high rating of importance of PA and supportive parents with sufficient resources appeared important factors.

#### Summary

The factors associated with participation in extra curricular PA were families who ranked PA important, parents who participated themselves frequently and either had a car or few other children under 18. The important child characteristics were an enjoyment of PA and a good self-perception of predilection and adequacy in PA irrespective of their motor performance ability. Furthermore, their self-perception did not change dramatically over time. Some

children with marked motor difficulties rated their enjoyment and predilection to take part in PA high. For them family attitude towards PA, car ownership, and fewer children under the age of 18 appeared to be particularly important prerequisites for these children to participate in extra curricular PA.

Factors associated with lower participation were higher rate of parent unemployment, more single parent families, fewer cars, higher rate of families with two or more children under 18 years, parents rating attitude to importance of PA low and parents rarely undertaking PA themselves.

#### 6.29 Themes from child's perspective interviews

Interview procedure was outlined in chapter five. A child from each group of motor ability, if possible one with a stable and one with a changing profile and one with AC and one without was selected. In addition, one participating in extra curricular physical activity and one not participated from each category was included. Ten children were interviewed and common themes emerged from children who participated and those who did not.

Themes that emerged from interviews with children who participated in extra curricular PA and those who did not are detailed in table 6.29 and table 6.30 respectively. Some of the main themes are described below.

#### 6.29.1 An encouraging adult:

Regardless of their motor ability, children who participated in extra curricular PA mentioned a supportive adult, somebody that took an interest, introduced them to a sport or played active games with them.

"When I was young my Dad erm said to me 'do you want to have a kick about?" and I had a kick about with him and I just got into it, erm I play. Me Dad got me, me Mum and Dad got me into a team and I now play for another team and I'm brilliant." Table 6.29 Mixed methods analysis grid for children who participated in PA (After Creswell & Plano Clark, 2011)

Face to face interview Themes	Survey questionnaires data
For children who participated in	
extra curricular PA	
1. Parental support: children talked of	Ranking of importance of PA to
parents taking an interest, organizing	parents (≥3)
enrolment in clubs, watching them or	Parents ranked importance 1-4
joining in, lifts to and from venues	(1 being not important)
2. Siblings to play with: children	Other children under 18 in family
talked of siblings, neighbours or	
cousins to play with	
3. Enjoyment of PA: children identified	Frequency of Parental
P.E. as fun, they talked of activities	participation in PA
they liked outside	(weekly/ twice weekly)
4. Participation in community team	Car transport, owned a bike
sport:	
football or rugby teams, sense of	
belonging, regular practice sessions,	
coaching with an interested adult	
5. Facilities nearby: children 9 and	Parent employment: one or both
over talked of getting to a play area	parent working
independently by bike, bus or walking	
e.g local swimming pool, playing field	
or somebody's garden or street, having	
a pet to play with	

Table 6.30 Mixed methods analysis grid for children who did not participate in PA (after Creswell & Plano Clark, 2011)

Face to face interview Themes	Survey questionnaires data
For children who did not participate	
in extra curricular PA	
1. <b>Parental support:</b> children did not mention parents taking an interest in	Ranking of importance of PA to parents (≤2)
PA, or joining in or playing with them	Parents ranked importance 1-4 (1 being not important)
2. Solitary activity (video games):	Other children under 18 in
children talked of mine craft or other	family, Associated
games, rarely of any friends	characteristics
3. Dislike of PA: some children	Frequency of Parental
identified P.E. as unpleasant. they	participation in PA
talked of fear of the ball, dislike of team games in the playground, others found	(rarely – twice monthly)
it boring or preferred other actvities	
4. Other interests: music, drama, or	Low competency in PA (MABC2
other interests the were motivated to	results)
do and feel a sense of mastery	
5. Difficulty finding playmates:	No transport, no bike, free
living to far away from other children	school meals, associated
the same age, difficulty with social	6112120101131103
interaction	

#### 6.29.2 Enjoyment of active outdoor games or P.E.

Another theme that emerged was the joy and fun element in movement and that play was incorporated in P.E. lessons. Many of the children cited P.E. as one of their favourite lessons irrespective of their motor ability. *"P.E. it's like different than most lessons, you get to move around and (pause) do fun things."* 

"Lots of good games and in P.E we go to play stuck in the mud, erm, dancing, erm then like we might erm, Simon Says and more good games."

"I like P.E. because I like to run around and run around"

#### 6.29.3 Somewhere to play

Children talked of the importance of somewhere to play nearby, preferably that they could reach without parents necessarily taking them.

"Ive got a back field so I've got massive, like massive grounds, massive back field with loads of football pitches and everything, so I can just go and play there myself"

#### 6.28.4 Somebody to play with

The children who participated in PA talked of having fun with siblings, cousins or children as close neighbours.

"We go up to the park or we go into the house and play on games"

*"I play my guitar, I like doing football, I like playing with my brother, I like writing, running and reading"* 

However, some of those that did not participate mentioned lack of someone to play with, either because their parents worked and childcare arrangements caused the difficulty or they did not live near children of a similar age.

"I go to me Nans almost every day so I don't get to have any friends where I live"

#### 6.29.5 Other interests

Another factor in the children that did not participate in PA was interest in other hobbies or activities and video games, X-Box and PlayStation all featured here. However, some children mentioned playing with pets, or a love of music, dance and drama that they would prefer to do. This was particularly the case for three of the five girls in the study.

"I don't mind P.E., its not like the best, but it's like one of those subjects which sometimes it's good, sometimes it's not good" ... I'd change the school curriculum and erm have instead of music and drama once a week I'd have it twice a week definitely"

#### Summary

Analysis of both the group and individual data indicated that children with typical motor ability and a stable self-perception were more likely to participate in PA. However, closer analysis of the individual data revealed that some children from all categories of motor ability were able to participate and enjoy PA, providing they had appropriate environmental support and opportunities. This support was even more important for the children with low motor ability and associated characteristics, as they are at greatest risk of the most disadvantages if they are not given opportunities for active participation in PA. This information offers areas for future intervention to support families to facilitate participation in PA for all children, especially those with DCD.

#### 6.30 Overall summary

Children with and without DCD aged 7-14 years were identified and assessed at baseline (T1) for a profile of their characteristics. When categorized by motor performance ability three distinct groups emerged, the children in the lowest fifth percentile of motor ability group had the greatest difficulty in everyday tasks, lower IQ, high prevalence of two or more associated characteristics and SEN. They also had a higher rate of first-degree relatives with developmental disorders. They also had a distinctive stable pattern of motor progression, with limited variability, over a two-year period. Their self-perception, however, did not differ from the other two groups, but they had a very low rate of participation in PA. The children with moderate motor impairment were indistinguishable from TDC in motor performance over time (by T3), but they had a highly variable pattern of motor progression. Interestingly, they also had a very low rate of participation in PA. The TDC had a less variable pattern of motor progression. Prevalence of participation in extra curricular PA was highest for those with typical motor ability, but case analysis identified that a child from all motor categories either with or without AC was able to participate, provided they had sufficient contextual support.

# CHAPTER 7 SUMMARY, DISCUSSION AND CONCLUSION

#### 7.1 Study aims

The aim of this study was to establish whether children with DCD with different severities of motor impairment had different profiles of characteristics. Another aim was to establish whether they had different motor progression over time. In addition, children's experiences of participation in physical activity were sought by choosing children with different characteristics and stable and changing motor profiles to better understand factors associated with their participation in extra curricular physical activity. Using a mixed quantitative and qualitative study design within the framework of Bronfenbrenner's bio ecological model permitted greater understanding of children's inter-individual differences and contextual influences. A number of questions were addressed regarding the profiles of children with DCD:

1. What are the characteristics of children with different severities of DCD compared to children without DCD?

These included their:

- Motor characteristics
- Associated characteristics
- IQ
- Ability to take part in everyday activities
- Socio-economic and family context
- Self-perception of their adequacy, enjoyment and predilection to participate in physical activity
- 2.1 How stable are the motor characteristics or do they change over time?

- 2.2 How consistent are their self-perceptions of their adequacy, enjoyment and predilection to participate in physical activity or do they change over time?
  - 3. How do the characteristics and stability or change in motor ability and self-perceptions impact experiences of participation in physical activity from the child's perspective?

Part one of the study examined the motor and other profiles of schoolchildren aged 7-14 years with and without motor difficulties. Part two followed their motor progression and child perceptions of their motor ability over time. Part three examined a smaller subset of children, identified with different characteristics and progression over time, to seek their experiences. The findings are used to inform better understanding of how different severities of motor impairment and the context of DCD can impact on children's experiences of participation in physical activity.

#### 7.2 Summary of study findings

A mainstream school based cohort of children with and without movement difficulties, identified by their teachers and identified with DCD according to DSM5 criteria were compared. Analysis comparing the children with and without movement difficulties did not identify differences in either socioeconomic status (SES) or age. However, when groups were categorized by their different motor performance ability, analysis of variance identified significant differences between the groups. The group in the lowest 5<sup>th</sup> percentile of motor ability had a very distinct profile of characteristics. Yet, there was no difference between the groups in the mean self-perception of their motor ability.

The second part of the study identified that the groups classified by motor ability progressed differently over time. Repeated measures analysis of variance identified that the children with different severities of motor impairment had distinct and different trajectories from each other. Those with the most severe motor difficulties were most likely to show persistent motor difficulties. The children with moderate motor impairment all showed improved motor performance over time and were no longer significantly different to typically developing children by the end of the study. Repeated measures analysis of variance of the children's self perception of motor ability over time, however, did not show any difference between the groups.

In the third part of the study children, who were typical and atypical cases from each of the motor ability groups, expressed different experiences of participation in physical activity. Participation and enjoyment of physical activity was not necessarily related to the children's motor ability or the progression of their motor ability over time, other factors were identified. The implications of these findings will be discussed in the following sections.

For data analysis a sample size analysis was performed using a significance level of 5%, a power of 80% and, based on a large effect size (d=.8); a sample size of 28 was required (Dunlop & Myers, 1997). The sample size for this study was 34 at baseline and therefore met these requirements. However, to enable statistical comparison of the three groups and in order to keep the power of 80%, the effect size was very large (d=1.5) and the sample size required was 8 in each group (Hinckle et al., 1994). The sample size of this study met these criteria for two out of the three groups (severe n=10, moderate n=7 & typical n=17). Repeated measures analysis of variance was used to investigate whether the groups differed in mean test scores and Bonferroni post hoc tests were performed to compare the mean scores of each group with each other. Alpha was set at 0.05.

Baseline data compared potential confounding variables of age and SES and revealed no significant differences between the groups, so that they could be considered to have an equal effect over time. However, IQ was also compared between the groups and showed that the children in the severe group (lowest

5<sup>th</sup> percentile of motor ability) had significantly lower composite and verbal IQ scores than TDC. Since this could potentially influence the children's progression, IQ scores were used as covariant in the final analysis but this did not change the outcome of results in that the groups still differed in their motor performance ability and progression over time.

#### 7.3 Profile of characteristics in DCD

### 7.3.1 What are the characteristics of children with different motor severities of DCD?

Children with DCD are often referred to as heterogeneous and, although a number of studies have searched for subtypes, there has been little consensus. One criticism is that most have investigated from information processing (IP) theoretical perspective and therefore narrowed the focus of investigation. Interest in motor development theory is moving away from a pure IP approach and starting to take account of the environmental context of development. This approach is encompassed by the dynamical systems theory, and some researchers are starting to propose combining both theories to take a hybrid approach to investigate the difficulties associated with DCD (Wilson et al., 2017b). Another source of heterogeneity in DCD is the presence of cooccurring conditions (often referred to as comorbidity). Although this has also been investigated it has not typically been studied in conjunction with the severity of the motor impairment. Indeed, the majority of DCD studies report results for children <15<sup>th</sup> percentile of motor ability and do not distinguish between different severity of motor impairment. Therefore, the main focus of this study was to examine the characteristics of children with and without DCD for different motor abilities.

The children in the lowest 5<sup>th</sup> percentile of motor ability appeared to be a very distinct group across a number of different domains.

#### 7.3.1.1 Motor characteristics

The motor difficulties in the severe group of this study were significantly different to the moderate and TDC groups in manual dexterity, aiming and catching and balance on MABC2 (P< .001), indicating that the severity and nature of motor difficulty is distinct. It is well established that the impact that this motor profile has on function in daily life is profound and interferes with activities of daily living, family life, schoolwork and play. Moreover, the evidence of wider deviance from typical development in cognitive and behavioural domains in this study and others (e.g. Schoemaker et al., 2013; Gillberg, 2010) indicates that the motor difficulties in the  $\leq 5^{th}$  motor percentile group of children may arise from different routes than those of children with more moderate motor impairment. For example, Peters, Maathuis & Hadders-Algra (2011) found that in a sample of (n=253) school age children (mean age 8 years) those with <5<sup>th</sup> percentile on MABC had a higher prevalence of more complex minor neurological dysfunction than those at the 5-15<sup>th</sup> percentile and TDC (54% vs. 17% vs 10%). This further supports the notion of a more atypical pattern of development in children with motor impairment in the lowest 5<sup>th</sup> percentile.

The children with moderate motor impairment (6-16<sup>th</sup> percentile) in this study did not differ significantly from the children in the typical motor ability group (>25<sup>th</sup> percentile) on balance or aiming and catching and only differed in their manual dexterity (p<. 001). However they were significantly different to the children in the lowest motor ability group across all motor domains (p<.001), adding further support that children with DCD in the lowest motor ability group have distinct profiles that differ from TDC and from children with moderate motor ability.

The children with the highest motor ability (>25<sup>th</sup> percentile) displayed typical performance across all motor domains.

#### 7.3.1.2 Associated characteristics

The tools in this study for screening profiles were carefully chosen psychometrically robust tools and a conservative 5 % cut-off used in order to minimize false positive results. Even so, the children in the lowest motor ability group had a much higher prevalence of associated characteristics (AC), of other developmental disorders such as ASD and ADHD, than either of the other groups. They were also more likely to have two or more ACs compared to the children in the other groups. In this study 90% had two or more ACs and this entire group were male. Other studies have found children with DCD with co-For example, Lingam et al. (2010) found 7-year-old occurring conditions. children in a large UK population cohort at risk of DCD, below 15<sup>th</sup> percentile of motor ability, likely to have additional traits of other developmental conditions. Gillberg (2010) also noted a higher prevalence of co-occurring disorders in a Swedish population sample children neurodevelopmental/ of with neuropsychiatric disorders and suggested that they are difficult to separate and share genes, environmental risk factors and symptoms. Therefore, the finding that 90% of children in the lowest 5<sup>th</sup> percentile of motor ability in this study had 2 or more ACs finds support in other studies.

The response rate for the parent questionnaire was lower for the group with moderate motor ability (6-16<sup>th</sup> percentile) and there were fewer children in this group, so the results should be interpreted with caution. However, two of the children in the moderate motor ability group also had associated characteristics but, in contrast to the lowest ability group, none had two or more ACs. Another interesting finding was that the typical motor ability group (>25<sup>th</sup> percentile) also contained 5 children with AC, one with 2 or more AC. From this we can conclude that children with ACs are distributed throughout the population of motor ability, but those with 2 or more ACs are much more likely to be concentrated in the children with DCD in the lowest 5<sup>th</sup> percentile of motor ability.

Recent analysis of children at risk of DCD aged between 7-9 years found a higher prevalence of co-occurring conditions in those  $<5^{th}$  percentile of motor ability (Schoemaker, Lingam et al., 2013) and supports this view. Schoemaker and colleagues used logistical regression in a large UK population based cohort (n=6959) to determine if the severity of motor difficulties related to co-morbidity in children at risk of DCD and found that the 7-9 year old children with severe motor difficulties (n=289) had a prevalence of 20-25% higher risk of problems in other developmental domains, even accounting for IQ and parent related confounding variables. Furthermore, they found that children with moderate motor difficulties (n=951) had a prevalence of 5% higher risk than TDC (n=5719). This further supports the findings that children with severe motor impairment are distinct from those with moderate impairment and from TDC and represents a much higher risk of developmental disadvantage.

#### 7.3.1.3 IQ

In addition there were significant group differences on mean IQ scores on KBIT2 (p<.001). The lowest motor ability group had the lowest mean composite and verbal IQ scores of all the groups, although post hoc tests revealed that only the difference between them and the children with typical motor performance (>25th percentile on MABC2) was significant for composite IQ and verbal IQ (p<.001). Other studies have also reported that children with DCD have lower IQ in comparison to those without DCD. For example, Flapper and Schoemaker (2013) found significantly lower IQ in 5-8 year old children with DCD < 15<sup>th</sup> percentile on MABC (according to DSM IV criteria) co-morbid with SLI than those with SLI alone. Furthermore, 66% of their sample of children co-morbid for SLI and DCD were in the lowest 5<sup>th</sup> percentile of MABC, adding further support to the present study findings that this group of children has very distinct characteristics.

This is important and may indicate that children in lowest 5<sup>th</sup> percentile of motor ability represent a separate subgroup of children with broader difficulties across

a number of domains including the motor domain, but this would need to be verified with more studies. There are also implications, which may indicate developmental differences relating to the CNS, as in this study the ≤5<sup>th</sup> percentile group had a 40% incidence of first-degree relatives with a developmental disorder in their family history, compared to 29.4% incidence for children who scored in range of TDC at baseline. However, and importantly over time, two children (11.76% of the group) of the TDC group with first-degree relatives with a developmental disorder subsequently changed group into the impaired range. Therefore, 7-14 year old children in lowest 5<sup>th</sup> percentile in this study were more likely to have a first-degree relative with a developmental condition than children whose motor ability remained in the typical range over time. This suggests a strong heritability associated with DCD with the lowest 5<sup>th</sup> percentile of motor ability. This was an interesting finding since the sample had been deliberately derived from a population based school sample, in order to avoid referral bias associated with the greater concentration of children with cooccurring conditions of clinic samples.

However, one consideration is that a high proportion of children in this study sample had symptoms of ASD and ADHD, both of which are thought to have high heritability and involve differences in neural substrates, which may account for some of the differences. Conversely, the one child without AC in the lowest motor ability group also displayed the same difficulties across domains (e.g. poor handwriting and spelling, lower verbal IQ, poor motor control in all subgroups and low score on DCDQ), which implies that the difficulties are associated with DCD and not ASD or ADHD per se. However, only similar studies with careful screening for ACs and larger sample size will determine whether this is the case.

The group of children with moderate motor difficulty (6-16<sup>th</sup> percentile MABC2) had lower mean IQ scores than the group with typical motor ability, post hoc analysis revealed that the difference between them just reached statistical significance (p=.053) for composite IQ score, and significant difference in their

mean verbal IQ scores (p<.001). These results may be erroneous due to the small sample size of this group. However, the trend showed that they had lower mean IQ scores than the group with typical motor ability and higher IQ scores than the group with lowest motor ability. Although the mean IQ score difference between them and the lowest motor ability group was not significant, it still points to the children with DCD in the lowest 5<sup>th</sup> percentile of motor ability having distinct characteristics. Unfortunately, it was difficult to ascertain the true frequency of first-degree relatives with a developmental disorder in the family for 6-16<sup>th</sup> percentile group, as few questionnaires were returned. However, of those returned, only 28.6% reported a relative with a disorder, which is still much less than that of the 5<sup>th</sup> percentile group. Taken together the higher prevalence of first-degree relatives with a developmental condition, the higher prevalence of two of more ACs and the lower mean IQ in the group with the lowest 5<sup>th</sup> percentile of motor ability indicates a group of children, not only distinct from TDC, but also distinct from those with moderate motor impairment.

#### 7.3.1.4 Ability to take part in everyday activities

The parents reported that the ability of the children with the lowest motor ability to take part in everyday activities was problematic. This group had significantly different scores to the other two groups when examined on DCDQ '07 (p= .001). However this may have resulted from difficulties arising from the symptoms of the AC as 90% of the children had co-occurring AC, only larger studies carefully detailing AC will be able to verify this. Although it is becoming apparent that DCD with AC occurs more frequently than not and is likely to be encountered in practice. Nevertheless, it does identify that children in the severe group encounter more difficulties in daily function that impacts on all areas of occupation at home than the other two groups and confirms the serious nature of the condition. School life was also affected as the children in this group had a much higher incidence of SEN and handwriting difficulties. Both of these potentially have a large impact on progress at school and seriously disadvantage children unless adequate provision is available for them. This group proved easier to identify, as teachers correctly identified 90% of the

children in the severe motor impaired group and 100% of the parents correctly identified that the children had problems, indicating that teachers and parents of the children in this study found it much easier to identify the children's difficulties in the severe motor impaired group.

The parents and teachers did not identify the difficulties as readily for the group with moderate motor difficulties, possibly as they were subtler in nature. However, studies indicate that these groups have difficulties resulting in reduced function and adverse consequences from their impaired motor ability (Wang, Tseng, Wilson & Hu, 2009). The mean DCDQ '07 score differed significantly from the severe group but not the TDC group. One possible explanation was that the TDC group also contained some children with AC and their parents highlighted difficulties associated with these. Interestingly, the moderate group was distinct from the severe group with significant differences in all motor domains, yet only differed significantly from the TDC in manual dexterity. Perhaps this was not sufficient to alert the attention of parents to functional problems.

### 7.3.1.5 Self perception of adequacy, enjoyment and predilection to take part in PA

The children's self rating of the adequacy of their motor ability, predilection to take part in physical activity (PA) and enjoyment of it (rated on CSAPPA) did not differ significantly between the groups at baseline. This is an interesting finding, since the extant literature reports that children with DCD have low self-efficacy in physical ability (Losse et al., 1991; Cantell, Smyth & Ahonen, 1994; Skinner & Piek, 2001; Poulsen et al., 2008; Cocks et al., 2009) and some report that children with DCD have significantly lower self-efficacy in physical ability than TDC (Cairney et al., 2005; Silman et al., 2011). This is thought to impact participation in physical activity. According to Harter's competence motivation theory (Harter, 1982) the assumption is that children with high-perceived competence in a specific domain are more motivated to participate in that domain, whilst those with low perceived competence are less motivated to

participate. One would therefore expect children with DCD to have low perceived competence in physical abilities, especially the children in the lowest 5<sup>th</sup> percentile of motor ability. However, this was not the finding in this study, despite the fact that the children with DCD participated less in PA.

Other studies have also found that children with motor difficulties do not necessarily have low perceived athletic competence and, further, have found no difference in perceived athletic competence between children with motor difficulties and TDC (Pless et al., 2002; Fliers et al., 2010; Noordstar et al., 2014; Toussaint et al., 2016).

It is possible that age related differences have influenced the results across studies. Harter (1988) differentiated between stages of self-awareness at different ages of children by describing the emergence of self-description at three developmental periods. The early period occurs at approximately at age 5, where the child observes and judges others, but does not realize that others can do the same to them and so is not able to critically evaluate the self directly. The second stage, around 6-7 years is where the child comes to appreciate that others are observing and evaluating the self and worry about criticism and being laughed at, but are still unable to critically observe the self directly. The third stage, usually around 8 years, where the child becomes interested in evaluating their own performance based on the standards that other people have for the self. They can begin to incorporate the observations of others into their own self-perceptions. They start to internalize these expectations into selfstandards and develop the capacity for self-criticism. This marks the emergence of the ability to compare oneself to others (Harter, 1988, p 61). However, as all the children in the present study were aged 8 or older, except one child, age and developmental stages of self-perception are unlikely factors in the results of this study.

Another area for consideration is the influence of children's self-esteem. Harter (1988) suggested that one's self-esteem is based on evaluations of judgments

of adequacy of the self across different domains in one's life. She suggested a domain specific approach to profile these differences in sense of adequacy across the domains. For example, a child may feel very competent at schoolwork, less adequate in peer relations and relatively incompetent in the domain of athletics. In addition, Harter (1988) suggested that whilst competence is considered one's source of self-worth, acceptance and positive regard are also important. Children with high self-worth feel that significant others (parents and peers) accept them, support them and hold them in high regard. Furthermore, perceived positive regard from significant others appear strongly related to self-worth, suggesting that children (aged 10-13) adopt the attitudes and opinions that they feel others hold toward the self. In addition, the degree that one feels competent in areas where success is important has the ability to discount the importance of domains where one feels less adequate is also an important determinant of self-worth (Harter, 1988, p72). This could point to the importance of the attitude of significant others (parents and friends) towards the child that can contribute to positive self-esteem and over ride the impact of motor difficulties on self-perception. Children of all levels of motor ability in this study described fun P.E. lessons, enjoying playing out with siblings, being a member of a team and reported parental encouragement and physical and moral support to take part in PA. It therefore suggests that the positive regard and attitude of significant others can have a powerful positive influence on self-perception and motivation towards PA.

Although the results in the extant literature appear equivocal, comparison of studies examining perceived self-efficacy in DCD is difficult due to methodological differences. Some of these include varying levels of adherence to DSM diagnostic criteria, different stages development of self-perception covered by large age range (6-17 years) of the children studied, few descriptions of any account of co-occurring conditions, different questionnaires used (e.g. Harter, CSAPPA, Dutch self perception profile for children) with questions that impact slightly different domains with different importance attached to the task described. Whether the questions include enjoyment or whether adequacy alone is questioned could also potentially impact results. The

context of the physical activity could also have a bearing, for example, competency at a lab based shuttle run could be perceived differently by children compared to the physical activity of their choice.

However, as this study indicated, children with all motor abilities can enjoy and express motivation to engage in physical activity. Children aged 8 and over are capable of making global judgments of their overall worth as a person and evidence supports two primary determinants of a child's self-worth (Harter, 1988). The first involves internalization of attitudes that significant others hold towards the self, the more one feels support and acceptance from others the higher one's self-worth. The second involves the degree to which one feels competent or adequate in areas where success is important, and can discount the importance of domains in which one is not competent results in high self worth (Harter, 1988). One's sense of self has a major impact on one's emotional life, which in turn influences children's motivation and behaviour (Harter, 1988, p 77). The results from this study indicated that factors other than motor ability were important in the children's perception of their adequacy, predilection and enjoyment of physical activity for the children in this study (mean age of 10 years, SD 1.66). The inclusive nature of the school and of P.E. lessons may have been important factors. Children's age may also be an important factor and this could potentially have implications for intervention, as it is easier to engage primary school children in physical intervention programmes before they become disillusioned and lose confidence and motivation to engage with physical activity.

#### Summary

It is evident from the data at baseline that the group of children with severe DCD, i.e. those in the lowest 5<sup>th</sup> percentile of motor ability, appeared to have very distinct characteristics from the other two groups in terms of their lower verbal and composite IQ scores, the increased co-occurrence of two or more AC and higher rate of SEN. Their motor difficulties tended to be across all subtests of MABC2, differed significantly from TDC in all subtests (p< .001) and

their difficulties were more likely to be identified by their teachers or parents. However, not necessarily identified by the children themselves, as there was no significant difference between the self-perceptions of the groups with different motor ability.

# 7.4 How stable are the motor characteristics or do they change over time?

The children from each group of motor severity were followed over time and any motor change was recorded using repeated measures of the MABC2. The same experienced OT administering the MABC2 in the same settings each time, which increased consistency and reliability of the repeated measures. Furthermore, motor change was not only quantified, but an effort was also made to capture clinically meaningful change in motor ability. Change was therefore determined using three methods: total MABC2 score, the MABC2 traffic light classification system and Holm et al. (2013) smallest detectable change (SDC) (+ or -10 points total MABC2 score).

# 7.4.1 The children in the group with lowest 5th percentile of motor ability

The children with the most severe motor impairment at baseline appeared to have a distinct pattern of motor change over time. The most striking feature was that none of the children in this group changed group membership over time, all remained in the most severe motor impaired group.

The next important finding was that repeated measures ANOVA of their motor ability showed that this group differed from the other two groups over time (p<.001). Even after accounting for the group mean lower IQ using ANCOVA this group was still statistically different from the other two groups (p<.001). Importantly, this indicates that they followed a different motor trajectory and further confirmed the differences found at the baseline of the study.

The persistent nature of the their motor difficulties was further supported by the fact that there the majority (77.8%) had no change in the individual scores for SDC (+ or - 10 points MABC2 total score) for each member of this group. This indicated that they had no clinical difference in motor ability from their baseline score over two years. Furthermore, although 22.2% of this group improved >10 points on MABC2 it was of insufficient magnitude to change group. However, no children from this group deteriorated over time. Therefore, the children with the lowest 5<sup>th</sup> percentile of motor ability had a general picture of stability of their motor ability over time as none improved sufficiently to leave the lowest 5<sup>th</sup> percentile of motor ability. This sets them apart as a distinctive group both at baseline and over time with persistent motor difficulties and answers two of the research objectives; 1) this group had distinctive profiles of characteristics and 2) their motor characteristics remained stable over time.

This may have important implications, both for identification of children in the lowest 5th percentile of motor ability and for intervention. First, they are much more likely to have co-occurring characteristics and a broader range of additional problems, which can seriously impact their function in daily life and these will require careful assessment. Kirby (2005) has long acknowledged the existence of conditions overlapping with DCD and called for screening across all developmental areas, preferably in schools. The evidence from this study suggests that this should particularly be targeted at the children with lowest 5<sup>th</sup> percentile of motor ability. Second, this evidence strongly suggests that the children in this group do not spontaneously clinically improve over time and so intervention is vital. Furthermore, because of their very different profiles and very different motor trajectories, they may require specific intervention, which is different to that required by children with moderate motor difficulties.

#### 7.4.2 The children in the group 6-16<sup>th</sup> percentile of motor ability

The children with moderate motor ability (6-16<sup>th</sup> percentile on MABC2) behaved very differently over time. Two secondary age children from this group were

lost to follow up and unfortunately the attrition rate in this group left insufficient numbers to give an adequate sample size for statistical analysis. Nevertheless, the analysis was undertaken to examine general trends and was augmented with case study contextual data and interview. Therefore, whilst interpreting these results with caution, they do give an insight into motor change over time.

All the remaining children in the moderate motor ability group improved over time and scored in the typically developing motor ability range by data point three, giving the impression of resolution of their motor difficulties. There was no statistical difference between them and the typically motor ability group over time, making them indistinguishable. Yet, the trajectories within this group were highly variable, nonetheless all improved their total MABC2 score by greater than 10 points over time and, by Holm's definition, showed a clinical difference in their motor performance. However, the pattern was not necessarily one of steady improvement for all, some children improved, then deteriorated then improved again. For example, the two girls started low and increased to the next percentile each data point into the typical range, whereas one boy with ASD started at the 9<sup>th</sup> percentile, dipped to the 5<sup>th</sup> percentile before ending on the 37<sup>th</sup> percentile. Another boy, with characteristics of ODD, showed less dramatic variation but still fluctuated over time, starting at the 9<sup>th</sup> percentile, moving to the 37th percentile, then the 25<sup>th</sup> before ending at 50<sup>th</sup> percentile for the final data point. This illustrates the extent of variability that the moderate group displayed in direct contrast to the stability displayed by the group in the lowest 5<sup>th</sup> percentile of motor ability.

It is difficult to draw useful conclusions about the reasons behind the stark difference in the relative trajectories of the two groups, apart from surmising that they represent different subgroups of children with motor difficulty. However, with such small group samples it would be unwise to accept this hypothesis without further testing with adequate sample sizes. Nevertheless, the girls in the group may provide a small insight into possible sex differences. Both girls improved incrementally over time to reach scores in the typical range, yet both rated their enjoyment and motor performance in the (impaired) suspect DCD range of the CSAPPA, despite improving their performance over time. This could be a case of low expectations and low self-confidence. Both girls continued to have low scores in aiming and catching over time and balance scores above the 50<sup>th</sup> percentile, but confessed they had little predilection to participate in physical activity.

In direct contrast, two boys in this group, with highly variable performance, expressed great enjoyment and predilection to take part in PA despite scoring similar motor scores to the girls. One had associated characteristics and the other did not. This may be less to do with presence or absence of AC and more to do with the effects of cultural influence on role expectations for boys and girls and warrants further study.

The conclusion we can draw from these data is that the children with moderate motor ability (6-16<sup>th</sup> percentile) in this study all improved their performance over time to reach typical motor ability by the 3rd data point. They may represent the lower end of a normal motor continuum and their motor ability appears variable. However, those that were tested for a 4th data point had deteriorated and we could conclude that it may be unstable because they were unable to sustain the higher level of motor performance.

# 7.4.3 The children in the group with 25<sup>th</sup> percentile and above of motor ability

The children who scored in the typical motor development (>25<sup>th</sup> percentile on MABC2) at baseline displayed yet another pattern of motor change over time. Over half (53%) showed no change in motor performance, they remained within <10 points MABC2 total score over time, demonstrating stability in their motor ability. One quarter of the children improved their motor performance and one quarter deteriorated over time. However, 2 of the 4 children that deteriorated over time had known diagnoses of ASD and ODD and both changed category
into the 6-16<sup>th</sup> percentile, moderate impairment range of motor ability. The other two had no known diagnoses, one remained in the typical range and the other (who had been adopted and had unknown family developmental history) changed to the 6-16<sup>th</sup> percentile. Although some children improved and some deteriorated, the majority 82% of this group remained in the typical range ( $\geq 25^{th}$  percentile of motor ability) over time and had stable motor ability. We can assume that some motor variability is to be expected in typical development, but that it does not generally impact on function. However, children at the lower end of typical motor ability appear more likely to encounter other difficulties, for example have AC, which may have a detrimental effect on their motor and other function.

# Summary

Distinct patterns of change emerged between the groups of children when they were categorized by their motor ability at the study baseline. Despite some children changing groups over time, none of the children from the lowest 5<sup>th</sup> percentile changed group, but persistently remained impaired at the lowest level of motor ability over time. Their motor performance was stable but poor, with no clinical improvement, in contrast to the 6-16<sup>th</sup> percentile group, which was highly variable, and 100% improved motor performance to  $\geq 25^{th}$  percentile over time. Conversely, the children who started  $\geq 25^{th}$  percentile of motor ability showed a mixture of patterns of change, but 82% remained in the typical range over time demonstrating relative stability of motor performance over time. These findings are of interest because this was not an intervention study and so any improvement in motor performance was spontaneous and reflected progression in motor development.

# 7.5 How stable are the self-perceptions over time?

Group statistical analysis of self-perceptions revealed little information to predict which children would improve their motor ability over time, as there was no significant difference between the groups for self-perception of enjoyment, adequacy or predilection to take part in PA at the start of the study or over time despite their differences in motor ability. Harter (1988) suggested that one's domain specific judgments and one's self-worth are moderately stable for most children over relatively short periods of time (e.g. a school year) and these results appear to support that.

However, examining the individual profiles of the children from each group revealed a different story. The three boys who initially scored in the typically developing range of motor ability, but then deteriorated over time, all ranked themselves in the at risk range of the CSAPPA by data point 3. They appeared to be aware of their poor motor performance. Two of the boys had ACs, one had ASD and the other ODD and difficulties in other areas. The other boy was adopted and had an unknown early history.

Conversely, two girls and a boy, who initially scored in the 6-16<sup>th</sup> percentile of MABC2 but improved over time to the typical motor ability range, ranked themselves in the at risk range of the CSAPPA by data point 3. Their self-perception had deteriorated over time, despite their improving motor performance. This suggests a lack of confidence in motor skills coupled with a lack of enjoyment or interest in PA had influenced their perception.

The children in the lowest motor ability group did not exhibit self-perceptions that matched their motor ability, all remained in the lowest motor category over time, yet individuals had a range of self-perceptions. For example, two boys improved their motor performance by more than 10 points over time, yet one rated his self-perception in the at risk range of the CSAPPA and the other did not, while some of the others rated themselves high on the CSAPPA. Clearly enjoyment and participation in PA was related to factors other than motor ability for them. Interestingly, Pless et al. (2002) also found a mismatch between self-perception and motor performance in their study of children with DCD. They found that high perception of self-competence did not necessarily relate to participation in PA, and some children with low self-perception did participate. Noordstar et al. (2014) used a Dutch self perception profile and did not find a difference in perceived athletic competence (PAC) between children aged 7-12

years with DCD (DSM IV criteria, IQ >70, <15<sup>th</sup> percentile on MABC) and matched TDC. In fact most children had a high PAC 64.5 % of the DCD group and 86.8% of the TDC group. However, there were large individual differences within the DCD group. They also found no difference in organized PA between the groups, but there was a significant difference between the groups for unorganized PA. They surmised from this that relationship between perceived athletic competence, motor ability and participation in PA is complex and is a supposition supported by this study.

Examining contextual information through an ecological lens permitted analysis of other potential influences on PA, such as parental attitude, available resources and suitability of the physical environment and the child's beliefs and motivations permitted analysis at the inter-individual level. In this study parent questionnaires, together with child interviews, facilitated collection of these data.

# 7.5.1 Ecological analysis with Bronfenbrenner's model:

At the macro system level of analysis:

On the surface the study neighbourhood appeared typical for the UK, as it had an average level of unemployment and an average utilisation of outdoor space for exercise when compared to England as a whole. However, the number of children in low income families was higher than average and the income deprivation indices, based on 2015 figures, was higher than average. Additionally, the borough had a higher than average number of children eligible for free school meals in 2014. So, although parents were in employment, their pay did not sufficiently cover their basic costs.

At the mesosystem level of analysis:

The school had robust SEN policy and a higher than average number of pupils with special educational needs, and also a very high level of pupils eligible for free school meals. It was well resourced and had committed and caring staff. The children attending the autism resource bases spent most of their time in classes with their year group in the main school and so had similar experiences in class.

At group level analysis the Red group had the lowest participation in extra curricular PA, less than half the level of TDC (20% compared to 52.9%). In addition, the Red group parents participated less in PA than parents of TDC (50% compared to 64%) but more than the Amber group. However, there were a greater proportion of families with siblings with developmental disorders in the Red group (40% compared to 29.4% in TDC), which could account for the parents' difficulty participating in PA.

Interestingly, the parents of children in the Red group ranked the importance of PA higher than those in Amber group and a little lower than TDC (50% compared to 58,8%). Although the parents in the Red group had a lower incidence of unemployment than those of TDC (10% compared to 35.3% in the Green group) and a similar rate of car ownership, their children participated much less in extra curricular PA. Thus, despite parents considering participation in PA important, the pressure of managing family life with 2 children with developmental disorders appeared to inhibit participation.

The microsystem level of analysis:

The children's characteristics results appear to agree with the concept of ESSENCE, proffered by Gillberg (2010), whereby the co-existence of disorders and the sharing of symptoms is the rule rather than the exception in developmental disorders. This was particularly evident in the Red group (those with lowest motor ability) as there was a much higher concentration of children with multiple symptoms, 80% had symptoms of ASD/ADHD/DCD and 10% had symptoms of ASD/DCD, whereas only 10% had symptoms DCD alone. Considering their ecological environment, 60% of this group had free school meals and 40% had a first-degree relative with a developmental disorder. It is evident that this group encountered multiple indices of adversity.

This compared to the Amber group, who had moderate motor impairment, and did not have any children with three symptoms of developmental conditions, 28.6% had ASD/DCD and 14.3% had ODD/DCD and 14.3% had hearing impairment/DCD. Adverse ecological environment was not as evident in this group, only 28.6% has free school means and 28.6% had a first-degree relative with a developmental disorder.

The green group had only one child with symptoms of two developmental disorders (ASD/ADHD) and only one child in each category of other developmental disorders, which showed that these developmental disorders could be present without movement impairment. The ecological context of this group is of some interest as 47.1% had free school meals (much higher than the amber group) and 29.4% had a first-degree relative with a developmental disorder (similar to the amber group), so demonstrating some indicators of ecological adversity, although not as severe as that of the Red group.

The high prevalence of children with ASD in the Red group introduced the possibility that poor comprehension might impact the study results, but the KBIT2 results did not bear this out. The Red group had 9 children with ASD but only 4 had low KBIT2 scores. The Amber group had 3 children with low KBIT2 scores, but none of them had ASD. The Green group had only one child with low KBIT2 score and he did not have ASD.

## Summary

Analysis of the ecological context of the children in the three groups revealed that the children in the Red group encountered more barriers to participation in PA than the other two groups.

However, group analysis of the children's self-perceptions appeared stable over time, as there was no difference between the groups in their self-perception according to the CSAPPA scores, although there were large inter-individual differences. These were investigated by cross case analysis.

# 7.6 The child's perspective of experiences of participation in physical activity

The themes that emerged from interview data and from the cross case analysis of cases of children who participated in PA, were present across all motor abilities and therefore all trajectories of motor progression. Bronfenbrenner's bio ecological framework helped to classify the elements at various levels: at the child level IQ and motor ability did not appear important, it was the enjoyment of active games and interest in playing that was important. At the microsystem level a positive parent attitude to PA was important, so that encouragement from parents or older sibling and car transport by a parent to venues, sufficient financial resources to permit this, someone to play with (whether parent, sibling, pet or friend). At the mesosystem level a space to play or easily accessible local amenities, well run local team sports, choice of activity and fun P.E. lessons at school that consistently emerged as facilitative.

The themes that emerged from the children who did not participate in PA also appeared not to be dependent upon motor ability or IQ at the child level, themes such as fear of being hit by a ball, lack of enjoyment of active games and preference for sedentary games. At the microsystem level themes such as lack of resources or transport, siblings with additional needs, lack of playmates and parental attitude that ranked PA of lower importance. Although the mesosystem environment was the same as that for the children who participated in PA it did not appear sufficient to facilitate participation for these children.

## Summary

Children without DCD participated in PA more than children with DCD and the children in the lowest 5<sup>th</sup> percentile of motor ability participated the least. However, irrespective of level of motor ability, presence or absence of DCD or AC, some children participated in PA, whilst others did not. From the interview data the most salient factor to facilitate participation was the presence of

encouragement from parents with adequate resources, or significant others, and a positive attitude to PA. This is encouraging and can inform intervention for children with movement difficulties.

# 7.7 Issues of consensus with the identification of motor difficulty

DCD is a complex heterogeneous condition, not always well understood or recognized by professionals. In this study teachers were asked to identify children thought to be typically developing and those with potential motor difficulties. Parents of all the children were then asked to complete questionnaires relating to movement difficulties (DCDQ '07) and all the children were assessed with MABC2 by an experienced paediatric OT. The study revealed differences between the identification of children's motor difficulties by parents, teachers and MABC2 with only 53% consensus between them. Interestingly, parents identified five children who scored above the 16th percentile on MABC2, but four of these children, when tested again over time, were subsequently found to have objectively measured movement difficulties. This only serves to illustrate that parents know their own children best and raises questions about awareness of DCD among teachers. Moreover, Barnett (2008) found that different tests identify different children because of their slightly different constructs. This further supports recommendations by Blank et al. (2012) that multiple sources are consulted for DCD diagnosis. However, Blank et al. (2012) do not recommend use of DCDQ for population based screening for DCD because of its low sensitivity and Missiuna et al. (2011) found that, even using the conservative 5% cut off for DCDQ and CSAPPA, they still found 29% false positive identification of DCD in a large population based study. See Schoemaker & Wilson (2015) for a review of the screening tools for DCD. Nevertheless, this study, although it used a school population rather than a clinic population, still targeted children identified by teachers rather than population screening. Furthermore, Rodger, Ziviani, Watter, Ozanne, Woodyatt & Springfield (2003) endorse the need for multiple sources of information to reflect the different perspectives (caregiver report, observation,

standardised test and self report) in order to address function and participation adequately in DCD.

Only seven of children in this study sample (20.6%) identified self-reported motor difficulties at the baseline, whereas 50% were identified by the MABC2. However, subsequent testing and follow up revealed ongoing motor difficulties in the children identified by the MABC2 and an additional three children subsequently rated difficulties over time (although not supported by MABC2 scores). This finding indicates that self-perception of motor ability and predilection and enjoyment of PA are dependent upon factors other than motor ability.

# Summary

DCD is not well recognized, among teachers, teachers did not identify one child in the severe motor impaired group with motor difficulties, whereas all the children with AC and motor difficulties were identified. It may point to the greater ease of identifying behavioural characteristics and calls for facilitating greater awareness of motor difficulties among teachers. This was apparently the case for parents too, as parents identified 5 children with DCDQ who scored above 16<sup>th</sup> percentile but had AC. The consensus for identifying DCD among teacher, parent & MABC2 was only 53%, which potentially leaves nearly half of children with DCD unrecognized.

# 7.8 Discussion of study variables and limitations of the study

Consideration of the choice of some of the variables and limitations in this study helped interpret some of the findings.

# Cohort size

The relatively small cohort of 34 participants dictated group comparison with large effect sizes for statistical analysis and was restricted by the number of children recruited to each group. The group in the 6-16<sup>th</sup> percentile of motor

ability did not have sufficient numbers for statistical analysis; however, supplementary data from mixed method analysis permitted another form of inter-group comparison. However, the other two groups were large enough to permit statistical analysis between the TDC and children in  $\leq 5^{\text{th}}$  percentile motor ability group and also benefitted from addition contextual data.

### Random sample

A random sample reduces the possibility of recruiting an unrepresentative sample and permits the generalisation of results. However, this generally requires large numbers of participants and more resources in terms of time and money, which is particularly true for longitudinal studies. Instead, children were investigated in depth using mixed methods incorporated in the design rather than a large random sample.

### Design

The design of this study was not randomized, because of the large numbers of children required to identify children with DCD and those with co-occurring disorders from the population. Instead, the study adopted a targeted approach where teachers, after receiving information on DCD, identified children they thought had motor difficulties and those they considered typically developing.

A limitation is that this may have resulted in recruiting fewer girls. This may be due to gender bias or lack of staff awareness of motor problems in girls, with a possibility that they were under represented in the study. However, the design enabled an ecological approach with much more detailed contextual data collection possible for each child. It also allowed more in depth testing of each child across a range of associated characteristics and also a specific level of verbal and non-verbal IQ was available for each child. Thus two confounding variables were accounted for in the study.

Furthermore, the ecological design was more in keeping with the dynamical systems theoretical approach adopted to investigate the motor progression over

time. It therefore suited the study better than a large random sample group of children with DCD with a control group, as it facilitated a different type of analysis, encompassing both the child and his or her motor progression in context. Therefore, although the sample size was not sufficiently large to be able to generalize the statistical findings, the additional contextual information supplemented the findings of the main patterns of motor progression and facilitated theoretical implications to be drawn from the results.

### Recruitment of participants

Recruiting a sample of children from mainstream schools was challenging, as schools are busy and have the pressure of following a prescriptive curriculum and external assessments. Physical space not in use and a slot in the school timetable to assess children proved difficult in secondary schools and resulted in fewer secondary children recruited to the study.

#### DSM5 criteria for DCD

The DSM5 criterion D requires ruling out other neurological conditions for a diagnosis of DCD. Zwicker & Michelson (2017) advocate examination by a paediatrician to ensure this, which was not possible in this study and so is a potential limitation. However, obtaining an early developmental and family history for each child helped to mitigate this.

#### Sample characteristics

There is a potential for recruitment bias as the majority of the children in the sample were recruited from one school in the UK, with an ASD unit attached. There was also an increased risk of a higher rate of co-occurrence of ASD and other developmental disorders. A comparative sample of 7-14 year old children randomly selected from a different area would support generalization of these findings to a wider population of children with DCD.

# Attrition

The attrition rate over the two years of the project was 14.7% and led to some missing data at re-testing, but 85.3% of the sample was available for all the three data points. Missing data was not imputed to estimate missing scores, as experience and the other results showed such variability of performance. In addition, the attrition occurred across each of the three groups of motor ability, so minimising impact on the results.

#### IQ

The range of IQ score in this study was quite large and included three children with scores below 70, which has been the usual cut-off score for previous studies. However, DSM5 criteria for DCD diagnosis does not stipulate an IQ threshold, only that the motor difficulties must be in excess of those attributed to any learning difficulty. The group with the lowest motor ability also had the lowest mean IQ score and included one child with IQ of 55, another with IQ of 58 and another with IQ of 62, which brought the mean score down for this group, but all had motor problems in excess of those expected. However, even when these scores were removed for the group, the remainder of the group still had a lower mean IQ than either the group with moderate motor difficulties or the group of TDC. Several other studies have also found significantly lower IQ scores in groups of children with severe motor difficulties, consistent with the findings in this study.

The finding that these children did not show sufficient motor progression over the two-year period to change groups calls into question the role of IQ and motor learning. Previous literature indicated that IQ might have an impact on progression of motor performance. For example Green et al. (2008) found that the children with DCD with higher verbal IQ responded better to intervention and progressed better over time. Furthermore, studies of other children with neurodevelopmental disabilities (Snowling, 2008) have found that children with higher IQ tend to find ways to circumvent their difficulties. We also know that there is a higher incidence of motor difficulties in children with lower IQ (Keogh & Sugden, 1985). Taken together with the findings from this study that the children with the lowest motor ability also had the lowest IQ may permit a different perspective.

If we interpret this from a dynamical systems theoretical stance and adopt an embodied cognition viewpoint we could suggest that the poor motor ability hinders cognitive development. Adolph & Kretch (2015) suggested that cognition builds on the foundation of perceptual knowledge. They quote Eleanor Gibson (1997) "Flexibility improves with learning and development. Expansion of the repertoire of available actions provides more opportunities for confronting Moreover, experiencing new environmental conditions novel challenges. provides opportunities to generalize old skills to new settings and to develop new solutions on the fly." Thus, we are presented with a chicken and egg situation, because poor motor control limits exploration and limited experience of exploration limits opportunity for learning. Another interpretation could be that the children with lowest motor ability have lower IQ and a higher prevalence of co-occurring neurodevelopmental conditions because they have atypical brain development and less capacity for learning. However, this does not explain why some children, at the individual level, with severe DCD had both average IQ and co-occurring conditions. However, the concepts of IQ and of measuring it have long been contentious.

Obtaining consent/assent for children with low IQ and communication difficulties Care was taken in this study to ensure that all the children understood the study information and consent procedure by help from a supportive adult, either a teaching assistant or parent. Future studies could use pictorial information to overcome comprehension difficulties for children or parents with literacy difficulties.

#### Associated characteristics

The extent of the role of co-occurring conditions and their AC on heterogeneity and motor progression in DCD is not fully understood. Visser (2003) advocated investigation of co-occurring conditions as a line of inquiry for heterogeneity, because the investigation of subtypes for DCD had proved inconclusive. The group with the lowest motor ability in this study also had the greatest prevalence of associated characteristics, a finding supported by other studies (Shoemaker et al., 2013; Kaplan et al., 2006) and so it is difficult to ascertain which had the greatest influence. A comparative population sample investigating children with pure DCD, ADHD, ASD and those with DCD with cooccurring conditions would be able to follow motor progression and investigate this.

A limitation to this study is that the group with the lowest motor ability had 90% of children with co-occurring ASD. The limited motor variability and lack of motor progress the group demonstrated over time could be attributed to ASD rather than severe DCD. However, a child in this group without ASD also displayed the same poor motor trajectory, suggesting that DCD was the culprit rather than ASD. This was also borne out by results from children who had ASD in the other motor groups, which had different, more favourable motor trajectories. Thus it is possible surmise that the high occurrence of ASD in this sample could be viewed as a strength of the study. However, this would need to be verified in larger studies with more children with severe DCD and no ASD.

Another potential limitation to this study is that DCD is known to co-occur with many different conditions, such as anxiety, Tourette's, dyslexia and dyscalculia, which were too numerous and not screened for in this study. There is a possibility that they could impact on the child's participation in extra curricular PA, but less likely to impact on the motor results.

## Over time

Methodological issues have made comparison of studies difficult and so not enough is known about how the children with DCD progress over time. Many studies used different motor cut-off points to identify and categorize the children

and others did not clarified how DSM criteria were met. Studies also used different methods to compare the children, for example Green at al. (2008) chose to categories according to perceptual deficits and others have used presence of co-occurring disorders. The length of follow up also varies substantially, from a year and a half to over 15 years, and the number and frequency of data points also varies considerably. The choice of method of analysis is widely different across studies and linear and non-liner methods have different emphasis and are likely to lead to different conclusions. Group comparison using GLM versus individual data can also have a different emphasis, as this study has shown. Even how change in motor ability is determined and measured differs, for example, the use of different motor tests (MABC, MABC2, MAND, Bruininks), and even within the same test the use of different levels of significant change. For example, LDD (Green et al, 2008) or MID (Wuang et al., 2012) making comparison across studies difficult. The theory underpinning the study can also make a difference in how the data is interpreted. For example, intra individual motor variability can be viewed either as 'noise' or as important data about the motor trajectory. We have also seen that development is a bidirectional process between the child and the environment and so the context in which the child develops over time becomes an important part of the analysis.

A potential limitation of this study is the length of follow up (2 academic years) which gave a brief window on child motor development. Future studies could increase the follow up time to obtain a more comprehensive picture of the motor stability and change.

### Age range

This study chose participants 7-14 years old at the start of the study. A limitation is that comparison with a narrower age band would ensure comparison of similar motor tasks and types of participation in PA. It would also remove any motor test ceiling effects and use only one age band in the motor test with exactly the same test components. The effects of maturation of the nervous system is more apparent in older children and leads to better motor

skills; however, the older children did not improve any more than younger children. Moreover, since MABC2 is a standardised test, age would not impact the motor results, but would be more likely to impact participation in PA.

## Family context & attitude

Bronfenbrenner's bio ecological model provided a useful framework with which to explore the child's context. The results showed that there was no difference in SES between the groups and all had a similar physical environment, yet very different motor progression. However, on the individual level, family attitude towards PA and the level of encouragement and support made a difference to the level of participation in PA of the child, irrespective of their level of motor ability. The role of culture is also important. Adolph & Hoch (2019) propose that motor development is encultured, social interactions are the impetus and caregiver support may constrain motor behaviour by structuring the physical environment in which motor skills develop. Cultural norms in turn influence social interactions, child rearing and home environments provide the backdrop for motor skill acquisition. This may account for the very different experiences of the children, despite their motor ability. It may also account for the under diagnosis of girls with DCD, as girls are not expected to engage as actively in physical play as boys. This may be a possible limitation in this study as cultural gender influences expected roles for girls, which could influence their self-perception in tools such as the CSAPPA. This could explain the selfperceptions of the girls in the moderate group in this study, who improved their motor performance but not their self-perceptions of their adequacy, enjoyment and predilection to take part in PA over time.

# 7.9 Theoretical and clinical implications from motor change over time

The findings from this study indicate that the children with DCD in the lowest 5<sup>th</sup> percentile of motor ability represent a group with distinct characteristics, which is supported by other recent studies (Schoemaker et al., 2013; Flapper &

Schoemaker, 2013). As a result they represent the group at the greatest risk of multiple adversities, because they are more likely to have AC and lower IQ (Gillberg, 2010), which puts greater strain on the families. Moreover, importantly, the findings from this study revealed that the motor characteristics for this group persisted over time. However, this was in contrast to the findings of Green et al. (2008) who reported improvement in motor performance for this group. However, that study differed in two important ways to the present study. First it was an intervention study, so the change noted could be attributed to the intervention that the children received and second it used a different method for measuring motor change. Motor change was acknowledged as ≥4 MABC total score points, (referred to as least detectable difference, LDD) as opposed to the present study, which used ≥10 MABC2 total score points and the MABC2 traffic light categories to indicate change. It is therefore possible that the improvement reported in that study might not register as change in the present study.

Moreover, previous longitudinal studies that categorized groups by initial motor severity have also found that children with DCD in the lowest 5<sup>th</sup> percentile of motor ability have relatively stable poor motor performance and ongoing motor difficulties over time. For example, Pless et al. (2002) and Cantell et al. (2003) both used DSM IV criteria to identify DCD and determined motor change as a change in MABC category, found similar results, that those with the poorest motor performance had continued difficulties over time and support the findings of this study.

However, unlike some previous longitudinal studies, the lack of improvement in motor performance in the present study could take account of context, including SES, presence of co-occurring characteristics and children's IQ as well as comparison with a group of typically developing peers. This therefore adds further support that this group differs both from typically developing children and from those with milder motor impairment over time, by confirming that the difference could not be attributed to IQ or SES. This poses questions about the

cause of the underlying difference and why the groups presented such different intra-individual variability over time. How should we interpret variability?

Viewing motor development from a dynamical systems perspective may permit a plausible explanation. The children with DCD with the lowest 5<sup>th</sup> percentile of motor ability demonstrated a lack of variability in their motor performance over time. Their poor balance and motor coordination, not conducive to allow them to unfreeze sufficient degrees of freedom, made it harder to adapt to new tasks or environments. It is possible that this in turn inhibited their active engagement with sufficient motor activities for them to learn new motor skills or accumulate the benefits from enhanced active participation. Indeed, this group had the lowest reported rates of participation in PA. The restricted movement repertoire further restricted new learning opportunities in motor and possibly social and other domains.

Paradoxically, researchers usually consider decrease in intra individual variability to be a sign of skilled performance and control as decrease in variability usually accompanies more consistent, accurate and economical performance. Indeed, the MABC2 results from the children who were without motor impairment were expected to reflect this. However, Adolph et al. (2014) warned that the endpoint of development is not elimination of variability, as intra individual variability only represents error for particular skills measured in particular ways. Adolph et al. (2014) cautioned that the paradigm in which basic motor skills are studied, in part, favours this conclusion. This is because the environment, the task constraints and participant's bodies are held constant and so a reduction in variability is presumed to be important. However, they argue that the role of variability is not so simple. This may well have a bearing on the results of motor testing in the present study because, although the tasks and environment remained constant, all the children's bodies grew and their opportunities for motor experience changed over the two-year period. This may account for the observed variability recorded across all children. However, the extent and nature of the variability differed between groups.

The presence of consistently low motor scores of the group of children who score in the lowest 5<sup>th</sup> percentile on the MABC2 indicates a lack of the appropriate strategies to gain from experience of every day activities and therefore contributes failure to adapt and develop new strategies. This may result in a limited repertoire and the children are thus less likely to tackle a real time problem with a satisfactory solution, which in turn restricts movement experience. In this sense the decrease in variability is seen as unhealthy or atypical development. Fig. 7.1 is a diagrammatic representation of a cycle of motor learning through experience and opportunities for practice in a child with typical motor development.

As previously mentioned, Adolph et al. (2014) explained that an increase in intra individual variability serves as an exploratory function, because sufficient experience in a variety of settings allows children to adapt movements simultaneously to changes in the body and the environment. In the course of coping with a task new different strategies arise and present more ways to tackle a problem, which increases the probability of selecting an appropriate solution. This opportunity is severely restricted in the children with DCD in the lowest 5<sup>th</sup> percentile of motor ability. Adolph et al. (2014) suggested that opportunities for learning variable movements in action create a variety of strategies, which in turn generate information about the self, the environment and, importantly, relations between them. Therefore, children with DCD in the lowest 5<sup>th</sup> percentile of motor ability miss out on many learning opportunities and are further disadvantaged by fewer opportunities to develop strategies and thereby develop an unhealthy cycle for motor proficiency.



Figure 7-1 Sufficient experience for motor learning for typically developing children

Schmidt & Wrisberg (2000, p241) sum this up by quoting Bernstein (1967, p134) "The process of practice towards the achievement of new motor habits essentially consists in the gradual success of a search for optimal motor solutions to the appropriate problems. Because of this, practice, when properly undertaken, does not consist of repeating the means of a solution to a motor problem time after time, but in the process of solving this problem again and again"

Children in the lowest 5<sup>th</sup> percentile appear to demonstrate that they are unable to benefit from normal practice, as they fail to learn the problem solving necessary to develop appropriate movement strategies and merely repeat the same solution time after time leading to lack of variability (stable) poor motor performance. Schoemaker and Smits-Engelsman (2015) argue that this is the case for DCD. They explain that when children with DCD were tested after participating in either intense play versus specific neurodevelopmental task training (NTT) with therapists, children with DCD improved significantly more and had more generalized learning after specific training with NTT. They concluded that more effective motor learning for children with DCD was established with NTT, where the children had been provided with explicit instruction and feedback in task orientated intervention. However, more research is required, as only one study has supported this at present.

Hadders-Algra (2010) advocated investigation of variability of motor performance to help discriminate children with typical from those with atypical motor development. She suggested that human development is characterized by the development of adaptive variability. Atypical children show limited variation and an impaired variability to task specific requirements, and therefore require much more practice to learn motor tasks (Hadders-Algra, 2010). Children with DCD in the lowest 5<sup>th</sup> percentile appear to have this impaired variability that corresponds to atypical development. Moreover, the higher prevalence of co-occurring characteristics and of first-degree relatives with developmental conditions in the 5<sup>th</sup> percentile group of children in this study appears to indicate they may have a possible genetic difference prompting a gene-environment interaction expressing as atypical motor development. This could infer a different underlying aetiology to the motor difficulties encountered by this group compared to the children with moderate motor difficulties. Future research may find this a fruitful line of inquiry. However, due to the plasticity of the CNS this does not necessarily infer a one-way developmental influence from underlying biological substrates, but could also include the influence of the environmental context.

Figure 7-2 Morton's model incorporating study findings (adapted from Sugden & Wade, 2013)

#### **Biological**

CNS, family genetics, TDC/DCD/DCD plus AC, IQ, level of motor impairment



#### Constraints

Organismic, environmental, task (Newell, 1986) Degrees of freedom, coordination structures, memory, attention, perception, enjoyment Environmental: family resources, siblings, physical/social/cultural environment, opportunities for practice & active movement participation



#### **Behavioural**

Self care skills, recreational skills, fine and gross motor skills, classroom skills Participation in PA, team member, social participation, confidence, selfperception, motivation to engage in PA, rate of motor learning

Sugden & Wade (2013) offered a modified Morton model to illustrate this in a dynamic system. This offers the potential to positively influence development by intervening and adapting motor tasks and the environment in order to permit a child's positive experience of active motor engagement and thereby promote learning opportunities otherwise denied to them. Motor scores alone do not provide sufficient information about motor development and the acquisition of motor skills in DCD, indeed Albaret & de Castelnau (2007) suggested that details of a child's context and impact on everyday life are also required. This study used Bronfenbrenner's bio ecological framework to provide another useful lens through which to view some of these contextual influences and examine the results in light of these. This lens was particularly helpful when considering some of the factors common to those children who were able to successfully

participate in physical activity and a further modification of Morton's model is shown in Fig. 7.2.

In contrast the children with moderate motor difficulties (6-16<sup>th</sup> percentile) all appeared to resolve their motor difficulties over time. Interestingly, Green et al. (2008) also found that many of the children identified with moderate motor difficulties in their study resolved without intervention over time. We can infer that they were able to adapt and improve their motor repertoire and learn from adapting to changes in their bodies, task or environment to widen their repertoire so that they became indistinguishable from their typically developing peers over time. However, the large degree of variability the group exhibited in this present study may indicate a lack of stability in their performance. For example, on the 4th data point a child's motor performance had reverted back to the 16<sup>th</sup> percentile perhaps indicating that the adaptive strategies had not become automatic so were not 'relatively permanent'. This could be tested with studies following children over more data points and longer periods. Previous studies (Pless et al., 2002; Cantell et al., 2003) have also found that this group of children can have variable results, with some improving so that they are indistinguishable from their typically developing peers, whilst others deteriorate over time. Even intervention studies have found a high level of variability. For example, Sugden and Chambers (2006) found that, despite children with DCD improving motor performance following intervention, they reverted back to low levels of performance once it had ceased and Green et al. (2008) found that some children with DCD had worse motor performance following intervention. So what can these observed differences in variability tells us about DCD?

Variability is inherent in biological systems as it ensures survival by having the flexibility to adapt to changes in order to optimize success, whilst maintaining optimal stability and performance (Smith & Wade, 2015). Furthermore, reduction in variability is seen as tantamount to a loss of flexibility to respond to changes in the demand of the environment (Smith & Wade, 2015). The lack of motor progress seen in the children with DCD in the lowest 5<sup>th</sup> percentile appears to relate to this lack of adaptability and flexibility in motor response.

They did not exhibit variability in performance, but maintained their low scores over time with impaired function.

On the other hand, the children with moderate motor impairment, who started in the 6-16<sup>th</sup> percentile of motor ability, but improved their performance over time to achieve typical motor performance exhibited wide variability. They showed that they could adapt and develop better motor skills, but that they were not necessarily stable, as displayed by the score reverting back to a lower motor category over more time. Here linear analysis with GLM did not tell full picture as it removed the temporal organization and masked the pattern. Smith & Wade (2015) assert that linear measures limit the explanation of performance variability, as they contain no information on the time-evolving nature of performance variability.

The majority of children who started with typical motor performance remained in this category over time, despite a range of patterns of progression and decrements, and maintained the stability of functional performance. The variability and inter-individual differences that they displayed achieved success and must have achieved an optimal level of variability and adaptability to maintain this level of performance over time. It is of interest that two of the three children who moved into the impaired range of motor ability over time also had additional problems of ACs and therefore could not be counted as TDC and the third had an unknown early history. We can therefore assume that TDC were able to manage optimal stability and performance, as their variability equated with flexibility, for managing constraints over time. Smith & Wade (2015) suggest that skilled performance demonstrates flexibility and adaptability, which equates with stability, and call this "good variability" because it is essential for consistent control. They explain that no two instances of a skilled action are identical because they are capable of managing changing environmental contexts.

Therefore by using both linear and non-linear methods for analysis in this study a more detailed picture could be compiled about the nature of the differences in change over time. The linear methods demonstrated group differences at base line and over time between children with different motor ability. However, it was the non-linear analysis that was able to demonstrate the importance of inter and intra-individual differences and point to some possible contextual influencing factors in progression and participation for children with DCD.

# 7.10 Conclusion

This prospective longitudinal study investigated the nature of DCD by examining the characteristics of children with and without DCD by classification of their motor performance ability. Careful identification of the children with DCD with DSM5 criteria ensured that the focus was on DCD. Furthermore, distinguishing between the children with moderate motor impairment (6-16<sup>th</sup> percentile MABC2), severe motor impairment ( $\leq$ 5<sup>th</sup> percentile MABC2) and TDC ( $\geq$ 25<sup>th</sup> percentile MABC2) permitted separate investigation of characteristics and their motor progression.

The first important finding indicated that the children in the lowest 5<sup>th</sup> percentile of motor ability were a group with distinct characteristics. Group comparison revealed potential biological differences, as the children with severe motor impairment had characteristics distinctive from the other groups. They not only had poor motor performance across all domains, but also had lower mean IQ scores than the children with moderate motor impairment (6-16<sup>th</sup> percentile) and TDC ( $\geq 25^{th}$  percentile). In addition, they were more likely to have two or more associated characteristics of other developmental disorders, particularly ASD and ADHD and a first degree relative with a developmental disorder. This indicated that they were at more serious disadvantage to start with.

To my knowledge this is the first study that investigated whether the motor characteristics of these different groups were stable or changed over time that also screened for associated conditions and IQ. The second important finding was that the children in the lowest 5<sup>th</sup> percentile did not change motor performance ability over time. Group comparison with GLM identified that the groups progressed differently over time. However, individual analysis allowed investigation of inter and intra group differences and revealed that children with severe motor impairment had stable motor progression and limited variability. They remained stable over two academic years, whereas the groups with moderate impairment and TDC showed more variation.

The study used an ecological design and was therefore able to investigate the children's context. The third finding was that despite having similar opportunities at school, physical environment, socio-economic status and self-perceptions about their physical ability to the children in the other two groups, the children in the lowest 5<sup>th</sup> percentile did not improve their motor performance over time. Some of this group even actively participated in sports and PA for leisure, yet still did not improve motor performance and remained in the lowest category over time. Importantly, the group with moderate impairment all improved over time to the level of typical motor performance in the same time period, whereas the severe group did not change.

The motor progression was viewed from a dynamical systems approach and this study highlighted the role of 'good' and 'bad' variability in motor development. This is emerging in the literature as an indicator for the presence or absence of atypical development. For example, children who can vary their motor performance and adapt to environmental constraints develop better motor skills, whereas lack of variability reduces the available responses repertoire and successful performance is less likely. The children in the lowest 5<sup>th</sup> percentile demonstrated lack of variability and lack of motor skill mastery, despite similar school experience and opportunity. This indicated that experience alone was not sufficient to improve their motor performance. Even at the individual level, the children who participated in extra curricular PA and had high self perceived motor competence and strong parental encouragement,

did not improve sufficiently to move out of the lowest 5<sup>th</sup> percentile. They were unable to adapt and learn by experience alone, indicating atypical motor development and suggest that specific training is required if they are to improve motor performance and generalize the learning. There is emerging literature supporting this view (Schoemaker & Smits-Engelsman, 2015, Preston et al., 2016) but more research is required to test this hypothesis.

On the other hand, the group with motor ability at 6-16<sup>th</sup> percentile all improved over the same time period without intervention, but demonstrated a high level of variability in motor performance over time. All improved and changed motor category on MABC2 by the third data point. However, some children that were tested a fourth time had subsequently deteriorated. This indicated instability and showed that the motor learning had not reached the 'relatively permanent' stage. The inference drawn from this is that motor learning took place to improve motor performance, but that it was not stable. It also shows that with the same opportunities at school, physical environment, socio-economic status and self-perceptions about their physical ability, they improved their motor performance over time. This indicated that experience and active engagement in PA was sufficient to improve motor performance over time without specific training. However, a question still remains about their retention of these skills long term and should be studied further.

The typically developing children, those  $\geq 25^{\text{th}}$  percentile of motor ability, also demonstrated variability in motor performance over time, but of less magnitude than the moderately motor impaired group. The majority remained in the typical motor development range over time. With the same opportunities, experience at school and physical environment they maintained typical levels of motor performance, indicating that these alone were sufficient for the acquisition and retention of age appropriate motor performance over time.

Therefore a difference in motor learning and skill acquisition between the children in the lowest group of motor ability and groups of children with different

levels of motor difficulty is indicated. The role of practice and experience in motor learning may well be different for children with the lowest 5<sup>th</sup> percentile of motor ability and that this warrants further investigation. It also has implications for policy, as the children with motor ability in the lowest 5<sup>th</sup> percentile will require careful screening for other developmental disorders and specific intervention if they are not going to be further disadvantaged.

A third important finding was that the children's self perception of their adequacy, predilection and enjoyment of PA were not dependent upon their motor performance ability and was unexpected. There was no difference between the groups in self-perception of adequacy, enjoyment and predilection to participate in PA, yet in general, the children with motor impairments participated less in PA. However, analysis at individual level revealed important differences for participation in extra curricular PA for the children with motor impairment. The findings that some children with both moderate and severe motor impairment enjoyed and participated in PA indicated a positive signal for future intervention programmes.

A fourth finding was that the children in the lowest 5<sup>th</sup> percentile of motor ability often faced additional barriers to participation in extra curricular PA, such as a sibling or parent with a developmental disorder, family financial pressures and lack of transport. The Bronfenbrenner framework facilitated a more in depth analysis of ecological reasons behind participation and non-participation in PA for children both with and without motor difficulties. The ecological study design permitted analysis both at the group and individual level, which facilitated a deeper understanding of the dynamic interaction between children and their context and gave insight into some influences in their development. This was important, as a dynamical systems approach to motor development requires greater understanding of these dynamic interactions. The findings have important policy implications, as additional parental support through inclusive schools or community services can help alleviate some of the barriers to participation and avoid the secondary consequences associated with nonparticipation in PA.

The fifth finding came from data from interviews with the children across all motor abilities. This gave further insight, from their perspective, and highlighted what they found to be supportive environments to enable participation in PA. Themes that emerged were that they enjoyed PA, had access to local amenities, had supportive parents who valued PA, their parents participated in PA themselves and had transport. This also has policy implications for community services, parents and teachers to help improve participation in PA for children. This is important, since active self-generated movement is crucial for motor development from a dynamical systems perspective. Therefore, an inclusive environment that facilitates this for children with the most severe motor impairment is imperative, given their additional problems.

Finally, the broader question of participation for children with DCD should be considered. This study demonstrated that children with DCD clearly have impairment, but the findings also indicated that participation in PA was possible, even for children with severe motor performance difficulties. However, they also crucially indicated that the children's ecological context was a vital part of their successful participation. The ICF emphasises participation as the desired outcome of intervention and so challenges both researchers and clinicians to consider appropriate outcome measures and study design that capture this important issue. This will help the future design of inclusive and supportive environments for children with DCD and other motor impairments.

### **Future directions**

The study should be repeated with a larger cohort of children to recruit more children in each category, especially those with pure DCD, to investigate whether the characteristics found in the lowest 5<sup>th</sup> percentile are consistent with the results from this study.

Knowledge transfer of information of the positive impact of supportive environments to facilitate participation in PA for children with motor difficulties should be disseminated to schools, parents, therapists and community services. The outcome of this should then be evaluated.

A future study should investigate the difference in motor learning and generalisation of motor skills between children with severe DCD (lowest 5<sup>th</sup> percentile) and those with moderate DCD (6-16<sup>th</sup> percentile) following a cross over intervention trial using a specific training method.

# REFERENCES

- Adams, J.A. (1971) A closed loop theory of motor learning. Journal of Motor Behaviour, 3, 111-149.
- Adolph, K.E. (2018) chapter Motor development in Bornstein, M. (Ed) (2018) Sage Encyclopedia of Lifespan Human Development, Thousand Oaks: Sage Publishing Inc. DOI: http://dx.doi.org/10.4135/9781506307633.n541
- Adolph, K.E., Weise, I. & Marin, L. (2003) Motor development in Encyclopedia of Cognitive Science, London: Macmillan.
- Adolph, K.E., Cole, W.G., Vereijken, B. (2014) Intraindividual variability in the development of motor skills in childhood, chapter 4 in Diehl, M., Hooker, K. & Sliwinski, M. (Eds) (2014) Handbook of Intra-individual Variability Across the lifespan. New York: Routledge/Taylor & Francis Group. Preprint formatted by authors.
- Adolph, K. & Kretch, K. (2015) Gibson's theory of perceptual learning in International Encyclopedia of the Social and Behavioural Sciences: 2<sup>nd</sup> edition, pp127-134. Elsevier inc.
- Adolph, K. & Hoch, T. (2019) Motor development: embodied, embedded, encultured and enabling. Annual Review of Psychology 2019, 70:26.1-26.24.

- Ahonen, T. (1990) Developmental coordination disorders in children: A developmental neuropsychological follow-up study. Jyvaskyla studies in education, psychology and social research, 78. Jyvaskyla, Finland: University of Jyvaskyla.
- Albaret, J.M. & De Castelnau, P. (2007) Diagnostic procedures for DCD in Geuze, R.H. (2007) Ed. Developmental coordination disorder: a review of current approaches. Marseille: Solal.
- Alderson, P., Morrow, V. (2011) The ethics of research with children and young people: a practical handbook, 2<sup>nd</sup> edition. London: Sage Publications.
- American Psychiatric Association (1987) DSM-III Diagnostic and Statistical Manual of Mental Disorders, third edition. Washington, DC, American Psychiatric Association.
- American Psychiatric Association (1994) DSM-IV Diagnostic and Statistical Manual of Mental Disorders, fourth edition. Washington, DC, American Psychiatric Association.
- American Psychiatric Association (1995) Diagnostic and Statistical Manual of Mental Disorders, fourth edition, International version. Washington, DC, American Psychiatric Association.
- American Psychiatric Association (2000) DSM-IV-TR Diagnostic and Statistical Manual of Mental Disorders, fourth edition. Washington, DC, American Psychiatric Association.

- American Psychiatric Association (APA) (2013) DSM5 Diagnostic and Statistical Manual of Mental Disorders, fifth edition. Arlington, VA: American Psychiatric Association.
- Anably, D., Law, M., Coster, W., Bedell, G., Khetani, M.Avery, L., & Teplicky, R. (2014) The mediating role of the environment in explaining participation of children and youth with and without disabilities across home, school and community. Archives of Physical Medicine and Rehabilitation, 2014; 95:908-17.
- Anderson, G. (1998) Case study, chapter 15 in Fundamentals of Educational Research, 2<sup>nd</sup> edition, London: Falmer Press.
- Arksey, H. & Knight, P. (1999) Interviewing for Social Scientists, London: Sage Publications Inc.
- Babic, M., Morgan, P., Plotnikoff, R. et al., (2014) physical activity and physical self-concept in youth: systematic review and meta-analysis. Sports Med 2014:1-13
- Baird, G. (2013) Classification of diseases and neurodevelopmental disorders: the challenge for DSM 5 and ICD 11, Developmental Medicine & Child Neurology (2013) Mar, 55(3): 200-1, doi: 10.1111/dmcn.12087.
- Bandura, A. (1982) Self-efficacy mechanisms in human agency. American Psychologist, 37 (2) 122-147.
- Barnett, L., Lai, S., Veldman, S., Hardy, L., Cliff, D., Morgan, P., Zask, A.,Lubans, D., Shultz, S., Ridgers, N., Rush, E., Brown, H. & Oakely, A.(2016) Correlates of gross motor competence in children and adolescents:

A systematic review and meta-analysis. Sports Med (2016) 46:1663-1688

- Barnett, A.L. (2008) Motor assessment in DCD: from identification to intervention, International Journal of Disability, Development and Education, Vol.55, No.2, 113-129.
- Barnett, A.L. (2011) Motor impairment in extremely preterm or low birthweight children. Developmental Medicine and Child Neurology 2011; 53: 9-10. DOI: 10.1111/j.1469-8749.2010.03779.x
- Barnett, A.L. (2014) Is there a "movement thermometer" for developmental coordination disorder? Curr Dev Disord Rep (2014) 1:132-139 DOI 10.1007/s40474-014-0011-9
- Barnett, A. & editorial panel (2015) DCD research: How are we moving along? Editorial/Human Movement Science 42 (2015) 289-292 http://dx.doi.org/10.1016/j.humov.2015.06.012
- Baron-Cohen, S., Scott, F., Allison, C., Williams, J., Bolton, P., Matthews, F & Brayne, C. (2009) Prevalence of autistic conditions: UK school-based population study. British Journal of Psychiatry 2009 Jun; 194 (6): 500-9.
- Batey, C.A., Missiuna, C.A., Timmons, B.W., Hay, J.A., Faught, B.E. & Cairney, J. (2014) Self-efficacy toward physical activity behaviour of children with and without developmental coordination disorder. Human Movement Science, 36 (2014) 258-271.
- Bedell, G., Coster, W., Law, M., Liljenquist, K., Ying-Chia Kao, B., Teplicky, R., Anaby, D. & Alunkal Khetani, M. (2013) Community participation, supports and barriers of school-age children with and without disabilities. Archives

of Physical Medicine and Rehabilitation, 2013; 94: 315-23.

- Bernstein, N. (1967) The coordination and regulation of movements. London: Pergamon.
- Bishop, D. V.M. (2002) Motor immaturity and specific speech and language impairment: evidence for a common genetic basis. Am. J. Med. Genet. 114:56-63 cited in Viholainen, H., Ahonen, T., Cantell, M., Lyytinen, P., Lyytinen, H. (2002) Development of early motor skills and language in children at risk from familial dyslexia. Developmental Medicine and Child Neurology, 2002, 44, 761-769.
- Bishop, D.V.M. (2003) The Children's Communication Checklist second edition CCC2 Manual, London: Harcourt Assessment, The Psychological Corporation.
- Blank, R. (2012) Information for parents and teachers on the European Academy for Childhood Disability (EACD) recommendations on Developmental Coordination Disorder. Developmental Medicine and child Neurology, 2012, 54:1-2.
- Blank, R., Smits-Engelsman, B., Polatajko, H., Wilson, P. (2012) European Academy for Childhood Disability (EACD): Recommendations on the definition, diagnosis and intervention of developmental coordination disorder (long version). Developmental Medicine and Child Neurology, 2012, 54: 54-93.
- Bo, J., Bastain, A.J., Kagerer, F.A., Contreras-Vidal, J.L., Clark, J.E. (2008) Temporal variability in continuous versus discontinuous drawing for children with developmental coordination disorder. Neurosci Lett. 2008, 6; 431 (3): 215-220.

- Bronfenbrenner, U. (1977) Toward an experimental ecology of human development. American Psychologist, 32, 513-531.
- Bronfenbrenner, U. (1979) The ecology of human development: Experiments by nature and design, Massachusetts: Harvard University Press.
- Brown, T. & Lalor, A. (2009) The Movement Assessment Battery for Children 2<sup>nd</sup> edition (MABC-2): Review and critique. Physical & Occupational Therapy in Paediatrics, 29:1, 86-103.
- Bruininks, R.H. (1978) Bruininks –Oseretski Test of Motor Proficiency. Circlepines, MN: American Guidance Service.
- Bryman, A. (2010) University of Leicester, School of Management Summer school, You tube lecture accessed on 22/2/12.
- Burgess, T. (2001) A Guide to the Design of Questionnaires: a general introduction to the design of questionnaires for survey research. University of Leeds
- Burton, A.W. & Miller, D.E. (1998) Movement skill assessment, Champagne, II: Human Kinetics.
- Bussing, R., Fernandez, M., Harwood, M., Hou, W., Wilson Garvin, C., Eyberg,
  S. (2008) Parent and teacher SNAP IV ratings of ADHD symptoms: psychometric properties and normative ratings for a school district sample. Assessment, vol. 15, No. 3, Sept. 2008, 317-328.

- Cairney, J., Hay, J., Faught, B., Mandigo, J. & Flouris, A. (2005) DCD, selfefficacy toward physical activity, and play: does gender matter? Adapted Physical Activity Quarterly, 2005, 22, 67-82.
- Cairney, J., Hay, J., Faught, B., Wade. T., Corna, L. & Flouris, A. (2005a) DCD, generalized self-efficacy toward physical activity, and participation in organized and free play activities. Journal of Paediatrics, October 2005; 147: 515-520.
- Cairney, J., Hay, J., Faught, B., Corna, L. & Flouris, A. (2006) DCD, age, and play: A test of the divergence in activity deficit with age hypothesis. Adapted Physical Activity Quarterly, 2006, 23, 261-276.
- Cairney, J., Hay, J., Mandigo, J., Wade, T., Faught, B.E. & Flouris. A. (2007)
  DCD and reported enjoyment of physical education in children. European
  Physical Education Review, 13 (1), 81-98.
  http://dx.doi.org/10.1177/1356336x07072678
- Cairney, J., Missiuna, C., Veldhuizen, S. & Wilson, B. (2008) Evaluation of psychometric properties of the developmental coordination questionnaire for parents (DCD-Q): Results from a community based study of schoolaged children. Human Movement Science, 27 (2008) 932-940.
- Cairney, J., Schmidt, L.A., Veldhuizen, S., Kurdyak, P., Hay, J. & Faught, B. (2008) Left-handedness and developmental coordination disorder. Canadian Journal of Psychiatry, 2008, Oct; 53(10) 696-9.
- Cairney, J., Hay, J., Veldhuizen, S., Missiuna, C., Faught, B. (2009) Comparing probable case identification of developmental coordination disorder using
the short form of the Bruininks-Oseretsky Test of Motor Proficiency and the Movement ABC. Child: care, health and development, 35, 402-408.

- Cairney, J., Hay, J., Veldhuizen, S., Missiuna, C., Faught, B. (2010) Developmental coordination disorder, sex, and activity deficit over time: a longitudinal analysis of participation trajectories in children with and without coordination difficulties. Developmental Medicine and Child Neurology, 2010, 52: e67 –e72.
- Cairney, J., Rigoli, D. & Piek, J. (2013) Developmental coordination disorder and internalizing problems in children: the environmental stress hypothesis elaborated. Dev Rev. 33, 224-238. Doi:101016/j.dr.2013.07.002
- Cairney, J. (2015) DCD, physical activity, and physical health: Results from the PHAST project, chapter 3 in Cairney, J. (2015) Ed. Developmental Coordination Disorder and its consequences, Toronto: University of Toronto Press.
- Cairney, J. (2015) Ed. Developmental Coordination Disorder and its consequences, Toronto: University of Toronto Press.
- Cantell, M. (1998) Developmental coordination disorder in adolescence: Perceptual motor, academic and social outcomes of early motor delay.
   Research reports on sport & health, 112. LIKES-Research Centre for Sport & Health Science: Jyvaskyla, Finland.
- Cantell, M., Smyth, M., Ahonen, A. (1994) Clumsiness in adolescence: Educational, motor and social outcomes of motor delay detected at 5 years. Adapted Physical activity Quarterly, 1994, 11, 115-129.

- Cantell, M. & Kooistra, L. (2002) Long term outcomes of DCD, chapter 2 in Cermak, S. & Larkin, D. (2002) Developmental Coordination Disorder, New York: Delmar Thomson Learning.
- Cantell, M., Smyth, M., Ahonen, T. (2003) Two distinct pathways for developmental coordination disorder: Persistence and resolution. Human Movement Science 22 (2003) 413-431.
- Cantin, N., Polatajko, H., Thach, T. & Jaglal, S. (2007) DCD: Exploration of the cerebellar hypothesis. Human Movement Science (2007) 491-509.
- Cantin, N., Ryan, J., Polatajko, H. (2014) Impact of task difficulty and motor ability on visual-motor task performance of children with and without developmental coordination disorder. Human Movement Science, 34 (2014) 217-232. http://dx.doi.org/10.1016/j.humov.2014.02.006
- Caron, C., & Rutter, M. (1991) Comorbidity in child psychopathology: concepts, issues and research strategies. Journal of Child Psychology and Psychiatry, Vol. 32, No. 7, pp1063-1080.
- Castelnau, P., Albaret, J., Chaix, Y., Zanone, P.et al (2007) Developmental coordination disorder pertains to a deficit in perceptuo-motor synchronization independent of attentional capacities. Human Movement Science 26 (2007) 477-490.
- Cermak, S., Gubbay, S., & Larkin, D. (2002) What is DCD? Chapter 1 in Cermak, S. & Larkin, D. (2002) Developmental coordination disorder, New York: Delmar, Thomson Learning.
- Chambers, M., Sugden D.A., Sinani, C. (2005) The nature of children with Developmental Coordination Disorder, Chapter 1 in Sugden, D.A.,

Chambers, M. (2005) (eds) Children with Developmental Coordination Disorder, London: Whurr publishers Ltd.

- Cocks, N., Barton, B. & Donelly, M. (2009) self-concept of boys with DCD. Physical and Occupational Therapy in Paediatrics, 29, 6-22.
- Cohen, N., Vallence, D., Barwick, M., Im, N., Menna, R., Horodezky, N., Isaacson, L. (2000) The interface between ADHD and language impairment: An examination of language, achievement and cognitive processing. J Child Psychol. Psychiatr. Vol. 41, No.3, pp353-362.
- Cohen, J. (1988) Statistical power analysis for the behavioural sciences (2<sup>nd</sup> edition) New York: Academic Press.
- Conti-Ramsden, G., Donlan, C, Grove, J. (1992) Children with specific language impairments: curricular opportunities and school performance. British Journal of Special Education, 19, 75-80 cited in Knox, E. (2002) Educational attainments of children with specific language impairment at year 6. Child Language Teaching and Therapy, June 2002, Vol. 18, No.2, 103-124.
- Creswell, J.W. & Plano Clark, V.L. (2011) Designing and conducting mixed methods research, London: Sage publications Ltd.
- Dare, M. & Gordon, N. (1970) Clumsy children: A disorder of perception and motor organization. Dev Med & Child Neurol, 12 178-185.

- Davids, K. (2010) The constraints based approach to motor learning, chapter 1 in Renshaw, I., Davids, K. & Savelsbergh, G. (2010) Motor learning in practice, a constraints-led approach, London: Routledge.
- Davids, K., Glazier, P., Araujo, D & Bartlett, R. (2003) Movement systems as dynamical systems: The functional role of variability and its implications for sports medicine. Sports Medicine, 2003:33 (4): 245-260.
- Deutsch, K. & Newell, K. (2005) Noise, variability, and the development of children's perceptual-motor skills. Developmental Review, 25, 155-180 in Adolph, K.E., Cole, W.G., Vereijken, B. (2014) Intraindividual variability in the development of motor skills in childhood, chapter 4 in Diehl ,M., Hooker, K. & Sliwinski, M. (Eds) (2014) Handbook of Intra-individual Variability Across the lifespan. New York: Routledge/Taylor & Francis Group. Preprint formatted by authors.
- Dewey, D., Kaplan, B. (1994) Subtyping of Developmental Motor Deficits. Developmental Neuropsychology, 1994, 10 (3). 265-284.
- Dewey, D., Wilson, B.N. (2001) Developmental Coordination Disorder: what is it? Physical and Occupational Therapy in Paediatrics, Vol. 20, No. 2/3, 2001, pp5-27.
- Dewey, D., Kaplan, B.J., Crawford, S.G., & Wilson, B. N. (2002) Developmental coordination disorder: Associated problems inattention, learning and psychosocial adjustment. Human Movement Science, 21, 905-918.
- Dewey, D. & Volkovinskaia, A. (2018) Health –related quality of life and peer relationships in adolescents with developmental coordination disorder and

attention-deficit-hyperactivity disorder. Developmental Medicine & Child Neurology 2018, 1-7, DOI:1111/dmcn.13753.

- Dunford, C. (2011) Goal-orientated Group Intervention for Children with Developmental Coordination Disorder. Physical and Occupational Therapy in Paediatrics, early online: 1-13, 2011. DOI: 10.3109/01946238.2011.565864
- Dunford, C., Missiuna, C., Street, E., Silbert, J. (2005) Children's perceptions of the impact of developmental coordination disorder on activities of daily living. British Journal of Occupational Therapy, 68, (5), 207-14
- Dunlop & Myers (1997) available at <u>http://imaging.mrc-</u> cbu.cam.ac.uk/statswiki/FAQ/effectsize accessed on 25.10.16
- Edelman, G.M. (1989) Neural Darwinism. The theory of Neuronal Group Selection, Oxford: Oxford University Press.
- Eisenhardt, K.M. (1989) Building theories from case study research. Academy of Management Review, 1989, Vol.14, No. 4, 532-550.
- Elllinoudis, T., Evaggelinou, C., Kourtessis, T., Konstantinidou, Z., Venetsanou, F. & Kambas, A. (2011) Reliability and validity of age band 1 of the Movement Assessment Battery for Children – second edition. Research in Developmental Disabilities, 32 (2011), 1046-1051.

- Farhat, F., Hsairi, I., Hamza, B., Cairney, J., Mchirgui, R., Masmoudi, K., Padulo, J., Triki, C & Moalla, W. (2015) Assessment of physical fitness and exercise tolerance with DCD. Research in Developmental Disabilities, 45-46 (2015) 210-219.
- Farhat, F., Hsairi, I., Baati, H., Smits-Engelsman, B.C.M., Masmoudi, K., Mchirgui, R., Triki, C., Moalla, W. (2016) The effect of motor skills training program in the improvement of practiced and non-practiced task performance in children with developmental coordination disorder. Human Movement Science 46 (2016) 10-22 <a href="http://dx.doi.org/10.1016/j.humov.2015.12.001">http://dx.doi.org/10.1016/j.humov.2015.12.001</a>
- Ferguson, G.D., Jelsma, D., Jelsma, J. & Smits-Engelsman, B.C.M. (2013) The efficacy of two task-orientated interventions for children with Developmental Coordination Disorder: Neuromotor Task Training and Nintendo Wii Fit training. Research in Developmental Disabilities, 34 (2013) 2449-2461.
- Field, A. (2013) Discovering statistics using IBM SPSS statistics, 4<sup>th</sup> edition. London: Sage.
- Field, A. (2018) Discovering statistics using IBM SPSS statistics, 5<sup>th</sup> edition. London: Sage.
- Fisk, J.L., Rourke, B.P. (1979) Identification of subtypes of learning-disabled children at three age levels: a neuropsychological, multivariate approach. Journal of Clinical Neuropsychology, 1979, No.4, 289-310.
- Fitts, P.M. & Posner, M.I. (1967) Human performance. Belmont CA: Brooks/Cole.

- Flapper, B. & Schoemaker, M. (2013) Developmental coordination disorder in children with specific language impairment: co-morbidity and impact on quality of life. Research in Developmental Disabilities 34 (2013) 756-763.
- Fliers, E., Marieke, L., de Hoog, A., Franke, B., Farone, S., Rommelse, N., Buitelaar, J. & Nijhuis-van der Sanden, M. (2010) Actual motor performance and self-perceived motor competence in children with attention-deficit hyperactivity disorder compared with healthy siblings and peers. Journal of Developmental & Behavioural Peadiatrics, 31, 35-40.
- Foulder-Hughes, L., Cooke, R. (2003) Do mainstream schoolchildren who were born preterm have motor problems? British Journal of Occupational Therapy, January, 2003, 66 (1) 9-16.
- Fox, M., Polatajko, H. (1994) International Consensus on children with developmental coordination disorder.
- Gaines, R., Missiuna, C. (2006) Early identification: are speech/languageimpaired toddlers at increased risk for Developmental Coordination Disorder? Child: Care, Health and development, 33, 3, 325-332.
- Germano, E., Gagliano, A., Curatolo, P. (2010) Comorbidity of ADHD and Dyslexia. Developmental Neuropsychology, 35 (5) 475-493.
- Gesell, A. (1933) Maturation and the patterning of behaviour. In Murchison, C.
   (Ed.) A Handbook of child psychology (2<sup>nd</sup> rev. ed., p2090235) Worcester
   MA: Clark University Press.

- Geuze, R.H. (2018) On constraints and affordances in motor development and learning – The case of DCD. A commentary on Wade & Kazek (2017), Human Movement Science 57 (2018) 505-509.
- Geuze, R., Jongmans, M., Schoemaker, M. et al (2001) Clinical and research diagnostic criteria for Developmental Coordination Disorder: a review and discussion. Human Movement Science 20 (2001) 7-47.
- Geuze, R. (2005) Chapter 2, motor impairment in DCD and activities of daily living in Sugden, D.A. & Chambers, M.E. (Eds.) (2005) Children with Developmental Coordination Disorder, London: Whurr publishers.
- Geuze, R., Borger, H. (1993) Children who are clumsy: five years later. Adapted Physical Quarterly, 1993, 10, 10-21.
- Geuze, R. & Kalverboer, A. (1994) Tapping a rhythm: a problem of timing for children who are clumsy and dyslexic. Adapted physical Quarterly, 11:203-213.
- Gibson, E. (1997) in Adolph, K. & Kretch, K. (2015) Gibson's theory of perceptual learning in International Encyclopedia of the Social and Behavioural Sciences: 2<sup>nd</sup> edition, pp127-134. Elsevier inc.
- Gibson, J.J. (1979) The ecological approach to visual perception. Boston: Houghton –Mifflin.

- Gibson, E.J. Adolph, K. & Eppler, M. (1999) Affordances in Wilson, R.A. & Keil, F.C. (Eds) (1999) The MIT encyclopedia of cognitive sciences, Cambridge, MA.: MIT Press.
- Gillberg, C. (2003) Deficits in attention, motor control, and perception: a brief review. Arch Dis Child 2003; 88, 904-910.
- Gillberg, C. (2003) Deficits in attention, motor control, and perception: a brief review. Arch Dis Child 2003; 88, 904-910.
- Gillberg, C. (2010) The ESSENCE in child psychiatry: Early Symptomatic Syndromes Eliciting Neurodevelopmental Clinical Examinations. Research in Developmental Disabilities, 31 (6) 1543-51.
- Goodgold-Edwards, S. & Cermak, S. (1990) Integrating motor control and motor learning concepts with neuropsychological perspectives on apraxia and developmental dyspraxia. The American Journal of Occupational Therapy, 1990, Vol. 44, No.5, 431-439.
- Gorard, S. (2003) Quantitative methods in social science: the role of numbers made easy, London: Continuum.
- Goulardins, J., Rigoli, D., Licari, M., Piek, J., Hasue, R., Oosterlaan, J. & Oliveira, J. (2015) Attention deficit hyperactivity disorder and developmental coordination disorder: Two separate disorders or do they share a common aetiology. Behavioural Brain Research , 292 (2015) 484-492.

Great Britain (1998) Data Protection Act. London: Stationery Office.

- Green, D., Baird, G., Barnett, A.L., Henderson, L., Huber, J., Henderson, S.E. (2002) The nature and severity of motor impairment in Asperger's syndrome: a comparison with Specific developmental disorder of motor function. Journal of child Psychology and Psychiatry, 43: 5 (2002), pp 655-668.
- Green, D., Baird, G. (2005) DCD and overlapping conditions, chapter 5 in Sugden, D.A., Chambers, M.E. (2005) Children with Developmental Coordination Disorder. London: Whurr Publishers.
- Green, D., Baird, G. & Sugden, D.A. (2006) A pilot study of psychopathology in developmental coordination disorder. Child: Care, Health and Development, 32, 741-750.

Green, D. (2007) unpublished PhD thesis, University of Leeds

- Green, D., Chambers, M.E., Sugden, D.A. (2008) Does subtype of developmental coordination disorder count: is there a differential effect on outcome following intervention? Human Movement Science, 27 (2008) 363-382.
- Hadders-Algra, M. (2000) The Neuronal Group Selection Theory: a framework to explain variation in normal motor development. Developmental Medicine & Child Neurology, 2000, 42: 566-572
- Hadders-Algra, M. (2008) Development of postural control, chapter 3 in Hadders-Algra. M. & Carlberg, E.B. (2008) Postural control: A key issue in developmental disorders, London: MacKeith Press.

Hadders-Algra, M. (2010) Variation and variability: key words in Human Motor
Development, Physical Therapy. 2010; 90: 1823-1837.
Doi:10.2522/ptj.20100006

Hall, D. (1988) Clumsy Children. British Medical Journal, 296:375-6.

- Hands, B. & Larkin, D. (2002) Physical fitness and DCD, Chapter 11 in Cermak,S. & Larkin, D. (2002) Developmental Coordination Disorder, New York:Delmar, Thomson Learning.
- Harris, S., Mickelson, E. & Zwicker, J. (2015) Diagnosis and management of DCD, CMAJ, 187(9): 659-665. DOI:10/1503/cmaj.140994.
- Harrowell, I., Hollen, L., Lingam, R. & Emond, A. (2017) Mental health outcomes of DCD in late adolescence. Developmental Medicine and Child Neurology 2017, 59; 973-979.
- Harrowell, I., Hollen, L., Lingam, R. & Emond, A. (2018) The impact of DCD on educational achievement in secondary school. Research in Developmental Disabilities 72 (2018) 13-22.
- Harter, S. (1982) The Perceived Competence Scale for Children. Child Development, 53, 87.
- Harter, S. (1987) The determinants and meditational role of global self worth in children. In Eisenberg, N., Contemporary topics in developmental psychology (pp.219-242). New York: Wiley. In Skinner, R.A., Piek, J.P.

(2001) Psychosocial implications of poor motor coordination in children and adolescents. Human Movement Science, 20, 73-94.

- Harter, S. (1988) Developmental processes in the construction of the self, chapter 3 in Yawkey, T.D. & Johnson, J.E. (Eds) Integrative processes and socialization: early to middle childhood (pp 45-78), Hillsdale NJ: Lawrence Erlbaum Associates. Inc.
- Hay, J. (1992) Adequacy in and predilection for Physical Activity in Children. Clinical Journal of Sport Medicine, 2: 192-201.
- Hay, J., Hawes, R. & Faught, B.E. (2004) Evaluation of a screening instrument for developmental coordination disorder. Journal of Adolescent Health, 34, 308-313.
- Hellgren, L., Gillberg, C., Gillberg, I., & Enerskog, I. (1993) Children with deficits in attention, motor control, and perception (DAMP) almost grown up: General health at 16 years. Developmental Medicine and Child Neurology, 35, 881-892.
- Henderson, S.E., Barnett, A. L. (1998) The classification of specific motor coordination disorders in children: some problems to be solved. Human Movement Science 17 (1998) 449-469.
- Henderson, S.E., Sugden, D.A. (1992) Movement Assessment Battery for Children. London: The Psychological Corporation.
- Henderson, S.E., Sugden, D.A., Barnett, A.L. (2007) Movement Assessment Battery for Children 2<sup>nd</sup> edition, London: Harcourt Assessment.

- Henderson, S.E. & Geuze, R.H. (2015) Ten conferences on developmental coordination disorder (DCD): A brief commentary on 20 years of research, Physical & Occupational therapy in Paediatrics, 35 (2): 97-102, 2015.
- Hill, E.L. (1998) A dyspraxic deficit in specific language impairment and developmental coordination disorder? Evidence from hand and arm movements. Developmental Medicine and Child Neurology, 1998, 40: 388-395.
- Hill, E.L. (2001) Non-specific nature of specific language impairment: a review of the literature with regard to concomitant motor impairments. International Journal of Language and Communication Disorders, 2001, Vol. 36, No. 2, 149-171.
- Hill, E.L. (2005) Cognitive explanations of the planning and organization of movement, chapter 3 in Sugden, D.A. & Chambers, M.E. (2005) Children with developmental coordination disorder, London: Whurr Publishers.
- Hill, E. (2006) Characteristics and comorbidity, seminar 3, Sugden, D.A., Chambers, M. & Utley, A. (2006) The Leeds Consensus statement ESRC seminar series.
- Hill, E.L. (2010) Motor difficulties in specific language impairment: evidence for the lverson account? – a commentary on lverson's 'Developing language in a developing body: the relationship between motor development and language development'. Journal of Child Language, 37 (2010), 287-292.
- Hinckle, D.E., Wiersma, W. & Jurs, S.G. (1994) Applied statistics for the behavioural sciences, 3<sup>rd</sup> edition, Boston: Houghton Mifflin Company.

- Hoare, D. (1994) Subtypes of Developmental Coordination Disorder. Adapted Physical Activity Quarterly, 1994, 11, 158-169.
- Holm, I., Tveter, A.T., Smith Aulie, V. & Stuge, B. (2013) High inter-rater chance variation of the Movement Assessment Battery for Children 2, age band 2. Research in Developmental Disabilities, 34 (2013) 795-800.
- Hulme, C. & Lord, R. (1986) Review: Clumsy children a review of recent research. Child Care: Health & Development, 12, 257-269.
- Hulme, C., Snowling, M. (2009) Developmental Disorders of Language and Cognition. Chichester: Wiley-Blackwell.
- Hulme, C. & Snowling, M.J. (2009) Chapter 1, Understanding Developmental Cognitive Disorders in Developmental Disorders of Language and Cognition. Chichester: Wiley- Blackwell.
- Humphries, T.W., Wright, M., McDougall, B., & Vertes, J. (1990) The efficacy of sensory integration therapy for children with learning disability. Physical and Occupational Therapy in Pediatrics, 10, 1-17.
- Imms, C., Adair, B., Keen, D., Ullenhag, A., Rosenbaum, P. & Granlund, M. (2016) 'Participation': a systematic review of language, definitions, and constructs used in research with children with disabilities. Developmental Medicine and Child Neurology, 2016, 58: 29-38.
- Iverson, S, Ellertsen, B., Tytlandsvik, A., Nodland, M. (2005) Intervention for 6year-old children with motor coordination difficulties: parental perspectives at follow up in middle childhood. Advances in Physiotherapy, 7 (2), 67-76.

- Jarus, T., Lourie-Gelberg, Y., Engel-Yeger, B. & Bart, O. (2011) Participation patterns of school-aged children with and without DCD. Research in Developmental Disabilities, 32 (2011) 1323-1331.
- Johnson, D., Wade, M. (2009) Children at risk for developmental coordination disorder: judgment of changes in action capabilities. Developmental Medicine and Child Neurology, Vol. 51, Issue 5, pp397-403.
- Jongmans, M., Mercuri, E., Dubowitz, L., Henderson, S.E. (1998) Perceptualmotor problems and their concomitants in six-year old children born prematurely. Human Movement Science, Vol. 17, Issue 4-5 (August 1998), p629-653.
- Jongmans, M., Smits-Engelsman, B., Schoemaker, M. (2003) Consequences of comorbidity of Developmental Coordination Disorder and Learning Disabilities for severity and patter of perceptual-motor dysfunction. Journal of Learning Disabilities, Vol.36, No.6, Nov/Dec 2003, p528-537.
- Kadesjo, B. & Gillberg, C. (1998) Attention deficits and clumsiness in Swedish
  7-year –old children. Developmental Medicine Child Neurology, 40, 796-804.
- Kadesjko, B., Gillberg, C. (2001) The comorbidity of ADHD in the general population in Swedish school-age children. Journal of Child Psychology and Psychiatry, Vol. 42, No. 4, pp. 487-492.

- Kaiser, M., Albaret, J. & Cantell, M. (2015) Assessment of the participation of children with a developmental coordination disorder (DCD): A review of the questionnaires addressed to parents and/or teachers. Journal of Child and Adolescent Behaviour, 2015, Vol. 3, Iss. 5. http://dx.doi.org/10.4172/2375-4494.1000234
- Kaplan, B., Wilson, B., Dewey, D. & Crawford, S. (1998) DCD may not be a discrete disorder. Human Movement Sciences, 17 (1998) 471-490.
- Kaplan, B., Dewey, D., Crawford, S., Wilson, B. (2001) The term comorbidity is of questionable value in reference to developmental disorders: Data and theory. Journal of Learning Disabilities, Vol. 34, No. 6, Nov/Dec 2001, 555-565.
- Kaplan, B., Crawford, S., Cantell, M., Kooistra, L., Dewey, D. (2006) Comorbidity, co-occurrence, continuum: what's in a name? Child: care, health and development, 32, 6, 723-731.
- Kaufman, A.S., & Kaufman, N.L. (2004) Kaufman brief Intelligence Test second edition, San Antonio: Pearson Inc.
- Keogh, J. & Sugden, D.A (1985) Movement skill development, New York: Macmillan Publishing Co.
- King, G., Law, M., King, S., Rosenbaum, P., Kertoy, M. & Young, N. (2003) A conceptual model affecting recreation and leisure participation of children with disabilities. Physical & Occupational Therapy in Paediatrics, Vol.23 (1) 2003, 63-90.

- King, B.R., Harring, J.R., Oliveira, M.A. & Clark, J.E. (2011) Statistically characterizing intra- and inter-individual variability in children with developmental coordination disorder. Research in developmental Disabilities 32 (2011) 1388-1398.
- King, G., Law, M., King, S., Rosenbaum, P., Kertoy, M. & Young, N. (2003) A conceptual model of the factors affecting the recreation and leisure participation of children with disabilities. Physical and Occupational Therapy in Paediatrics, Vol. 23 (1) 63-90.
- Kirby, A. (2005) Overlapping conditions- overlapping management: services for individuals with Developmental Coordination Disorder, chapter 13 in Sugden, D.A. & Chambers, M.E. (2005) (eds) Children with Developmental Coordination Disorder, London: Whurr Publishers.
- Kirby, A., Sugden, D.A., Beveridge, S., Edwards, L. (2008) Developmental coordination disorder (DCD) in adolescents and adults in further and higher education. Journal of Research in Special Educational Needs, Vol. 8, No.3, 2008, 120-131.
- Kirby, A., Edwards, L. & Sugden, D.A. & Rosenblum, S. (2010) The development and standardization of the adult developmental coordination disorders/dyspraxia checklist (ADC). Research in Developmental Disabilities, 31, 131-139.
- Kirby, A., Edwards, L., & Sugden, D.A. (2011) Emerging adulthood in developmental coordination disorder: parent and young adult perspectives.
   Research in Developmental Disabilities 32(4): 1351-60

- Knox, E. (2002) Educational attainments of children with specific language impairment at year 6. Child Language Teaching and Therapy, June 2002, Vol. 18, No.2, 103-124.
- Knuckey, N., Gubbay, S. (1983) Clumsy children: a prognostic study. Aust. Paediatr.J. (1983), 19: 9-13.
- Kwan, M., Cairney, J., Hay, J. & Faught, B. (2013) Understanding physical activity motivation for children with DCD: An investigation using the Theory of Planned Behaviour. Research in Developmental Disabilities, 34 (2013), 3691-3698.
- Larkin, D. & Cermak, S. (2002) Issues in identification and assessment of DCD, chapter 6 in Larkin, D. & Cermak, S. (2002) (eds) Developmental Coordination Disorder, New York: Delmar, Thomson Learning Inc.
- Larkin, D. & Rose, E. (2005) Assessment of DCD, chapter 7 in Sugden, D.A. & Chambers, M.E. (2005) Developmental Coordination Disorder, London: Whurr publishers.
- Larsson, H., Dilshad, R., Lichtenstein, P., Barker, E. (2011) Developmental trajectories of DSM-IV symptoms of attention-deficit/hyperactivity disorder: genetic effects, family risk and associated psychopathology. Journal of Child Psychology and Psychiatry, 52: 9 (2011), pp 954-963.
- Law, M., Haight, M., Milroy, B, Willms, D., Stewart, D. & Rosenbaum, P. (1999) Environmental factors affecting the occupations of children with physical disabilities. Journal of Occupational Science, 6 (3), 102-110.

- Leonard, H.C. (2016) The impact of poor motor skills on perceptual, social and cognitive development: The case of developmental coordination disorder. Frontiers in Psychology 7:311. Doi: 10.3389/fpsyg.2016.00311.
- Leonard, H. & Hill, E. (2015) Executive difficulties in developmental coordination disorder: methodological issues and future directions. Curr Dev Disord Rep (2015) 2: 141-149.
- Lerner, R.M. (2005) Foreword in Bronfenbrenner, U. (2005) Making Human Beings Human, bio ecological perspectives on human development. London: Sage Publications.
- Lingam, R., Hunt, L., Golding, J., Jongmans, M., Emond, A. (2009) prevalence of Developmental Coordination Disorder using the DSM-IV at 7 years of age: A UK population based study. Paediatrics, vol.123, No.4, April 2009, e693-e700. DOI. 10.1542/peds.2008-1770.
- Lingam, R., Golding, J., Jongmans, M., Hunt, L., Ellis, M., Emond, A. (2010) The association between developmental coordination disorder and other developmental traits. Paediatrics, Vol. 126, No.5, Novmber, 2010, e1109e1118. DOI:10.1542/peds.2009-2789.
- Lingam, R.P., Novak, C., Emond, A. & Coad, J.E. (2013) The importance of identity and empowerment to teenagers with developmental coordination disorder. Child: Care, health and development, 40, 3, 309-318.
- Losse, A., Henderson, S., Elliman, D., Knight, E., Jongmans, M. (1991) Clumsiness in children – do they grow out of it? A 10 year follow up study. Developmental Medicine and Child Neurology, 1991, 33, 55-68.

- Macnab, J., Miller, L., Polatajko, H. (2001) The search for subtypes of DCD: is cluster analysis the answer? Human Movement Science 20 (2001) 49-72.
- Magalhaes, L.C., Cardoso, A.A. & Missiuna, C. (2011) Activities and participation in children with DCD: A systematic review. Research in Developmental Disabilities, 32 (2011) 1309-1316.
- Mancini, V., Rigoli, D., Cairney, J., Roberts, D. & Piek, J. (2016) The elaborated environmental stress hypothesis as a framework for understanding the association between motor skills and internalizing problems: A mini-review. Frontiers in Psychology 7:239 doi:10.3389/fpsyg.2016.00239.
- Mandich, A.D., Polatajko, H.J., Rodger, S (2003) Rites of passage: understanding participation of children with developmental coordination disorder. Human Movement Science, 22 (4-5), 583-595.
- Martin, N.C., Piek, J., Baynam, G., Levy, F., Hay, D. (2010) An examination of the relationship between movement problems and four common developmental disorders. Human Movement Science 29 (2010) 799-808.
- McArthur, G.M., Hogben, J.H., Edwards, V.T., Heath, S.M., Mengler, E.D. (2000) On the "specifics" of specific reading disability and specific language impairment. J Child Psychol. Psychiat. Vol. 41, No.7, pp 869-874.
- McCarron, L.T. (1982) MAND McCarron Assessment of Neuromuscular Development (revised edition), Dallas TX: Common Market Press.

- McGraw, M. (1935) Growth: a study of Johnny and Jimmy, New York: Appleton-Century.
- Menard, S.W. (2002) Longitudinal Research 2<sup>nd</sup> edition, Sage University paper series: Quantitative Applications in the Social Sciences, No. 07-76, London: Sage Publications Inc.
- Michelson, S.I. (2006) Social consequences of cerebral palsy, PhD thesis, Faculty of Health Science, University of Copenhagen, National Institute of Public Health Copenhagen.
- Miller, L.J., Polatajko, H.J., Missiuna, C., Mandich, A. & Macnab, J.J. (2001) A pilot trail of a cognitive treatment for children with developmental coordination disorder. Human Movement Science, 20 (1-2), 183-210.
- Mon Williams, M. (2005) Why robot systems could help children with developmental coordination disorder in Sugden, D.A. (2005) Leeds consensus statement: Developmental coordination disorder as a specific learning difficulty, ESRC research seminar series 2004-2005, DCD-UK.
- Missiuna, C., Mandich, A., Polatajko, H. & Malloy-Miller, T. (2001) Children with DCD, strategies for success. Physical and Occupational Therapy in Paediatrics, Vol.20, No.2/3, 69-81.
- Missiuna, C., Pollock, N. & Law, M. (2004) The perceived efficacy and goal setting system, San Antonio: Harcourt Assessments Inc.

- Missiuna, C., Gaines, R., Soucie, H. & McClean, J. (2006) Parental questions about developmental coordination disorder: A synopsis of current evidence. Paediatric child Health, Vol. 11, No.8, 507-512.
- Missiuna, C., Moll, S., Law, M., King, S., King, G. (2006) Mysteries and mazes: parents experiences of children with developmental coordination disorder. Canadian Journal of Occupational Therapy, 73 (1), 7-17.
- Missiuna, C., Rivard, L. & Bartlett, D. (2006) Exploring assessment tools and the target of intervention for children with DCD. Physical and Occupational Therapy in Paediatrics, Vol. 26 (1/2), 2006.
- Missiuna, C., Moll, S., King, S., King, G., Law, M. (2007) A trajectory of troubles:Parent's impressions of the impact of developmental coordination disorder.Physical and Occupational Therapy in Paediatrics, Vol. 27 (1), 81-101.
- Missiuna, C., Moll, S., King, G., Stewart, D., Macdonald, K. (2008) Life experiences of young adults who have coordination difficulties. Canadian Journal of Occupational Therapy, Vol. 75, No.3, June 2008, 157-165.
- Missiuna, C., Pollock, N., Egan, M., DeLaat, D, Gaines, R., Soucie, H. (2008)
  Enabling occupation through facilitating the diagnosis of Developmental
  Coordination Disorder. Canadian Journal of Occupational Therapy, Vol.
  75, No. 1, February 2008, 29-34.
- Missiuna, C., Cairney, J., Pollock, N., Russell, D., Macdonald, K., Cousins, M., Veldhuizen, S., Schmidt, L. (2011) A staged approach for identifying children with developmental coordination disorder from the population. Research in Developmental Disabilities, 32 (2011) 549-559.

- Missiuna, C. & Campbell, W. (2014) Psychological aspects of developmental coordination disorder: can we establish causality? Curr Dev Disord Rep (2014) 1: 125-131.
- Missiuna, C., Cairney, J., Pollock, N., Campbell, W., Russell, D., Macdonald, K., Schmidt, L., Heath, N., Veldhuizen, S. & Cousins, M. (2014) Psychological distress in children with DCD and ADHD. Research in Developmental Disabilities, 35 (2014) 1198-1207.
- Missiuna, C. (2015) Prospecting in the field of developmental coordination disorder, Physical & Occupational therapy in Paediatrics, 35 (2): 93-96, 2015.
- Miyahara, M. (1994) Subtypes of students with learning disabilities based on gross motor functions. Adapted Physical activity Quarterly, 1994, 11, 368-382.
- Morse, J. (1991) Approaches to qualitative-quantitative methodological triangulation. Nursing Research, Vol.40, No. 2, 120-123.
- Morton, J. (2004) Understanding developmental disorders: a causal modeling approach. Oxford: Blackwell Publishing.
- Newell, K.M. (1991) Motor skill acquisition. Annual Review Psychology, 1991, 42:213-37.
- Newell, K.M. (1986) Constraints on the development of coordination. In Wade, M.G. & Whiting, H.T.A. (Eds), Motor development in children: aspects of coordination and control. Dordrecht: Martinus Nijhoff, pp3341-61.

- Newell, K.M. & Wade, M.G. (2018) Physical growth, body scale and perceptualmotor development, chapter 7 in Advances in Child Development and Behaviour, volume 55, Elsevier Inc. ISSN 0065-2407
- NICE Public Health collaborating Centre-Physical activity (2007). Physical activity and children, Review 3- The views of children on the barriers and facilitators to participation in physical activity: a review of qualitative studies, version 4. Available at: http://www.nice.org.uk/guidance/ph17/documents/promoting-physical-activity-for-children-consulatation-on-the-evidence4, accessed on: 18.3.18
- Noordstar, J, Stuive, I., Herweijer, H., Holty, L., Oudenampsen, C., Schoemaker, M. & Reinders-Messelink, H. (2014) Perceived athletic competence and physical activity in children with DCD who are clinically referred, and control children. Research in Developmental Disabilities, 35 (2014) 3591-3597.
- Norbury, C., Nash, M., Baird, G., Bishop, D. (2004) Using a parental checklist to identify diagnostic groups in children with communication impairment: a validation of the children's communication checklist -2. Int. J. Lang. Comm. Dis. July-September 2004, Vol. 39, No. 3, 345-364.
- Pannekoek, L., Rigoli, D., Piek, J, Barrett, N. & Schoemaker, M. (2012) The revised DCDQ: Is it a suitable screening measure for motor difficulties in adolescents? Adapted Physical Activity Quarterly, 2012, 29, 81-97.
- Pearsall-Jones, J.G., Piek, J.P., Levy, F. (2010a) Developmental Coordination Disorder and cerebral palsy: categories or a continuum? Human Movement Science 29 (2010) 787-798.

- Pearsall-Jones, J.G., Piek, J.P., Levy, F. (2010b) Etiological pathways for developmental coordination disorder and attention deficit/hyperactivity disorder: shared or discrete? Expert Review Neurother. 10 (4), 491-494 (2010).
- Perezgonzalez, J.D. (2015) The meaning of significance in data testing. Frontiers in Psychology, August 2015, Volume 6, article 1293, doi: 10.3389/fpsyg.2015.01293
- Peters, J., Henderson, S. E. (2008) Understanding Developmental Coordination Disorder and its impact on families: the contribution of single case studies. International journal of Disability, Development and Education, Vol. 1, No.2, June 2008, 97-111.
- Peters, L., Maathuis, C., Hadders-Algra, M. (2010) Limited motor performance and minor neurological dysfunction at school age Acta Paediatrica, http:/doi.org.10.1111/j.1651-2227.2010.01998x
- Piek, J., Dyck, M., Nieman, A., Anderson, M. Hay, D., et al. (2004) The relationship between motor coordination, executive functioning and attention in school aged children. Archives of Clinical Neuropsychology 19 (2004) 1063-1076.
- Piek, J., Dyck, M., Francis, M., Conwell, A. (2007) Working memory, processing speed, and set shifting in children with developmental coordination disorder. Developmental Medicine and Child Neurology 2007, 49: 678-683.

- Pless, M., Carlsson, M., Sundelin, C. & Persson, K. (2002) Preschool children with DCD: a short-term follow up of motor status at seven to eight years of age. Acta Peadiatr 91:521-528, 2002.
- Polatakjo, H.J., Mandich, A.D., Miller, L.T., & Macnab, J.J. (2001) Cognitive orientation to daily occupational performance (CO-OP); part II The evidence. Physical and Occupational Therapy in Pediatrics, 20, 83-106.
- Polderman, T., Boomsma, D., Verhulst, F., Huizink, A. (2010) A systematic review of prospective studies on attention problems and academic achievement. Acta Psychiatrica Scandanavica 2010: 122: 271-284.
- Poulsen, A., Ziviani, J., Cuskelly, M. & Smith, R. (2007) Boys with DCD: Loneliness and team sports participation. American Journal of Occupational Therapy, 61 (4) 451-462.
- Poulsen, A.A., Ziviani, J.M. & Cuskelly, M. (2007) Perceived freedom in leisure and physical coordination ability: impact on out-of-school activity participation and life satisfaction. Child: Care, health and development, 33, 4, 432-440.
- Poulsen, A.A., Ziviani, J.M., Johnson, H. & Cuskelly, M. (2008) Loneliness and life satisfaction of boys with developmental coordination disorder: The impact of leisure participation and perceived freedom in leisure. Human Movement Science, 27 (2), 325-343.
- Poulsen, A.A., Ziviani, J.M., & Cuskelly, M (2008) Leisure time physical activity energy expenditure in boys with DCD: The role of peer relations selfconcept perceptions. OTJR: Occupation, participation and Health, 28, 30-39.

- Pratt, M.L., Hill, E.L. (2011) Anxiety profiles in children with and without developmental coordination disorder. Research in Developmental Disabilities, 32 (2011) 1253-1259.
- Preston, N., Magallon, S., Hill, L.J.B., Andrew, E., Ahern, S.M. & Mon Williams, M. (2017) A systematic review of high quality randomized controlled trials investigating motor skill programmes for children with DCD. Clinical Rehabilitation 2017, Vol. 31(7) 857-870.
- Prunty, M., Barnett, A., Wilmut, K. & Plumb, M. (2013) Handwriting speed in children with DCD: Are they really slower? Research in Developmental Disabilities, 34 (9) 2927-2936.
- Przysucha, E., Boyle, A. & Zerpa, C. (2016) Reliability of Movement Assessment Battery for Children, Age band 2 (second edition): Exploratory study. RA Journal of Applied Research, Vol. 2, Issue 09, 611-619. ISSN (e): 2394-6709, <u>www.rajournas.in</u>
- Public Health England (2019) Wider determinants of health: Marmot indicators available at httpp://fingertips.phe.org.uk/profile/wider-determinants/data (accessed 29.7.19)
- Rasmussen, P., Gillberg, C. (2000) Natural outcome of ADHD and Developmental Coordination Disorder at age 22 years: A controlled, longitudinal, community-based study. J Am. Acad. Child Adolesc. Pyschiatry, 31: 11, November 2000.
- Reinhart, N., McGinley, J. (2010) Is motor dysfunction core to autistic spectrum disorder? Developmental Medicine and Child Neurology, 2010, 52: e697. DOI:10.1111/j.1469-8749.2009.03606.x.

- Rivalis, I., Hay, J., Cairney, J., Klentrou, P., Liu, J & Faught, B. (2011) Physical activity and fitness in children with DCD: A systematic review. Research in Developmental Disabilities 32 (2011) 894-910.
- Rivard, L., Missiuna, C., McCauley, D. & Cairney, J. (2014) Descriptive and factor analysis of the Developmental Coordination Disorder Questionnaire (DCDQ'07) in a population –based sample of children with and without Developmental Coordination Disorder. Child: care, health & development, 2014; 40(1): 42-9. doi:10.1111/j.1356-221425.x
- Roberts, G., Anderson, P., Davis, N., De Luca, C. et al (2011) Developmental coordination disorder in geographic cohorts of 8 year old children born extremely preterm extremely low birth weight in the1990s. Developmental Medicine and Child Neurology 2011, 53:55-60.
- Roeber, B., Tober, C., Bolt, D., Pollak, S. (2012) Gross motor development in children adopted from orphanage settings. Developmental Medicine and Child Neurology, 2012, 54: 527-531.
- Rodger, S., Ziviani, J., WAtter, P., Ozanne, A., Woodyatt, G. & Springfield, E. (2003) Motor and functional skills of children with developmental coordination disorder: A pilot investigation of measurement issues. Human Movement Science 22 (2003) 461-478.
- Rosenbaum, D.A. (2010) Human Motor Control, 2<sup>nd</sup> edition, San Diego CA: Elsevier.
- Rosenblum, S. (2006) The development and standardization of the Children Activity Scales (ChAS-P/T) for the identification of children with DCD. Child: Care, Health and Development, 32(6), 619-632.

- Rosenblum, S. (2015) Do motor ability and handwriting kinematic measures predict organizational ability among children with DCD? Human Movement Science (2015) 43: 201-215.
- Rutter, M. (2011) Research Review: child psychiatric diagnosis and classification: concepts, findings, challenges and potential. Journal of Child Psychology and Psychiatry, 52: 6 (2011) pp647-660.
- Schmidt, R.A. (1975) A Schema theory of discrete motor skill learning. Psychological Review, Vol. 82, No.4, 225-260.
- Schmidt, R.A. & Wrisberg, C.A. (2000) 2<sup>nd</sup> edition Motor learning and performance: a problem-based learning approach, Champagne, IL: Human Kinetics.
- Schmidt, R.A. & Lee, T.D. (2011) Motor control and learning: A behavioural emphasis (5<sup>th</sup> ed.), Champagne II: Human Kinetics in Adolph, K.E., Cole, W.G., Vereijken, B. (2014) Intraindividual variability in the development of motor skills in childhood, chapter 4 in Diehl ,M., Hooker, K. & Sliwinski, M. (Eds) (2014) Handbook of Intra-individual Variability Across the lifespan. New York: Routledge/Taylor & Francis Group. Preprint formatted by authors.
- Schoemaker, M. & kalverboer, A. (1994) Social and affective problems in children who are clumsy: How early do they begin? Adapted Physical Activity Quarterly, 11, 130-140.
- Schoemaker, M.M., Hijlkema, M.G. & Kalverboer, A.F. (1994) Physiotherapy for clumsy children: An evaluation study. Developmental Medicine & Child Neurology, 36, 143-155.

- Schoemaker, M. & Smits-Engelsman, B.C.M. (2005) Neuromotor Task Training: a new approach to treat children with DCD, chapter 11 in Sugden, D.A. & Chambers, M.E. (2005) Children with Developmental Coordination Disorder, London: Whurr publishers.
- Schoemaker, M., Lingam, R., Jongmans, M., Van Heuvelen, M. & Eland, A. (2013) Is severity of motor coordination difficulties related to co-morbidity in children at risk for developmental coordination disorder? Research in Developmental Disabilities, 34 (2013) 3084-3091.
- Schoemaker, M. & Wilson, B. (2015) Screening for DCD in school age children, chapter 6 in DCD and its consequences, Cairney, J. Ed. (2015) Toronto: University of Toronto Press.
- Schoemaker, M.M. & Smits-Engelsman, B.C.M (2015) Is treating motor problems in DCD just a matter of practice and more practice? Current Developmental Disorders Report (2015) 2; 150-156.
- Schulz, J., Henderson, S.E., Sugden, D.A., Barnett, A, L. (2011) Structural validity of the Movement ABC-2 test: Factor structure comparisons across three age groups. Research in Developmental Disabilities 32 (2011) 1361-1369.
- Shannon, C.E & Weaver, W. (1949) The mathematical theory of communication. Urbana: University of Illinois Press.
- Shumway-Cook, A. & Woollacott, M.H. (2001) Motor control, theory and practical applications, second edition. Philadelphia: Lippincott Williams & Wilkins.

- Shirley, M.M. (1931) The first two years, a study of twenty-five babies: I. Postural and locomotor development. Minneapolis, MN: University of Minnesota Press
- Silman, A., Cairney, J., Hay, J., Klentrou, P. & Faught, B. (2011) Role of physical activity and perceived adequacy on peak aerobic power in children with DCD. Human Movement Science, 30 (2011) 672-681.
- Silverman, D. (1997) Qualitative Research, theory, method and practice, London: Sage publications.
- Sinani, C., Sugden, D.A. & Hill, E.L. (2011) Gesture production in school vs. clinical samples of children with developmental coordination disorder (DCD) and typically developing children. Research in Developmental Disabilities 32 (2011) 1270-1282.
- Skinner, R. & Piek, J. (2001) Psychosocial implications of poor motor coordination in children and adolescents. Human Movement Science, 20 (1-2), 73-94.
- Slykerman, S., Ridgers, N., Stevenson, C. & Barnett, L. (2016) How important is young children's actual and perceived movement skills competence to their physical activity? Journal of Science and Medicine in Sport, Vol. 19, Iss.6, 488-492.
- Smith, P.K., Cowie, H. and Blades, M. (2003) Understanding children's development, 4<sup>th</sup> edition, Oxford: Blackwell publishing.

- Smith, L.B. & Thelen, E. (2003) Development as a dynamic system. Trends in Cognitive Science, Vol. 7, No. 8, 343 348.
- Smith, T.J. & Wade, M.G. (2015) Variability in human motor and sport performance, chapter 2 in Smith, T.J., Henning, R., Wade, M.G. & Fisher, T. (2015) Variability in human performance, Boca Raton: Taylor & Francis Group Inc.
- Smits- Engelsman, B., Niemeijer, A. & Van Galen, G. (2001) Fine motor deficiencies in children diagnosed as DCD based on poor grapho-motor ability. Human Movement Science, 20 (1-2), 161-182.
- Smits-Engelsman, B., Schoemaker, M., Delabastita, T., Hoskens, J. & Geuze, R. (2015) Diagnostic criteria for DCD: past and future. Human Movement Science, Vol. 42, 293-306.
- Snijders, T.A.B & Bosker, R.J. (2002) Multilevel Analysis: an introduction to basic and advanced multilevel modeling (reprint edition). London: Sage Publications
- Snowling, M. J. (2008) Specific disorders and broader phenotypes: the case of dyslexia. The Quarterly Journal of Experimental Psychology, 2008, 61 (1), 142-156.
- Soorani-Lunsing, R., Hadders-Algra, M., Huisjes, H.& Towen, B. (1994) Neurobehavioural relationships after the onset of puberty. Developmental Medicine and Child Neurology, 36, 334-343.
- Spironello, C., Hay, J., Missiuna, C., Faught, B., Cairney, J. (2010) Concurrent and construct validation of the short form of the Bruininks-Oseretsky Test of Motor Proficiency and the Movement ABC when administered in the

field: implications for screening. Child: care, health and development, 36, 4, 499-507.

- Stake, R. E. (1994) Cases studies, chapter 14 in Denzin, N. & Lincoln, Y. (1994) (editors) A Handbook of Qualitative Research, London: Sage Publishers.
- Stephenson, E. & Chesson, R. (2008) 'Always the guiding hand': parents' accounts of the long-term implications of DCD for their children and families. Child: Care, Health and Development, 34, 335-343.
- Stewart, D. & Rosenbaum, P. (2003) The International Classification of Functioning, Disability and Health (ICF) A global model to guide clinical thinking and practice in childhood disability. Available at <u>http://www.canchild.ca/en/canchildresources/internationalclassificationoffu</u> <u>nctioning</u>. Accessed on 26.4.2011.
- Stodden, D., Goodway, J., Langendorfer, S., Roberton, M., Rudisill, M., Garcia,
  C. & and Garcia, L. (2008) Developmental Perspective on the Role of
  Motor Skill Competence in Physical Activity: An Emergent Relationship.
  Quest, 2008, 60, 290-306.
- Sugden, D.A. (2018) Developmental coordination disorder and its cause: The road less travelled, commentary by David Sugden, Human movement Science 57 (2018) 501-504.
- Sugden, D.A. (2014) The aggregation of marginal gains in relation to ecological intervention and support for children with additional needs. Hillary Place Papers, Issue 1, University of Leeds. Available at www.hpp.education.leeds.ac.uk accessed on 21.3.18.

- Sugden, D.A. & Wann, C. (1987) The assessment of motor impairment in children with moderate learning difficulties. British Journal of Educational Psychology, Vol. 57, Issue 2, (June 1987), p225-236.
- Sugden, D.A. & Chambers, M.E. (2003) Intervention in children with DCD the role of parents and teachers. British Journal of Educational Psychology, 73: 545-61.
- Sugden, D.A. & Chambers, M.E. (2005) (Eds) Developmental Coordination Disorder, London: Whurr publishers.
- Sugden, D.A. & Chambers, M.E. (2006) Stability and change in children with DCD. Child: Care, health and development, 33, 5, 520-528.
- Sugden, D.A., Chambers, M., Utley, A. (2006) Leeds Consensus Statement, ESRC Seminar series (2004-2005).
- Sugden, D.A. & Dunford, C. (2007) Intervention and the role of theory, empiricism and experience in children with motor impairment. Disability and Rehabilitation, 2007; 29(1): 3-11.
- Sugden, D.A. & Wade, M. (2013) Typical and atypical motor development, London: Mac Keith Press.
- Sugden, D.A & Soucie, H. (2017) Motor Development, chapter 4 in Armstrong,
   N. & Van Mechelen, W. (2017) Oxford textbook of children's sport and exercise medicine, Oxford: Oxford University Press.

- Summers, J., Larkin, D., Dewey, D. (2008) Activities of daily living in children with developmental coordination disorder: Dressing, personal hygiene, and eating skills. Human Movement Science, 27 (2008) 215-229.
- Tallal, P., Allard, L., Miller, S., Curtiss, S. (1997) Academic outcomes of language impaired children. Chapter 10 in Hulme, C., Snowling, M. (eds) (1997) Dyslexia: biology, cognition and intervention. London: Whurr Publishers.
- Thelen, E. (2000) Motor development as foundation and future of developmental psychology. International Journal of Behavioural Development, 2000, 24 (4), 385-397.
- Toussaint, L., Van der Cammon-van Zijp, M., Jannsen, A., Tibboel, D., Van Heijst, A., Ijsseistijn, H. (2016) Perceived motor competence differ from actual performance in 8 year old neonatal ECMO survivors. Paediatrics, Vol. 137, No.3, March 2016, e20152724.
- Trafimow, D. & Marks, M. (2015) Editorial, Basic and Applied Psychology, 37:1-2, 2015 DOI:10.1080/01973533.2015.1012991
- Tsiotra, G. D., Flouris, A. D., Koutedakis, Y., Faught, B. E., Nevill, A. M., Lane, A. M., & Skenteris, N. (2006) A comparison of developmental coordination disorder prevalence rates in Canadian and Greek children, The Journal Of Adolescent Health: Official Publication Of The Society For Adolescent Medicine, vol. 39, no. 1, pp. 125–127.
- Van Waelvelde, H., Oostra, A., Dewitte, G., Van Den Broeck, C. (2010) Stability of motor problems in young children with or at risk of autism spectrum

disorders, ADHD, and or developmental coordination disorder. Developmental Medicine and Child Neurology, 2010, 52: e74-e178. DOI:10.1111/j.1469-8749.2009.03606.x.

- Veldhuizen, S. & Cairney, J. (2015) Chapter 7, Methodological issues in fieldbased research: case identification and study design in Cairney, J. (ed.)
  2015 Developmental coordination disorder and its consequences. London: University of Toronto Press.
- Viholainen, H., Ahonen, T., Cantell, M., Lyytinen, P., Lyytinen, H. (2002) Development of early motor skills and language in children at risk from familial dyslexia. Developmental Medicine and Child Neurology, 2002, 44, 761-769.
- Visser, J. (1998) Clumsy adolescents: a longitudinal study on the relationship between physical growth and sensorimotor skills of boys with and without DCD. Unpublished PhD thesis, University of Groningen.
- Visser, J., Geuze, R., Kalverboer, A. (1998) The relationship between physical growth, the level of activity and the development of motor skills in adolescence: differences between children with DCD and controls. Human Movement Science 17 (1998) 573-608.
- Visser, J. (2003) Developmental Coordination Disorder: a review of research on subtypes and comorbidities. Human Movement Science 22 (2003) 479-493.
- Visser, J. (2007) Suybtypes and comorbidities, chapter 3 in Geuze, R. (2007) (ed) Developmental coordination disorder: a review of current approaches. Marseille: Solal.
- Wade, M. & Whiting, H. (1986) eds. Motor development in children: Aspects of coordination and control. Boston: Martinus Nijhoff in Newell, K.M. (1991)
   Motor skill acquisition. Annual Review Psychology, 1991, 42:213-37.
- Wade, M. & Kazek, M. (2018) Developmental coordination disorder and its cause: The road less travelled, Human movement Science 57 (2018) 489-500.
- Walker, I. (2008) Statistics for Psychology, making sense of our world through analysis. Available at www.staff.bath.ac.uk accessed on 5.10.16
- Wang, T., Tseng, M., Wilson, B. & Hu, F. (2009) Functional performance of children with DCD at home and at school. Developmental Medicine and Child Neurology, 2009, 51(10): 817-825.
- Welk, G. (1999) The youth physical activity promotion model: A conceptual bridge between theory and practice. QUEST, 1999, 51, 5-23.
- Welk, G., Wood, K & Morss, G. (2003) Parental influences on physical activity in children: an exploration of potential mechanisms. Pediatric Exercise Science, Vol.15, No.1, p19-33.
- Welsby, M. & Paxman, P. (2014) Developing embodied cognition: insights from children's concepts and language processing. Frontiers in Psychology, Vol.8, 506, 1-10. Doi: 10.3389/fpsyg.2014.00506
- Williams, J., Hyde, C. & Spittle, A. (2014) Developmental coordination disorder and cerebral palsy: is there a continuum? Curr Dev Disord Rep (2014) 1: 118-124.

- Wilmut, K. & Byrne, M. (2014) Grip selection for sequential movements in children and adults with and without DCD. Human Movement Science (2014) 36: 272-284.
- Wilson, B., Dewey, D., Campbell, A. (1998) Developmental Coordination Disorder Questionnaire. Calgary Canada: published by the authors.
- Wilson, B., Kaplan, B., Crawford, S., Campbell, A. & Dewey, D. (2000)
   Reliability and validity of a parent questionnaire on childhood motor skills.
   American Journal of Occupational Therapy, 54 (5), 484-493.
- Wilson, B. & Crawford, S. (2007) The Developmental Coordination Questionnaire 2007 available at <u>www.dcdq.ca</u> accessed 12.2.19
- Wilson, B., Crawford, S., Green, D., Roberts, G., Aylott, A. & Kaplan, B. (2009)
   Psychometric properties of the revised Developmental Coordination
   Questionnaire. Physical and Occupational Therapy in Paediatrics 29 (2):
   182-202.
- Wilson, B., Neil, K., Kemps, P. et al. (2013) Awareness and knowledge of DCD among physicians, teachers and parents. Child Care Health Dev 2013, 39: 296-300
- Wilson, P., McKenzie, B. (1998) Information processing deficits associated with Developmental Coordination Disorder : A meta-analysis of research findings. J. Child Psychol. Psychiat. Vol. 39, No.6, pp829-840.

- Wilson, P., Maruff, P., Lum, J. (2003) Procedural learning in children with developmental coordination disorder. Human Movement Science 22 (2003) 515-526.
- Wilson, P., Ruddock, S., Smits-Engelsman, B., Polatajko, H. & Blank, R. (2013) Understanding performance deficits in DCD: a meta-analysis of recent research. Developmental Medicine and Child Neurology, 55 (3), 217-228.
- Wilson, P.H., Adams, I.L.J., Caeyenberghs, K., Thomas, P., Smits-Engelsman,
  B. & Steenbergen, B. (2016) Motor imagery training enhances motor skill in children with DCD: a replication study. Research in Developmental Disabilities 57 (2016) 54-62.
- Wilson, P., Smits-Engelsman, B., Caeyenberghs, K., Steenbergen, B., Sugden, D., Clark, J, Mumford, N. & Blank, R. (2017) Cognitive and neuroimaging findings in developmental coordination disorder: new insights from a systematic review of recent research. Developmental Medicine & Child Neurology, 2017, 59: 1117-1129.
- Wilson, P.H., Smits-Engelsman, B., Caeyenberghs, K. & Steenbergen, B. (2017b) Toward a hybrid model of developmental coordination disorder, Curr Dev Disord Rep (2017) 4:64-71.
- Wolpert, D.M., Miall, C. & Kawato, M. (1998) Internal models of the cerebellum, trends in Cognitive Science, Vol.2, No.9, 338-347.
- World Health Organization (1993) Classification of mental and behavioural disorders: diagnostic criteria for research. Geneva: World Health Organization.

- World Health Organization (1992) ICD 10 Classification of mental and behavioural disorders: clinical descriptions and diagnostic guidelines. Geneva: World Health Organization.
- World Health Organization (WHO) (2001) International classification of functioning, disability and health (ICF). Geneva: World Health Organization.
- Wright, H., Sugden, D.A. (1996) The nature of Developmental Coordination Disorder: Inter- and intragroup differences. Adapted Physical Activity Quarterly, 1996, 13, 357-371.
- Wright, H., Sugden, D.A. (1996) A two-step procedure for the identification of children with DCD in Singapore. Developmental Medicine and Child Neurology 1996, 38, 1099-1105.
- Wright, H. C. (1997) Children with Developmental Coordination Disorder A review. European Journal of Physical Education, 1997, 2, 5-22.
- Wright, K., Furzer, B., Licari, M., Thornton, A., Dimmock, J., Naylor, L, Reid, S.
  & Kwan, S. (2018) Physiological characteristics, self perceptions and parental support of physical activity in children with, or at risk of developmental coordination disorder. Research in Developmental Disabilities (2018) http://doi.org/10.1016/j.ridd.2018.05.013
- Wuang, Y., Su, J. & Su, C. (2012) Reliability and responsiveness of the Movement Assessment Battery for Children – second edition test in children with developmental coordination disorder. Developmental Medicine and Child Neurology, 2012: 54, 160-165. DOI:10.1111/j.1469-8749.2011.04177.x

- Yin, R. K. (2009) Case Study Research, design and methods 4<sup>th</sup> edition, applied social research methods volume 5, London: Sage Publications Inc.
- Zecevic, C.A., Tremblay, L., Lovsin, T & Michel, L. (2010) Parental influence on young children's physical activity. International Journal of Paediatrics, Vol. 2010, article ID 468526, 9 pages. Doi:10.1155/2010/468526
- Zwicker, J.G. & Harris, S.R. (2009) A reflection of motor learning in paediatric occupational therapy practice. Canadian Journal of Occupational Therapy, 2009, 76 (1): 29-37
- Zwicker, J., Missiuna, C., Harris, S. & Boyd, L. (2012) Developmental coordination disorder: A review and update. European Journal of Paediatric Neurology, 16 (2012) 573-581.
- Zwicker, J., Harris, S., & Klassen, A. (2013) Quality of life domains affected in children with DCD: A systematic review. Child: Care, Health and Development 30(4): 562-580.
- Zwicker, J. & Mickelson, E. (2017) Differential diagnoses for DCD. Paediatrics Child Health, June 2017, Vol.22 (suppl.1):e5
- Zwicker, J., Suto, M., Harris, S., Vlasakova, N. & Missiuna, C. (2018) DCD is more than a motor problem: children describe the impact of daily struggles on their quality of life. British Journal of Occupational Therapy 2018, vol. 81(2)65-73.

# Appendix 1 Ethical approval

Performance, Governance and Operations Research & Innovation Service Charles Thackrah Building 101 Clarendon Road Leeds LS2 9LJ Tel: 0113 343 4873



Email: ResearchEthics@leeds.ac.uk

Vicky McQuillan

School of Education

University of Leeds

Leeds, LS2 9JT

# AREA Faculty Research Ethics Committee

**University of Leeds** 

3 October 2019

Dear Vicky

# Title of study:The progression over time of profiles of children with<br/>Developmental Coordination Disorder (DCD)

Ethics reference: AREA 12-121, response 2

I am pleased to inform you that the above research application has been reviewed by the ESSL, Environment and LUBS (AREA) Faculty Research Ethics Committee and following receipt of your response to the Committee's comments, I can confirm a favourable ethical opinion as of the date of this letter. The following documentation was considered:

Document	Version	Date
AREA 12-121 response2.docx	1	20/01/14
AREA 12-121 Dec Final ethical consent and information sheets.docx	1	09/12/13
AREA 12-121 Ethical review form final version.doc	1	12/06/13
AREA 12-121 Interview schedule.docx	1	12/06/13
AREA 12-121 Child Information sheet.docx	1	12/06/13
AREA 12-121 parent information sheet Movement development and children.docx	1	12/06/13
AREA 12-121 Appendix 2 parent consent form.docx	1	12/06/13

Committee members still had concerns about the information leaflet for children who are only 8 years old and urge you to further adapt the information leaflet for younger children. It should be much more user friendly and talk directly to the child. Reviewers suggested showing it to a primary school teacher and perhaps carrying out a pilot with some children of a similar age before it is used.

Please notify the committee if you intend to make any amendments to the original research as submitted at date of this approval, including changes to recruitment methodology. All changes must receive ethical approval prior to implementation. The amendment form is available at <a href="http://ris.leeds.ac.uk/EthicsAmendment">http://ris.leeds.ac.uk/EthicsAmendment</a>.

Please note: You are expected to keep a record of all your approved documentation, as well as documents such as sample consent forms, and other documents relating to the study. This should be kept in your study file, which should be readily available for audit purposes. You will be given a two week notice period if your project is to be audited. There is a checklist listing examples of documents to be kept which is available at http://ris.leeds.ac.uk/EthicsAudits.

We welcome feedback on your experience of the ethical review process and suggestions for improvement. Please email any comments to <u>ResearchEthics@leeds.ac.uk</u>.

Yours sincerely

Jennifer Blaikie

Senior Research Ethics Administrator, Research & Innovation Service On behalf of Dr Andrew Evans, Chair, <u>AREA Faculty Research Ethics</u> <u>Committee</u>

CC: Student's supervisor(s)

# Appendix 2 Study flyer



Up to 5% of schoolchildren have difficulty with motor coordination that is not due to any medical condition. This can seriously interfere with daily life at home and school. We have a project to investigate it and would like to invite you to take part.

# **2014** Project

Vicky McQuillan PhD candidate University of Leeds

University of Leeds School of Education Hilary Place, Leeds

Professor David Sugden University of Leeds

University of Leeds Dr Mary Chambers & Dr Ruth Swanwick

CONTACT:

vickym@liv.ac.uk

It often goes undiagnosed and yet is a longterm condition remaining throughout childhood and into adulthood often causing difficulty with handwriting, tool use, ball skills and physical activity. It is called Developmental Coordination Disorder (sometimes known as dyspraxia). It can have a negative impact on a child's confidence and is frequently accompanied by other difficulties such as poor attention, reading or language difficulties and problems with social interaction.

Without appropriate identification and intervention the children are at risk of being unable to participate in physical activities with their peers, also risk obesity, psychological distress and are more vulnerable to bullying. However, with effective identification and management many of the difficulties can be addressed and the secondary consequences avoided.

VICKY MCQUILLAN M ED, DIP COT Principal Investigator University of Liverpool, School of Health Sciences The aim of this project is to identify two groups aged 8-16 years. One group suspected by their teachers of having movement difficulties and the other group with no movement or learning difficulties (typically developing). To help us understand factors in their development, the children will be assessed for motor and associated characteristics and then their motor development and progress will be monitored over 20 months (3 contacts). The project has ethical approval from Leeds University ethics committee.

Initially we will send a letter inviting parents and their children to take part in the project. Those who agree to take part will be invited to complete short questionnaires and the children (if they agree) will be invited to take part in activities that assess verbal, non-verbal ability and motor ability and a short questionnaire about enjoyment and participation in physical activity. They will be reassesde very 6 months. EACH CHILD PARTICIPATING:

Will receive a short summary of their results and the parents and teachers of any children identified with difficulties will be given written advice.

# Appendix 3 Letter to head teachers and information for teachers

School of Education University of Leeds Hilary Place Leeds

Dear Headteacher,

I am an occupational therapist and previously worked in the NHS at Woodview Child Development Centre in Widnes before taking up my post at Liverpool University.

I am currently researching a childhood condition called developmental coordination disorder (DCD), also known as dyspraxia. It is of great significance to teachers because it is so commonly found in the school age population but so little understood and yet can impair participation and progress at school.

I would like to invite you and children attending your school to participate in the project, which has received ethical approval from Leeds university research ethics committee.

The aim is to invite parents and their children to take part by letter and then recruit children aged 8-14 years old who have been identified by their teachers as having possible coordination difficulties (may manifest as poor handwriting, poor balls skills, difficulty with tools or equipment) and another group of children without coordination difficulties or learning difficulties for comparison. Research so far has very little information about how children with the condition change during adolescence and we suspect that girls may well be under represented.

We are therefore including both primary and secondary age children in the project.

My team and I will assess all the children who consent to be involved using questionnaires and a standardized movement assessment which present tasks as a series of games. Their parents will also be asked to complete short questionnaires. The children will be reassessed every five months to document change (on 3 occasions) and a small sample of children and parents will be invited to an interview about their experiences. All results will be anonymous and kept confidential.

The results will inform us about the factors associated with change in movement ability and the impact on participation. We will run workshops for professionals and parents to share our findings and help inform future intervention. I am happy to meet you and your staff to explain the research further.

Thank you for taking the time to read this and I look forward to hearing from you.

Yours sincerely,

Vicky McQuillan M Ed, Dip COT vickym@liv.ac.uk

#### Instruction for teachers:

- Introduce the research study explaining the importance of investigating how children's movement skills impact on progress and participation in daily life (see research project flyer "Children with movement difficulties in schools")
- Explain the need to investigate some children who have difficulty with motor skills as well as some children who do not display any difficulty
- Identify any children that you suspect may be encountering difficulty with motor coordination (see leaflet "secondary school children with possible DCD")
- Identify children without motor coordination difficulty and who also do not have any special educational needs and try to **match as closely** as possible to the same year group, age and sex as possible.
- Obtain parental consent and consent from the child to participate in the study
- Collect consent forms to return to the researcher
- Parent questionnaires will then be distributed via school
- The researcher can be contacted by email to answer questions that the parents may have. It is possible to arrange to be available for a drop in session at school if parents or teachers want additional information.
- The researcher will then arrange to collect the questionnaires and arrange a convenient time to carry out the assessment with the child
- A short summary report will be given to each parent to share with their child
- The child assessment will then be repeated the following two terms
- The school will receive a copy of summary of the projects findings and staff will be invited to workshop to discuss the findings and potential implications.

# Children with possible Developmental Coordination Disorder (DCD) in secondary school:

Children with DCD have an unexplained difficulty with the development of their motor coordination, i.e. their difficulties cannot be attributed to a known medical condition, such as cerebral palsy. Either their fine motor coordination or their gross motor coordination, or even both, can be affected.

Children who have reached secondary school without receiving a diagnosis for DCD may well have covered their difficulties with a range of strategies that are effective in primary school and a less demanding environment, but this starts to break down as more demands are expected.

Sometimes the symptoms of DCD may only manifest as increased pressure on speed and volume of writing occurs. Teachers might notice some of the following:

- Very slow handwriting and difficulty completing timed work.
- Disorganized work and very untidy handwriting
- Good verbal ability but paucity of written out put.
- Avoidance of P.E. and if they do attend, they show poor ball skills and difficulty orientating the body to equipment or positioning themself on the sports field. They may be much slower at getting changed.
- Clumsy manipulation of equipment and frequent spills in technical subjects and science labs.
- Difficulty organizing their time and or belongings.
- Difficulty opening packaging e.g. items in a packed lunch
- Girls and boys may react to difficulties differently; often girls tend to withdraw and keep 'under the radar' and tend not to present behaviour problems at school.

The benefits of identifying DCD

- 1. Assessment and correct diagnosis acknowledges the children's difficulties, promotes self-understanding and helps the adults supporting them to advocate for them and alleviate some of the frustration.
- 2. It reinforces that the child is not at fault or 'not trying'
- 3. Task analysis and appropriate adaptive strategies can be adopted to help the child master every day activities and participate
- 4. Alternative forms of recording written work and exam concessions can be adopted if appropriate to enable the children to reach their potential
- 5. In more severe cases, referral to therapists for direct intervention may be required

# Appendix 4 Study information sheets and consent forms

**Parent information** 

Movement development and children

Vicky McQuillan David Sugden

Mary Chambers

Dear Parent,

We are working with children, some of whom have movement difficulties, and we are interested in how they cope with movement activities that they require for everyday life. We are asking for support to work with your child and help you make a decision on permission. We are outlining the study, what your child will do and what we will do with the results.

The results of this work will inform us about patterns of movement development and the collective results will aid other children with movement difficulties who are showing similar characteristics both in this country and internationally.

# Background to the project:

Children who experience movement difficulties, but with no apparent medical reason may meet the criteria for a condition called developmental coordination disorder (DCD). Children with DCD show great variety in their motor development. This impacts on their lives in areas such as mobility and balance leading to reduced participation in everyday activities and recreation. It can also impact on handwriting and tool use. Many children with DCD also have associated difficulties, for example with attention or learning to read, whilst others have no difficulties.

### Aims of the project:

- 1. To identify a group of children who may be at risk of DCD or have DCD and those who do not.
- 2. To profile in detail these children to identify their movement capabilities and any associated characteristics over a period of 21 months.
- 3. To ascertain how these capabilities and characteristics influence activities of daily living, academic progress and enjoyment of physical activity.
- 4. To examine how they develop and change over time.
- 5. To examine which patterns of movement difficulty and which characteristics hinder progress.

### What the children, teachers, parents and researchers will do:

- 1. Parents will be given short questionnaires asking about their children's movement abilities and about their child's family history and communication.
- 2. Children will be given an assessment of their motor skills by using the Movement Assessment Battery for Children (MABC2) (approximately 40 minutes). They will also be given an assessment of their verbal and non-verbal ability (KBIT2). They will also be asked to complete a questionnaire about their enjoyment of physical activity (approximately 10 minutes). Parents will be given a short summary of results and if any difficulties are identified they will be provided with a information sheet, details of where to seek support and an appropriate pathway of referral.
- 3. We would like to interview a small number of teachers, parents and children to ascertain which types of activities the children may experience difficulties and their views (approximately 30 minutes). We would like to interview a small number again over the 21 months (maximum of 3 times). You will be given the opportunity to indicate whether you would prefer not to be contacted for an interview on the consent form.
- 4. After 6-8 months the all the children will be given another assessment of their motor skills and a questionnaire on their enjoyment and participation in physical activity and this will be repeated after another 6-8 months (a maximum of 3 assessments).
- 5. From the results of this investigation and from the research, and professional literature on DCD and motor learning, we will gain better understanding of how children with DCD progress, and what helps or hinders progress. From this recommendations will be made to help with the planning of appropriate interventions for children with DCD and associated characteristics.

6. A short summary of results will be given to the parents of each child who takes part in the study. Finally, workshops will be run by Vicky McQuillan, Professor David Sugden and Dr Mary Chambers at the end of the project for parents and professionals to share the findings and plan a way forward.

Vicky McQuillan January 2014

PhD candidate

School of Education

University of Leeds

Leeds LS2 9JT

Email: edvam@leeds.ac.uk

# Parent consent form

Please circle the appropriate response	Yes	or No
• I confirm that I have read and understood the information letter dat 2013 explaining the research project.	ed Septe <b>Yes</b>	ember <b>No</b>
• I understand that our participation is voluntary and that we are free at any time without giving any reason and without there being any r consequences. In addition, should I wish not to answer any particular questions, I am free to decline.	to witho legative ar quest <b>Yes</b>	draw ion or <b>No</b>
• I understand my responses will be kept strictly confidential.	Yes	No
• I agree for the data collected from me, and my child to be used in rel research and for our data to be held securely in a data archiving facilities includes audio extracts for teaching and conferences/seminars	evant fu lity. Thi <b>Yes</b>	ture s No
• I understand that I am free to withdraw from the study at any time, I data already collected and anonymised will remain in the study and the overall results	out that contribu <b>Yes</b>	any ite to <b>No</b>
• I confirm that I give my permission for the researcher to approach m	iy child <b>Yes</b>	No
• I agree to being contacted for an interview	Yes	No
• I agree to take part in the above research project and will inform the researcher should my contact details change	lead Yes	No
Vicky McQuillan PhD candidate School of Education University of Leeds Leeds LS2 9JT		
January 2014		

# Child information sheet

### Movement development and children

Vicky McQuillan David Sugden

Mary Chambers

#### Invitation

You are being invited to be part of a research study. This research study tries to understand how children get better at movement skills as they get older. These are skills such as balance, throwing and catching and handling small objects. These skills are needed to do every day activities.

It is up to you if you want to take part in this study. Even if you agree now to be part of the study you can change your mind later. No one will be annoyed with you if you choose not to be part of the study.

#### Why are we doing it?

Some children have difficulties with movement skills and this stops them doing some of the things that they want to, even some things that other children do not find difficult.

This study is trying to find out which children have difficulty with movement and the type of challenges they have. It will follow how the children change over time to compare with children who do not have movement difficulties. This will help us to understand how children get better at movement skills and explore some ideas that may help the children who have difficulty with movement.

### What will happen?

If you agree to be in this study, a researcher will come to your school once a term for about an hour on three different occasions. Each visit you will be

asked to do activities involving balance, throwing, catching and drawing and answer some questions on whether you enjoy or take part in physical activity.

If you decide that you do not want to continue in the study you can stop at any time, but the information that we have already collected will stay in the study.

A small number of children will be invited to have interviews about their movement experiences. If you would rather not be asked to an interview then please circle **No** on the attached form.

### Who is doing the study?

Vicky McQuillan, other therapists and therapy students will come to your school.

### Who will know I am in the study?

Only your parents, teachers and the researchers will know you are in the study. When the study is finished Vicky will write a report about what has been learned. This report will not say your name or that you were in the study.

### When do I have to decide?

It would be a good idea to discuss your decision with your parents. Then let Vicky know if you want to take part in two weeks time.

If you would like to be part of the study then put your name at the end of the attached form.

Thank you for taking the time to read this and for your help.

Vicky McQuillan PhD candidate School of Education University of Leeds Leeds LS2 9JT

January 2014 Email: edvam@leeds.ac.uk (Adapted from Child & Family Institute Canada, March 2010, www.CFRI.Ca)

# Child consent sheet

# Movement development and children

	Vicky McQuillan	David Sugden	Mary Cham	bers	
Plea	se circle your answer			Yes	or No
•	I have read and unders explaining the research	tand the information project.	letter dated September 2	2013 <b>Yes</b>	No
•	I understand that I am f without there being any particular question or c	free to stop at any tin y problem.  In additio questions, I am free to	ne without giving any rea n, if I wish not to answer o not to.	ison and any <b>Yes</b>	d No
•	I understand my respo	nses will be kept stric	tly private.	Yes	No
•	I understand that I am f already collected will st	free to leave the stud tay in the study and c	y at any time, but that an ontribute to the overall r	y inforn esults <b>Yes</b>	nation <b>No</b>
•	I agree to being contact	ed for an interview		Yes	No
	Vicky McQuillan PhD candidate School of Education University of Leeds Leeds LS2 9JT				
	Email: <u>edvam@leeds.ac</u>	<u>c.uk</u>			
	January 2014				

# Child interview information

My name is Vicky McQuillan. I am an occupational therapist working in a university. I am doing research about how children develop their movement skills.

I would really like to hear about you and your experiences with movement in every day life, for example, at home and school.

Our talk would be in private but your ideas will help us to understand more about how children develop movement skills in every day life. I would like to come back to hear your ideas again after the next school term and then one more time the following term (3 times in total).

You can ask for the interview to stop at any time. It will not take longer than an hour.

You can say yes or no. It is up to you whether you take part and nobody will be annoyed with you if decide that you do not want to take part.

If you would like to take part in the study by talking to me please sign the attached form and return it to school.

If you want to know more information about the project please contact me at the address below.

Thank you for taking the time to read this and for your help.

Vicky McQuillan PhD candidate School of Education University of Leeds Leeds LS2 9JT Email: edvam@leeds.ac.uk

January 2014

# Child interview consent sheet

# Movement development and children

	Vicky McQuillan	David Sugden	Mary Chan	nbers	
Pleas	se circle your answer			Yes	or No
•	I have read and unders the research project.	tand the information l	etter dated January 201	l 4 expla <b>Yes</b>	ining <b>No</b>
•	I understand that I am f without there being any particular question or c	free to stop at any time y problem. In addition juestions, I am free to	e without giving any rea 1, if I wish not to answer not to.	ason and r any <b>Yes</b>	d No
•	I understand my respo	nses will be kept strict	ly private.	Yes	No
•	I understand that I am f already collected will st	free to leave the study ay in the study and co	at any time, but that an intribute to the overall	ıy inforr results <b>Yes</b>	nation <b>No</b>
•	I agree to being contact	ed for another intervi	ew	Yes	No
	Vicky McQuillan PhD candidate School of Education University of Leeds Leeds LS2 9JT Email: <u>edvam@leeds.ac</u>	<u>uk</u>			

January 2014

# Appendix 5 Responses to ethics committee

# The progression over time of profiles of children with Developmental Coordination Disorder (DCD)

# Ethics Reference: AREA 12-121

# 2<sup>nd</sup> summary response for Leeds University ethics committee (19.01.14)

Please find attached the changes (highlighted in yellow) to the parent information (Appendix 1), child information (Appendix 3) and parent consent (Appendix 2) sheets following advice from Leeds ethics committee and discussion with David Sugden on 9.1.14.

The changes include:

- 1. A sentence asking parents to give permission for the researcher to approach their child
- 2. An amended child information sheet with language easier for an 8 year old
- 3. The support available to children identified with DCD is made more explicit
- 4. The workshops will be facilitated by experienced professionals

I will also use two information sheets for parents/schools about DCD from the Can Child organization at McMaster University based on their research on DCD, which will be provided for the children identified with movement difficulties in this study. The two flyers will be Advice for P.E. educators and Accommodations at school

These are available from: <a href="http://dcd.canchild.ca/en/EducationalMaterials/school.asp">http://dcd.canchild.ca/en/EducationalMaterials/school.asp</a>.

#### DCD project: ethical reference AREA 12-121

Title: The progression over time of profiles of children with Developmental Coordination Disorder (DCD)

# 1<sup>st</sup> Summary response for Research ethics committee University of Leeds (15.9.13)

A9, A10 and C7 (i) (ii)

The cohort of children identified by teachers as having, or not having movement difficulties do not already have a diagnosis. This is crucial to the study, as previous research has shown that clinic populations and school populations of children with DCD (i.e. not known to a clinic and therefore probably undiagnosed) contain quite different types of children:

- Clinic children are more likely to have associated characteristics and meet diagnostic criteria for one or more other developmental disorder. It is likely too that they may have more severe symptoms (Kaplan et al. 2001)
- School population children may have better coping strategies or other characteristics that have enabled them to avoid coming to the attention of referral agents before the age of 8-16 years. This is an area that we will be exploring in this study and we hope will give us insight into helping children in the future.

Part of a teacher's role is to evaluate a child's profile of skills and identify any specific needs for support throughout the curriculum. These are then usually discussed with parents during term, or at particular meetings, such as parent's evenings. This is not generally thought to raise anxiety, rather raise awareness for understanding of the nature of the problem and how to support improved learning, which leads to better outcomes for the child. It is appropriate for teachers to highlight potential difficulties, but it should be noted that it would not be until the child has undergone the battery of assessments that the nature of any difficulty will be able to be determined.

Each school has a Special Educational Needs Coordinator (SENCO) who is able to advise and support the teachers if they suspect that any child may have additional special needs. Furthermore, the SENCOs have regular meetings and in-service training with paediatric therapists from the local community team to discuss movement difficulties and children with medical diagnoses, so they already have a reasonable awareness of developmental disorders.

Parents will be given information about the study via the school and can opt out straight away by informing the teacher if they do not want their child to take part. The remaining parents will then be given consent forms for their child to take part in the study and undergo assessment. All parents will have to give consent to have their child assessed. Any child whose parent who decides that they would not like to have their child assessed will be excluded from the study.

Please refer to the decision tree for an explanation of the pathway for the consent process.

C 7(ii) The information sheet covers all categories because it will not be known which category the child will meet until the battery of assessments is complete. However, once the child has been assessed the parents and the child will be informed of the results and consequently which arm of the study they will be in.

A 10 After discussion it was decided to provide each school and clinic with a brief written summary of progress of the project rather than through Facebook.

C 2 Interview information:

The aim of the interviews is to explore the subjective experiences of the children with DCD and DCD plus associated characteristics and those of the adults that support them. The theoretical stance will be phenomenological and will form will be semi-structured interviews, so that issues that arise can be explored in more depth with the used of grand tour questions such as "tell me about what it is like...". This will be informed by previous qualitative studies in

DCD, the majority of which have covered the parent's perspective of a child with DCD (Iverson et al. 2005; Mandich et al. 2003; Missiuna et al. 2006; Missiuna et al. 2007), some the young adult perspective with DCD (Missiuna et al. 2008; Kirby et al. 2008) and fewer the child's perspective (Dunford et al. 2005). An interview schedule, such as that proposed by Missiuna et al. (2008) comprising the main anticipated issues from previous research on DCD and areas that focus specifically on the research question for this study will cover two of the ICF categories of participation in self care and community, social and civic life:

- Life and progress in school, participation in physical activity, sports, friendships, academic strengths/weaknesses, organization skills, best/worst times, strategies
- Leisure pursuits, type/importance /enjoyment/motivation/frequency
- Self care- differences /problems/strategies
- Use of probes "can you tell me more about that?" "what helps?" "what do you like/dislike about that?"

•

This schedule will be used for both the experiences of the child with DCD (and DCD plus associated characteristics) and their parents.

The teacher interviews however, will focus on their perspective and include:

- Differences both positive and negative/frustrations/unexpected outcomes
- Effects on child/class
- Strategies
- Information that I wish I had before
- C 3 Follow up interviews for parents and teachers may take place over the phone for their convenience but all child interviews will be face to face.
- C11 There is sparse longitudinal data on DCD and there will be both quantative and qualitative from this study, which will be valuable to compare with future studies. Should the same participants be invited to take part in a future study the appropriate permission would be sought and new ethical approval and consent would be sought before it could go forward.
- C15 The interviews:

- a) Those with the parents will be conducted by VM, an experienced paediatric OT with 18 years of experience of talking and listening to parents about sensitive subjects concerning their children. The interviews will take place in a private room, either at school or in the parent's home if preferred, and will take approximately half an hour. The topic of the interviews have been outlined above and will aim to gain the parent's perspective of how the movement and other characteristics effect family life and their child's ability to participate in daily life. We will be particularly interested in strategies that families have found helpful to them and areas of particular difficulty.
- b) Those with children will also be conducted by VM. The interviews will take place in school in a private room and will take approximately half an hour as outlined above. The child's perspective is important and may well differ from the parent's. They will be reassured that they can stop at any time, or take a break if any sensitive subjects arise.
- c) Those with teachers will be conducted by VM and will differ from that of the child or parent interviews in that they will seek the perspective of the teacher on the effect on the class, any strategies or information that has been helpful or unhelpful.
- C 15 A different information sheet will be used for interviews.
- C6, C 18 Risk assessment is required for lone working
- C 19 Personal records will include clinic records for the clinic cohort and school records for the school cohort and these will be assessed by VM. These will be required to determine the nature of the medical diagnoses in the clinic population and any intervention they may have had, and the school records will be required to determine school progress.
- Appendix 3 After discussion the child information sheet was considered appropriate for children of 8-16 years. Pictures or much simpler language may be considered patronizing for children of 8 years old.

# Appendix 6 Parent Questionnaire Parent Questionnaire

#### About you and your family

- 1. Name: 2. Postcode:
- 2. Have you or any close relative been diagnosed with one of the following? Please tick more than one if appropriate:

ADHD	DCD/dyspraxia	Dyslexia	Autistic	Language	Other	None
attention	movement	reading	Spectrum D	impairment		

3. Who lives at home? Please tick and state number of children in each group

Mother	Father	Children	Children aged	Children aged	Young people
		under 5	5-11	12-18	18+

4. How would you describe your current health? Please tick one box

Healthy	Health problems	Registered disabled

5. Employment for you and your partner: Full time (FT), Part time (PT) please tick one box

Both parents FT	One FT or both PT	One parent FT	One parent PT	Not working

6. Highest level of your education (mother only): please tick one box

High	High school with	'A' level	College	H E	Post grad
school	GCSE		diploma	Degree	degree

7. Are your children entitled to free school meals? Please circle: Yes/no

8. Family transport: do you have a car(s)? Please tick one box

0	1	2 (or more)

- 9. How important is sport or physical activities to your family? Rank 1= not very to 4= very important: please circle: 1 2 3 4
- 10. It is hard to find time sometimes but how much physical activity/sport are you able to do regularly? Please tick one box

Twice weekly	Weekly	Twice monthly	Monthly	Rarely

11. Rank how much the level of your child's coordination interferes with family life: 1=not at all<br/>4= very much:1234

# Appendix 7 Interview schedule Child Interview:

#### Child's name:

#### Date of interview:

Information sheet given, voice recorder explained:	yes/no
Consent form signed:	yes/no

#### **Child Interview Schedule Prompts:**

- 1. Tell me about yourself
- 2. What do you like doing?
- 3. What is good about being you?
- 4. What is school like for you? Friends/classwork/playground
- 5. What do you enjoy?
- 6. Is there anything you struggle with at school? Or home?
- 7. How do you cope?
- 8. Who/what helps you most?
- 9. You have already told me that you enjoy/don't enjoy physical activity & P.E, why is this?
- 10. What do you do after school and at the weekends? Organized? Or free play?
- 11. What does your mum or dad think of you?
- 12. If you could wave a magic wand to change something what would it be?

After Lingam et al., 2013.

# Appendix 8 Publications and conference presentations arising from the study

McQuillan, V.A. et al. (2015) Intervention and support in DCD: from research to practice Journal of Comorbidity, 5 (32). Pp. 32-109. DOI: 10.15256/joc.2015.5.52

**McQuillan, V.A.** (2015) *Stability and change over time in children with movement difficulties.* Hillary Place Papers, 2nd edition, 2. Online Journal

**McQuillan, V.A.** et al. (2013) *The progression over time of profiles of children with developmental coordination disorder.* (2013) Brazilian Journal of Motor Behaviour, 7. p. 8. ISSN ISSN 1980-5586

### **Conference presentations (unpublished)**

**McQuillan, V.A., Swanwick, R.A., Chambers, M.E. & Sugden, D.A.** (2019) What do detailed motor and non-motor profiles of children with DCD tell us about motor progression over time? DCD 13<sup>th</sup> International Conference 5-8<sup>th</sup> June 2019, Jyvaskyla, Finland.

**McQuillan, V.A., Swanwick, R.A., Chambers, M.E. & Sugden, D.A.** (2018) *How detailed profiles of children with Developmental Coordination Disorder* (*DCD*) *influence participation in physical activity (PA). School based evidence.* 7<sup>th</sup> Biennial Academic and practitioner conference on DCD, Brunel University, 29<sup>th</sup> -30<sup>th</sup> June 2018. **McQuillan, V.A., Sugden, D.A., Chambers, M.E., Swanwick, R.** (2017) *How* can we broaden the horizons for children with developmental coordination disorder? -Evidence from a collaborative study. 'Broadening Horizons: *Celebrating Collaborative Practice*' Royal College of Occupational Therapy Children Young People and Families, specialist section conference, Leeds 2017.

**McQuillan, V.A., Sugden, D.A., Chambers, M.E., Swanwick, R.** (2017) *Progression and participation in children with and without DCD and associated characteristics: A longitudinal study.* DCD 12<sup>th</sup> International conference, 5-8<sup>th</sup> July 2017 Perth Australia.

**McQuillan, V.A., Sugden, D.A., Chambers, M.E., Swanwick, R.** (2017) *DCD and associated characteristics: what facilitates participation in physical activity?* DCD 12<sup>th</sup> International conference, 5-8<sup>th</sup> July 2017 Perth, Australia.

**McQuillan, V.A., Sugden, D.A. , Chambers, M.E., Swanwick, R.** (2016) Stability and change in children with and without development coordination disorder. Oral presentation: 6<sup>th</sup> Biennial Conference DCD UK, University of Leeds, 22 & 23<sup>rd</sup> July 2016.

**McQuillan, V.A.** (2013) An investigation of the influence of associated characteristics on the progression of children with developmental coordination disorder. In: Symposium on motor impairment, 3.6.2013, University of Leeds.

**McQuillan, V.A**. (2012) The progression of profiles of children with developmental coordination disorder over time. In: Salford Postgraduate Research Conference, May 2012, Salford University.